Trace Formulas in Noncommutative Geometry

Eva-Maria Hekkelman

A thesis in fulfilment of the requirements for the degree of

Doctor of Philosophy



School of Mathematics and Statistics

Faculty of Science

June 2025

Abstract

Trace formulas appear in many forms in noncommutative geometry (NCG). In the first part of this thesis, we obtain results for asymptotic expansions of trace formulas like heat trace expansions by adapting the theory of Multiple Operator Integration to NCG. More broadly, this construction provides a natural language for operator integrals in NCG, which systematises and simplifies operator integral arguments throughout the literature. Towards this end, we construct a functional calculus for abstract pseudodifferential operators and generalise Peller's construction of multiple operator integrals to this abstract pseudodifferential calculus. In the process, we obtain a noncommutative Taylor formula.

In the second part of this thesis, we shift our attention to Dixmier trace formulas. First, we provide an approximation of the noncommutative integral for spectrally truncated spectral triples in the Connes–Van Suijlekom paradigm of operator system spectral triples. Our approximation has a close link to Quantum Ergodicity, which we will use to state an NCG analogue of the fundamental result that ergodic geodesic flow implies quantum ergodicity. Furthermore, we provide a Szegő limit theorem in NCG. Next, we provide a Dixmier trace formula for the density of states, a measure originating in solid state physics that can be associated with an operator on a geometric space. We first provide this formula in the setting of discrete metric spaces, and then in the setting of manifolds of bounded geometry. The latter leads to a Dixmier trace formula for Roe's index on open manifolds.

Acknowledgement

With heartfelt gratitude and admiration to my PhD supervisor Fedor Sukochev, and co-supervisors Edward McDonald and Dmitriy Zanin. You have not only the highest standards for research, but also the utmost confidence in your students to meet them. No encouragement is more effective. Thank you for your guidance, I will remember my time with you very fondly. In particular, I would like to thank Edward McDonald for the many hours you set aside to keep an eye on my progress. You are an inspiration to me, and I am proud to call myself a student of yours.

I owe much to Teun van Nuland as well. Thank you for taking me along in a new research direction when I was looking for something fresh. I learned a lot from you, and it has been fantastic to work together.

A sincere thanks to Nigel Higson is in order, I am immensely grateful for your support. It has been inspirational to witness your joyful lectures.

I would like to thank Nigel Higson, Marius Junge, Edward McDonald, Teun van Nuland, and Walter van Suijlekom for their hospitality during my visits abroad, as well as the staff at the Pennsylvania State University, Delft University of Technology, and the Radboud University Nijmegen. Thank you to Nigel Higson, Fedor Sukochev, and the Australian Mathematical Society for providing funds for these travels.

For the work this thesis was based on, I am glad to acknowledge my co-authors Nurulla Azamov, Edward McDonald, Teun van Nuland, Fedor Sukochev, and Dmitriy Zanin, as well as helpful communications with Magnus Goffeng, Nigel Higson, Bruno Iochum, Eric Leichtnam, Qiaochu Ma, and Raphaël Ponge. Their contributions will be indicated in the relevant chapters as well.

For the brave task of proofreading this thesis, I thank and apologise to Fedor Sukochev, Edward McDonald, and Teun van Nuland. Of course, I claim full responsibility for any mistakes remaining. Furthermore, I am thankful to the referees who volunteered their time for reading and assessing this dissertation, as well as the anonymous referees of the preprints and journal articles on which this is all based.

I thoroughly enjoyed my trips down to the University of Wollongong for the OANCG seminars. Thank you to Adam Rennie, Aidan Sims, Angus Alexander, Ada Masters,

Alexander Mundey, Abraham Ng and the others for the warm welcomes, the cold drinks, and of course the maths.

Many thanks to Thomas Scheckter for answering silly questions of mine, and for helping me navigate the scary world of academic paperwork. This PhD was partially supported by Australian Research Council grant FL170100052.

Cheers, Abirami Srikumar and Haley Stone, for helping with a bit of Python coding in the final stretch of work on this thesis.

Finally, I wish to thank my parents, my sibling, my grandmother, and my dear friends for your love and support. These past few years have been quite a ride, and I will never forget that you were there for me through it all.

> someone will remember us I say even in another time

> > Sappho, Fragment 147 [Voi71] Trans. Anne Carson [Car02]

List of Publications

- [Aza+22] N. Azamov, E.-M. Hekkelman, E. McDonald, F. Sukochev, and D. Zanin. An application of singular traces to crystals and percolation. J. Geom. Phys. 179 (2022), Paper No. 104608.
- [HM24a] E.-M. Hekkelman and E. McDonald. A general Dixmier trace formula for the density of states on open manifolds. SIGMA Symmetry Integrability Geom. Methods Appl. 20 (2024), Paper No. 007, 34.
- [HM24b] E.-M. Hekkelman and E. McDonald. A noncommutative integral on spectrally truncated spectral triples, and a link with quantum ergodicity. *arXiv* preprint arXiv:2412.00628 (2024).
- [HMN24] E.-M. Hekkelman, E. McDonald, and T. D. H. van Nuland. Multiple operator integrals, pseudodifferential calculus, and asymptotic expansions. *arXiv* preprint arXiv:2404.16338 (2024).

Contents

\mathbf{A}	Abstract Acknowledgement								
A									
Li	List of Publications								
Co	Contents								
1 Introduction									
	1.1	Noncommutative geometry	2						
		1.1.1 Spectral geometry	2						
		1.1.2 Noncommutative geometry	7						
	1.2	Pseudodifferential operators	11						
	1.3	8 Multiple operator integrals							
	1.4	1.4 Connes' integral formula							
	1.5 Density of states								
	1.6	Preliminary material	32						
Ι	I The Joys of MOIs								
2	2 Functional calculus for abstract pseudodifferential operators								

2.1	Abstract pseudodifferential calculus	. 39			
	2.1.1 Review	. 39			
	2.1.2 Adjoints	. 41			
	2.1.3 Elliptic operators	. 45			
2.2	2 Functional calculus for elliptic operators				
2.3 Functional calculus for zero-order operators					
Mu	tiple operator integrals as abstract pseudodifferential operators	65			
3.1	Operator integrals	. 66			
3.2	Multiple operator integrals as abstract pseudodifferential operators	. 69			
3.3	Divided differences	. 76			
3.4	MOI identities	. 82			
3.5	Asymptotic expansions	. 84			
3.6	Trace expansions and NCG	. 90			
Di	xmier Trace Formulas	100			
Con	nes' integral formula and quantum ergodicity	101			
4.1	Integration on truncated spectral triples	. 105			
4.2	Szegő limit theorem	. 113			
4.3	Fröhlich functional	. 115			
4.4	Noncommutative ergodicity	. 117			
The density of states on discrete spaces 131					
5.1	Metric condition	. 137			
	5.1.1 Solid matter	. 139			
	 2.1 2.2 2.3 Mull 3.1 3.2 3.3 3.4 3.5 3.6 Di Com 4.1 4.2 4.3 4.4 The 5.1 	2.1 Abstract pseudodifferential calculus 2.1.1 Review 2.1.2 Adjoints 2.1.3 Elliptic operators 2.2 Functional calculus for elliptic operators 2.3 Functional calculus for zero-order operators 3.4 Multiple operator integrals as abstract pseudodifferential operators 3.2 Multiple operator integrals as abstract pseudodifferential operators 3.3 Divided differences 3.4 MOI identities 3.5 Asymptotic expansions 3.6 Trace expansions and NCG 3.7 Dixmier Trace Formulas Connes' integral formula and quantum ergodicity 4.1 Integration on truncated spectral triples 4.3 Fröhlich functional 4.4 Noncommutative ergodicity 4.4 Noncommutative ergodicity			

		5.1.3	The integer lattice	142		
		5.1.4	Quasicrystals	143		
		5.1.5	Percolation	144		
	5.2	Dixmi	er trace formula for the DOS	146		
	5.3	Problems of measurability				
	5.4					
		5.4.1	Translation equivariance on lattice graphs $\ldots \ldots \ldots \ldots \ldots$	155		
		5.4.2	Ergodic operators	157		
6	The density of states on manifolds					
	6.1	Preliminaries				
		6.1.1	Operator theory	168		
		6.1.2	Preliminaries on manifolds	170		
	6.2	Proof of Theorem 6.0.4				
	6.3	Manife	olds of bounded geometry	178		
	6.4	Roe's	index theorem	195		
	6.5	An exa	ample with a random operator	201		
	6.6	Discre	te metric spaces revisited	205		

Chapter 1

Introduction

Do not forget our motto "work harder", and act accordingly.

Fedor Sukochev

The thesis before you covers a variety of topics in spectral theory, tied together by Alain Connes' philosophy of noncommutative geometry [Con94]. These will be served in two parts: the first concerns pseudodifferential operators and multiple operator integrals based on the paper [HMN24], the second part revolves around Dixmier trace formulas related to Connes' integral formula based on the papers [Aza+22; HM24a; HM24b]. All four papers are joint works with Edward McDonald, and furthermore Nurulla Azamov, Teun van Nuland, Fedor Sukochev and Dmitriy Zanin feature as co-author on one paper each.

Along the way we will additionally encounter heat trace expansions, quantum ergodicity, and a little index theory. While this mix hopefully provides an entertaining read, as a consequence there is a lot of background material to discuss. I have made an attempt to summarise the five main themes of this dissertation in five sections. The experts and the brave could skip ahead as the mathematical content of each chapter is meant to stand alone — with the exception of Chapter 3 which depends on Chapter 2. In comparison with the papers this thesis draws from, the presentation here is made more accessible, with additional background information and some novel results.

1.1 Noncommutative geometry

Over the past decades noncommutative geometry has developed into a rich field of mathematics, bringing together differential geometry, functional analysis, spectral theory, *K*theory, representation theory, quantum field theory, and many more areas. It allows the study of 'manifolds' on which coordinate functions do not necessarily commute, exotic spaces which appear in various areas of mathematics and physics. As any brief summary of this subject must, the overview here leaves out many interesting and important aspects of NCG. Recommended expositions which do more justice to the field can be found in [Con94; GVF01; Sui25]. The book [EI18] is an excellent reference for many of the analytical details that we will use. For a more algebraic introduction to noncommutative geometry, see [Kha13].

1.1.1 Spectral geometry

As is taught in kindergarten, operators associated with geometric spaces carry geometric information in their spectra which can be carefully extracted. Indeed, (hearing) children can tell that smaller objects tend to produce higher-pitched sounds, hollow objects sound different from solid ones, and can blindly distinguish between the sound of strings, drums, and blocks. What their ears are picking up is the spectrum of the Laplace operator.

Modeling an object as a bounded domain (open connected set) in *d*-dimensional Euclidean space $\Omega \subseteq \mathbb{R}^d$, the sound it produces when lightly struck decomposes into frequencies which are determined by the eigenvalues of the Laplace operator $\Delta := \sum_{j=1}^d \partial_j^2$ on Ω with Dirichlet boundary conditions, that is, they are the eigenvalues of those $\lambda \in \mathbb{R}_{\geq 0}$ for which there exists a solution of the Helmholtz equation

$$\begin{cases} -\Delta u(x) &= \lambda u(x), \quad x \in \Omega; \\ u|_{\partial \Omega} &\equiv 0. \end{cases}$$

Writing $N(\lambda)$ for the number of such eigenvalues (counting multiplicities) less than λ ,

Weyl's law [Wey12; Cha84; Ivr16] gives that

$$N(\lambda) \sim \frac{\omega_d}{d(2\pi)^d} \operatorname{Vol}(\Omega) \lambda^{\frac{d}{2}}, \quad \lambda \to \infty,$$
 (1.1)

where ω_d is the volume of the unit sphere $\mathbb{S}^{d-1} \subseteq \mathbb{R}^d$, and ~ means that the ratio converges to 1. This indicates that in a perfect world, our ears can detect the dimension d and the volume of an object. For $d \geq 2$ and with technical assumptions on Ω (smooth boundary, set of periodic billiards has measure zero), Duistermaat, Guillemin and Ivrii have shown [DG75; Ivr80; Ivr16] that even the d-1-dimensional volume of the boundary $\partial\Omega$, Vol $(\partial\Omega)$, can be 'heard' as indicated by the Weyl law with error terms

$$N(\lambda) \sim \frac{\omega_d}{d(2\pi)^d} \operatorname{Vol}(\Omega) \lambda^{\frac{d}{2}} - \frac{\omega_{d-1}}{4(d-1)(2\pi)^{d-1}} \operatorname{Vol}(\partial\Omega) \lambda^{\frac{d-1}{2}} + o(\lambda^{\frac{d-1}{2}}), \quad \lambda \to \infty.$$
(1.2)

This error term was already conjectured by Weyl himself [Wey13]. However, note that spheres satisfy neither these technical conditions nor this version of Weyl's law [Ivr16]. See Figure 1.1 for an illustration of the effect of the error term in Weyl's law.

More is true: even information about an object's curvature is hidden inside the spectrum of the Laplace operator. Taking now a closed (oriented) Riemannian manifold (M,g)with Laplace–Beltrami operator Δ_g , Minakshisundaram and Pleijel [MP49] proved that the heat trace admits an asymptotic expansion

$$\operatorname{Tr}(\exp(t\Delta_g)) \sim \sum_{k=0}^{\infty} a_k(\Delta_g) t^{\frac{k-d}{2}}, \quad t \to 0,$$

the first coefficients of which can be given as [Ber68; Gil75]

$$a_0(\Delta_g) = (4\pi)^{-\frac{d}{2}} \operatorname{Vol}(M);$$

$$a_2(\Delta_g) = -\frac{1}{6} (4\pi)^{-\frac{d}{2}} \int_M R \, d\nu_g,$$

where ν_g is the Riemannian volume form of M and R is the scalar curvature. The coefficients $a_k(\Delta_g)$ vanish for odd k, and the higher coefficients $a_k(\Delta_g)$ are integrals over M of (complicated) expressions involving the metric of M [Gil75].

These facts raise the question how far this philosophy can be pushed. And indeed it is tradition in any text on spectral geometry to cite an influential essay by Mark Kac from



Figure 1.1: Weyl's law and the spectral counting function for the Laplacian on a disk and a 5-pointed star, with Dirichlet boundary conditions. Both domains have area equal to 1, making the first-order term in Weyl's law the same, $N(\lambda) \approx \frac{1}{4\pi} \lambda$, plotted in blue. The 5-pointed star has a longer perimeter than the disk, resulting in a larger deviation from Weyl's law (1.2). These eigenvalues were generated in Python with the finite-element method, using FEniCSx [Bar+23].

1966 with the title "Can one hear the shape of a drum?" [Kac66]. In this text, Kac asked whether a bounded domain in \mathbb{R}^2 can be determined up to isometries by the eigenvalues of the Laplace operator, and explored various aspects surrounding this question. For Riemannian manifolds the answer was known to be negative [Mil64] — only much later two isospectral domains in \mathbb{R}^2 were identified [GWW92], answering Kac's question in the negative, see Figure 1.2.

In some sense, the Laplacian is not too far off from pinning down a compact Riemannian manifold, however. Let us say that the Riemannian manifolds (M_1, g_1) and (M_2, g_2) are isometric if there exists a diffeomorphism $\phi : M_1 \to M_2$ such that $\phi^* g_2 = g_1$, where ϕ^* denotes the pull-back via ϕ . Arendt, Biegert and Ter Elst proved that the data $(L_2(M), \Delta_q)$



Figure 1.2: Two domains in \mathbb{R}^2 with spectrally indistinguishable Laplacians [GWW92]. This figure was created by Jitse Niesen and is in public domain.

in combination with the cone of positive elements $L_2(M)_+$, that is, those $\phi \in L_2(M)$ corresponding to positive real-valued functions, determines the manifold up to isometries.

Theorem 1.1.1 ([ABE12]). Let (M_1, g_1) and (M_2, g_2) be connected compact Riemannian manifolds, with corresponding Laplace–Beltrami operators Δ_1 and Δ_2 . Then the following are equivalent:

- 1. the Riemannian manifolds (M_1, g_1) and (M_2, g_2) are isometric;
- 2. there exists a bounded linear map $U: L_2(M_1) \to L_2(M_2)$ such that

$$\phi \in L_2(M_1)_+ \iff U\phi \in L_2(M_2)_+;$$

 $U\Delta_1 = \Delta_2 U.$

Still, the cone of positive elements $L_2(M)_+$ is in some sense not operator theoretical, and one can wonder if there is a clever and natural way to nail down a compact Riemannian manifold using only operators.

And indeed there is an algebra of operators on $L_2(M)$ that we can add to the data $(L_2(M), \Delta_g)$ to do this job. It is the commutative C^* -algebra C(M) of continuous

complex-valued functions on M, which represents canonically as bounded operators on $L_2(M)$ via pointwise multiplication: for $f \in C(M)$ we write

$$M_f: L_2(M) \to L_2(M)$$

 $g \mapsto fg.$

Then the triple $(C(M), L_2(M), \Delta_g)$ uniquely determines the Riemannian manifold.

Corollary 1.1.2. Let (M_1, g_1) and (M_2, g_2) be connected compact Riemannian manifolds, with corresponding Laplace–Beltrami operators Δ_1 and Δ_2 . Then the following are equivalent:

- 1. the Riemannian manifolds (M_1, g_1) and (M_2, g_2) are isometric;
- there exists a unital *-isomorphism ψ : C(M₁) → C(M₂) and a unitary operator
 U : L₂(M₁) → L₂(M₂) such that

$$UM_f = M_{\psi(f)}U, \quad f \in C(M_1)$$
$$U\Delta_1 = \Delta_2 U.$$

Proof. (1) \Rightarrow (2) is trivial. For the reverse direction, suppose we have a unitary U: $L_2(M_1) \rightarrow L_2(M_2)$ as in (2). Because the manifolds M_i are assumed to be compact, the kernel of Δ_i consists of constant functions. Since U intertwines the Laplacians Δ_i , we can therefore assume without loss of generality that $U1_{L_2(M_1)} = 1_{L_2(M_2)}$. By Theorem 1.1.1, we need only check that

$$\phi \in L_2(M_1)_+ \iff U\phi \in L_2(M_2)_+.$$

Indeed, we have

$$\phi \in L_2(M_1)_+ \iff \langle M_f 1_{L_2(M_1)}, \phi \rangle_{L_2(M_1)} \ge 0 \quad \forall f \in C(M_1)_+$$
$$\iff \langle M_{\psi(f)} 1_{L_2(M_2)}, U\phi \rangle_{L_2(M_2)} \ge 0 \quad \forall f \in C(M_1)_+$$
$$\iff U\phi \in L_2(M_2)_+.$$

1.1.2 Noncommutative geometry

Comfortable with the idea that operators on a Hilbert space can encode geometrical data, we can now take the radical step that Alain Connes laid out and generalise geometry beyond the commutative world.

With Gelfand duality in mind (the correspondence between commutative C^* -algebras and locally compact Hausdorff spaces [Mur90]), the study of noncommutative C^* -algebras can be thought of as 'noncommutative topology'. Similarly, the theory of noncommutative von Neumann algebras can be thought of as 'noncommutative measure theory'. These analogies are more than amusing observations. Such noncommutative C^* -algebras and von Neumann algebras have proven their value as two of the most versatile objects in operator theory, appearing in countless areas of mathematics and physics. Topological and measure theoretic constructions and arguments can inform analogues in C^* -algebras and von Neumann algebras and vice versa.

As a concrete example, K-theory was developed by Atiyah and Hirzebruch [AH61] as a theory of groups constructed from vector bundles on topological spaces (now called topological K-theory). This was used with great success in a proof of the Atiyah–Singer index theorem [AS68]. Soon after, K-theory was generalised to (noncommutative) C^* algebras, where it became indispensable for operator algebraists, with landmark results being Elliott's program for the classification of C^* -algebras [Ell76; Ell93] and Brown– Douglas–Fillmore's development of K-homology and the classification of extensions of C^* -algebras [BDF77] (see [Weg93; Bla98; RLL00] for books on K-theory).

In the wake of the developments and ideas surrounding K-theory and the Atiyah–Singer index theorem, Connes postulated his noncommutative differential geometry [Con82; Con85; Con94]. He saw that a noncommutative *-algebra of bounded operators, in combination with a self-adjoint operator satisfying certain properties, can be interpreted as a noncommutative generalisation of a (spin) manifold. These ideas and techniques are relevant for badly behaved spaces like the unitary dual of a finitely generated non-abelian discrete or Lie group, the leaf space of a foliated manifold, or the orbit space of group actions on manifolds, which are more easily described with noncommutative *-algebras than as a usual point-set topological space.

Explicitly, a 'noncommutative space' is described by a spectral triple. This has its origin in a paper on Kasparov's KK-theory by Baaj and Julg [BJ83] where spectral triples are introduced as a type of 'unbounded K-cycle'. Spectral triples were later popularised by Alain Connes who recognised their potential as a characterisation of manifolds.

Definition 1.1.3. A unital spectral triple $(\mathcal{A}, \mathcal{H}, D)$ consists of:

- 1. a Hilbert space \mathcal{H} ;
- 2. a unital *-algebra \mathcal{A} represented faithfully as bounded operators $\pi : \mathcal{A} \to B(\mathcal{H})$, where $\pi(1) = 1_{B(\mathcal{H})}$;
- 3. a self-adjoint operator D on \mathcal{H} with compact resolvent,

such that $\pi(a) \operatorname{dom}(D) \subseteq \operatorname{dom}(D)$ and $[D, \pi(a)]$ extends to a bounded operator for all $a \in \mathcal{A}$. We usually identify $a \in \mathcal{A}$ with its image $\pi(a) \in B(\mathcal{H})$, and omit writing π altogether.

The spectral triple is called 'even' if equipped with a \mathbb{Z}_2 -grading γ on \mathcal{H} such that $D\gamma = -\gamma D$ and $a\gamma = \gamma a$ for all $a \in \mathcal{A}$.

The basic commutative example of a spectral triple has roots in the Atiyah–Singer index theorem. It is not dissimilar from the triple $(C(M), L_2(M), \Delta_g)$ that we arrived at in Section 1.1.1. We simply replace C(M) by $C^{\infty}(M)$, and Δ_g by the Dirac operator, an elliptic first-order differential operator which functions as a 'square root' of Δ_g . Such an operator does not exist on $L_2(M)$, but it does on the Hilbert space of square-integrable sections of the spinor bundle of so-called 'spin manifolds'. An oriented Riemannian manifold is called *spin* if there exists a complex Hermitian vector bundle $S \to M$ (the spinor bundle) such that End(S) is isomorphic to a certain Clifford algebra, and there exists a charge conjugation on S [LM89, Chapter II][GVF01, Chapter 9][Sui25, Chapter 4]. The precise details do not matter much for this thesis, what is important is that the triple then takes the form $(C^{\infty}(M), L_2(S), D_S)$, where $L_2(S)$ are the square-integrable sections of Sand D_S is the Dirac operator associated with S. The terminology 'even' spectral triple is motivated by the fact that for an even-dimensional spin manifold M, the canonical spectral triple $(C^{\infty}(M), L_2(S), D_S)$ admits a natural grading γ_M making it an even spectral triple [GVF01, Chapter 9][Sui25, Chapter 4].

The claim that D_S functions as a square root of the Laplacian is motivated by the Lichnerowicz formula [LM89, Theorem II.8.8][GVF01, Theorem 9.16]

$$D_S^2 = -\Delta_S + \frac{1}{4}s,$$

where Δ_S is the Laplacian associated with the spin bundle S and s is the scalar curvature of M. In line with this formula, for general spectral triples $(\mathcal{A}, \mathcal{H}, D)$, the operator D^2 is commonly interpreted as an analogue of a Laplace operator.

In light of Section 1.1.1 it is not surprising that the canonical spectral triple of a compact Riemannian spin manifold is a unique invariant for the spin manifold. However, for spectral triples we even have an explicit *reconstruction* theorem [Con13]. From an abstract unital spectral triple $(\mathcal{A}, \mathcal{H}, D)$ with specific additional properties like \mathcal{A} being commutative, one can construct a compact spin manifold M which realises the spectral triple concretely as $(C^{\infty}(M), L_2(S), D_S)$. This highly non-trivial result requires recognising when the *-algebra \mathcal{A} is of the form $C^{\infty}(M)$ for a smooth manifold M, using only how \mathcal{A} and Dare situated relative to each other as operators on the abstract Hilbert space \mathcal{H} . This puts solid ground under the claim that spectral triples can be considered noncommutative generalisations of (spin) manifolds.

Interest in NCG has not been limited to mathematics. In physics, noncommutative spaces have been used to explain features of the quantum Hall effect [BES94] and topological insulators [Sch16], and there is extensive literature on applications of NCG in Quantum Field Theory, see e.g. [Sza03; CM08; Sui25] for overviews. The Standard Model of particle physics, minimally coupled to gravity, can be entirely derived through an action principle [CC96] from the description of the universe as a noncommutative space [CCM07;

CHAPTER 1. INTRODUCTION

Sui25]. This unified geometrical treatment of both Quantum Field Theory and General Relativity makes for an attractive approach to quantum gravity, and extends beyond the Standard Model to, for example, the Pati–Salam model [CCS13; CCS15].

Wrapping up this overview of NCG, we will now give two examples of noncommutative spaces which will appear throughout this dissertation. This exposition is taken from the joint paper of the author with Edward McDonald [HM24b]; for more details on the noncommutative torus (on which there is much literature) we refer to [HLP19a; HLP19b] and [GVF01, Section IV.12.3], for details on almost commutative manifolds to [Sui25, Chapter 10].

Example 1.1.4. Let $d \ge 2$ and let θ be a real $d \times d$ antisymmetric matrix. The noncommutative torus is the universal C^* -algebra $C(\mathbb{T}^d_{\theta})$ generated by a family of unitary elements $\{u_n\}_{n\in\mathbb{Z}^d}$ subject to the relations

$$u_n u_m = e^{\frac{i}{2} \langle n, \theta m \rangle} u_{n+m}, \quad n, m \in \mathbb{Z}^d.$$

The functional

$$\tau_{\theta}\left(\sum_{k\in\mathbb{Z}^d}c_ku_k\right):=c_0$$

extends to a continuous faithful tracial state on $C(\mathbb{T}^d_{\theta})$. The smooth subspace $C^{\infty}(\mathbb{T}^d_{\theta})$ is the subalgebra of $x \in C(\mathbb{T}^d_{\theta})$ for which $\hat{x}(k) = \tau_{\theta}(xu_k^*)$ is a rapidly decaying sequence on \mathbb{Z}^d . The Hilbert space in the GNS representation corresponding to τ_{θ} is denoted $L_2(\mathbb{T}^d_{\theta})$, and $\{u_n\}_{n\in\mathbb{Z}^d}$ is an orthonormal basis for $L_2(\mathbb{T}^d_{\theta})$. The self-adjoint operators D_j , $j = 1, \ldots, d$ on $L_2(\mathbb{T}^d_{\theta})$ are defined on the basis by

$$D_j u_k := k_j u_k, \quad k = (k_1, \dots, k_d) \in \mathbb{Z}^d.$$

The operator $D = \sum_{j=1}^{d} D_j \otimes \gamma_j$ on $L_2(\mathbb{T}^d_{\theta}) \otimes \mathbb{C}^{N_d}$, where γ_j are standard Clifford matrices on \mathbb{C}^{N_d} with $N_d = 2^{\lfloor \frac{d}{2} \rfloor}$, gives a spectral triple

$$(C^{\infty}(\mathbb{T}^d_{\theta}), L_2(\mathbb{T}^d_{\theta}) \otimes \mathbb{C}^{N_d}, D),$$

where we represent $C^{\infty}(\mathbb{T}^d_{\theta})$ as operators on $L_2(\mathbb{T}^d_{\theta}) \otimes \mathbb{C}^{N_d}$ by acting on the first component [GVF01, Section 12.3]. We write $\Delta := -\sum_{j=1}^d D_j^2$ as an operator on $L_2(\mathbb{T}^d_{\theta})$, so that $|D| = \sqrt{-\Delta} \otimes 1_{\mathbb{C}^{N_d}}$. **Example 1.1.5.** Given the canonical spectral triple $(C^{\infty}(M), L_2(S), D_M)$ of an evendimensional spin manifold with natural grading γ_M , and a finite spectral triple $(\mathcal{A}_F, \mathcal{H}_F, D_F)$, meaning that \mathcal{H}_F and \mathcal{A}_F are finite-dimensional, the product spectral triple

$$(C^{\infty}(M) \otimes \mathcal{A}_F, L_2(S) \otimes \mathcal{H}_F, D_M \otimes 1 + \gamma_M \otimes D_F).$$

is called an almost-commutative manifold.

1.2 Pseudodifferential operators

Pseudodifferential operators naturally emerged from the study of partial differential equations (PDEs) [KN65; Hör67], where they became a crucial tool in determining qualitative properties of solutions to PDEs. For us, they are indispensable in obtaining Weyl laws and in determining coefficients in heat trace expansions like those covered in Section 1.1. See [Tay81; Shu01; Hör07] for classic books on the subject.

Let us loosely sketch a path leading to the definition of a classical pseudodifferential operator, with no respect for technical details. Suppose we want to study a PDE on Euclidean space,

$$\sum_{|\alpha| \le k} a_{\alpha}(x) \partial^{\alpha} u(x) = f(x), \quad x \in \mathbb{R}^d,$$

for nice enough functions f and a_{α} . The operator $L := \sum_{|\alpha| \le k} M_{a_{\alpha}} \partial^{\alpha}$, complicated as it is, is much easier to study after a Fourier transform. For the Fourier transform, we will use the convention

$$\hat{u}(\xi) := \mathcal{F}(u)(\xi) := \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{-i\langle x,\xi\rangle} u(x) \, dx, \quad \xi \in \mathbb{R}^d, u \in L_2(\mathbb{R}^d).$$

In the frequency domain, the operator L then corresponds to multiplying with the polynomial $p_L(x,\xi) := \sum_{|\alpha| \le k} a_{\alpha}(x) (i\xi)^{\alpha}$,

$$Lu(x) = \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{i\langle x,\xi\rangle} p_L(x,\xi) \hat{u}(\xi) d\xi.$$

In an ideal scenario, when $p_L(x,\xi) = p_L(\xi)$ is non-zero and independent of x, we can simply solve the PDE from before by putting

$$u(x) = L^{-1}f(x) = \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{i\langle x,\xi\rangle} \frac{1}{p_L(\xi)} \hat{f}(\xi) \, d\xi.$$

Likewise, the heat equation

$$Lu(t, x) = \partial_t u(t, x);$$
$$u(0, x) = f(x),$$

can then be solved by putting

$$u(t,x) = e^{tL} f(x) := \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{i\langle x,\xi\rangle} e^{tp_L(\xi)} \hat{f}(\xi) \, d\xi.$$

The idea behind pseudodifferential operators is that even though these operators L^{-1} and e^{tL} are not differential operators, the analytical properties of an operator of the form

$$T_a u(x) = \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{i\langle x,\xi\rangle} a(x,\xi) \hat{u}(\xi) \,d\xi,$$

depend not so much on its symbol *a* being a *polynomial*, but rather on the qualitative properties of $a(x,\xi)$ like its asymptotic growth as $|\xi| \to \infty$. Studying properties of these more general operators, and deriving similar norm and regularity estimates as for differential operators, lets us conclude properties about solutions to PDEs, as the examples L^{-1} and e^{tL} above show.

With the theory of pseudodifferential operators, it can be made precise when and in what ways the operator L^{-1} behaves like an order -m 'differential' operator, e^{tL} an operator of order $-\infty$, and an operator like $(1-\Delta)^{\frac{s}{2}}$ for $s \in \mathbb{R}$ an operator of order s.

The oldest and most 'standard' theory of pseudodifferential operators is that of classical pseudodifferential operators on \mathbb{R}^d [KN65; Hör67]. Its definition is quite technical; we follow [Tay81, Chapter II], and write $\langle \xi \rangle := (1 + |\xi|^2)^{\frac{1}{2}}$.

Definition 1.2.1. The symbol class $S_{1,0}^m(\mathbb{R}^d)$, $m \in \mathbb{R}$, is defined as those smooth functions $a \in C^{\infty}(\mathbb{R}^d \times \mathbb{R}^d)$ such that for all multi-indices $\alpha, \beta \in \mathbb{N}^d$, there exists a constant $C_{\alpha,\beta} \geq 0$ such that

$$|\partial_x^{\alpha}\partial_{\xi}^{\beta}a(x,\xi)| \le C_{\alpha,\beta}\langle\xi\rangle^{m-|\alpha|}, \quad x,\xi \in \mathbb{R}^d.$$

We say that $a \in S_{1,0}^m(\mathbb{R}^d)$ is a classical symbol, denoted $a \in S_{cl}^m(\mathbb{R}^d)$, if there exist smooth functions $a_{m-j} \in C^\infty(\mathbb{R}^d \times \mathbb{R}^d)$ for all $j \in \mathbb{N}$ which are homogeneous of order m-j,

$$a_{m-j}(x,\lambda\xi) = \lambda^{m-j}a_{m-j}(x,\xi), \quad \lambda, |\xi| \ge 1,$$

with the property that

$$a(x,\xi) \sim \sum_{j \ge 0} a_{m-j}(x,\xi),$$

meaning that

$$a(x,\xi) - \sum_{j=0}^{N} a_{m-j}(x,\xi) \in S_{1,0}^{m-N-1}(\mathbb{R}^d), \quad N \in \mathbb{N}.$$

It is no coincidence that for the differential operator L from before, if each coefficient $a_{\alpha}(x)$ is smooth and it and its derivatives are bounded, the symbol $p_L(x,\xi) = \sum_{|\alpha| \le k} a_{\alpha}(x)(i\xi)^{\alpha}$ is in the symbol class $S_{1,0}^k(\mathbb{R}^d)$. It is furthermore a classical symbol, as we can write

$$p_L(x,\xi) = \sum_{j=0}^k \sum_{|\alpha|=j} a_\alpha(x)(i\xi)^\alpha$$

where each component $\sum_{|\alpha|=j} a_{\alpha}(x)(i\xi)^{\alpha}$ is homogeneous of order j.

Definition 1.2.2. The Schwartz functions $S(\mathbb{R}^d)$ are defined as those $f \in C^{\infty}(\mathbb{R}^d)$ for which all seminorms

$$||f||_{\alpha,\beta} := \sup_{x \in \mathbb{R}^d} |x^{\alpha} \partial_x^{\beta} f(x)|, \quad \alpha, \beta \in \mathbb{N}^d,$$

are finite.

Definition 1.2.3. A pseudodifferential operator of order $m \in \mathbb{R}$, denoted $T \in \Psi^m(\mathbb{R}^d)$, is an operator $T : \mathcal{S}(\mathbb{R}^d) \to \mathcal{S}(\mathbb{R}^d)$ such that

$$Tf(x) = \frac{1}{(2\pi)^{\frac{d}{2}}} \int_{\mathbb{R}^d} e^{i\langle x,\xi\rangle} a(x,\xi) \hat{f}(\xi) \, d\xi, \quad f \in \mathcal{S}(\mathbb{R}^d),$$

for some $a \in S_{1,0}^m(\mathbb{R}^d)$ — see [Hör07, Theorem 18.1.6] for a proof that $Tf \in \mathcal{S}(\mathbb{R}^d)$. If furthermore $a \in S_{cl}^m(\mathbb{R}^d)$, we write $T \in \Psi_{cl}^m(\mathbb{R}^d)$, and call T a classical pseudodifferential operator. Finally,

$$\Psi^{\infty}(\mathbb{R}^d) := \bigcup_{m \in \mathbb{R}} \Psi^m(\mathbb{R}^d), \quad \Psi^{-\infty}(\mathbb{R}^d) := \bigcap_{m \in \mathbb{R}} \Psi^m(\mathbb{R}^d).$$

Similarly we define $\Psi_{cl}^{\infty}(\mathbb{R}^d) := \bigcup_{m \in \mathbb{R}} \Psi_{cl}^m(\mathbb{R}^d).$

With these constructions, in many ways the pseudodifferential operators $\Psi^{\infty}(\mathbb{R}^d)$ behave like differential operators, and $\Psi^{\infty}_{cl}(\mathbb{R}^d)$ mimics this behaviour even more closely. For $S \in \Psi^m(\mathbb{R}^d)$ and $T \in \Psi^{m'}(\mathbb{R}^d)$, we have

$$S \circ T \in \Psi^{m+m'}(\mathbb{R}^d), \quad [S,T] \in \Psi^{m+m'-1}(\mathbb{R}^d),$$

as is the case for differential operators. Furthermore, we can define the Bessel-potential Sobolev spaces (see e.g. [Hör90, Section VII.9] or [Tri78, Section 2.3.3])

$$\mathcal{H}^s := \overline{\operatorname{dom}(1-\Delta)^{\frac{s}{2}}}^{\|\cdot\|_s}, \quad \|f\|_s := \|(1-\Delta)^{\frac{s}{2}}f\|_{L_2(\mathbb{R}^d)}, \quad s \in \mathbb{R},$$

which form Hilbert spaces. To be very precise, we take Δ to be the Friedrich's extension of the Laplacian, and dom $(1 - \Delta)^{\frac{s}{2}}$ indicates the domain of the corresponding operator on $L_2(\mathbb{R}^d)$ defined via functional calculus [Sch12, Section 5.3]. For $s \geq 0$, this domain is complete in the norm $\|\cdot\|_s$, but for s < 0 taking the completion in the norm $\|\cdot\|_s$ is necessary. For positive integer s, the space \mathcal{H}^s consists of the s times weakly differentiable L_2 -functions. For the differential operator $L = \sum_{|\alpha| \leq k} M_{a_{\alpha}} \partial^{\alpha}$ as above, if all coefficients a_{α} and their derivatives are uniformly bounded, then L extends to a bounded operator

$$L: \mathcal{H}^{s+k} \to \mathcal{H}^s, \quad s \in \mathbb{R}.$$

Pseudodifferential operators have the same property: if $T \in \Psi^m(\mathbb{R}^d)$, then T extends to a bounded operator

$$T: \mathcal{H}^{s+m} \to \mathcal{H}^s, \quad s \in \mathbb{R}.$$

Intuitively, T therefore 'removes' m degrees of regularity from a function. Note, however, that $m \in \mathbb{R}$ need not be an integer — or even positive.

All these objects can similarly be constructed on smooth (not necessarily Riemannian) manifolds [Shu01, Section I.4][Hör07, Chapter XVIII]. This takes some care, as we need to think about how to handle the Fourier transform on manifolds. Commonly, pseudodifferential operators are defined locally in coordinate patches via the definitions for \mathbb{R}^d above, and stitched together to define global operators on the manifold. However, symbols can be defined intrinsically as functions on the cotangent bundle as well [Wid78; Wid80; Saf97]. The principal symbol of a pseudodifferential operator is of particular importance, as it can tell us a lot about the operator's (approximate) invertibility. For compact manifolds, we have a short exact sequence of vector spaces

$$0 \to \Psi_{cl}^{m-1}(M) \to \Psi_{cl}^m(M) \xrightarrow{\sigma_0} C^{\infty}(S^*M) \to 0,$$

where the map σ_0 is called the symbol map, and $S^*M \subseteq T^*M$ is the cotangent sphere. The manifold M is not required to be equipped with a Riemannian metric in order for this to make sense: the cotangent sphere S^*M can be defined independently of such a metric as the rays in the cotangent bundle $T^*M \setminus M$, which is T^*M with the zero vectors removed. The intuitive explanation of this exact sequence is that an essentially homogeneous function $f \in C^{\infty}(T^*M)$ of order m, which satisfies locally (up to a compactly supported error),

$$f(x,\lambda\xi) = \lambda^m f(x,\xi), \quad \lambda, |\xi| \ge 1,$$

clearly defines a function on S^*M . The map σ_0 in the exact sequence selects the part of the symbol of the classical pseudodifferential operator that is essentially homogeneous of order m and produces the corresponding function on S^*M .

Turning this around, such exact sequences can be used to *define* principal symbols. For example, since on compact manifolds the operators $\Psi^{-1}(M)$ are compact, a map like σ_0 above, now mapping from order zero operators into the Calkin algebra $B(\mathcal{H})/K(\mathcal{H})$, is the basis for a paradigm called the C^* -algebraic approach to the principal symbol [Cor79; SZ18; MSZ19; KSZ24a].

Furthermore, it should be emphasised that there are by now many different pseudodifferential calculi, in different settings, for different purposes, all generalising differential operators in a different way. Fundamentally, these various pseudodifferential calculi are crafted for studying different classes and aspects of PDEs. On both Euclidean space and on manifolds, various symbol classes have been used as a base (variations of Definition 1.2.1) [BF73; Bon13; NR10]. Other directions of research include pseudodifferential calculi on the Heisenberg group [BFG12; FR14], Lie groups [Mel83; Tay84; MS86; RT10], Lie groupoids [MP97; NWX99], and types of manifolds like manifolds with boundaries [Mel91; Mel93; MM98], Heisenberg manifolds [BG88; Tay84; Pon08], and filtered manifolds [EY19; AMY22; FFF24].

In this thesis, we will focus on a very general abstract pseudodifferential calculus which strips away any reference to a geometric space. It is the purely operator theoretical skeleton at the basis of any pseudodifferential calculus, and as such it is difficult to give accurate credits towards its invention. Elements of this theory even *predate* the study of pseudodifferential operators via the symbol calculus by Kohn–Nirenberg [KN65] and Hörmander [Hör67]. The abstract Sobolev spaces in this theory already appeared in their current form in work by S. G. Krein where they are called Hilbert scales, see the reviews [Mit61; KP66] and references therein. Later, Guillemin [Gui85], Connes–Moscovici [CM95] and Higson [Hig04] amongst others have used general pseudodifferential calculi. Of these, Connes and Moscovici introduced the formalism in NCG.

The basic principle is as follows. We take one operator Θ as reference, a self-adjoint boundedly invertible operator on a separable Hilbert space \mathcal{H} . With this operator, define the Sobolev spaces

$$\mathcal{H}^{s}(\Theta):=\overline{\mathrm{dom}\,\Theta^{s}}^{\|\cdot\|_{s}}, \quad \|\xi\|_{s}:=\|\Theta^{s}\xi\|_{\mathcal{H}}, \quad s\in\mathbb{R},$$

clearly inspired by the Bessel potential Sobolev spaces from before. And as before, here dom Θ^s is defined via the functional calculus for unbounded self-adjoint operators [Sch12, Section 5.3], which is complete in the norm $\|\cdot\|_s$ for $s \ge 0$ but not for s < 0. Our 'pseudodifferential operators' are simply operators on \mathcal{H} which extend or restrict to bounded maps

$$T: \mathcal{H}^{s+r}(\Theta) \to \mathcal{H}^s(\Theta), \quad s \in \mathbb{R},$$

in a consistent and well-defined way. These operators are denoted by $op^r(\Theta)$, sometimes called operators of analytic order (less than) r [Hig04].

In this calculus, there is no notion of symbols, which is at once a strength and a weakness: we will not be able to use any techniques based on symbol manipulations, but as a consequence results are valid in any pseudodifferential calculus. And, importantly, this makes the pseudodifferential calculus perfectly adapted for noncommutative geometry where we do not want to interact directly with a geometric base space in the first place.

Some properties of this calculus have been compiled in [Uuy11]. The following concrete examples of this abstract pseudodifferential calculus are borrowed from the exposition in [HMN24] (joint work of the author with Edward McDonald and Teun van Nuland).

- As remarked before, taking $\Theta = (1 \Delta)^{\frac{1}{2}}$ on $L_2(\mathbb{R}^d)$ gives the classical (Bessel potential) Sobolev spaces $\mathcal{H}^s = W_2^s(\mathbb{R}^d)$. (Pseudo)differential operators of order k, and (unbounded) Fourier multipliers T_{ϕ} with symbols $|\phi(\xi)| \leq (1 + |\xi|)^k$, $\xi \in \mathbb{R}^d$ are contained in $\operatorname{op}^k(1 \Delta)^{\frac{1}{2}}$. Note though that $\operatorname{op}(1 \Delta)^{\frac{1}{2}}$ is a much larger class than this, and for example also contains translation operators.
- For spectral triples $(\mathcal{A}, \mathcal{H}, D)$, taking $\Theta = (1 + D^2)^{\frac{1}{2}}$ recovers the calculus of Connes and Moscovici as used in noncommutative geometry [CM95].
- Taking $\Theta = (1 \Delta)^{\frac{1}{2}}$, where Δ is the sub-Laplacian on a stratified Lie group, gives the Sobolev spaces defined by Folland and Stein [FS74; RS76].
- Related to the previous example is the anharmonic oscillator $\Theta^2 = 1 + \Delta^{2l} + |x|^{2k}$ on \mathbb{R}^d for integers $l, k \geq 1$ and generalisations thereof, which define Sobolev spaces and a pseudodifferential calculus that appear in the study of sub-Laplacian operators on stratified Lie groups too [CDR21]. The special case where Θ^2 is the harmonic oscillator gives Shubin's Sobolev spaces $Q^s(\mathbb{R}^d)$ [Shu01, Section IV.25] and the Γ pseudo-differential operators in [NR10, Chapter 2], see also [BT06].
- In the aforementioned pseudodifferential calculus in Van Erp–Yuncken [EY19], and Androulidakis–Mohsen–Yuncken [AMY22] on filtered manifolds, Sobolev spaces are constructed with a similar procedure.
- A similar calculus has been constructed for quantum Euclidean spaces [GJM22].
- Finally, we note the case where $\Theta = 1_{\mathcal{H}}$, which gives that $\mathcal{H}^s = \mathcal{H}$ for all $s \in \mathbb{R}$, and $\operatorname{op}^r(1_{\mathcal{H}}) = B(\mathcal{H}), r \in \mathbb{R}$.

It is a basic fact that the square of an operator in $\operatorname{op}^m(\Theta)$ is an operator in $\operatorname{op}^{2m}(\Theta)$, but what can be said of its square root? More generally, one can wonder for which type of abstract pseudodifferential operators and class of functions there exists a well-behaved functional calculus within the framework of this abstract pseudodifferential calculus. In Chapter 2, we resolve this question by defining abstract *elliptic* pseudodifferential operators, and provide a functional calculus for these operators based on the preprint [HMN24].

1.3 Multiple operator integrals

To an extent, we have physicists to thank for the field of multiple operator integrals. To see their ingenuity in manipulating operator integrals, simply open any book on Quantum Field Theory. For example, chances are that you will encounter the time-dependent operator

$$V(t) = e^{itH_0}e^{-it(H_0 + H_I)}, \quad t \in \mathbb{R},$$

where H_0 and H_I are unbounded self-adjoint operators on a Hilbert space, which is expanded by the Dyson series [Fol08, Chapter 6]

$$V(t) \sim I + \sum_{n=1}^{\infty} V_n(t),$$

where

$$V_n(t) = \frac{1}{i^n} \int_0^t \int_0^{\tau_n} \cdots \int_0^{\tau_2} e^{i\tau_n H_0} H_I e^{i(\tau_{n-1} - \tau_n)H_0} H_I \cdots e^{i(\tau_1 - \tau_2)H_0} H_I e^{-i\tau_1 H_0} d\tau_1 \cdots d\tau_n.$$

Of course, the symbols '=' and '~' in these formulas mean something different to physicists than they do to mathematicians, and making these identities mathematically rigorous requires some effort. The subject of Multiple Operator Integration provides a rigorous foundation for identities like the Dyson series above and generalisations thereof. We will mostly follow [Pel16; ST19] in this overview, but see also [PWS02; BS03; ACDS09] for important developments of the subject.

Suppose that we are interested in the Gâteaux derivative

$$\left. \frac{d}{dt} \right|_{t=0} f(A+tB) = \lim_{h \to 0} \frac{f(A+hB) - f(A)}{h},$$

for $A, B \in B(\mathcal{H})$ and some holomorphic function $f : \mathbb{C} \to \mathbb{C}$ (using the holomorphic functional calculus), taking the limit with respect to the operator norm. Picking a contour γ in the complex plane which encircles counterclockwise the union of the spectra of A + tBfor $t \in (-\varepsilon, \varepsilon)$, we have

$$f(A+tB) = \frac{1}{2\pi i} \int_{\gamma} f(z)(z-A-tB)^{-1} dz, \quad t \in (-\varepsilon, \varepsilon),$$

as a $B(\mathcal{H})$ -valued Bochner integral. Using the resolvent identity, we have that

$$\frac{d}{dt}\Big|_{t=0} (z - A - tB)^{-1} = \lim_{t \to 0} \frac{(z - A - tB)^{-1} - (z - A)^{-1}}{t}$$
$$= \lim_{t \to 0} (z - A - tB)^{-1} B(z - A)^{-1}$$
$$= (z - A)^{-1} B(z - A)^{-1}.$$

Similarly, for the holomorphic function f above we can see that

$$\frac{1}{n!} \frac{d^n}{dt^n} \bigg|_{t=0} f(A+tB) = \frac{1}{2\pi i} \int_{\gamma} f(z)(z-A)^{-1} B(z-A)^{-1} B\cdots (z-A)^{-1} dz, \qquad (1.3)$$

where we have an alternating product of n + 1 factors of $(z - A)^{-1}$ and n factors of B. In case A and B commute, this formula simply reduces to

$$\left. \frac{d^n}{dt^n} \right|_{t=0} f(A+tB) = f^{(n)}(A)B^n.$$

When they do not, we can devise a way to describe the right-hand side of (1.3). First define the divided differences $f^{[n]}: \mathbb{C}^{n+1} \to \mathbb{C}$ by

$$f^{[0]}(\lambda) := f(\lambda),$$

$$f^{[n]}(\lambda_0, \dots, \lambda_n) := \frac{f^{[n-1]}(\lambda_0, \dots, \lambda_{n-1}) - f^{[n-1]}(\lambda_1, \dots, \lambda_n)}{\lambda_0 - \lambda_n}, \quad \lambda_0, \dots, \lambda_{n-1}, \lambda_n \neq \lambda_0 \in \mathbb{C},$$

defining $f^{[n]}$ with an appropriate limit when $\lambda_0 = \lambda_n$, so that

$$f^{[n]}(\lambda,\ldots,\lambda) = \frac{1}{n!}f^{(n)}(\lambda), \quad \lambda \in \mathbb{C}.$$

By similar resolvent manipulations as before, it can be shown that for our holomorphic function f, for a contour γ encircling $\lambda_0, \ldots, \lambda_n \in \mathbb{C}$ counterclockwise,

$$f^{[n]}(\lambda_0,\ldots,\lambda_n) = \frac{1}{2\pi i} \int_{\gamma} f(z)(z-\lambda_0)^{-1}\cdots(z-\lambda_n)^{-1} dz.$$

Hence, intuitively, the expression on the right-hand side of (1.3) is something like $f^{[n]}(A, \ldots, A)$ with factors of B spliced 'in-between' the n + 1 copies of A.

Richard Feynman dealt with the issue [Fey51] by defining a clever system of notation

$$\overset{2}{A}\overset{1}{B}\overset{3}{C} := BAC.$$

In this notation, we formally have

$$\frac{d^n}{dt^n}f(A+tB) = f^{[n]}(\overset{1}{A},\overset{3}{A},\ldots,\overset{2n+1}{A})\overset{2}{B}\overset{4}{B}\cdots\overset{2n}{B}.$$

Obviously, this needs serious work to make well-defined and rigorous, see [Fey51; Mas76][KM93, Appendix I][Jef04, Chapter 7]. Daletskii has made some historical comments on this matter [Dal98].

A different approach, originally developed for studying evolution equations in Hilbert spaces, was devised by Daletskii and S. G. Krein [DK51; DK56; Dal98], elements of which had already appeared independently in work by Löwner [Löw34]. Their work was continued by Birman–Solomyak [BS66; BS67; BS73; BS03], Pavlov [Pav71], Sten'kin [SS71; Ste77], and Peller [Pel06; Pel16]. See also [ACDS09; PWS02; ST19]. Given a function $\phi : \mathbb{R}^{n+1} \to \mathbb{C}$, they define operators corresponding to the formal expression

$$T_{\phi}^{H_0,\dots,H_n}(V_1,\dots,V_n) := \int_{\sigma(H_0)\times\dots\times\sigma(H_n)} \phi(\lambda_0,\dots,\lambda_n) \, dE_0(\lambda_0) V_1 dE_1(\lambda_1)\cdots V_n dE_n(\lambda_n),$$

where dE_i is the spectral measure of the self-adjoint operator H_i . While, like Feynman's approach, this too is only formal notation, in various settings it can be made precise. An important class of examples is the following [Pel06; Pel16]. Assume that $\phi : \mathbb{R}^{n+1} \to \mathbb{C}$ can be written as

$$\phi(\lambda_0, \dots, \lambda_n) = \int_{\Omega} a_0(\lambda_0, \omega) \cdots a_n(\lambda_n, \omega) \, d\nu(\omega), \qquad (1.4)$$

where (Ω, ν) is a finite measure space, and the functions $a_i : \mathbb{R} \times \Omega \to \mathbb{C}$ are bounded and measurable. Then we can define

$$T_{\phi}^{H_0,\dots,H_n}(V_1,\dots,V_n)\psi := \int_{\Omega} a_0(H_0,\omega)V_1a_1(H_1,\omega)\cdots V_na_n(H_n,\omega)\psi\,d\nu(\omega), \quad \psi \in \mathcal{H},$$
(1.5)

as an \mathcal{H} -valued Bochner integral, where $V_1, \ldots, V_n \in B(\mathcal{H})$ and H_0, \ldots, H_n are selfadjoint. Peller proved that this defines a well-defined bounded operator, independent of the representation (1.4) chosen [Pel06; Pel16]. Furthermore, he proved that for functions $f : \mathbb{R} \to \mathbb{C}$ in the Besov space $B^n_{\infty,1}(\mathbb{R})$, we have that the divided difference $f^{[n]}$ is of the appropriate form and hence $T^{H_0,\ldots,H_n}_{f^{[n]}}(V_1,\ldots,V_n)$ can be defined. If furthermore $f \in B^n_{\infty,1}(\mathbb{R}) \cap B^1_{\infty,1}(\mathbb{R})$ and $f^{[n]}$ is bounded, the formula

$$\frac{1}{n!} \frac{d^n}{dt^n} \bigg|_{t=0} f(A+tB) = T^{A,\dots,A}_{f^{[n]}}(B,\dots,B)$$

can be justified for unbounded self-adjoint A and bounded self-adjoint B [Pel06, Theorem 5.6]. The operators $T_{\phi}^{H_0,\ldots,H_n}(V_1,\ldots,V_n)$ are called multiple operator integrals, and when n = 1 they are often called double operator integrals.

Multiple and double operator integrals (MOIs and DOIs) are not only useful for studying derivatives, but they also come with the identities

$$[f(A), B] = T_{f^{[1]}}^{A, A}([A, B]),$$

$$f(A) - f(B) = T_{f^{[1]}}^{A, B}(A - B),$$
(1.6)

as well as higher-order analogues. This indicates that DOIs and MOIs can appear in many settings, and this framework has indeed been a powerful formalism in functional analysis. The technique has been crucial in Potapov and Sukochev's proof of the longstanding conjecture that Lipschitz functions are operator Lipschitz on the Schatten classes \mathcal{L}_p , 1 [PS11], that is, the estimate

$$||f(A) - f(B)||_p \le C_{p,f} ||A - B||_p, \quad 1$$

The case 0 has been studied in [MS22]. DOIs have similarly been used to providedifferent kinds of sharp operator estimates [AP10; AP16; CMPS14; CPSZ19]. Furthermore, MOIs are used in the study of spectral flow and the spectral shift function [ACS07;PSS13], and in many other areas of functional analysis like Biane–Speicher's stochasticanalysis and their free Itô formula [BS98; Nik22].

A useful consequence of the two identities (1.6) (and their higher-order analogues) is a

noncommutative Taylor expansion:

$$f(D+V) \sim \sum_{n,m=0}^{\infty} \sum_{m_1+\dots+m_n=m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \delta_D^{m_1}(V) \cdots \delta_D^{m_n}(V) f^{(n+m)}(D),$$
(1.7)

where $\delta_D(V) := [D, V]$, and C_{m_1,\dots,m_n} are some combinatorial constants specified below in Section 3.5. If the operators D and V happen to commute, this reduces to a standard Taylor expansion. These Taylor expansions should for now be interpreted as purely formal, but they have been made rigorous for classical pseudodifferential operators on manifolds and in Banach algebras [Pay07; Pay11; HL24]. We too will prove a precise version of the noncommutative Taylor expansion for pseudodifferential operators in Chapter 3, using the abstract pseudodifferential calculus described in Section 1.2.

For us, MOIs connect to the other themes of this dissertation due to their appearance in noncommutative geometry, which has at times gone unrecognised. Take for example the JLO cocycle [JLO88], which functions in NCG as the noncommutative analogue of the Chern character in differential geometry. Given an even spectral triple $(\mathcal{A}, \mathcal{H}, D)$ with grading γ , the JLO cocycle is a collection of functionals $(\phi^0, \phi^2, \phi^4, \ldots)$ where ϕ^n : $\mathcal{A}^{\otimes n+1} \to \mathbb{C}$ is defined as

$$\phi^n(a_0 \otimes a_1 \dots \otimes a_n) := \int_{\Sigma^n} \operatorname{Tr}(\gamma a^0 e^{-t_0 D^2} [D, a_1] e^{-t_1 D^2} \dots [D, a_n] e^{-t_n D^2}) dt.$$

Here, Σ_n is the *n*-simplex with normalised Lebesgue measure *dt*. Comparing this expression with (1.5), it is not difficult to see that we have

$$\phi^n(a_0, a_1, \dots, a_n) = \operatorname{Tr}(\gamma a_0 T_{f^{[n]}}^{D^2, \dots, D^2}([D, a_1], \dots, [D, a_n])),$$

where $f(x) = \exp(-x)$.

This cocycle is a cornerstone of the local index formula, a landmark result in NCG which generalises the Atiyah–Singer index theorem to noncommutative spaces. For this result by Connes and Moscovici [CM95], there exist by now several different proofs [Hig04; CPRS06a; CPRS06b; CPRS08; CPRS13]. Stemming from manipulations of the JLO cocycle, these papers contain many elaborate and technical arguments involving operator integrals. The theory of MOIs can potentially be used to simplify these arguments. In fact, the original statement of the local index formula already suggests this possibility. **Theorem 1.3.1** (Local index formula [CM95]). Let $(\mathcal{A}, \mathcal{H}, D)$ be an odd unital spectral triple with simple dimension spectrum, such that $D^{-1} \in \mathcal{L}_{p,\infty}$ for some p > 0. The following functionals $\phi^n : \mathcal{A}^{\otimes n+1} \to \mathbb{C}$ define a cocycle $(\phi^1, \phi^3, \phi^5, \ldots)$ in the (b, B)bicomplex of \mathcal{A} :

$$\phi^{n}(a_{0},\ldots,a_{n}) = \sqrt{2i} \sum_{q \ge 0, k_{j} \ge 0} c_{n,k,q} \tau_{q} \Big(a_{0} \delta_{D^{2}}^{k_{1}}([D,a_{1}]) \cdots \delta_{D^{2}}^{k_{n}}([D,a_{n}]) |D|^{-(n+2\sum k_{j})} \Big),$$

where $c_{n,k,q}$ are some constants, and $\tau_q(T) := \operatorname{res}_{z=0} z^q \operatorname{Tr}(T|D|^{-2z})$. The cohomology class of the cocycle $(\phi^1, \phi^3, \phi^5, \ldots)$ in $HC^{odd}(\mathcal{A})$ coincides with the cyclic cohomology Chern character $ch_*(\mathcal{A}, \mathcal{H}, D)$.

Even without knowing what a simple dimension spectrum, (B, b)-cocycles, cyclic cohomology, or the Chern character are, one can see that the cocycle defined in this theorem involves something resembling a noncommutative Taylor expansion (1.7). In fact, it more closely resembles an expansion of a single MOI, which we will see in Chapter 3 of this thesis (Proposition 3.5.2).

The occurrence of MOI arguments in NCG is not limited to proofs of the local index formula. They also appear in expansions of the spectral action [Skr14; Skr18; Sui11; NS22] and the heat trace (often studied for a noncommutative analogue of curvature) [CT11; FK12; FK13; CM14; DS15; LM16; KS18; Liu18; CF19; Liu22; NSZ25]. See [FK19] for a review of the topic of curvature in NCG. An important tool in the area of these curvature computations is the rearrangement lemma [CT11; CM14; Les17; HL24], which is closely related to MOIs. See [NSZ25] for an approach to curvature via multiple operator integrals.

The theory of MOIs can potentially be a useful tool in all these contexts in NCG, but there is a hurdle to be taken. This is best illustrated by seeing what happens when one takes a spectral triple $(\mathcal{A}, \mathcal{H}, D)$ and applies an identity like (1.6) when the relevant operators are D^2 and $a \in \mathcal{A}$ like for the JLO cocycle:

$$[f(D^2), a] = T_{f^{[1]}}^{D^2, D^2}([D^2, a]).$$

Such manipulations of MOIs are (implicitly) common in the literature, and indeed necessary to interpret the local index formula, Theorem 1.3.1, as a MOI expansion. The problem is that the operator $[D^2, a]$ appearing as an argument in the MOI on the righthand side is typically *unbounded* in NCG, and the theory of MOIs as in [Pel16; ST19] is only equipped to deal with bounded arguments. In the NCG literature, many ad-hoc arguments can be found to justify such integral manipulations [CPRS06a; CM95; Hig04], but an overarching theory is missing.

In Chapters 2 and 3 of this thesis, this is resolved by developing a theory of multiple operator integrals for abstract pseudodifferential operators (introduced in Section 1.2), based on joint work [HMN24] with Edward McDonald and Teun van Nuland. Our results extend those by Paycha [Pay11] who proved a noncommutative Taylor expansion for classical pseudodifferential operators on manifolds, and are comparable to those by Hartmann and Lesch [HL24] on Banach spaces. Our results are more general than those of Paycha, both in terms of the functions and the operators considered, and more closely adapted to operators appearing in NCG than those in [HL24]. Furthermore, while the Paycha, Hartmann and Lesch papers prove Taylor expansions, our fundamental construction of MOIs also systematises the underlying techniques involving operator integrals that are widespread in NCG, allowing to justify the arguments illustrated above involving unbounded operators. As a demonstration of the power and adaptability of this formalism, Chapter 3 resolves an open question regarding the existence of certain asymptotic expansions in spectral triples, posed by Eckstein and Iochum [EI18].

1.4 Connes' integral formula

The fact that Riemannian manifolds can be completely captured by spectral triples, indicates that there must be a way to integrate functions on manifolds spectrally. Alain Connes recognised that Dixmier traces, or more general singular traces, are the appropriate tool.

The Dixmier trace was originally constructed as an example by Dixmier [Dix66], which settled in the negative the question whether the operator trace is (up to constants) the only trace that can be defined on $B(\mathcal{H})$. With trace we mean a linear functional $\phi : \mathcal{J} \to \mathbb{C}$ on a two-sided ideal $\mathcal{J} \subseteq B(\mathcal{H})$ which vanishes on commutators,

$$\phi([A, B]) = 0, \quad A \in \mathcal{J}, B \in B(\mathcal{H})$$

See [SU16] for details on Dixmier's original construction. We will take a different approach, following [LSZ21; LMSZ23].

Definition 1.4.1. The weak Schatten class $\mathcal{L}_{p,\infty} \subseteq B(\mathcal{H})$, p > 0, consists of those compact operators A for which

$$\lambda(k, |A|) = O(k^{-\frac{1}{p}}), \quad k \to \infty,$$

where for a compact operator T we write $\{\lambda(k,T)\}_{k=0}^{\infty}$ for an eigenvalue sequence of T counted with algebraic multiplicity, ordered in decreasing modulus. These sets are two-sided ideals in $B(\mathcal{H})$. The ideal $\mathcal{L}_{1,\infty}$ is called the weak trace-class ideal.

Definition 1.4.2. An extended limit $\omega \in (\ell_{\infty}(\mathbb{N}))^*$ is a continuous positive linear functional on $\ell_{\infty}(\mathbb{N})$ such that $\omega(1) = 1$, and for each sequence $x \in \ell_{\infty}(\mathbb{N})$ that converges to zero, we have $\omega(x) = 0$.

Associated with each extended limit ω , we can define the Dixmier trace $\operatorname{Tr}_{\omega} : \mathcal{L}_{1,\infty} \to \mathbb{C}$ by

$$\operatorname{Tr}_{\omega}(A) := \omega \left(\left\{ \frac{1}{\log(n+2)} \sum_{k=0}^{n} \lambda(k,A) \right\}_{n=0}^{\infty} \right)$$

In general, the value of $\operatorname{Tr}_{\omega}(A)$ may depend on the choice of ω . If it does not, we call A Dixmier measurable. Note that there is some inconsistency in the literature on the definition of Dixmier measurable operators, we follow [LSZ21]. See also [SUZ13; Usa13].

At first sight, it might come as a surprise that the Dixmier traces are linear maps or even traces. It is furthermore a crucial fact that they vanish on the finite-rank operators (a forteriori, on \mathcal{L}_1), a property which characterises so-called singular traces.

There is a rich mathematical theory surrounding Dixmier traces and their more general cousins the singular traces. Originally Dixmier [Dix66], and later Connes [Con88][Con94, Section 4.2], required additional properties of the extended limits ω like dilation invariance in their definitions of the Dixmier trace, and defined these on a larger ideal $\mathcal{M}_{1,\infty}$ which

is now sometimes called the Macaev–Dixmier ideal. There are many subtle points to be made regarding these differing definitions, or the question when different extended limits define the same Dixmier trace, or what the sufficient and necessary conditions are for an operator to be Dixmier measurable and more. For these subtleties we refer to the books [LSZ21; LMSZ23] and the many references therein.

The most important result illustrating that Dixmier traces can be used for integration, comes from Connes' trace theorem [Con88]. This provides that the noncommutative residue (sometimes called Wodzicki residue or Wodzicki–Guillemin residue) for order -d classical pseudodifferential operators on d-dimensional manifolds can be given by a Dixmier trace.

If we equip M with a Riemannian metric g, and consider the operator $M_f(1 - \Delta_g)^{-\frac{a}{2}}$ on $L_2(M)$, where $f \in C_c^{\infty}(M)$ and Δ_g is the Laplace–Beltrami operator, Connes' trace theorem reduces to Connes' integral formula.

Theorem 1.4.3 (Connes' integral formula [Con88]). Let M be a d-dimensional Riemannian manifold. For all $f \in C_c^{\infty}(M)$, we have that $M_f(1 - \Delta_g)^{-\frac{d}{2}} \in \mathcal{L}_{1,\infty}$, and for all extended limits $\omega \in \ell_{\infty}^*$

$$\operatorname{Tr}_{\omega}(M_f(1-\Delta_g)^{-\frac{d}{2}}) = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d(2\pi)^d} \int_M f \, d\nu_g, \quad f \in C_c^{\infty}(M).$$
(1.8)

For compact manifolds and \mathbb{R}^d , this theorem admits a generalisation to *all* singular traces on $\mathcal{L}_{1,\infty}$ (not just Dixmier traces), and all $f \in L_2(M)$, see [LPS10; KLPS13], [LSZ21, Chapters 7, 8] and [LMSZ23, Chapters 2, 3]. Interestingly, $f \in L_2(M)$ is both necessary and sufficient for (1.8) to hold [LPS10; KLPS13], illustrating that an extension to $L_1(M)$ is impossible as mistakenly claimed in [GVF01, Corollary 7.22].

The state of the art for Connes' integral formula can be found in [LSZ20b], supplemented by [ZS23]. By modifying the left-hand side of (1.8) to the symmetric expression

$$\phi((1-\Delta_g)^{-\frac{d}{4}}M_f(1-\Delta_g)^{-\frac{d}{4}}),$$

where ϕ is a singular trace on the aforementioned Dixmier–Macaev ideal $\mathcal{M}_{1,\infty}$, these papers extend Connes' integral formula to functions f in certain function spaces which
include the space $L_p(M)$ for p > 1 (on a compact manifold M). Even in this approach, an extension to $L_1(M)$ is demonstrably impossible [LPS10, Lemma 5.7].

In the field of noncommutative geometry, Connes' integral formula is of significant philosophical importance. It justifies that for *d*-summable spectral triples $(\mathcal{A}, \mathcal{H}, D)$ (meaning that $(1+D^2)^{-\frac{d}{2}} \in \mathcal{L}_{1,\infty}$), the functional

$$a \mapsto \operatorname{Tr}_{\omega}(a(1+D^2)^{-\frac{a}{2}}), \quad a \in \mathcal{A}$$

can be interpreted as a noncommutative integral, which allows us to talk about integration on a noncommutative space. After all, for a spectral triple the operator D^2 functions as the analogue of the Laplace operator (see Section 1.1). We refer to [LS10; LSZ21; LMSZ23] for thorough studies of the noncommutative integral in this context.

The philosophy goes deeper, however. The integral formula shows that the Lebesgue measure can be completely 'recovered' from operator based computations with Dixmier traces. It has led Connes to craft a paradigm where he considers compact operators to be infinitesimals, with the weak trace-class $\mathcal{L}_{1,\infty}$ being infinitesimals of order 1, and the Dixmier trace being the integral [Con94]. This 'quantised calculus' is a completely operator-theoretical approach to integration theory. In the words of Alain Connes: "I believe that *most* integrals that we know of, are special cases of this" [Con21], i.e. are computable via a Dixmier trace.

These Dixmier trace formulas have some advantages over the usual approach to integration. Due to the nature of the Dixmier trace, properties like convergence or positivity of an integral are easy to see on the operator-theoretical side. Furthermore, its robustness towards trace-class perturbations (recall that $\text{Tr}_{\omega}(T) = 0$ if T is trace-class) can lead to deductions about similar robustness properties for the integral in question [Con94, p.558].

It should be remarked that in the noncommutative geometry literature, a popular alternative definition for the noncommutative integral is the functional

$$a \mapsto \int a(1+D^2)^{\frac{s}{2}} := \operatorname{res}_{s=0} \operatorname{Tr}(a(1+D^2)^{\frac{1+s}{2}}), \quad a \in \mathcal{A}.$$

Whenever $s \mapsto \zeta_{a,D}(s) := \text{Tr}(a(1+D^2)^{\frac{1+s}{2}})$ can be extended meromorphically to a neighbourhood of the origin, we have for all extended limits $\omega \in \ell_{\infty}^*$ [LSZ21, Theorem 9.1.5(a)]

$$\operatorname{res}_{s=0} \operatorname{Tr}(a(1+D^2)^{\frac{1+s}{2}}) = \operatorname{Tr}_{\omega}(a(1+D^2)^{\frac{1}{2}}).$$

In general, however, it need not be the case that $\zeta_{a,D}$ allows such an extension, in which case $\int a(1+D^2)^{\frac{1}{2}}$ is not well-defined. Counterexamples are given in [CS12, Lemma 17][KLPS13, Corollary 6.34, Corollary 7.23][LSZ21, Example 9.3.3], of which [KLPS13, Corollary 6.34, Corollary 7.23] is notable for providing an example on a *d*-dimensional manifold of a compactly supported pseudodifferential operator of order -d with such behaviour. See also [LSZ21, Chapters 8, 9] for extensive analysis on the relation between the Dixmier trace and residue formulas. Throughout this thesis, to maintain maximal generality, we will stick with the Dixmier trace formulation of the noncommutative integral.

In Chapter 4, the noncommutative integral is studied in relation to the paradigm of spectral truncations in NCG due to Connes and Van Suijlekom [CS21; CS22]. The result provides a vehicle to relate NCG to a field of mathematics called Quantum Ergodicity. In Chapters 5 and 6, we provide a Dixmier trace formula for integrals with respect to a measure called the density of states, on respectively discrete metric spaces and open manifolds based on the papers [Aza+22; HM24a], validating Connes' mantra in this context.

1.5 Density of states

Parts of this section are borrowed from [Aza+22; HM24a], respectively a joint work with Nurulla Azamov, Edward McDonald, Fedor Sukochev and Dmitriy Zanin, and a joint work with Edward McDonald.

Many electrical and thermal properties of a material can be deduced from its associated density of states (DOS), a construction in solid-state physics. Loosely speaking, the DOS describes how many quantum states are admitted for the electrons in the material at each energy level per unit volume. Over the years the topic has kept of plenty of mathematicians and mathematical physicists occupied, see for example the books [PF92; Ves08; AW15].

The rigorous definition of the DOS already takes some effort, and there exist differing approaches. In this thesis we follow Simon [Sim82, Section C]. Given a (possibly unbounded) lower-bounded self-adjoint operator H on the Hilbert space $L_2(X)$ where X is some metric space with a Borel measure written as $|\cdot|$, we consider the limits

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(f(H) M_{\chi_{B(x_0, R)}}), \quad f \in C_c(\mathbb{R}),$$

where $B(x_0, R)$ denotes the closed ball with center $x_0 \in X$ and radius R, and $\chi_{B(x_0,R)}$ is its indicator function. When these limits exist (this includes assuming that $f(H)M_{\chi_{B(x_0,R)}}$ is trace-class in the first place), we have a positive linear functional on $C_c(\mathbb{R})$ and hence, via the Riesz-Markov-Kakutani theorem, we obtain a Borel measure ν_H on \mathbb{R} [Sim82, Proposition C.7.2] such that

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(f(H) M_{\chi_{B(x_0, R)}}) = \int_{\mathbb{R}} f \, d\nu_H, \quad f \in C_c(\mathbb{R}).$$

This measure, if it exists, is what we call the density of states of the operator H. This essentially coincides with similar definitions elsewhere in the literature, at least for Schrödinger operators on Euclidean space [Sim82]. Important to observe is that while we define the DOS via increasing balls, in principle we could also consider taking another increasing sequence of sets $\{\Omega_n\}_{n\in\mathbb{N}}$. However, the DOS in general depends on the choice of this increasing sequence of sets, even for Schrödinger operators with radially homogenous potentials on \mathbb{R}^d [AMSZ22].

Apart from its origin in physics, the density of states can be considered as a substitute for the spectral counting function [Str12] which featured in Section 1.1,

$$N(\lambda) := \#\{k : \lambda_k \le \lambda\} = \operatorname{Tr}(\chi_{(-\infty,\lambda]}(H)),$$

in case H does not have discrete spectrum. Indeed, when X has finite diameter and $|X| < \infty$, every self-adjoint operator H with compact resolvent admits a DOS and we have

$$\nu_H(-\infty,\lambda] = \frac{N(\lambda)}{|X|}.$$

Common research areas for studying the DOS are its existence [AW15; BS91; CL90; DIM01; PF92; Shu79; Sim82], the analytical properties of the function $\lambda \mapsto \nu_H(-\infty, \lambda]$ [AW15;

BK13; CL90; PF92] and its asymptotic behaviour as λ approaches boundaries of the support of ν_H [AW15; BK13; CL90; Lan91; PF92]. The study of the Anderson localisation phenomenon, of interest to mathematical physicists, is inseparably related to the study of the density of states [AW15; CL90; Lan91].

The DOS can be connected to NCG, which will be the focus of the studies of the DOS in this dissertation. Connes' mantra that most integrals can be computed with Dixmier traces was mentioned in Section 1.4, and since the DOS is a measure associated with operators, it is a good case study for Connes' philosophy. This was first observed and investigated by Nurulla Azamov, Edward McDonald, Fedor Sukochev and Dmitriy Zanin on \mathbb{R}^d for Schrödinger operators [AMSZ22].

Let us consider the Laplacian on \mathbb{R}^d , $\Delta = \sum_{j=1}^d \partial_j^2$. We know that the classical Weyl law covered in Section 1.1 (equation (1.1)) gives for the Laplacian Δ_{Ω} on a bounded domain $\Omega \subseteq \mathbb{R}^d$ with Dirichlet boundary conditions, that

$$\operatorname{Ir}(\chi_{(-\infty,\lambda]}(-\Delta_{\Omega})) \sim \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d(2\pi)^d} \operatorname{Vol}(\Omega) \lambda^{\frac{d}{2}}, \quad \lambda \to \infty.$$

It might therefore not come as a surprise that we have the limit

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(\chi_{[-\infty, \lambda)}(-\Delta) M_{\chi_{B(x_0, R)}}) = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d(2\pi)^d} \lambda^{\frac{d}{2}}, \quad \lambda > 0,$$

where we now have an exact equality instead of an asymptotic identity. See [Str12] for a precise proof. In fact, the Laplacian admits a DOS $\nu_{-\Delta}$ which satisfies

$$\nu_{-\Delta}(-\infty,\lambda] = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d(2\pi)^d}\lambda^{\frac{d}{2}}, \quad \lambda > 0,$$
(1.9)

and more generally,

$$d\nu_{-\Delta}(\lambda) = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{2(2\pi)^d} \lambda^{\frac{d}{2}-1} d\lambda, \quad \lambda > 0,$$
(1.10)

where $d\lambda$ is the Lebesgue measure on \mathbb{R} , see e.g. [Sim82; Str12; AMSZ20]. The function (1.9) is sometimes called the integrated DOS.

The following argument was presented in [AMSZ22]. Recall Connes' integral formula (Theorem 1.4.3), which gives for any extended limit $\omega \in \ell_{\infty}^*$,

$$\operatorname{Tr}_{\omega}(M_f(1-\Delta)^{-\frac{d}{2}}) = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d(2\pi)^d} \int_{\mathbb{R}^d} f(x) \, dx, \quad f \in C_c(\mathbb{R}^d).$$

Since the Dixmier trace is invariant under unitary transformations, we can perform a Fourier transform on the left-hand side to deduce that

$$\operatorname{Tr}_{\omega}(f(-i\nabla)M_{\langle x\rangle}^{-d}) = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d(2\pi)^d} \int_{\mathbb{R}^d} f(x) \, dx, \quad f \in C_c(\mathbb{R}^d).$$

Suppose now that f is radially symmetric and $f(x) = g(|x|^2)$ for a continuous compactly supported function $g \in C_c[0,\infty)$. Then, $f(-i\nabla) = g(-\Delta)$, and switching to spherical coordinates we get

$$\operatorname{Tr}_{\omega}(g(-\Delta)M_{\langle x \rangle}^{-d}) = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d(2\pi)^d} \int_{\mathbb{R}^d} g(|x|^2) \, dx$$
$$= \frac{\left(\operatorname{Vol}(\mathbb{S}^{d-1})\right)^2}{d(2\pi)^d} \int_0^\infty g(r^2) r^{d-1} \, dr$$
$$= \frac{\left(\operatorname{Vol}(\mathbb{S}^{d-1})\right)^2}{2d(2\pi)^d} \int_0^\infty g(\lambda) \lambda^{\frac{d}{2}-1} \, d\lambda.$$

Hence, we find that

$$\operatorname{Tr}_{\omega}(g(-\Delta)M_{\langle x\rangle}^{-d}) = \frac{\operatorname{Vol}(\mathbb{S}^{d-1})}{d} \int_{\mathbb{R}} g(\lambda) \, d\nu_{-\Delta}(\lambda), \quad g \in C_c(\mathbb{R}),$$
(1.11)

which is a Dixmier trace formula for the DOS of the Laplacian. The results of [AMSZ22] extend the identity (1.11) to Schrödinger operators $-\Delta + M_V$ which admit a DOS and where $V \in L_{\infty}(\mathbb{R}^d)$ is real-valued.

Chapters 5 and 6 in this thesis focus on extending formula (1.11) to discrete metric spaces and manifolds respectively, based on the papers [Aza+22; HM24a]. The DOS in these settings has previously been studied, as discrete spaces can serve as a model or a discrete approximation [AW15; AO82; BK13; CL90; Cha+86; Hof93; KM06; PF92; Ves05; Weg81], and manifolds allow studying the DOS in even greater generality [AS93; LPPV08; LPV04; PV02; Ves08].

Furthermore, a link between the DOS and Dixmier traces has previously appeared in the influential work by Bellissard, van Elst and Schulz-Baldes [BES94], which provides an NCG explanation of the quantum Hall effect. Noting the relation between Dixmier traces and ζ -function residues [LSZ21], a result by Bourne and Prodan [BP18, Lemma 6.1] also bears some resemblance to formula (1.11).

CHAPTER 1. INTRODUCTION

A notorious aspect of the DOS is that the existence of the DOS cannot be guaranteed in general situations. One advantage of the Dixmier trace formula (1.11) is that the Dixmier trace on the left-hand side is guaranteed to be well-defined even if the DOS itself has not been shown to exist. In fact, it can be interpreted as a generalisation of the definition of the DOS. This is unlikely to be any good if this 'DOS' depends on the choice of extended limit, so one might at least require $g(H)M_{\langle x \rangle}^{-d}$ to be Dixmier measurable. In Chapter 5 we will confirm on discrete spaces that this is a strictly weaker requirement than existence of the DOS, but in other settings and for Schrödinger type operators it is not yet precisely known what the relation is between Dixmier measurability and existence of the DOS.

Finally, the Dixmier trace formula allows questions regarding the DOS to be answered with operator theoretical machinery. We will see examples of this in Chapters 5 and 6, where questions like translation invariance of the DOS are answered in this manner. We will also see a connection between the Dixmier trace formula for the DOS and index theory, namely, Roe's index theorem on open manifolds.

1.6 Preliminary material

This section serves as a brief overview of common notation, convention, and definitions used throughout the rest of this dissertation. The covered material is standard and can be found in [LSZ21; LMSZ23; Sim05]. Some of the content has been copied from [HM24a].

The natural numbers are by convention $\mathbb{N} = \{0, 1, 2, ...\}$. For functions $f, g : \mathbb{R} \to \mathbb{R}_{>0}$ (or with other appropriate domains and co-domains) we write

$$f(x) \sim g(x), \quad x \to \infty,$$

to mean that

$$\lim_{x \to \infty} \frac{f(x)}{g(x)} = 1.$$

All Hilbert spaces are assumed to be complex separable Hilbert spaces, denoted \mathcal{H} , on which the inner product $\langle \cdot, \cdot \rangle$ is linear in its second component. The set of bounded op-

erators on \mathcal{H} is denoted by $B(\mathcal{H})$ and the ideal of compact operators by $K(\mathcal{H})$. The operator norm on $B(\mathcal{H})$ is denoted $\|\cdot\|$. For any compact operator $T \in K(\mathcal{H})$, an eigenvalue sequence $\lambda(T) = \{\lambda(k,T)\}_{k=0}^{\infty}$ is a sequence of the eigenvalues of T listed with algebraic multiplicity, ordered such that $\{|\lambda(k,T)|\}_{k=0}^{\infty}$ is non-increasing. The singular value sequence $\mu(T) = \{\mu(k,T)\}_{k=0}^{\infty}$ of T is defined by

$$\mu(k,T) := \lambda(k,|T|), \quad k \ge 0.$$

Equivalently,

$$\mu(k,T) = \inf\{\|T - R\| : \operatorname{rank}(R) \le k\}.$$

Let ℓ_{∞} denote the space of complex-valued bounded sequences indexed by \mathbb{N} . For $x \in \ell_{\infty}$, we will denote by $\mu(x) = {\{\mu(k, x)\}_{k=0}^{\infty} \in \ell_{\infty} \text{ the decreasing rearrangement of } {||x_k||_{k=0}^{\infty}}$. This is consistent with the notation for the singular value sequence of an operator in the sense that if diag(x) is the operator given by a diagonal matrix with entries ${\{x_k\}_{k=0}^{\infty}}$, then $\mu(\text{diag}(x)) = \mu(x)$.

For $p, q \in (0, \infty)$, we define the Lorentz sequence spaces $\ell_{p,q}$, $\ell_{p,\infty}$ and $\ell_{\infty,q}$ as the spaces of sequences $x \in \ell_{\infty}$ such that

$$\|x\|_{p,q} := \left(\sum_{k=0}^{\infty} (k+1)^{\frac{q}{p}-1} \mu(k,x)^{q}\right)^{\frac{1}{q}} < \infty,$$
$$\|x\|_{p,\infty} := \sup_{k>0} (k+1)^{\frac{1}{p}} \mu(k,x) < \infty,$$

and

$$||x||_{\infty,q} := \left(\sum_{k=0}^{\infty} (k+1)^{-1} \mu(k,x)^q\right)^{\frac{1}{q}} < \infty,$$

respectively. The space $\ell_{\infty,\infty}$ is defined as $\ell_{\infty,\infty} := \ell_{\infty}$, and $\ell_{p,p}$ is denoted as ℓ_p . Using the previously defined singular value sequences, we give the definition of the Lorentz ideals $\mathcal{L}_{p,q}$ for $p,q \in (0,\infty]$ as the quasi-normed spaces of compact operators T such that

$$||T||_{p,q} := ||\mu(T)||_{\ell_{p,q}} < \infty.$$

Like for the sequence spaces, $\mathcal{L}_{p,p}$ is denoted as \mathcal{L}_p , and $\mathcal{L}_{\infty,\infty} := B(\mathcal{H})$. Indeed, consistent with these definitions, the operator norm $\|\cdot\|$ on $B(\mathcal{H})$ is sometimes denoted by $\|\cdot\|_{\infty}$ if confusion might arise with other norms.

The spaces $\mathcal{L}_{p,q}$, $0 < p,q \leq \infty$ form two-sided ideals in $B(\mathcal{H})$. The ideals \mathcal{L}_p are called Schatten classes, \mathcal{L}_1 is the familiar ideal of trace-class operators (on which we can define the usual operator trace Tr), and $\mathcal{L}_{1,\infty}$ is called the ideal of weak trace-class operators. The ideal \mathcal{L}_2 is called the Hilbert–Schmidt class. All of these quasi-norms have the property that, for $p,q \in (0,\infty]$,

$$\|ABC\|_{p,q} \le \|A\|_{\infty} \|B\|_{p,q} \|C\|_{\infty}, \quad B \in \mathcal{L}_{p,q}, A, C \in B(\mathcal{H}).$$

In similar spirit, we define the following function spaces.

Definition 1.6.1. Let (X, Σ, ν) be a measure space. We write $L_{\infty}(X)$ for the quotient space

$$L_{\infty}(X) := \left\{ f : X \to \mathbb{C} \ measurable \mid ||f||_{\infty} < \infty \right\} / \sim$$

where $\|\cdot\|_{\infty}$ indicates the essential supremum, and $f \sim g$ if $f(x) = g(x) \nu$ -almost everywhere. We define similarly for 0 ,

$$L_p(X) := \left\{ f: X \to \mathbb{C} \ \text{measurable} \ | \ \|f\|_p := \left(\int_X |f|^p \, d\nu \right)^{\frac{1}{p}} < \infty \right\} \Big/ \sim .$$

There is a way to define the spaces $L_{p,q}(X)$ as well, and to furthermore see the spaces $L_{p,q}(X)$, $\mathcal{L}_{p,q}$ and $\ell_{p,q}$ as instances of a more general integration theory on von Neumann algebras [DPS23]. We will not delve into that matter in this thesis though.

Let \mathcal{J} be a two-sided ideal in $B(\mathcal{H})$. A linear functional $\phi : \mathcal{J} \to \mathbb{C}$ is called a trace if it satisfies

$$\phi(BT) = \phi(TB), \quad T \in \mathcal{J}, B \in B(\mathcal{H}).$$

This condition is equivalent with requiring

$$\phi(U^*TU) = \phi(T), \quad T \in \mathcal{J}, U \in U(\mathcal{H}),$$

where $U(\mathcal{H})$ are the unitary operators on \mathcal{H} . Traces ϕ on \mathcal{J} that vanish on finite-rank operators called singular traces.

All traces on $\mathcal{L}_{1,\infty}$ are singular traces. In contrast with the situation on \mathcal{L}_1 , continuous traces on $\mathcal{L}_{1,\infty}$ are far from unique. A particular kind of continuous traces can be defined

on $\mathcal{L}_{1,\infty}$ which are called Dixmier traces, as covered in Section 1.4, repeated here for convenience.

Definition 1.6.2. An extended limit $\omega \in (\ell_{\infty}(\mathbb{N}))^*$ is a continuous positive linear functional on $\ell_{\infty}(\mathbb{N})$ such that $\omega(1) = 1$, and for each sequence $x \in \ell_{\infty}(\mathbb{N})$ that converges to zero, we have $\omega(x) = 0$.

Associated with each extended limit ω , we can define the Dixmier trace $\operatorname{Tr}_{\omega} : \mathcal{L}_{1,\infty} \to \mathbb{C}$ by

$$\operatorname{Tr}_{\omega}(A) := \omega \left(\left\{ \frac{1}{\log(n+2)} \sum_{k=0}^{n} \lambda(k,A) \right\}_{n=0}^{\infty} \right).$$

In general, the value of $\operatorname{Tr}_{\omega}(A)$ may depend on the choice of ω . If it does not, we call A Dixmier measurable [LSZ21].

More can be read about singular and Dixmier traces in [LSZ21; LMSZ23], for the topic of measurability see [LSZ21; Usa13; SUZ13].

Definition 1.6.3. Given an operator $0 \le V \in B(\mathcal{H})$, an operator $A \in B(\mathcal{H})$ is said to be *V*-modulated if

$$\sup_{t>0} t^{\frac{1}{2}} \|A(1+tV)^{-1}\|_2 < \infty.$$
(1.12)

As can be seen from the definition, a V-modulated operator is necessarily Hilbert–Schmidt. The importance of V-modulated operators comes from the following theorem, see [LSZ21, Theorem 7.1.3]. We will make use of this theorem in Chapters 4 and 5.

Theorem 1.6.4. If $V \in \mathcal{L}_{1,\infty}(\mathcal{H})$ is strictly positive, T is V-modulated and $\{e_k\}_{k=0}^{\infty}$ is an orthonormal basis such that $Ve_k = \mu(k, V)e_k$, then as $n \to \infty$,

$$\sum_{k=0}^{n} \lambda(k,T) = \sum_{k=0}^{n} \langle e_k, Te_k \rangle + O(1).$$
 (1.13)

Note that if $0 \leq V \in \mathcal{L}_{1,\infty}(\mathcal{H})$, then V is automatically V-modulated [LSZ21, Lemma 7.3.4], and if A is bounded and T is V-modulated, then AT is V-modulated, which directly follows from (1.12).

All manifolds in this thesis are smooth, oriented manifolds. The unique volume form on an (oriented) Riemannian manifold (M, g) is written as ν_g , which is given in local coordinates as

$$\nu_g = \sqrt{|\det(g)|} dx^1 \wedge \dots \wedge dx^d,$$

where d is the dimension of M. We then define $L_2(M)$ as in Definition 1.6.1. On $L_2(M)$ we will often consider the Laplace–Beltrami operator Δ_g , which is given in local coordinates as

$$\Delta_g f = \frac{1}{\sqrt{|\det(g)|}} \partial_i \left(\sqrt{|\det(g)|} g^{ij} \partial_j f \right), \quad f \in C_c^\infty(M).$$
(1.14)

When not mentioned otherwise, we will consider this operator on $L_2(M)$ as the Friedrichs extension of (1.14) [Sch12, Chapter 10]. Explicitly, we put

$$\operatorname{dom} \Delta_q = H^2(M),$$

where $H^k(M)$ are the standard Sobolev spaces on Riemannian manifolds, see e.g. [Tri92, Chapter 7]. With this domain, Δ_g is a negative self-adjoint operator on $L_2(M)$. Part I

The Joys of MOIs

Chapter 2

Functional calculus for abstract pseudodifferential operators

Take care, and work harder.

Fedor Sukochev

This chapter has appeared in slightly modified form as part of the preprint [HMN24], joint work with Edward McDonald and Teun van Nuland. We thank Dmitriy Zanin for help with Proposition 2.3.4. The review of the abstract pseudodifferential calculus in Section 2.1.1 contains established results, the results and definitions in the other sections are novel. The main results of this chapter are the functional calculus for positive-order pseudodifferential operators in Theorem 2.2.3, and the zero-order case in Theorem 2.3.7.

Much of the motivation for this chapter has already been covered in Sections 1.2 and 1.3. We will look at a scale of Hilbert spaces $\{\mathcal{H}_s\}_{s\in\mathbb{R}}$ defined by a single operator Θ , which is a very plain generalisation of Sobolev spaces. On this scale, operators can be defined which map in some well-defined and consistent way

$$T: \mathcal{H}^{s+m} \to \mathcal{H}^s, \quad s \in \mathbb{R},$$

which forms a bare-bones abstract pseudodifferential calculus. We will develop a functional

calculus for these operators, which will therefore be applicable to any pseudodifferential calculus. Though this is interesting in its own right, we are motivated by developing multiple operator integrals for these operators in Chapter 3, which in turn is motivated by applications in noncommutative geometry.

Usually, defining a functional calculus for a pseudodifferential calculus involves elliptic operators [Str72; MS87; Bon13]. We too will follow this strategy, although it breaks down for zero-order operators. For the latter, we can instead directly apply work by Davies [Dav95b; Dav95c].

2.1 Abstract pseudodifferential calculus

2.1.1 Review

First, the details of the abstract pseudodifferential calculus that we use. See also [KP66; Dal67; CM95; Uuy11].

Definition 2.1.1. Let Θ be a possibly unbounded invertible positive self-adjoint operator on a separable Hilbert space \mathcal{H} . Define the Hilbert spaces $\mathcal{H}^s := \overline{\mathrm{dom}\,\Theta^s}^{\|\cdot\|_s}$ for $s \in \mathbb{R}$ where $\|\phi\|_s := \|\Theta^s \phi\|$ – noting that taking this completion is unnecessary for $s \ge 0$. We write $\mathcal{H}^\infty := \bigcap_{s \ge 0} \mathcal{H}^s$, which is dense in \mathcal{H} .

This definition dates back to work by S. G. Krein, see the reviews [Mit61; KP66] and references therein. Further developments were made in [Dal67; Bon67]. Furthermore, the space \mathcal{H}^{∞} is a 'countably Hilbert space' in the terminology of [GV64, Section I.3].

A quick first observation is that the embedding $\mathcal{H}^s \hookrightarrow \mathcal{H}^t$ is continuous for all $s \ge t$. For s > 0, there is a pairing between \mathcal{H}^s and \mathcal{H}^{-s} given by

$$\langle u, v \rangle_{(\mathcal{H}^s, \mathcal{H}^{-s})} := \langle \Theta^s u, \Theta^{-s} v \rangle_{\mathcal{H}}, \quad u \in \mathcal{H}^s, v \in \mathcal{H}^{-s}.$$

This pairing identifies \mathcal{H}^{-s} with the (continuous) anti-linear dual space of \mathcal{H}^{s} and vice-versa.

The space $\mathcal{H}^{\infty} = \bigcap_{s \in \mathbb{R}} \mathcal{H}^s$ is a Fréchet space equipped with the norms $\|\cdot\|_s$, $s \in \mathbb{R}$. By construction, $\mathcal{H}^{\infty} \subseteq \mathcal{H}^s$ for any $s \in \mathbb{R}$, and in fact \mathcal{H}^{∞} is dense in \mathcal{H}^s . Since a subspace of a separable metric space is itself separable it follows that every \mathcal{H}^s admits an orthonormal basis consisting of vectors in \mathcal{H}^{∞} .

We define $\mathcal{H}^{-\infty}$ as the continuous anti-linear dual space of \mathcal{H}^{∞} , which can be identified with

$$\mathcal{H}^{-\infty} = \bigcup_{s \in \mathbb{R}} \mathcal{H}^s.$$
(2.1)

This is an *LF*-space, in the sense of [Trè67, Chapter 13]. From this perspective \mathcal{H}^{∞} can be interpreted as a Schwartz space and $\mathcal{H}^{-\infty}$ as a space of distributions. In case $\Theta^{-1} \in \mathcal{L}_s$ for some s > 0, we have that the Gelfand triples $\mathcal{H}^{\infty} \subset \mathcal{H} \subset \mathcal{H}^{-\infty}$ and $\mathcal{H}^s \subset \mathcal{H} \subset \mathcal{H}^{-s}$ are 'rigged Hilbert spaces' in the terminology of [GV64, Section I.4].

Given $u \in \mathcal{H}^{\infty}$ and $v \in \mathcal{H}^{-\infty}$ it follows from (2.1) that $v \in \mathcal{H}^{-s}$ for some particular $s \in \mathbb{R}$. It is immediate that $u \in \mathcal{H}^s$, and we have

$$\langle u, v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})} = \langle u, v \rangle_{(\mathcal{H}^{s}, \mathcal{H}^{-s})}.$$

Proposition 2.1.2. The Sobolev spaces \mathcal{H}^s in Definition 2.1.1 form an exact interpolation scale. That is, let $s_0 \leq s_1$, $r_0, r_1 \in \mathbb{R}$, and let $0 < \theta < 1$. Set

$$s_{\theta} := (1-\theta)s_0 + \theta s_1, \quad r_{\theta} = (1-\theta)r_0 + \theta r_1.$$

If T is a bounded linear map

$$T: \mathcal{H}^{s_0} \to \mathcal{H}^{r_0}, \quad T|_{\mathcal{H}^{s_1}}: \mathcal{H}^{s_1} \to \mathcal{H}^{r_1},$$

then $T|_{\mathcal{H}^{s_{\theta}}}$ is bounded from $\mathcal{H}^{s_{\theta}}$ to $\mathcal{H}^{r_{\theta}}$ for every θ . Moreover we have

$$\|T\|_{\mathcal{H}^{s_{\theta}} \to \mathcal{H}^{r_{\theta}}} \leq \|T\|_{\mathcal{H}^{s_{0}} \to \mathcal{H}^{r_{0}}}^{1-\theta} \|T\|_{\mathcal{H}^{s_{1}} \to \mathcal{H}^{r_{1}}}^{\theta}.$$

Proof. See [KP66]. Or, after identifying \mathcal{H}^s with a weighted L_2 -space through the spectral theorem, this follows from the Stein–Weiss interpolation theorem for L_p -spaces [BL76, Theorem 5.4.1].

Definition 2.1.3. We say that a linear operator $A : \mathcal{H}^{\infty} \to \mathcal{H}^{\infty}$ is in the class $\operatorname{op}^{r}(\Theta)$ (it has analytic order $\leq r$) if A extends to a continuous operator

$$\overline{A}^{s+r,r}:\mathcal{H}^{s+r}\to\mathcal{H}^s$$

for all $s \in \mathbb{R}$. If no confusion can arise, we often write

$$A: \mathcal{H}^{s+r} \to \mathcal{H}^s.$$

Furthermore, $\operatorname{op}(\Theta) := \bigcup_{r \in \mathbb{R}} \operatorname{op}^{r}(\Theta)$, and $\operatorname{op}^{-\infty}(\Theta) := \bigcap_{r \in \mathbb{R}} \operatorname{op}^{r}(\Theta)$. We define $\operatorname{OP}^{r}(\Theta) \subseteq \operatorname{op}^{r}(\Theta)$ as those $A \in \operatorname{op}^{r}(\Theta)$ for which $\delta_{\Theta}^{n}(A) \in \operatorname{op}^{r}(\Theta)$ for each $n \ge 0$, where $\delta_{\Theta}(A) := [\Theta, A]$.

Since $\mathcal{H}^s \hookrightarrow \mathcal{H}^t$ is continuous for all $s \ge t$, it follows that $\operatorname{op}^r(\Theta) \subseteq \operatorname{op}^t(\Theta)$ for $r \le t$. Both $\operatorname{op}(\Theta)$ and $\operatorname{OP}(\Theta) := \bigcup_{r \in \mathbb{R}} \operatorname{OP}^r(\Theta)$ form a filtered algebra, as $\operatorname{op}^r(\Theta) \cdot \operatorname{op}^t(\Theta) \subseteq \operatorname{op}^{r+t}(\Theta)$ and $\operatorname{OP}^r(\Theta) \cdot \operatorname{OP}^t(\Theta) \subseteq \operatorname{OP}^{r+t}(\Theta)$.

Commonly, this paradigm is also used when dealing with unbounded operators

$$T: \operatorname{dom}(T) \to \mathcal{H}.$$

In this case, one writes $T \in \operatorname{op}^{r}(\Theta)$ if $\mathcal{H}^{\infty} \subseteq \operatorname{dom}(T), T(\mathcal{H}^{\infty}) \subseteq \mathcal{H}^{\infty}$, and

$$T|_{\mathcal{H}^{\infty}} \in \mathrm{op}^{r}(\Theta).$$

Conversely, for $T \in \mathrm{op}^r(\Theta)$ with $r \ge 0$,

$$\overline{T}^{r,0}:\mathcal{H}^r\subseteq\mathcal{H}\to\mathcal{H}$$

can be interpreted as an unbounded operator. Furthermore, note that $\operatorname{op}^{-r}(\Theta) \subseteq B(\mathcal{H})$ for $r \geq 0$ in the sense that for $A \in \operatorname{op}^{-r}(\Theta)$ we have $\overline{A}^{-r,0}|_{\mathcal{H}} \in B(\mathcal{H})$. Similarly if $\Theta^{-1} \in \mathcal{L}_s(\mathcal{H})$, a Schatten class, then $\operatorname{op}^{-s}(\Theta) \subseteq \mathcal{L}_1(\mathcal{H})$.

2.1.2 Adjoints

Here we discuss various ways of defining the adjoint in $op^r(\Theta)$.

Definition 2.1.4. Let $A \in op^r(\Theta)$ so that A extends to a bounded operator

$$A:\mathcal{H}^{s+r}\to\mathcal{H}^s$$

for all $s \in \mathbb{R}$.

1. The adjoint of A as an endomorphism of the topological vector space \mathcal{H}^{∞} we denote

$$A^{\dagger}:\mathcal{H}^{-\infty}\to\mathcal{H}^{-\infty}$$

defined by the identity $% \left(f_{i}^{2} + f_{i}^{2} +$

$$\langle Au, v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})} = \langle u, A^{\dagger}v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})}, \quad u \in \mathcal{H}^{\infty}, v \in \mathcal{H}^{-\infty}.$$

2. In similar fashion we denote the adjoint

$$A'^s:\mathcal{H}^{-s}\to\mathcal{H}^{-s-r}$$

defined by the relevant identity

$$\langle Au, v \rangle_{(\mathcal{H}^s, \mathcal{H}^{-s})} = \langle u, A'^s v \rangle_{(\mathcal{H}^{s+r}, \mathcal{H}^{-s-r})}, \quad u \in \mathcal{H}^{s+r}, v \in \mathcal{H}^{-s}.$$

3. We define the Hermitian adjoint

$$A^{\flat_s}:\mathcal{H}^s\to\mathcal{H}^{s+r}$$

via the identity

$$\langle Au, v \rangle_{\mathcal{H}^s} = \langle u, A^{\flat_s} v \rangle_{\mathcal{H}^{s+r}}, \quad u \in \mathcal{H}^{s+r}, v \in \mathcal{H}^s.$$

4. In case $r \geq 0$, the map

$$A:\mathcal{H}^{s+r}\subseteq\mathcal{H}^s\to\mathcal{H}^s$$

is an unbounded operator on the Hilbert space \mathcal{H}^s , so we define another Hermitian adjoint

$$A^{*_s}: \mathcal{D}_s \to \mathcal{H}^s,$$

with domain

$$\mathcal{D}_s := \{ u \in \mathcal{H}^s \mid \exists v \in \mathcal{H}^s \forall \phi \in \mathcal{H}^{s+r} : \langle u, T\phi \rangle_{\mathcal{H}^s} = \langle v, \phi \rangle_{\mathcal{H}^s} \},\$$

such that

$$\langle Au, v \rangle_{\mathcal{H}^s} = \langle u, A^{*s}v \rangle_{\mathcal{H}^s}, \quad u \in \mathcal{H}^{s+r}, v \in \mathcal{D}_s.$$

These adjoints are related in the following way.

Proposition 2.1.5. Let $A \in op^r(\Theta)$. Then, for all $s \in \mathbb{R}$,

1. $A'^{s} = A^{\dagger} \Big|_{\mathcal{H}^{-s}};$ 2. $A^{\flat_{s}} = \Theta^{-2s-2r} A^{\dagger} \Theta^{2s} \Big|_{\mathcal{H}^{s}}.$

If $r \geq 0$,

3.
$$A^{*_s} = \Theta^{-2s} A^{\dagger} \Theta^{2s} \Big|_{\mathcal{D}_s}.$$

Proof. 1. Take $u \in \mathcal{H}^{\infty} \subseteq \mathcal{H}^{s+r}$ and $v \in \mathcal{H}^{-s} \subseteq \mathcal{H}^{-\infty}$. Then

$$\begin{split} \langle Au, v \rangle_{(\mathcal{H}^s, \mathcal{H}^{-s})} &= \langle u, A'^s v \rangle_{(\mathcal{H}^{s+r}, \mathcal{H}^{-s-r})} \\ &= \langle u, A'^s v \rangle_{(\mathcal{H}^\infty, \mathcal{H}^{-\infty})}. \end{split}$$

We also have

$$\begin{aligned} \langle Au, v \rangle_{(\mathcal{H}^s, \mathcal{H}^{-s})} &= \langle Au, v \rangle_{(\mathcal{H}^\infty, \mathcal{H}^{-\infty})} \\ &= \langle u, A^{\dagger}v \rangle_{(\mathcal{H}^\infty, \mathcal{H}^{-\infty})}. \end{aligned}$$

Hence it follows that

$$A'^{s}v = A^{\dagger}v \in \mathcal{H}^{-\infty}, \quad v \in \mathcal{H}^{-s}$$

2. Take $u \in \mathcal{H}^{\infty}, v \in \mathcal{H}^{s}$. Then on the one hand,

$$\begin{split} \langle Au, v \rangle_{\mathcal{H}^s} &= \langle Au, \Theta^{2s} v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})} \\ &= \langle u, A^{\dagger} \Theta^{2s} v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})}, \end{split}$$

and on the other hand

$$\begin{split} \langle Au, v \rangle_{\mathcal{H}^s} &= \langle u, A^{\flat_s} v \rangle_{\mathcal{H}^{s+r}} \\ &= \langle u, \Theta^{2s+2r} A^{\flat_s} v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})}. \end{split}$$

We therefore find

$$A^{\flat_s} = \Theta^{-2s-2r} A^{\dagger} \Theta^{2s} \Big|_{\mathcal{H}^s}.$$

3. Take $u \in \mathcal{H}^{\infty}, v \in \mathcal{D}_s \subseteq \mathcal{H}^s \subseteq \mathcal{H}^{-\infty}$. Then

$$\begin{aligned} \langle Au, v \rangle_{\mathcal{H}^s} &= \langle Au, \Theta^{2s} v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})} \\ &= \langle u, A^{\dagger} \Theta^{2s} v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})} \end{aligned}$$

and

$$\langle Au, v \rangle_{\mathcal{H}^s} = \langle u, A^{*s}v \rangle_{\mathcal{H}^s}$$
$$= \langle u, \Theta^{2s}A^{*s}v \rangle_{(\mathcal{H}^\infty, \mathcal{H}^{-\infty})}.$$

Hence

$$A^{*_s} = \Theta^{-2s} A^{\dagger} \Theta^{2s} \Big|_{\mathcal{D}_s}.$$

An important take away from this proposition is that if $A: \mathcal{H}^{\infty} \to \mathcal{H}^{\infty}$, we have a priori that

$$A^{\dagger}: \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty},$$

but if $A \in \mathrm{op}^r(\Theta)$ we have in fact that $A^{\dagger} \in \mathrm{op}^r(\Theta)$ (or, more precisely, $A^{\dagger}|_{\mathcal{H}^{\infty}} \in \mathrm{op}^r(\Theta)$).

It is now also clear that the Hermitian adjoints A^{\flat_s} and A^{\ast_s} in general cannot be regarded as operators in $op(\Theta)$, as the operators A^{\flat_s} and A^{\flat_t} do not agree on the intersection $\mathcal{H}^s \cap \mathcal{H}^t$ for $s \neq t$, and the same holds for A^{\ast_s} .

Proposition 2.1.6. If $A \in op^r(\Theta)$, $r \ge 0$, then

$$A:\mathcal{H}^r\subseteq\mathcal{H}^0\to\mathcal{H}^0$$

is symmetric if and only if $A = A^{\dagger}$.

Proof. Suppose that $A = A^{\dagger}$. Let $u \in \mathcal{H}^{\infty}, v \in \mathcal{H}^{r}$. Then

$$\begin{split} \langle Au, v \rangle_{\mathcal{H}^0} &= \langle Au, v \rangle_{(\mathcal{H}^\infty, \mathcal{H}^{-\infty})} \\ &= \langle u, A^{\dagger}v \rangle_{(\mathcal{H}^\infty, \mathcal{H}^{-\infty})} \\ &= \langle u, Av \rangle_{\mathcal{H}^0}. \end{split}$$

By density of $\mathcal{H}^{\infty} \subseteq \mathcal{H}^r$, the above equality holds for $u \in \mathcal{H}^r$ as well, and hence $A : \mathcal{H}^r \subseteq \mathcal{H}^0 \to \mathcal{H}^0$ is symmetric.

On the other hand, if $A: \mathcal{H}^r \subseteq \mathcal{H}^0 \to \mathcal{H}^0$ is symmetric, then for $u, v \in \mathcal{H}^\infty$,

$$\begin{split} \langle u, A^{\dagger}v \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})} &= \langle Au, v \rangle_{\mathcal{H}^{0}} \\ &= \langle u, Av \rangle_{\mathcal{H}^{0}} \\ &= \langle u, Av \rangle_{(\mathcal{H}^{\infty}, \mathcal{H}^{-\infty})}, \end{split}$$

showing that $A^{\dagger}v = Av \in \mathcal{H}^{\infty}$ which implies that $A = A^{\dagger} \in \mathrm{op}^{r}(\Theta)$.

2.1.3 Elliptic operators

To prepare the way for a functional calculus on this Hilbert scale, we will define a notion of ellipticity called Θ -ellipticity in this subsection. We then show that for $A \in \operatorname{op}^{r}(\Theta)$, $r \geq 0$, Θ -elliptic and symmetric we have that A as an operator on \mathcal{H} is self-adjoint with domain \mathcal{H}^{r} . Furthermore, if A is invertible in an appropriate sense, then $A^{-1} \in \operatorname{op}^{-r}(\Theta)$.

Definition 2.1.7. We say that an operator $A \in \operatorname{op}^{r}(\Theta)$ is Θ -elliptic if there exists a parametrix for A of order -r, that is, there exists an operator $P \in \operatorname{op}^{-r}(\Theta)$ such that

$$AP = 1_{\mathcal{H}^{\infty}} + R_1;$$
$$PA = 1_{\mathcal{H}^{\infty}} + R_2,$$

where $R_1, R_2 \in op^{-\infty}(\Theta) = \bigcap_{s \in \mathbb{R}} op^s(\Theta)$.

We emphasise that the notion of Θ -ellipticity depends on Θ and on the order $r \in \mathbb{R}$.

Remark 2.1.8. In the definition of Θ -ellipticity above, we could have equivalently required the formally weaker condition of A having a right-parametrix $P_1 \in \operatorname{op}^{-r}(\Theta)$ and a leftparametrix $P_2 \in \operatorname{op}^{-r}(\Theta)$,

$$AP_1 = 1_{\mathcal{H}^{\infty}} + R_1;$$
$$P_2 A = 1_{\mathcal{H}^{\infty}} + R_2,$$

where $R_1, R_2 \in op^{-\infty}(\Theta)$. Namely, it is not difficult to deduce that this would imply $P_1 - P_2 \in op^{-\infty}(\Theta)$.

We will now provide a lemma which lets us formally weaken the condition of Θ -ellipticity in Definition 2.1.7.

For a pseudodifferential operator $T \in op(\Theta)$, we say that there exists an asymptotic expansion

$$T \sim \sum_{k=0}^{\infty} T_k,$$

if

$$T - \sum_{k=0}^{N} T_k \in \mathrm{op}^{m_N}(\Theta), \quad m_N \downarrow -\infty.$$

Lemma 2.1.9 (Borel lemma). Let $\{A_k\}_{k=0}^{\infty}$ be a sequence of linear operators from \mathcal{H}^{∞} to \mathcal{H}^{∞} for which $A_k \in \operatorname{op}^{m_k}(\Theta)$ such that $m_k \downarrow -\infty$ as $k \to \infty$. There exists a linear operator $A \in \operatorname{op}^{m_0}(\Theta)$ such that

$$A \sim \sum_{k=0}^{\infty} A_k.$$

Proof. Let $\eta \in C_c^{\infty}(\mathbb{R})$ be equal to 1 in a neighbourhood of zero, and let $\{\varepsilon_k\}_{k=0}^{\infty}$ be a sequence of positive numbers tending to zero in a manner to be determined shortly. Formally we define

$$A := \sum_{k=0}^{\infty} A_k (1 - \eta(\varepsilon_k \Theta)).$$

We will prove that $\{\varepsilon_k\}_{k=0}^{\infty}$ can be chosen such that this series makes sense and $A \in op^{m_0}(\Theta)$ with the desired asymptotic expansion.

Let $\xi \in \mathcal{H}^{\infty}$. Then for every $k \geq 0$ and $n \in \mathbb{Z}$, we have

$$\begin{aligned} \|A_k(1-\eta(\varepsilon_k\Theta))\xi\|_{\mathcal{H}^n} &\leq \|A_k\|_{\mathcal{H}^{n+m_k}\to\mathcal{H}^n} \|(1-\eta(\varepsilon_k\Theta))\xi\|_{\mathcal{H}^{n+m_k}} \\ &\leq \|A_k\|_{\mathcal{H}^{n+m_k}\to\mathcal{H}^n} \|1-\eta(\varepsilon_k\Theta)\|_{\mathcal{H}^{n+m_0}\to\mathcal{H}^{n+m_k}} \|\xi\|_{\mathcal{H}^{n+m_0}}. \end{aligned}$$

Let a > 0 be a number such that $a < \Theta$. The norm of $1 - \eta(\varepsilon_k \Theta)$ from \mathcal{H}^{n+m_0} to \mathcal{H}^{n+m_k} is determined by functional calculus as

$$\sup_{t>a} t^{m_k - m_0} (1 - \eta(\varepsilon_k t)) \le \varepsilon_k^{m_0 - m_k} \sup_{s>0} s^{m_k - m_0} (1 - \eta(s)) \le C_\eta \varepsilon_k.$$

for some constant C_{η} , and for k sufficiently large so that $m_0 - m_k \ge 1$. Now we choose ε_k sufficiently small such that

$$0 \le \varepsilon_k C_\eta \max_{|n| \le k} \{ \|A_k\|_{\mathcal{H}^{n+m_k} \to \mathcal{H}^n} \} < 2^{-k}.$$

With this choice of sequence $\{\varepsilon_k\}_{k=0}^{\infty}$, we have just proved that the series

$$A\xi = \sum_{k=0}^{\infty} A_k (1 - \eta(\varepsilon_k \Theta))\xi$$

converges in every \mathcal{H}^n , and defines a bounded linear operator

$$A: \mathcal{H}^{n+m_0} \to \mathcal{H}^n, \quad n \in \mathbb{Z}.$$

Since this holds for every $n \in \mathbb{Z}$, it follows that $A : \mathcal{H}^{\infty} \to \mathcal{H}^{\infty}$ and by interpolation (Proposition 2.1.2) $A \in \operatorname{op}^{m_0}(\Theta)$. Note that with this fixed choice of $\{\varepsilon_k\}_{k=0}^{\infty}$, we have proved the stronger result that the 'tail' of A

$$\sum_{k=N+1}^{\infty} A_k (1 - \eta(\varepsilon_k \Theta))$$

converges in every \mathcal{H}^s and defines a linear operator in $\operatorname{op}^{m_{N+1}}(\Theta)$.

Now we prove that A has the desired asymptotic expansion. For every N > 0 we have

$$A - \sum_{k=0}^{N} A_k = -\sum_{k=0}^{N} A_k \eta(\varepsilon_k \Theta) + \sum_{k=N+1}^{\infty} A_k (1 - \eta(\varepsilon_k \Theta))$$

Since η is compactly supported, it is easy to see that the first summand has order $-\infty$ for every $N \ge 0$, and the second summand has order at most m_{N+1} due to the result just proved.

Corollary 2.1.10. Suppose that $A \in \operatorname{op}^{r}(\Theta)$ has an inverse $B \in \operatorname{op}^{-r}(\Theta)$ up to order -1. That is,

$$AB = 1_{\mathcal{H}^{\infty}} + R_1;$$
$$BA = 1_{\mathcal{H}^{\infty}} + R_2$$

where R_1, R_2 have order -1. Then A is Θ -elliptic.

Proof. Since R_1^j has order at most -j, we can use the Borel lemma to construct an operator B' such that

$$B'\sim \sum_{k=0}^\infty (-1)^j BR_1^j.$$

Then AB' - 1 has order $-\infty$. Similarly we can construct a left inverse.

The condition for Θ -ellipticity in Corollary 2.1.10 corresponds to a definition of ellipticity by Guillemin in an abstract pseudodifferential calculus [Gui85].

Proposition 2.1.11. Let $A \in op^r(\Theta)$ be Θ -elliptic. If the bounded extension

$$A:\mathcal{H}^{s_0+r}\to\mathcal{H}^{s_0}$$

admits a bounded inverse

$$A^{-1}: \mathcal{H}^{s_0} \to \mathcal{H}^{s_0+r},$$

for a specific $s_0 \in \mathbb{R}$, then $A^{-1}\Big|_{\mathcal{H}^{\infty}} \in \operatorname{op}^{-r}(\Theta)$. Simply writing $A^{-1} = A^{-1}\Big|_{\mathcal{H}^{\infty}}$, we have that

$$A^{-1}A = AA^{-1} = 1_{\mathcal{H}^{\infty}}.$$

Proof. Let P be a parametrix of A and take $x \in \mathcal{H}^{\infty}$, so that

$$A^{-1}x = (PA - R_2)A^{-1}(AP - R_1)x$$

= $PAPx - R_2Px - PR_1x + R_2A^{-1}R_1x$.

Observe that for $y \in \mathcal{H}^t$, $t \in \mathbb{R}$, we have $A^{-1}R_1y \in \mathcal{H}^{s_0+r}$, so that $R_2A^{-1}R_1y \in \mathcal{H}^{\infty}$. Hence,

$$R_2 A^{-1} R_1 \in \mathrm{op}^{-\infty}(\Theta)$$

and therefore

$$A^{-1} = PAP - R_2P - PR_1 + R_2A^{-1}R_1$$

$$\in \operatorname{op}^{-r}(\Theta) + \operatorname{op}^{-\infty}(\Theta) = \operatorname{op}^{-r}(\Theta).$$

Proposition 2.1.12. Let $A \in op^r(\Theta)$ be Θ -elliptic of order $r \ge 0$. Then the unbounded operator

$$A:\mathcal{H}^{s+r}\subseteq\mathcal{H}^s\to\mathcal{H}^s$$

is closed for each $s \in \mathbb{R}$.

Proof. Define the graph norm of A on dom $(A) = \mathcal{H}^{s+r}$ as

$$||x||_{G(A)} := ||Ax||_s + ||x||_s, \quad x \in \mathcal{H}^{s+r}.$$

By definition, A is a closed operator if and only if dom(A) is complete with respect to this graph norm. We will show that for Θ -elliptic operators, the graph norm is equivalent to $\|\cdot\|_{s+r}$, which immediately implies that \mathcal{H}^{s+r} is complete with respect to the graph norm. First, we have that

$$\begin{aligned} \|Ax\|_s + \|x\|_s &\leq \|A\|_{\mathcal{H}^{s+r} \to \mathcal{H}^s} \|x\|_{s+r} + \|\Theta^{-r}\|_{\mathcal{H}^s \to \mathcal{H}^s} \|\Theta^r x\|_s \\ &\lesssim \|x\|_{s+r}. \end{aligned}$$

Next, let P be a parametrix for A, and take $x \in \mathcal{H}^{\infty}$ so that

$$||x||_{s+r} \le ||PAx||_{s+r} + ||R_2x||_{s+r}$$

$$\le ||P||_{\mathcal{H}^s \to \mathcal{H}^{s+r}} ||Ax||_s + ||R_2||_{\mathcal{H}^s \to \mathcal{H}^{s+r}} ||x||_s$$

$$\lesssim ||Ax||_s + ||x||_s.$$

The assertion of the proposition is now immediate.

Elliptic operators have a property which is often called elliptic regularity, or maximal subellipticity.

Proposition 2.1.13. Let $A \in \operatorname{op}^{r}(\Theta)$ be a Θ -elliptic operator. If $x \in \mathcal{H}^{-\infty}$ is such that $Ax \in \mathcal{H}^{s}$ for an $s \in \mathbb{R}$, then $x \in \mathcal{H}^{s+r}$.

Proof. Take $x \in \mathcal{H}^{-\infty}$, and suppose that $Ax \in \mathcal{H}^s$. Then $PAx \in \mathcal{H}^{s+r}$, which implies that

$$x = PAx - R_2 x \in \mathcal{H}^{s+r}.$$

Finally we will now show that if $A \in \operatorname{op}^r(\Theta)$, $r \ge 0$ is Θ -elliptic and symmetric, then A is self-adjoint with domain \mathcal{H}^r . Recall from Section 2.1.2 that $A^{\dagger}|_{\mathcal{H}^{\infty}} \in \operatorname{op}^r(\Theta)$, and for any $s \in \mathbb{R}$, we have that

$$\langle Au, v \rangle_{(\mathcal{H}^s, \mathcal{H}^{-s})} = \langle u, A^{\dagger}v \rangle_{(\mathcal{H}^{s+r}, \mathcal{H}^{-s-r})}, \quad u \in \mathcal{H}^{s+r}, v \in \mathcal{H}^{-s}.$$
(2.2)

And, by Proposition 2.1.6, for $r \ge 0$ we have that $A = A^{\dagger}$ if and only if $\overline{A}^{r,0} : \mathcal{H}^r \subseteq \mathcal{H} \to \mathcal{H}$ is symmetric.

Proposition 2.1.14. Let $A \in op^r(\Theta)$, $r \ge 0$, be a Θ -elliptic and symmetric operator. Then A is self-adjoint with domain \mathcal{H}^r .

Proof. To prove that A is self-adjoint, we need to show that the Hermitian adjoint of the closed operator $A : \mathcal{H}^r \subseteq \mathcal{H}^0 \to \mathcal{H}^0$, writing $A^{*_0} := \left(\overline{A}^{r,0}\right)^*$,

$$A^{*_0}$$
: dom $(A^{*_0}) \subseteq \mathcal{H}^0 \to \mathcal{H}^0$,

has domain dom $(A^{*_0}) = \mathcal{H}^r$. Recall that, by definition,

 $\operatorname{dom}(A^{*_0}) := \{ u \in \mathcal{H}^0 : \exists v \in \mathcal{H}^0 \text{ such that } \forall \phi \in \mathcal{H}^r \ \langle u, A\phi \rangle_{\mathcal{H}^0} = \langle v, \phi \rangle_{\mathcal{H}^0} \}.$

If $u, v \in \mathcal{H}^0$ and $\phi \in \mathcal{H}^r$, then by (2.2) and Proposition 2.1.6,

$$\begin{aligned} \langle u, A\phi \rangle_{\mathcal{H}^0} &= \langle A^{\dagger}u, \phi \rangle_{(\mathcal{H}^{-r}, \mathcal{H}^{r})} \\ &= \langle Au, \phi \rangle_{(\mathcal{H}^{-r}, \mathcal{H}^{r})}; \\ \langle v, \phi \rangle_{\mathcal{H}^0} &= \langle v, \phi \rangle_{(\mathcal{H}^{-r}, \mathcal{H}^{r})}. \end{aligned}$$

Since \mathcal{H}^r separates the points of \mathcal{H}^{-r} , we have that for $u, v \in \mathcal{H}^0$,

$$\langle Au, \phi \rangle_{(\mathcal{H}^{-r}, \mathcal{H}^{r})} = \langle v, \phi \rangle_{(\mathcal{H}^{-r}, \mathcal{H}^{r})}, \quad \forall \phi \in \mathcal{H}^{r},$$

if and only if

$$Au = v \in \mathcal{H}^{-r}.$$

Hence,

$$dom(A^{*_0}) = \{ u \in \mathcal{H}^0 : \exists v \in \mathcal{H}^0 \text{ such that } Au = v \}$$
$$= \{ u \in \mathcal{H}^0 : Au \in \mathcal{H}^0 \}.$$

By elliptic regularity (Proposition 2.1.13) it follows that $dom(A^{*_0}) = \mathcal{H}^r$, completing the proof.

2.2 Functional calculus for elliptic operators

Due to the results in the previous section, we know that for a Θ -elliptic symmetric operator $A \in \operatorname{op}^{r}(\Theta), r \geq 0$, the operator $\overline{A}^{r,0} : \mathcal{H}^{r} \subseteq \mathcal{H} \to \mathcal{H}$ is self-adjoint. Hence, we can ask when the operator $f(\overline{A}^{r,0})$ defined via the Borel functional calculus is an operator in $\operatorname{op}(\Theta)$. An obstacle for a naive approach is that $f(\overline{A}^{s+r,s})$ is not easily defined for $s \neq 0$ as an operator on \mathcal{H}^{s} , as $\overline{A}^{s+r,s}$ is generally *not* normal or symmetric as an operator on \mathcal{H}^{s} when $\overline{A}^{r,0}$ is on \mathcal{H} . Indeed, Proposition 2.1.5 tells us that this is only the case when

$$A = \Theta^{-2s} A \Theta^{2s}.$$

The main idea in this section is the following. A functional calculus for the self-adjoint Θ itself as a pseudodifferential operator is not difficult to construct. Namely, for measurable $f : \mathbb{R} \to \mathbb{C}$, one can define $f(\Theta)$ as an unbounded operator on \mathcal{H} [Sch12, Theorem 5.9], with operator norms

$$\|f(\Theta)\|_{\mathcal{H}^{s+m}\to\mathcal{H}^s} = \|\Theta^s f(\Theta)\Theta^{-s-m}\|_{\mathcal{H}\to\mathcal{H}} = \|f(x)x^{-m}\|_{L_{\infty}(E)}$$

which gives a quick and appropriate condition for $f(\Theta) \in \operatorname{op}^m(\Theta)$ (here, E is the spectral measure of Θ , see Definition 2.2.2 below for a precise definition of $L_{\infty}(E)$). We can further exploit this by the insight that for a Θ -elliptic symmetric operator $A \in \operatorname{op}^r(\Theta)$, r > 0, in fact the norms $\|\Theta^s \xi\|$ and $\|\langle A \rangle^{\frac{s}{r}} \xi\|$ are equivalent, where $\langle x \rangle := (1 + |x|^2)^{\frac{1}{2}}$. If we can provide that

$$\mathcal{H}^{s}(\Theta) = \mathcal{H}^{s}(\langle A \rangle^{\frac{1}{r}}), \quad \mathrm{op}^{s}(\Theta) = \mathrm{op}^{s}(\langle A \rangle^{\frac{1}{r}}),$$

we can then obtain a functional calculus for A on the Sobolev spaces $\mathcal{H}^{s}(\langle A \rangle^{\frac{1}{r}})$ as easily as for Θ above, which lets us conclude that $f(A) \in \mathrm{op}(\Theta)$.

Proposition 2.2.1. Let $A \in \operatorname{op}^{r}(\Theta)$, r > 0 be Θ -elliptic and symmetric. Then $\langle A \rangle^{\frac{1}{r}} := (1 + A^{2})^{\frac{1}{2r}}$ extends to an invertible positive self-adjoint operator on \mathcal{H} , and

$$\operatorname{dom} \Theta^s = \operatorname{dom} \left(\langle A \rangle^{\frac{1}{r}} \right)^s.$$

The norms $\|\Theta^s \xi\|_{\mathcal{H}}$ and $\|\langle A \rangle^{\frac{s}{r}} \xi\|_{\mathcal{H}}$ are equivalent on this subspace of \mathcal{H} . Therefore, Θ and $\langle A \rangle^{\frac{1}{r}}$ define the same Sobolev scale

$$\mathcal{H}^{s}(\Theta) = \mathcal{H}^{s}(\langle A \rangle^{\frac{1}{r}}),$$

and we have

$$\operatorname{op}^{t}(\Theta) = \operatorname{op}^{t}\left(\langle A \rangle^{\frac{1}{r}}\right).$$

Proof. The first statement follows from Proposition 2.1.14. For the remaining statements, it suffices to prove that

$$(1+A^2)^{\alpha}\Theta^{-2\alpha r}$$

extends to a bounded operator on \mathcal{H} for all $\alpha \in \mathbb{R}$, as this would imply for $\xi \in \mathcal{H}^{\infty}$,

$$\|\langle A \rangle^{\frac{s}{r}} \xi \|_{\mathcal{H}} \le \|\langle A \rangle^{\frac{s}{r}} \Theta^{-s} \|_{\infty} \|\Theta^{s} \xi \|_{\mathcal{H}} \lesssim \|\Theta^{s} \xi \|_{\mathcal{H}},$$

and an analogous estimate in the other direction.

Let P be a parametrix for A, so that AP = 1 + R with $R \in op^{-\infty}(\Theta)$. Since

$$(1+A^2)P^2 = P^2 + A(1+R)P = 1 + P^2 + R + ARP$$

and $P^2 + R + ARP \in \text{op}^{-2r}(\Theta)$ and similarly for $(1 + A^2)P^2$, it follows that the operator $1 + A^2$ is also Θ -elliptic due to Corollary 2.1.10. Since A is self-adjoint with domain \mathcal{H}^r , applying Proposition 2.1.11 gives that $(1 + A^2)^{-1} \in \text{op}^{-2r}(\Theta)$. We therefore have $(1 + A^2)^k \in \text{op}^{2kr}(\Theta), k \in \mathbb{Z}$. This in turn gives that $\mathcal{H}^{\infty} \subseteq \text{dom}(1 + A^2)^z$ for any $z \in \mathbb{C}$.

We use the Hadamard three-line theorem, so define the function

$$F(z) := \langle x, (1+A^2)^{mz} \Theta^{-2mzr} y \rangle_{\mathcal{H}}, \quad z \in \mathbb{C},$$

where $m \in \mathbb{Z}$ and $x, y \in \mathcal{H}^{\infty}$ are fixed. Let $\{e_n\}_{n \in \mathbb{N}} \subseteq \mathcal{H}^{\infty}$ be an orthonormal basis of \mathcal{H} , then

$$F(z) = \sum_{n=0}^{\infty} \langle x, (1+A^2)^{mz} e_n \rangle_{\mathcal{H}} \langle e_n, \Theta^{-2mzr} y \rangle_{\mathcal{H}}.$$
(2.3)

Using the dominated convergence theorem, it can be seen that $z \mapsto \langle x, (1+A^2)^{mz} e_n \rangle_{\mathcal{H}}$ and $z \mapsto \langle e_n, \Theta^{-2mzr} y \rangle_{\mathcal{H}}$ are continuous maps. Applying the Cauchy–Schwarz inequality to the series (2.3) yields

$$\sum_{n=0}^{\infty} |\langle x, (1+A^2)^{mz} e_n \rangle_{\mathcal{H}}| |\langle e_n, \Theta^{-2mzr} y \rangle_{\mathcal{H}}| \le \|(1+A^2)^{m\overline{z}} x\|_{\mathcal{H}} \|\Theta^{-2mzr} y\|_{\mathcal{H}}$$

which is uniformly bounded on compact subsets of \mathbb{C} due to the continuity of the righthand side. We can therefore apply the dominated convergence theorem again to deduce that F(z) is a continuous function itself. Furthermore, this uniform boundedness yields through Fubini's theorem that if γ is a closed loop in \mathbb{C} then

$$\int_{\gamma} F(z) \, dz = \sum_{n=0}^{\infty} \int_{\gamma} \langle x, (1+A^2)^{mz} e_n \rangle_{\mathcal{H}} \langle e_n, \Theta^{-2mzr} y \rangle_{\mathcal{H}} \, dz$$

Using Fubini's theorem once more, we have that

$$\int_{\gamma} F(z) dz = \sum_{n=0}^{\infty} \int_{\sigma(\Theta^{-2r})} \int_{\sigma(1+A^2)} \int_{\gamma} (\lambda \mu)^{mz} dz \langle x, dE^{1+A^2} e_n \rangle_{\mathcal{H}} \langle e_n, dE^{\Theta^{-2r}} y \rangle_{\mathcal{H}} = 0,$$

so that we can conclude by Morera's theorem that F(z) is holomorphic.

Since $1 + A^2$ and Θ are positive operators and

$$\sup_{x>0} |x^{it}| = 1,$$

it follows from the Borel functional calculus that for $s \in \mathbb{R}$,

$$|F(is)| = |\langle (1+A^2)^{-ims}x, \Theta^{-2imsr}y \rangle_{\mathcal{H}}|$$
$$\leq ||x||_{\mathcal{H}} ||y||_{\mathcal{H}}.$$

Likewise,

$$|F(1+is)| = |\langle (1+A^2)^{-ims}x, (1+A^2)^m \Theta^{-2mr} \Theta^{-2imsr}y \rangle_{\mathcal{H}}|$$
$$\leq ||(1+A^2)^m \Theta^{-2mr}||_{\mathcal{H} \to \mathcal{H}} ||x||_{\mathcal{H}} ||y||_{\mathcal{H}},$$

which we know to be finite since m is an integer.

The Hadamard three-line theorem (see e.g. [BL76, Lemma 1.1.2]) now gives that for $\alpha \in (0, 1)$,

$$\begin{aligned} |F(\alpha)| &\leq \max_{s \in \mathbb{R}} |F(\alpha + is)| \\ &\leq \Big(\max_{s \in \mathbb{R}} |F(is)| \Big)^{(1-\alpha)} \Big(\max_{s \in \mathbb{R}} |F(1+is)| \Big)^{\alpha} \\ &\leq \| (1+A^2)^m \Theta^{-2mr} \|_{\mathcal{H} \to \mathcal{H}}^{\alpha} \|x\|_{\mathcal{H}} \|y\|_{\mathcal{H}}. \end{aligned}$$

Hence, with $\alpha \in (0, 1)$ and $m \in \mathbb{Z}$,

$$\|(1+A^2)^{m\alpha}\Theta^{-2m\alpha r}\|_{\mathcal{H}\to\mathcal{H}} \le \|(1+A^2)^m\Theta^{-2mr}\|_{\mathcal{H}\to\mathcal{H}}^{\alpha},$$

which proves that $(1 + A^2)^{m\alpha} \Theta^{-2m\alpha r}$ extends to a bounded operator on \mathcal{H} for all $m \in \mathbb{Z}$, $\alpha \in [0, 1]$.

Definition 2.2.2. Let E be a spectral measure on \mathbb{R} with the Borel sigma algebra. For a Borel measurable function $f : \mathbb{R} \to \mathbb{C}$ we define the essential supremum seminorm

$$||f||_{L_{\infty}(E)} := \sup\{y \in \mathbb{R} : E(|f|^{-1}((y,\infty))) = 0\},\$$

which defines $L_{\infty}(E)$ in the usual way, namely as the quotient of the set of measurable functions with finite seminorm, by the set of those of zero seminorm. In the same way we define $L_{\infty}^{\beta}(E)$ by the seminorm

$$\|f\|_{L^{\beta}_{\infty}(E)} := \|x \mapsto f(x) \langle x \rangle^{-\beta}\|_{L_{\infty}(E)}$$

for any $\beta \in \mathbb{R}$, where $\langle x \rangle = (1 + |x|^2)^{1/2}$.

Theorem 2.2.3. Let $A \in \operatorname{op}^{r}(\Theta)$, r > 0, be Θ -elliptic and symmetric, and let E denote its spectral measure. If $f \in L^{\beta}_{\infty}(E), \beta \in \mathbb{R}$, then

$$f(\overline{A}^{r,0}) \in \mathrm{op}^{\beta r}(\Theta),$$

and we simply write $f(A) := f(\overline{A}^{r,0})$. More precisely,

$$\|f(A)\|_{\mathcal{H}^{s+\beta r}\to\mathcal{H}^s} \le C_{s,A} \|f\|_{L^{\beta}_{\infty}(E)}.$$

Proof. Let $A \in \operatorname{op}^{r}(\Theta)$, r > 0 be Θ -elliptic and symmetric. Using Proposition 2.2.1, we replace Θ by $\langle A \rangle^{\frac{1}{r}}$ so that $A \in \operatorname{op}^{r}(\langle A \rangle^{\frac{1}{r}})$ is Θ -elliptic and symmetric. By Proposition 2.1.14, the operator

$$\overline{A}^{r,0}:\mathcal{H}^r(\langle A\rangle^{\frac{1}{r}})\subseteq\mathcal{H}\to\mathcal{H}$$

is self-adjoint; we denote its spectral measure by E. Then for $f \in L^{\beta}_{\infty}(E)$, using Borel functional calculus to define $f(\overline{A}^{r,0})$, we have

$$\|\langle A\rangle^{\frac{s}{r}}f(\overline{A}^{r,0})\langle A\rangle^{-\frac{s}{r}-\beta}\|_{\mathcal{H}\to\mathcal{H}} = \|f(\overline{A}^{r,0})\langle A\rangle^{-\beta}\|_{\mathcal{H}\to\mathcal{H}} = \|f\|_{L^{\beta}_{\infty}(E)} < \infty,$$

which shows that $f(\overline{A}^{r,0})|_{\mathcal{H}^{\infty}} \in \mathrm{op}^{\beta r}(\langle A \rangle^{\frac{1}{r}})$. Converting this estimate back into an estimate on the spaces $\mathcal{H}^{s}(\Theta)$ introduces the constant $C_{s,A}$.

Theorem 2.2.3 has a converse in the following sense. If $A \in \operatorname{op}^{r}(\Theta)$, r > 0 is an arbitrary Θ -elliptic symmetric operator and if $f : \mathbb{R} \to \mathbb{C}$ is such that $f(A) \in \operatorname{op}^{\beta r}(\Theta)$, then the proof of Proposition 2.2.1 gives that $f(A)(1+A^2)^{-\beta/2}$ is a bounded operator on \mathcal{H} . This happens if and only if $f\langle x \rangle^{-\beta} \in L^0_{\infty}(E)$ [Sch12, Theorem 5.9], i.e. $f \in L^{\beta}_{\infty}(E)$.

Corollary 2.2.4. If $A \in \operatorname{op}^r(\Theta)$, r > 0 is symmetric and Θ -elliptic and if $f : \mathbb{R} \to \mathbb{C}$ is a bounded Borel measurable function, then for any $t \in \mathbb{R}$ we have

$$||f(tA)||_{\mathcal{H}^s \to \mathcal{H}^s} \le C_{s,A} \sup_{x \in \mathbb{R}} |f(x)|,$$

independent of $t \in \mathbb{R}$.

where we wrot

Proof. Like in the proof Theorem 2.2.3, we have

$$\|f(tA)\|_{\mathcal{H}^{s}(\langle A \rangle^{\frac{1}{r}}) \to \mathcal{H}^{s}(\langle A \rangle^{\frac{1}{r}})} = \|f_{t}\|_{L^{0}_{\infty}(E)} \leq \sup_{x \in \mathbb{R}} |f(x)|,$$

e $f_{t}(x) := f(tx).$

We say that two normal (potentially unbounded) operators A, B strongly commute if all their respective spectral projections commute.

The functional calculus constructed in Theorem 2.2.3 can easily be extended to a larger class of operators. For example, on \mathbb{R}^d we have that $i\frac{d}{dx}$ is not $(1-\Delta)^{\frac{1}{2}}$ -elliptic, but

it does commute strongly with a $(1 - \Delta)^{\frac{1}{2}}$ -elliptic symmetric operator, namely $(1 - \Delta)^{\frac{1}{2}}$ itself. The following proposition shows that a functional calculus for $i\frac{d}{dx}$ does exist in $op(1 - \Delta)^{\frac{1}{2}}$ for this reason.

Proposition 2.2.5. Let A be a self-adjoint operator on \mathcal{H} with spectral measure E. If there exists a Θ -elliptic symmetric operator $H \in \operatorname{op}^{h}(\Theta)$, h > 0 such that A strongly commutes with $\overline{H}^{h,0} : \mathcal{H}^{h} \subseteq \mathcal{H} \to \mathcal{H}$, then for $f \in L^{0}_{\infty}(E)$ we have that $f(A) \in \operatorname{op}^{0}(\Theta)$. If $A \in \operatorname{op}^{r}(\Theta)$ itself for some $r \in \mathbb{R}$, we have that $f(A) \in \operatorname{op}^{\beta r}(\Theta)$ for $f \in L^{\beta}_{\infty}(E)$, $\beta \geq 0$.

Proof. In light of Proposition 2.2.1, we can assume without loss of generality that $H = \Theta$. If $f \in L^0_{\infty}(E)$, then $f(A) : \mathcal{H} \to \mathcal{H}$ is a bounded operator, and for $\xi \in \mathcal{H}^{\infty}$ we have

$$\|f(A)\xi\|_{\mathcal{H}^k} = \|\Theta^k f(A)\xi\|_{\mathcal{H}} \le \|f(A)\|_{\infty} \|\xi\|_{\mathcal{H}^k}, \quad k \in \mathbb{Z},$$

which shows that $f(A) \in \operatorname{op}^0(\Theta)$ through interpolation (Proposition 2.1.2). The second part of the proposition is proved similarly, after the observation that the Hadamard threeline argument in the proof of Proposition 2.2.1 goes through for A if $m \in \mathbb{N}$, i.e. $(1 + A^2)^{\alpha} \Theta^{-2\alpha r}$ is bounded for $\alpha \ge 0$.

Let us now compare the functional calculus of Theorem 2.2.3 with other examples. For classical pseudodifferential operators, pseudodifferential operators in the Beals–Fefferman calculus, and a generalisation thereof on manifolds, general functions in $f \in L^{\beta}_{\infty}(E)$ are much too rough to guarantee that f(T) is again a pseudodifferential operator of the same class [Str72; Bon13]. To emphasise, our results provide that for such rough functions f, the operator f(T) is well-defined and maps boundedly between appropriate Sobolev spaces, but nothing more. The typical function class that allows concluding that f(T) is again a pseudodifferential operator of the right type, is the following [Str72; Bon13].

Definition 2.2.6. For $I \subseteq \mathbb{R}$ an interval and $\beta \in \mathbb{R}$, we define $S^{\beta}(I)$ as the class of smooth functions $f: I \to \mathbb{C}$ such that

$$\|f\|_{S^{\beta}(I),k} := \sup_{x \in I} |f^{(k)}(x)| \langle x \rangle^{k-\beta} < \infty, \quad k \in \mathbb{N}.$$

The quantities above are seminorms.

From this perspective, the operator class $OP(\Theta)$ (recall Definition 2.1.3) behaves more like typical pseudodifferential operators.

Theorem 2.2.7. Let $A \in OP^r(\Theta)$, r > 0, be Θ -elliptic and symmetric, with spectrum $\sigma(A)$. If $f \in S^{\beta}(\sigma(A)), \beta \in \mathbb{R}$, then the operator $f(A) \in op^{\beta r}(\Theta)$ defined in Theorem 2.2.3, satisfies

$$f(A) \in \bigcap_{\varepsilon > 0} \operatorname{OP}^{\beta r + \varepsilon}(\Theta).$$

To prove this theorem, we need to estimate the appropriate operator norms of the repeated commutators $\delta_{\Theta}^n(f(A))$. As mentioned in Section 1.3, this is exactly what multiple operator integrals are for. The proof of this theorem will hence be provided in Chapter 3, as a consequence of Theorem 3.4.2.

Remark 2.2.8. It is unknown to the author whether the conclusion of Theorem 2.2.7 can be strengthened to

$$f(A) \in \mathrm{OP}^{\beta r}(\Theta).$$

2.3 Functional calculus for zero-order operators

In Section 2.2 we proved that Θ -elliptic symmetric operators in $\operatorname{op}^r(\Theta)$ for r > 0 admit a functional calculus. The approach of that section does not apply for the case r = 0. To illustrate how different the zero-order case is, consider the situation where $\Theta = (1 - \Delta)^{\frac{1}{2}}$ on $L_2(M)$ where M is a compact subset of \mathbb{R}^d . We have that for $\phi : M \to \mathbb{R}$, the multiplication operator

$$M_{\phi}: L_2(M) \to L_2(M)$$

 $\xi \mapsto \phi \cdot \xi,$

where $\phi \cdot \xi(x) = \phi(x)\xi(x)$, is in $\operatorname{op}^0(1-\Delta)^{\frac{1}{2}}$ if and only if ϕ is smooth. If $f(M_{\phi}) \in \operatorname{op}^0(1-\Delta)^{\frac{1}{2}}$ for all $M_{\phi} \in \operatorname{op}^0(1-\Delta)^{\frac{1}{2}}$, then the identity

$$f(M_{\phi}) = M_{f \circ \phi}$$

shows that the function f has to be smooth itself and no functional calculus with general functions in $L^{\beta}_{\infty}(E)$ is possible; smoothness is a necessary condition.

For $op^0(\Theta)$ we therefore use a different strategy altogether. An approach by Davies [Dav95c; Dav95b] on the construction of a functional calculus using almost analytic extensions directly applies. The technique was introduced by Hörmander [Hör69; Hör70], and subsequently used in various contexts by many authors, though the formula is often called the Helffer–Sjöstrand formula due to their independent rediscovery in [HS88]. For detailed notes on the historical origins, see [Hör69]. Using almost analytic extensions to obtain a functional calculus for pseudodifferential operators has precedent in the works of, amongst others, Hörmander [Hör69], Dynkin [Dyn70; Dyn72], Helffer–Sjöstrand [HS88], Dimassi–Sjöstrand [DS99, Chapter 8], and Bony [Bon13]. In an interesting twist, some of the earliest applications by Helffer and Sjöstrand of this formula occured in the study of the Schrödinger operator and its density of states [HS88; HS89; Sjö91], the main subject of Chapters 5 and 6 in this thesis (but in which these extensions make no appearance). For the details of this construction we follow Davies [Dav95b][Dav95c, Section 2.2].

Definition 2.3.1 ([Dav95b]). Let $f \in C_c^{\infty}(\mathbb{R})$. Recall that $\langle x \rangle := (1 + |x|^2)^{\frac{1}{2}}$. We define an extension $\tilde{f} : \mathbb{C} \to \mathbb{C}$ by

$$\tilde{f}(x+iy) := \tau(y/\langle x \rangle) \sum_{k=0}^{N} f^{(k)}(x) \frac{(iy)^k}{k!},$$

where $N \ge 1$ and $\tau : \mathbb{R} \to \mathbb{R}$ is a smooth bump function with $\tau(s) = 0$ for |s| > 2, $\tau(s) = 1$ for |s| < 1. Then we have

$$f(x) = -\frac{1}{\pi} \int_{\mathbb{C}} \left(\frac{\partial \tilde{f}}{\partial \overline{z}}(z) \right) (z - x)^{-1} dz, \quad x \in \mathbb{R},$$

independent of the choice of τ and N. We refer to \tilde{f} as an almost analytic extension of f. Definition 2.3.2. We define $T^{\beta}(I)$ as the space of smooth functions $f: I \to \mathbb{C}$ such that

$$\|f\|_{T^{\beta}(I),k} := \int_{I} |f^{(k)}(x)| \langle x \rangle^{k-\beta-1} \, dx < \infty, \quad k \in \mathbb{N}.$$

Recall that we previously put

$$||f||_{S^{\beta}(I),k} = \sup_{x \in I} |f^{(k)}(x)| \langle x \rangle^{k-\beta} < \infty, \quad k \in \mathbb{N}.$$

Hence, in case $I = \mathbb{R}$, we note the inclusions

$$\bigcup_{\alpha < \beta} S^{\alpha}(\mathbb{R}) \subsetneq T^{\beta}(\mathbb{R}) \subsetneq S^{\beta}(\mathbb{R}) \subsetneq L_{\infty}^{\beta}(E), \quad \beta \in \mathbb{R},$$

for any spectral measure E.

Theorem 2.3.3 ([Dav95b]). Let $f \in C_c^{\infty}(\mathbb{R})$ with almost analytic extension \tilde{f} as in Definition 2.3.1, so that

$$f(x) = -\frac{1}{\pi} \int_{\mathbb{C}} \left(\frac{\partial \tilde{f}}{\partial \overline{z}}(z) \right) (z-x)^{-1} dz, \quad x \in \mathbb{R}.$$

For any closed, densely defined operator H with $\sigma(H) \subseteq \mathbb{R}$, if for some $\alpha \in \mathbb{R}_{\geq 0}$ we have the estimate

$$\|(z-H)^{-1}\|_{\infty} \le C \frac{1}{|\Im(z)|} \left(\frac{\langle z \rangle}{|\Im(z)|}\right)^{\alpha}, \quad z \in \mathbb{C} \subseteq \mathbb{R},$$

then we have that

$$f(H) := -\frac{1}{\pi} \int_{\mathbb{C}} \frac{\partial \tilde{f}}{\partial \overline{z}} (z - H)^{-1} dz$$

defines a bounded operator on \mathcal{H} independent of the choice of $N > \alpha$ and τ in the construction of the extension \tilde{f} , with

$$\|f(H)\|_{\infty} \leq \sum_{k=0}^{N+1} \|f\|_{T^{0}(\mathbb{R}),k}.$$

The integral should be interpreted as a $B(\mathcal{H})$ -valued Bochner integral. In case H is selfadjoint, this agrees with the continuous functional calculus.

We thank Dmitriy Zanin for providing a key step in the following proof, which is an adaptation of an argument by Beals [Bea77, Lemma 3.1].

Proposition 2.3.4. Let $X \in op^r(\Theta)$ be such that $[\Theta, X] \in op^r(\Theta)$. If the extension

$$X:\mathcal{H}^{s_0+r}\to\mathcal{H}^{s_0}$$

has a bounded inverse

$$X^{-1}: \mathcal{H}^{s_0} \to \mathcal{H}^{s_0+r}$$

for one particular $s_0 \in \mathbb{R}$, then $X^{-1}\Big|_{\mathcal{H}^{\infty}} \in \operatorname{op}^{-r}(\Theta)$. We have $XX^{-1}|_{\mathcal{H}^{\infty}} = X^{-1}X|_{\mathcal{H}^{\infty}} = 1_{\mathcal{H}^{\infty}}$. In particular, if $X \in \operatorname{op}^{r}(\Theta)$ and $[\Theta, X] \in \operatorname{op}^{r}(\Theta)$ with $r \geq 0$, then we have as (unbounded) operators

$$\sigma(X:\mathcal{H}^{s_0+r}\subseteq\mathcal{H}^{s_0}\to\mathcal{H}^{s_0})=\sigma(X:\mathcal{H}^{s+r}\subseteq\mathcal{H}^s\to\mathcal{H}^s)$$

for all $s \in \mathbb{R}$, where σ denotes the spectrum of the operator.

Proof. Since $op^r(\Theta) = op^0(\Theta) \cdot \Theta^r$, it suffices to prove the proposition for r = 0.

Suppose that $X \in \operatorname{op}^0(\Theta)$ is a bijection on $\mathcal{H}^{s_0} \to \mathcal{H}^{s_0}$ and write $X^{-1} : \mathcal{H}^{s_0} \to \mathcal{H}^{s_0}$. Then X restricts to a necessarily injective map $\mathcal{H}^{s_0+1} \to \mathcal{H}^{s_0+1}$. We now prove that $X : \mathcal{H}^{s_0+1} \to \mathcal{H}^{s_0+1}$ is also surjective.

We follow [Bea77, Lemma 3.1], filling in some omitted details. Take $v \in \mathcal{H}^{s_0+1}$, then there exists $u \in \mathcal{H}^{s_0}$ with Xu = v. Let $\varepsilon > 0$, then $\frac{\Theta}{1+\varepsilon\Theta} \in \operatorname{op}^0(\Theta)$, and

$$\frac{\Theta}{1+\varepsilon\Theta}u = \frac{\Theta}{1+\varepsilon\Theta}X^{-1}v$$
$$= X^{-1}\frac{\Theta}{1+\varepsilon\Theta}v + X^{-1}\left[\frac{\Theta}{1+\varepsilon\Theta}, X\right]X^{-1}v.$$

Now, the \mathcal{H}^{s_0} norm of the right-hand side is bounded independent of ε :

$$\begin{split} \left\| \frac{\Theta}{1+\varepsilon\Theta} \right\|_{\mathcal{H}^{s_0+1} \to \mathcal{H}^{s_0}} &\leq \|\Theta\|_{\mathcal{H}^{s_0+1} \to \mathcal{H}^{s_0}};\\ \left\| \left[\frac{\Theta}{1+\varepsilon\Theta}, X \right] \right\|_{\mathcal{H}^{s_0} \to \mathcal{H}^{s_0}} &= \| (1+\varepsilon\Theta)^{-1} [\Theta, X] (1+\varepsilon\Theta)^{-1} \|_{\mathcal{H}^{s_0} \to \mathcal{H}^{s_0}} \leq \| [\Theta, X] \|_{\mathcal{H}^{s_0} \to \mathcal{H}^{s_0}}. \end{split}$$

This implies that $u \in \mathcal{H}^{s_0+1}$, a fact that can be quickly verified with the spectral theorem and Fatou's lemma. Therefore,

$$X: \mathcal{H}^{s_0+1} \to \mathcal{H}^{s_0+1}$$

is a bijection. By induction and interpolation (Proposition 2.1.2), the same assertion holds for each \mathcal{H}^s , $s \ge s_0$.

Finally, it is a basic fact that the adjoint of a bijective operator is bijective, i.e.

$$X'^{s_0} = X^{\dagger}|_{\mathcal{H}^{-s_0}} : \mathcal{H}^{-s_0} \to \mathcal{H}^{-s_0}$$

is a bijection (recall the notation from Section 2.1.2). Since

$$[\Theta, X^{\dagger}] = -[\Theta, X]^{\dagger} \in \mathrm{op}^{0}(\Theta),$$

we can apply the same arguments as above to deduce that

$$X^{\dagger}: \mathcal{H}^{-s} \to \mathcal{H}^{-s}$$

is a bijection for all $-s \ge -s_0$. This implies that

$$X = X^{\dagger\dagger} : \mathcal{H}^s \to \mathcal{H}^s$$

is a bijection for all $s \leq s_0$.

As an aside, one may wonder how for r > 0 the condition $A, [\Theta, A] \in \operatorname{op}^{r}(\Theta)$ compares to the condition of A being Θ -elliptic used in the previous section. The following consequence of Proposition 2.3.4 shows that, under the assumption that A is self-adjoint with domain \mathcal{H}^{r} , Θ -ellipticity is a weaker condition.

Corollary 2.3.5. Let $A \in op^r(\Theta), r > 0$, be such that

$$A:\mathcal{H}^r\subseteq\mathcal{H}^0\to\mathcal{H}^0$$

has a non-empty resolvent set (for example if A is self-adjoint with domain \mathcal{H}^r), and suppose that $[\Theta, A] \in \operatorname{op}^r(\Theta)$ as well. Then A is Θ -elliptic.

Proof. By assumption, there exists $z \in \mathbb{C}$ such that

$$z - A : \mathcal{H}^r \subseteq \mathcal{H}^0 \to \mathcal{H}^0$$

is invertible, and because $[\Theta, A] \in \operatorname{op}^{r}(\Theta)$ it follows that $(z - A)^{-1} \in \operatorname{op}^{-r}(\Theta)$ by Proposition 2.3.4. Now,

$$A(z-A)^{-1} = -1 + z(z-A)^{-1}.$$

In other words, $-(z - A)^{-1}$ is an inverse of A modulo $\operatorname{op}^{-r}(\Theta)$. Corollary 2.1.10 gives that A is Θ -elliptic.

This following type of estimate on the resolvent also appears in L_p -boundedness problems, see [Dav95a; Dav95b; JN94].

Lemma 2.3.6. Let $A \in \operatorname{op}^{0}(\Theta)$ be such that $[\Theta, A] \in \operatorname{op}^{0}(\Theta)$ and $\overline{A}^{0,0} : \mathcal{H} \to \mathcal{H}$ is self-adjoint. Then for all $s \in \mathbb{R}$, there is a constant $C_s > 0$ such that

$$\|(z-A)^{-1}\|_{\mathcal{H}^s \to \mathcal{H}^s} \le C_s \frac{1}{|\Im(z)|} \left(\frac{\langle z \rangle}{|\Im(z)|}\right)^{2^{|s|}-1}, \quad z \in \mathbb{C} \setminus \mathbb{R}.$$

Proof. The proof is by induction and interpolation (Proposition 2.1.2). For s = 0, the estimate holds by self-adjointness of $\overline{A}^{0,0}$. Note that $(z - A)^{-1} \in \operatorname{op}^0(\Theta)$ due to Proposition 2.3.4.

Suppose the inequality is proved for a fixed $s \in \mathbb{R}_{\geq 0}$. Then for $z \in \mathbb{C} \setminus \mathbb{R}$,

$$\begin{split} \|(z-A)^{-1}\|_{\mathcal{H}^{s+1}\to\mathcal{H}^{s+1}} \\ &= \|\Theta(z-A)^{-1}\Theta^{-1}\|_{\mathcal{H}^{s}\to\mathcal{H}^{s}} \\ &\leq \|(z-A)^{-1}\|_{\mathcal{H}^{s}\to\mathcal{H}^{s}} + \|(z-A)^{-1}[\Theta,A](z-A)^{-1}\Theta^{-1}\|_{\mathcal{H}^{s}\to\mathcal{H}^{s}} \\ &\leq \|(z-A)^{-1}\|_{\mathcal{H}^{s}\to\mathcal{H}^{s}} \left(1 + \|[\Theta,A](i+A)^{-1}\|_{\mathcal{H}^{s}\to\mathcal{H}^{s}}\|(i+A)(z-A)^{-1}\|_{\mathcal{H}^{s}\to\mathcal{H}^{s}}\|\Theta^{-1}\|_{\mathcal{H}^{s}\to\mathcal{H}^{s}}\right). \end{split}$$

Note that $(i + A)^{-1} \in \operatorname{op}^0(\Theta)$ by Proposition 2.3.4, so that for some constant $B_s > 0$,

 $\|[\Theta, A](i+A)^{-1}\|_{\mathcal{H}^s \to \mathcal{H}^s}\|\Theta^{-1}\|_{\mathcal{H}^s \to \mathcal{H}^s} \le B_s.$

Using the resolvent identity, we have

$$(i+A)(z-A)^{-1} = (i+z)(z-A)^{-1} - 1$$

and therefore

$$||(i+A)(z-A)^{-1}||_{\mathcal{H}^s \to \mathcal{H}^s} \le 1 + |z+i|||(z-A)^{-1}||_{\mathcal{H}^s \to \mathcal{H}^s}.$$

This yields

$$\begin{aligned} \|(z-A)^{-1}\|_{\mathcal{H}^{s+1}\to\mathcal{H}^{s+1}} &\leq \|(z-A)^{-1}\|_{\mathcal{H}^s\to\mathcal{H}^s}(1+B_s) \\ &+ |z+i|B_s\|(z-A)^{-1}\|_{\mathcal{H}^s\to\mathcal{H}^s}^2 \\ &\leq (1+B_s)\|(z-A)^{-1}\|_{\mathcal{H}^s\to\mathcal{H}^s} \cdot (1+|z+i|\|(z-A)^{-1}\|_{\mathcal{H}^s\to\mathcal{H}^s}). \end{aligned}$$
This estimate also holds with |z - i| on the right-hand side, and $\min(|z + i|, |z - i|) \le \langle z \rangle$, so that

$$\|(z-A)^{-1}\|_{\mathcal{H}^{s+1}\to\mathcal{H}^{s+1}} \le (1+B_s)\|(z-A)^{-1}\|_{\mathcal{H}^s\to\mathcal{H}^s} \cdot (1+\langle z \rangle \|(z-A)^{-1}\|_{\mathcal{H}^s\to\mathcal{H}^s}),$$

from which the claimed estimate follows. Induction and interpolation now provide the estimate for all $s \ge 0$.

The case $s \leq 0$ is proved in the same manner, using induction in the negative direction. Namely, the norm

$$\|(z-A)^{-1}\|_{\mathcal{H}^{s-1}\to\mathcal{H}^{s-1}} = \|\Theta^{-1}(z-A)^{-1}\Theta\|_{\mathcal{H}^s\to\mathcal{H}^s}$$

can be estimated as before.

We now arrive at the main result of this section, a functional calculus for $op^0(\Theta)$.

Theorem 2.3.7. Let $A \in \operatorname{op}^{0}(\Theta)$ be such that $[\Theta, A] \in \operatorname{op}^{0}(\Theta)$ and $\overline{A}^{0,0} : \mathcal{H} \to \mathcal{H}$ is self-adjoint. For $f \in C^{\infty}(\mathbb{R})$,

$$f(A) \in \mathrm{op}^0(\Theta).$$

Specifically, we have the estimate

$$||f(A)||_{\mathcal{H}^s \to \mathcal{H}^s} \le \sum_{k=0}^{\lceil 2^{|s|} \rceil + 1} ||f||_{T^0(\mathbb{R}),k}.$$

Proof. Without loss of generality, we assume that $f \in C_c^{\infty}(\mathbb{R})$. As $A \in op^0(\Theta)$, it extends to a bounded operator

$$\overline{A}^{s,s}:\mathcal{H}^s\to\mathcal{H}^s,\quad s\in\mathbb{R}.$$

Furthermore, by Proposition 2.1.11, we have

$$(z-A)^{-1} \in \operatorname{op}^0(\Theta), \quad z \in \mathbb{C} \setminus \mathbb{R}.$$

Theorem 2.3.3 and Lemma 2.3.6 combined give that

$$f(\overline{A}^{s,s}): \mathcal{H}^s \to \mathcal{H}^s, \quad s \in \mathbb{R},$$

is a bounded operator with the norm bound as claimed. By construction, for $\xi\in\mathcal{H}^\infty$ we have that

$$f(\overline{A}^{s,s})\xi = \int_{\mathbb{C}} \frac{\partial \tilde{f}}{\partial \overline{z}} (z - \overline{A}^{s,s})^{-1} \xi \, dz \in \mathcal{H}^s,$$

as an \mathcal{H}^s -valued Bochner integral. It is clear that

$$(z - \overline{A}^{s,s})^{-1}\Big|_{\mathcal{H}^{\infty}} = (z - A)^{-1},$$

and therefore these arguments show that for $\xi \in \mathcal{H}^{\infty}$ the integral

$$\int_{\mathbb{C}} \frac{\partial \tilde{f}}{\partial \overline{z}} (z - A)^{-1} \xi \, dz$$

can be evaluated as a Bochner integral in each Sobolev space \mathcal{H}^s . Hence

$$\xi \mapsto \int_{\mathbb{C}} \frac{\partial \tilde{f}}{\partial \overline{z}} (z - A)^{-1} \xi \, dz$$

forms a bounded linear map on \mathcal{H}^{∞} ; denote this operator by $f(A) : \mathcal{H}^{\infty} \to \mathcal{H}^{\infty}$. Then since f(A) agrees with $f(\overline{A}^{s,s})$ on \mathcal{H}^{∞} , we must have

$$\overline{f(A)}^{s,s} = f(\overline{A}^{s,s}),$$

and thus we have $f(A) \in \operatorname{op}^0(\Theta)$.

Similarly to the Θ -elliptic case, we also provide a theorem for $OP^0(\Theta)$ (recall Definition 2.1.3) for which the proof will follow in Chapter 3 using a multiple operator integral technique. Specifically, it is a consequence of Theorem 3.4.2.

Theorem 2.3.8. Let $A \in OP^0(\Theta)$ be such that $\overline{A}^{0,0} : \mathcal{H} \to \mathcal{H}$ is self-adjoint. For $f \in C^{\infty}(\mathbb{R})$,

$$f(A) \in \mathrm{OP}^0(\Theta).$$

Theorem 2.3.7 and Theorem 2.3.8 are consistent with [Bon13, Theorem 3], which provides a functional calculus for zero-order pseudodifferential operators on manifolds for $f \in C^{\infty}(\mathbb{R})$. Since our pseudodifferential calculus is vastly more general, our results are both more general and less specific than that in [Bon13].

Chapter 3

Multiple operator integrals as abstract pseudodifferential operators

Thanks again, and work even harder!!!

Fedor Sukochev

Like Chapter 2, the content and exposition in this chapter has appeared previously as part of the preprint [HMN24], joint work with Edward McDonald and Teun van Nuland. For this chapter, we are indebted to discussions with Fedor Sukochev and Dmitriy Zanin. The main result of this chapter, Theorem 3.2.5, is a construction of multiple operator integrals adapted to the abstract pseudodifferential calculus of Connes and Moscovici. The other main results of this chapter are a more specific version of this MOI construction in Theorem 3.3.3, a noncommutative Taylor expansion in Theorem 3.5.5, and a result on the existence of asymptotic trace expansions in spectral triples in Corollary 3.6.6.

Due to the results in Chapter 2, we have a functional calculus for abstract pseudodifferential operators $A \in op(\Theta)$. As explained in Section 1.3, it would be incredibly

CHAPTER 3. MULTIPLE OPERATOR INTEGRALS AS ABSTRACT PSEUDODIFFERENTIAL OPERATORS

useful in the field of noncommutative geometry if we can define multiple operator integrals (MOIs) for this abstract pseudodifferential calculus. There are two closely related constructions we could employ here: Peller's approach [Pel06; Pel16] and Azamov–Carey– Dodds–Sukochev's [ACDS09]. The former works with a very general class of symbol functions, constructing MOIs for operators on Hilbert spaces. The latter considers a more restricted class of functions and achieves stronger results as a consequence (Fréchet versus Gateaux differentiability), and notably achieves this in the more general setting of semi-finite von Neumann algebras. Since we wish to keep our symbol functions as general as possible and the results in Peller's work suffice for our purposes, we will use Peller's approach as a blueprint to construct MOIs of pseudodifferential operators.

In Section 3.5 we will provide asymptotic expansions for these MOIs including a noncommutative Taylor expansion, which we will apply in Section 3.6 to noncommutative geometry, answering a previously open question on the existence of asymptotic expansions of heat traces in spectral triples.

Throughout this chapter, we fix a positive invertible operator Θ on the separable Hilbert space \mathcal{H} , which yields a scale of separable Hilbert spaces $\{\mathcal{H}^s\}_{s\in\mathbb{R}}$ by Definition 2.1.1.

3.1 Operator integrals

We start with standard definitions and results on measurability and integrability of operator valued functions.

Definition 3.1.1. Let $\mathcal{H}_0, \mathcal{H}_1$ be separable Hilbert spaces and let (Ω, Σ, ν) be a measure space with complex measure. A function $f : \Omega \to B(\mathcal{H}_1, \mathcal{H}_0)$ is called weak operator topology measurable (weakly measurable for short) if for all $\eta \in \mathcal{H}_0, \xi \in \mathcal{H}_1$ the complexvalued function

$$\omega \mapsto \langle \eta, f(\omega)\xi \rangle_{\mathcal{H}_0}, \quad \omega \in \Omega,$$

is measurable. Similarly, f is said to be weak operator topology integrable if for all ξ and η the above map is integrable.

Lemma 3.1.2. [LMSZ23, Lemma 1.4.2] Let $\mathcal{H}_0, \mathcal{H}_1, (\Omega, \Sigma, \nu)$ be as above, and let $f : \Omega \to B(\mathcal{H}_1, \mathcal{H}_0)$ be weakly measurable. Then the norm function

$$\omega \mapsto \|f(\omega)\|_{\mathcal{H}_1 \to \mathcal{H}_0}, \quad \omega \in \Omega,$$

is measurable. If moreover

$$\int_{\Omega} \|f(\omega)\|_{\mathcal{H}_1 \to \mathcal{H}_0} \, d|\nu|(\omega) < \infty,$$

then there exists a unique $I_f \in B(\mathcal{H}_1, \mathcal{H}_0)$ such that

$$\langle \eta, I_f \xi \rangle_{\mathcal{H}_0} = \int_{\Omega} \langle \eta, f(\omega) \xi \rangle_{\mathcal{H}_0} \, d\nu(\omega), \quad \eta \in \mathcal{H}_0, \xi \in \mathcal{H}_1,$$

and

$$\|I_f\|_{\mathcal{H}_1 \to \mathcal{H}_0} \le \int_{\Omega} \|f(\omega)\|_{\mathcal{H}_1 \to \mathcal{H}_0} \, d|\nu|(\omega).$$

We then write $I_f = \int_{\Omega} f(\omega) d\nu(\omega)$.

Proposition 3.1.3. [DDSZ20, Lemma 3.11] Let (Ω, Σ, ν) be a σ -finite measure space, let $a : \mathbb{R} \times \Omega \to \mathbb{C}$ be measurable and bounded, and let H be an (unbounded) self-adjoint operator on \mathcal{H} . Then

$$\omega \mapsto a(H,\omega)$$

is weakly measurable.

Proof. Though [DDSZ20, Lemma 3.11] is only formulated for bounded H, the unbounded case follows with the same proof.

Lemma 3.1.4. Let $\mathcal{H}_0, \ldots, \mathcal{H}_{2n+1}$ be separable Hilbert spaces, $X_i \in B(\mathcal{H}_{2i}, \mathcal{H}_{2i-1})$, and let $f_i : \Omega \to B(\mathcal{H}_{2i+1}, \mathcal{H}_{2i})$ be weakly measurable functions. Then

$$\Omega \to B(\mathcal{H}_{2n+1}, \mathcal{H}_0)$$
$$\omega \mapsto f_0(\omega) X_1 f_1(\omega) \cdots X_n f_n(\omega)$$

is weakly measurable. Furthermore, if

$$\int_{\Omega} \|f_0(\omega)\|_{\mathcal{H}_1 \to \mathcal{H}_0} \cdots \|f_n(\omega)\|_{\mathcal{H}_{2n+1} \to \mathcal{H}_{2n}} \, d|\nu|(\omega) < \infty$$

the map

$$B(\mathcal{H}_1, \mathcal{H}_0) \times \cdots \times B(\mathcal{H}_{2n-1}, \mathcal{H}_{2n-2}) \to B(\mathcal{H}_{2n}, \mathcal{H}_0)$$
$$(X_1, \dots, X_n) \quad \mapsto \int_{\Omega} f_0(\omega) X_1 f_1(\omega) \cdots X_n f_n(\omega) \, d\nu(\omega),$$

whose existence follows from Lemma 3.1.2, is so-continuous when restricted to the unit ball in each argument $B(\mathcal{H}_{2i}, \mathcal{H}_{2i-1})$.

Proof. The first part of the lemma is a consequence of the fact that the pointwise product of weakly measurable functions is weakly measurable, see [DDSZ20, Lemma 3.7].

The so-continuity follows from the joint continuity of the multiplication

$$(X_1,\ldots,X_n)\mapsto a_0(H_0,\omega)X_1a_1(H_1,\omega)\cdots X_na_n(H_n,\omega)$$

in the strong operator topology when restricting to the unit balls [Bla06, Section I.3.2], in combination with the Dominated Convergence Theorem for the Bochner integral of Hilbert space-valued functions [DS88, Corollary III.6.16]. \Box

Recall the definition of $L^{\beta}_{\infty}(E)$ in Definition 2.2.2, and that we use the notation $\|\cdot\|_{\infty}$ for the essential supremum.

Lemma 3.1.5. Let (Ω, Σ, ν) be a σ -finite measure space, let $a : \mathbb{R} \times \Omega \to \mathbb{C}$ be measurable and bounded, and let E be a spectral measure on \mathcal{H} . Then the functions

$$\omega \mapsto \|a(\cdot,\omega)\|_{\infty}, \qquad \omega \mapsto \|a(\cdot,\omega)\|_{L^{0}_{\infty}(E)}$$

are measurable.

Proof. Both claims can be proved with the Fubini–Tonelli Theorem or by combining Lemma 3.1.2 and Proposition 3.1.3. It is vital that $\|\cdot\|_{\infty}$ is the essential supremum, and that E is a spectral measure on a *separable* Hilbert space as pointed out in [Nik23, Remark 4.1.3].

3.2 Multiple operator integrals as abstract pseudodifferential operators

In this section we construct multiple operator integrals following Peller [Pel06; Pel16]. His MOIs are constructed with symbols ϕ in integral projective tensor products $L_{\infty}(E_0)\hat{\otimes}_i \cdots \hat{\otimes}_i L_{\infty}(E_n)$ for spectral measures E_0, \ldots, E_n . A precise study of this integral tensor product can be found in [Nik23]. We first generalise this integral tensor product.

Definition 3.2.1. Let $\Gamma_0, \ldots, \Gamma_n$ be function spaces of bounded measurable functions $\mathbb{R} \to \mathbb{R}$ equipped with (semi)norms $\|\cdot\|_{\Gamma_i,k}$, $k \in \mathbb{N}$. We define $\Gamma_0 \boxtimes_i \cdots \boxtimes_i \Gamma_n$ as the set of functions $\phi : \mathbb{R}^{n+1} \to \mathbb{C}$ for which there exists a decomposition

$$\phi(\lambda_0, \dots, \lambda_n) = \int_{\Omega} a_0(\lambda_0, \omega) \cdots a_n(\lambda_n, \omega) \, d\nu(\omega)$$
(3.1)

where (Ω, ν) is a σ -finite measure space, $a_i : \mathbb{R} \times \Omega \to \mathbb{C}$ is measurable, $a_i(\cdot, \omega) \in \Gamma_i$, the functions $\omega \mapsto ||a_i(\cdot, \omega)||_{\Gamma_i, k}$ are measurable for each i and k, and

$$\int_{\Omega} \|a_0(\cdot,\omega)\|_{\Gamma_0,k_0} \cdots \|a_n(\cdot,\omega)\|_{\Gamma_n,k_n} \, d|\nu|(\omega) < \infty.$$
(3.2)

We define the seminorm

$$\|\phi\|_{\Gamma_0\boxtimes_i\cdots\boxtimes_i\Gamma_n,k_0,\ldots,k_n}, \quad k_0,\ldots,k_n \in \mathbb{N}$$

to be the infimum of the quantity (3.2) over all representations (3.1).

Remark 3.2.2. We have that

$$\Gamma_0 \otimes \cdots \otimes \Gamma_n \subseteq \Gamma_0 \boxtimes_i \cdots \boxtimes_i \Gamma_n$$

where \otimes denotes the algebraic tensor product for topological vector spaces. We refrain from answering the question whether the space in Definition 3.2.1 is in general the completion of the algebraic tensor product under the given seminorms.

Note that due to Lemma 3.1.5, if $a : \mathbb{R} \times \Omega \to \mathbb{C}$ is measurable and $a(\cdot, \omega) \in S^{\beta}(\mathbb{R})$, we have that $\omega \mapsto ||a(\cdot, \omega)||_{S^{\beta}(\mathbb{R}),k}$ is measurable, and the same claim holds for $T^{\beta}(\mathbb{R})$ and $L^{\beta}_{\infty}(E)$ (recall the definitions of the spaces $S^{\beta}(\mathbb{R})$, $T^{\beta}(\mathbb{R})$ and $L^{\beta}_{\infty}(E)$ respectively in Definition 2.2.6, Definition 2.3.2 and Definition 2.2.2). Hence the construction in Definition 3.2.1 can be applied for these spaces without this extra assumption.

For $L^{\beta}_{\infty}(E)$ with E a spectral measure, this gives the integral projective tensor product

$$L_{\infty}^{\beta_0}(E_0) \boxtimes_i \cdots \boxtimes_i L_{\infty}^{\beta_n}(E_n) = L_{\infty}^{\beta_0}(E_0) \hat{\otimes}_i \cdots \hat{\otimes}_i L_{\infty}^{\beta_n}(E_n),$$

which appears in particular for $\beta_0 = \cdots = \beta_n = 0$ in the works by Peller [Pel06, pp.6, 7].

Remark 3.2.3. Observe that

$$S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R}) \subseteq L^{\beta_0}_{\infty}(E_0) \hat{\otimes}_i \cdots \hat{\otimes}_i L^{\beta_n}_{\infty}(E_n)$$

no matter what spectral measures E_i are taken.

Proposition 3.2.4. If $\phi \in S^{\alpha_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\alpha_n}(\mathbb{R})$ and $\psi \in S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R})$, then

$$\Phi(\lambda_0,\ldots,\lambda_n) := \phi(\lambda_0,\ldots,\lambda_n)\psi(\lambda_0,\ldots,\lambda_n) \in S^{\alpha_0+\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\alpha_n+\beta_n}(\mathbb{R}).$$

An analogous statement holds for the spaces $L^{\beta}_{\infty}(E)$.

Proof. According to Definition 3.2.1, we can find σ -finite measure spaces (Ω, ν) , (Σ, μ) and measurable functions $a_i : \mathbb{R} \times \Omega \to \mathbb{C}$, $b_i : \mathbb{R} \times \Sigma \to \mathbb{C}$ such that

$$\phi(\lambda_0, \dots, \lambda_n) = \int_{\Omega} a_0(\lambda_0, \omega) \cdots a_n(\lambda_n, \omega) \, d\nu(\omega);$$

$$\psi(\lambda_0, \dots, \lambda_n) = \int_{\Sigma} b_0(\lambda_0, \sigma) \cdots b_n(\lambda_n, \sigma) \, d\mu(\sigma).$$

As observed above, the maps

$$\omega \mapsto \|a_i(\cdot,\omega)\|_{S^{\alpha_i}(\mathbb{R}),k}$$

are measurable, similarly for the functions b_i .

Using Tonelli's theorem,

Hence, by Fubini's theorem

$$\Phi(\lambda_0,\ldots,\lambda_n) = \int_{\Omega\times\Sigma} a_0(\lambda_0,\omega) b_0(\lambda_0,\sigma) \cdots a_n(\lambda_n,\omega) b_n(\lambda_n,\sigma) d(\nu\times\mu)(\omega,\sigma).$$

The fact that $\Phi \in S^{\alpha_0+\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\alpha_n+\beta_n}(\mathbb{R})$ now follows from the computation

$$\|a_k(\cdot,\omega)b_k(\cdot,\sigma)\|_{S^{\alpha_k+\beta_k},m} \leq \sum_{j=0}^m \binom{m}{j} \|a_k(\cdot,\omega)\|_{S^{\alpha_k},j} \|b_k(\cdot,\sigma)\|_{S^{\beta_k},m-j}.$$

The following theorem is the first main result of this chapter. It is a construction of multiple operator integrals adapted to the abstract pseudodifferential calculus by Connes and Moscovici which we have seen in Chapter 2.

Theorem 3.2.5. Let $H_i \in \operatorname{op}^{h_i}(\Theta)$, $h_i > 0$, $i = 0, \ldots, n$, be Θ -elliptic symmetric operators with spectral measures E_i , and let $\phi \in L^{\beta_0}_{\infty}(E_0) \boxtimes_i \cdots \boxtimes_i L^{\beta_n}_{\infty}(E_n)$, given with explicit representation

$$\phi(\lambda_0,\ldots,\lambda_n) = \int_{\Omega} a_0(\lambda_0,\omega)\cdots a_n(\lambda_n,\omega) \,d\nu(\omega), \quad \lambda_i \in \sigma(H_i), i = 0,\ldots,n.$$

Then the integral

$$T_{\phi}^{H_0,\dots,H_n}(X_1,\dots,X_n)\psi := \int_{\Omega} a_0(H_0,\omega)X_1a_1(H_1,\omega)\cdots X_na_n(H_n,\omega)\psi\,d\nu(\omega), \quad \psi \in \mathcal{H}^{\infty},$$

for $X_1, \ldots, X_n \in \operatorname{op}(\Theta)$, converges as a Bochner integral in \mathcal{H}^s for every $s \in \mathbb{R}$, and defines an n-multilinear map

$$T_{\phi}^{H_0,\dots,H_n}: \operatorname{op}^{r_1}(\Theta) \times \dots \times \operatorname{op}^{r_n}(\Theta) \to \operatorname{op}^{\sum_j r_j + \sum_j \beta_j h_j}(\Theta)$$

depending on Ω and the functions a_0, \ldots, a_n only through the symbol ϕ . For $s \in \mathbb{R}$ we have the estimate

$$\left\|T_{\phi}^{H_0,\dots,H_n}(X_1,\dots,X_n)\right\|_{\mathcal{H}^{s+\sum_j r_j+\sum_j \beta_j h_j} \to \mathcal{H}^s} \lesssim \prod_{j=1}^n \|X_j\|_{\mathcal{H}^{s_j+r_j} \to \mathcal{H}^{s_j}} \|\phi\|_{L^{\beta_0}_{\infty}(E_0)\boxtimes_i \dots \boxtimes_i L^{\beta_n}_{\infty}(E_n)}$$

for some specific $s_1, \ldots, s_n \in \mathbb{R}$ depending on h_j, r_j, β_j .

For $\Theta = 1_{\mathcal{H}}$ and $\beta_0 = \cdots = \beta_n = 0$, it is immediate that $a_j(H_j, \omega) \in \text{op}(1_{\mathcal{H}}) = B(\mathcal{H})$ and

$$\|a_j(H_j,\omega)\|_{\mathcal{H}^s \to \mathcal{H}^s} = \|a_j(\cdot,\omega)\|_{L^0_\infty(E_j)},$$

and the above theorem reduces to Peller's construction of MOIs [Pel06; Pel16]. Hence, this theorem is a strict generalisation of his results. The proof is a subtle modification of (aspects of) the proofs presented in [ACDS09; Pel06].

Proof of Theorem 3.2.5. Fix the Θ -elliptic symmetric operators $H_i \in \operatorname{op}^{h_i}(\Theta), h_i > 0$, $i = 0, \ldots, n$, with spectral measures E_i , and the function $\phi : \mathbb{R}^{n+1} \to \mathbb{C}$

$$\phi(\lambda_0,\ldots,\lambda_n) = \int_{\Omega} a_0(\lambda_0,\omega)\cdots a_n(\lambda_n,\omega) \, d\nu(\omega),$$

where (Ω, ν) is a finite measure space and the functions $a_j : \mathbb{R} \times \Omega \to \mathbb{C}$ are measurable and such that $(x, \omega) \mapsto a_j(x, \omega) \langle x \rangle^{-\beta_j}$ is $E_j \times \nu$ -a.e. bounded for $\beta_j \in \mathbb{R}$.

By Theorem 2.2.3, we have that $\mathcal{H}^{\infty} \subseteq \text{dom } a_j(H_j, \omega)$. Now fix $\omega \in \Omega$ and take $\eta, \xi \in \mathcal{H}^{\infty}$. Then [Sch12, Theorem 4.13] gives that

$$a_j(H_j,\omega)\xi = \lim_{n \to \infty} a_j(H_j,\omega)\chi_{[-n,n]}(H_j)\xi,$$

where $\chi_{[-n,n]}$ is the indicator function of the interval [-n,n], because $\operatorname{ess\,sup}_{|\lambda| \leq n} |a_j(\lambda,\omega)| < \infty$. Now Proposition 3.1.3 gives that

$$\omega \mapsto \langle \eta, a_j(H_j, \omega) \xi \rangle_{\mathcal{H}^s} = \lim_{n \to \infty} \langle \Theta^{2s} \eta, a_j(H_j, \omega) \chi_{[-n,n]}(H_j) \xi \rangle_{\mathcal{H}^s}$$

is measurable for all $s \in \mathbb{R}$.

Let now $X_i \in op^{r_i}(\Theta)$, i = 1, ..., n. Fix $s \in \mathbb{R}$ and define $s_0, ..., s_{2n+1} \in \mathbb{R}$ with

$$s_0 := s, \quad s_{2n+1} := s + \sum_{i=0}^n \beta_i h_i + \sum_{i=1}^n r_i,$$

so that the operators $a_j(H_j, \omega)$ and X_j extend to bounded operators

$$a_j(H_j,\omega) \in B(\mathcal{H}^{s_{2j+1}},\mathcal{H}^{s_{2j}}),$$
$$X_j \in B(\mathcal{H}^{s_{2j}},\mathcal{H}^{s_{2j-1}}).$$

By the previous argument,

$$\omega \mapsto a_j(H_j, \omega) \in B(\mathcal{H}^{s_{2j+1}}, \mathcal{H}^{s_{2j}})$$

is weakly measurable since \mathcal{H}^{∞} is dense in both $\mathcal{H}^{s_{2j}}$ and $\mathcal{H}^{s_{2j+1}}$.

Using the functional calculus in Theorem 2.2.3,

$$\|a_j(H_j,\omega)\|_{\mathcal{H}^{s_{2j+1}}\to\mathcal{H}^{s_{2j}}} \leq C_{s,H_j}\|a_j(\cdot,\omega)\|_{L^{\beta_j}_{\infty}(E_j)},$$

and so we have that

$$\begin{split} \int_{\Omega} \|a_0(H_0,\omega)X_1a_1(H_1,\omega)\cdots X_na_n(H_n,\omega)\|_{\mathcal{H}^{s_{2n+1}}\to\mathcal{H}^{s_0}}\,d\nu(\omega) \\ \lesssim \prod_{j=1}^n \|X_j\|_{\mathcal{H}^{s_{2j}}\to\mathcal{H}^{s_{2j-1}}}\int_{\Omega} \prod_{j=0}^n \|a_j(\cdot,\omega)\|_{L^{\beta_j}_{\infty}(E_j)}\,d|\nu|(\omega)<\infty, \end{split}$$

where Lemma 3.1.5 ensures the right-hand side is defined. This is a finite quantity since $a_j(x,\omega)\langle x \rangle^{-\beta_j}$ is $E_j \times \nu - a.e.$ bounded and ν is a finite measure space. Therefore, Lemma 3.1.2 provides that

$$\int_{\Omega} a_0(H_0,\omega) X_1 a_1(H_1,\omega) \cdots X_n a_n(H_n,\omega) \, d\nu(\omega)$$

defines an operator in the weak sense in $B(\mathcal{H}^{s_{2n+1}}, \mathcal{H}^{s_0})$ with

$$\left\| \int_{\Omega} a_0(H_0,\omega) X_1 a_1(H_1,\omega) \cdots X_n a_n(H_n,\omega) \, d\nu(\omega) \right\|_{\mathcal{H}^{s_{2n+1}} \to \mathcal{H}^{s_0}}$$

$$\lesssim \prod_{j=1}^n \|X_j\|_{\mathcal{H}^{s_{2j}} \to \mathcal{H}^{s_{2j-1}}} \int_{\Omega} \prod_{j=0}^n \|a_j(\cdot,\omega)\|_{L^{\beta_j}_{\infty}(E_j)} \, d|\nu|(\omega).$$
(3.3)

With Pettis' theorem [VTC87, Propositions 1.9 and 1.10], it now follows that for $\psi \in \mathcal{H}^{s_{2n+1}}$,

$$\omega \mapsto a_0(H_0, \omega) X_1 a_1(H_1, \omega) \cdots X_n a_n(H_n, \omega) \psi \in \mathcal{H}^s$$

is Bochner integrable in \mathcal{H}^s . This holds in particular for $\psi \in \mathcal{H}^\infty$, and as $s \in \mathbb{R}$ was taken arbitrarily it follows that for $\psi \in \mathcal{H}^\infty$,

$$\int_{\Omega} a_0(H_0,\omega) X_1 a_1(H_1,\omega) \cdots X_n a_n(H_n,\omega) \psi \, d\nu(\omega) \in \mathcal{H}^{\infty}.$$

It is therefore clear that we have a well-defined operator

$$\int_{\Omega} a_0(H_0,\omega) X_1 a_1(H_1,\omega) \cdots X_n a_n(H_n,\omega) \, d\nu(\omega) \in \operatorname{op}^{\sum_j r_j + \sum_j \beta_j h_j}(\Theta).$$

The claim that this operator is independent of the chosen representation of ϕ

$$\phi(\lambda_0,\ldots,\lambda_n) = \int_{\Omega} a_0(\lambda_0,\omega)\cdots a_n(\lambda_n,\omega) \, d\nu(\omega)$$

permits taking the infimum over all representations on the right-hand side of the estimate (3.3) giving the norm estimate in the assertion of this theorem. This independence follows from the proof of [ACDS09, Lemma 4.3]. Namely, given $\eta, \xi \in \mathcal{H}^{\infty}$, it is easy to check that $\theta_{\eta,\xi} : \mathcal{H}^{\infty} \to \mathcal{H}^{\infty}$ defined by

$$\theta_{\eta,\xi}(\psi) := \langle \eta, \psi \rangle_{\mathcal{H}} \xi, \quad \psi \in \mathcal{H}^{\infty},$$

is an element of $op^{-\infty}(\Theta)$. The computations in [ACDS09, Lemma 4.3] give that, for $\eta_k, \xi_k \in \mathcal{H}^{\infty}$, the integral

$$\int_{\Omega} a_0(H_0,\omega)\theta_{\eta_1,\xi_1}a_1(H_1,\omega)\cdots\theta_{\eta_n,\xi_n}a_n(H_n,\omega)\,d\nu(\omega)\in B(\mathcal{H})$$

does not depend on the chosen representation of ϕ , and so neither does

$$\int_{\Omega} a_0(H_0,\omega)\theta_{\eta_1,\xi_1}a_1(H_1,\omega)\cdots\theta_{\eta_n,\xi_n}a_n(H_n,\omega)\,d\nu(\omega)\Big|_{\mathcal{H}^{\infty}}\in \mathrm{op}^{-\infty}(\Theta).$$

The so-density of the span of $\{\theta_{\eta,\xi} : \eta, \xi \in \mathcal{H}^{\infty}\}$ in $B(\mathcal{H}^{s_{2i}}, \mathcal{H}^{s_{2i-1}})$ combined with Lemma 3.1.4 concludes the proof.

Proposition 3.2.6. The MOI constructed in Theorem 3.2.5 is linear in its symbol:

$$T^{H_0,\dots,H_n}_{\alpha\phi+\beta\psi}(X_1,\dots,X_n) = \alpha T^{H_0,\dots,H_n}_{\phi}(X_1,\dots,X_n) + \beta T^{H_0,\dots,H_n}_{\psi}(X_1,\dots,X_n), \quad \alpha,\beta \in \mathbb{C}.$$

Proof. If both $\phi, \psi : \mathbb{R}^{n+1} \to \mathbb{C}$ have an integral representation of the required form over measure spaces Ω and Σ respectively, then $\alpha \phi + \beta \psi$ can be decomposed appropriately as an integral over the disjoint union $\Omega \sqcup \Sigma$. The assertion then follows by elementary arguments.

Remark 3.2.7. The MOI constructed in Theorem 3.2.5 is independent of the operator Θ defining the abstract pseudodifferential calculus in the following sense. If H_i and X_i are operators on \mathcal{H} such that $X_i|_{\mathcal{H}^{\infty}} \in \operatorname{op}^{r_i}(\Theta)$ and $a_i(H_i, \omega)|_{\mathcal{H}^{\infty}(\Theta)} \in \operatorname{op}^{\beta_i h_i}(\Theta)$ satisfying the conditions of Theorem 3.2.5, then the proof of Theorem 3.2.5 shows that we can define $T_{\phi}^{H_0,\ldots,H_n}(X_1,\ldots,X_n): \mathcal{H}^{\sum_i r_i + \sum_i \beta_i h_i} \to \mathcal{H}$ by

$$T_{\phi}^{H_0,\dots,H_n}(X_1,\dots,X_n)\psi = \int_{\Omega} a_0(H_0,\omega)V_1a_1(H_1,\omega)\cdots V_na_n(H_n,\omega)\psi\,d\nu(\omega) \in \mathcal{H},$$
$$\psi \in \mathcal{H}^{\sum_i r_i + \sum_i \beta_i h_i},$$

which is a map that, apart from the definition of its domain, does not depend on Θ .

Furthermore, using the functional calculus for $op^0(\Theta)$ in Theorem 2.3.7, we can similarly define multiple operator integrals for $op^0(\Theta)$.

Theorem 3.2.8. Let $H_i, [\Theta, H_i] \in \operatorname{op}^0(\Theta)$, $i = 0, \ldots, n$, be such that each $\overline{H_i}^{0,0}$ is selfadjoint. For $\phi \in T^0(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i T^0(\mathbb{R})$ with corresponding representation

$$\phi(\lambda_0,\ldots,\lambda_n) = \int_{\Omega} a_0(\lambda_0,\omega)\cdots a_n(\lambda_n,\omega) \, d\nu(\omega),$$

the integral

$$T_{\phi}^{H_0,\dots,H_n}(X_1,\dots,X_n)\psi := \int_{\Omega} a_0(H_0,\omega)X_1a_1(H_1,\omega)\cdots X_na_n(H_n,\omega)\psi\,d\nu(\omega), \quad \psi \in \mathcal{H}^{\infty}$$

for $X_1, \ldots, X_n \in \operatorname{op}(\Theta)$, converges as a Bochner integral in \mathcal{H}^s for every $s \in \mathbb{R}$, and defines an n-multilinear map $T_{\phi}^{H_0,\ldots,H_n} : \operatorname{op}^{r_1}(\Theta) \times \cdots \times \operatorname{op}^{r_n}(\Theta) \to \operatorname{op}^{\sum_j r_j}(\Theta)$.

Proof. By definition of $T^0(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i T^0(\mathbb{R})$ we have for all $k_0, \ldots, k_n \in \mathbb{N}$,

$$\int_{\Omega} \|a_0(\cdot,\omega)\|_{T^0(\mathbb{R}),k_0} \cdots \|a_n(\cdot,\omega)\|_{T^0(\mathbb{R}),k_n} d|\nu|(\omega) < \infty.$$

The proof of the theorem is then identical to the proof of Theorem 3.2.5, using that

$$\|a_j(H_j,\omega)\|_{\mathcal{H}^s \to \mathcal{H}^s} \leq \sum_{k=0}^{\lceil 2^{|s|}\rceil+1} \|a_j(\cdot,\omega)\|_{T^0(\mathbb{R}),k}$$

see Theorem 2.3.7.

3.3 Divided differences

The conditions that appear in Theorem 3.2.5 and Theorem 3.2.8 on the symbol $\phi : \mathbb{R}^{n+1} \to \mathbb{C}$ need to be analysed more closely in order to prepare these multiple operator integral constructions for practical applications. The main result of this section is Lemma 3.3.1, which gives that for $f \in T^{\beta}(\mathbb{R}), \beta \in \mathbb{R}$, the divided difference $f^{[n]}$ has an integral representation satisfying the conditions of the MOI construction for Θ -elliptic operators in Theorem 3.2.5, and that for $f \in C^{\infty}(\mathbb{R})$ the divided difference $f^{[n]}$ can be used in the MOIs for zero-order operators in Theorem 3.2.8.

First of all, for functions $\phi : \mathbb{R}^{n+1} \to \mathbb{C}$ it is an equivalent condition to admit a representation

$$\phi(\lambda_0,\ldots,\lambda_n) = \int_{\Omega} a_0(\lambda_0,\omega)\cdots a_n(\lambda_n,\omega) d\nu(\omega)$$

for a finite complex measure space (technically: of finite variation) (Ω, ν) and measurable functions $a_j : \mathbb{R} \times \Omega \to \mathbb{C}$ such $(x, \omega) \mapsto a_j(x, \omega) \langle x \rangle^{-\beta_j}$ is $E_j \times \nu$ -a.e. bounded for $\beta_j \in \mathbb{R}$, or a representation

$$\phi(\lambda_0,\ldots,\lambda_n) = \int_{\Sigma} b_0(\lambda_0,\sigma)\cdots b_n(\lambda_n,\sigma) d\mu(\sigma),$$

where Σ is a σ -finite measure space, $b_j : \mathbb{R} \times \Omega \to \mathbb{C}$ measurable, and

$$\int_{\Sigma} \|b_0(\cdot,\sigma)\|_{L^{\beta_0}_{\infty}(E_0)} \cdots \|b_n(\cdot,\sigma)\|_{L^{\beta_n}_{\infty}(E_n)} d|\mu|(\sigma) < \infty.$$

Namely, given the second representation, a representation of the first type can be obtained by putting [ST19, p.48]

$$a_j(\lambda_j,\sigma) := \frac{b_j(\lambda_j,\sigma)}{\|b_j(\cdot,\sigma)\|_{L^{\beta_j}_{\infty}(E_j)}}, \quad \nu(\sigma) := \|b_0(\cdot,\sigma)\|_{L^{\beta_0}_{\infty}(E_0)} \cdots \|b_n(\cdot,\sigma)\|_{L^{\beta_n}_{\infty}(E_n)} \mu(\sigma).$$

We now again use the construction of an almost analytic extension (see Definition 2.3.1) to provide an explicit integral representation for $f \in T^{\beta}(\mathbb{R})$. Recall the definition of the product \boxtimes_i , Definition 3.2.1.

Lemma 3.3.1. 1. Take $n \in \mathbb{N}$, let α be some real number with $-1 \leq \alpha \leq n$, and consider any collection of real numbers $-1 \leq \beta_0, \ldots, \beta_n \leq 0$ such that $\sum \beta_j = \alpha - n$.

Then

$$f \in T^{\alpha}(\mathbb{R}) \Rightarrow f^{[n]} \in S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R}),$$

where for each $k_0, \ldots, k_n \in \mathbb{N}$ we have

$$\|f^{[n]}(\lambda_0,\ldots,\lambda_n)\|_{S^{\beta_0}(\mathbb{R})\boxtimes_i\cdots\boxtimes_i S^{\beta_n}(\mathbb{R}),k_0,\ldots,k_n} \lesssim \sum_{r=0}^{n+\sum_{j=0}^n k_j+2} \|f\|_{T^{\alpha}(\mathbb{R}),r}.$$

2. Let $\alpha \leq n$. Then

$$f \in T^{\alpha}(\mathbb{R}) \Rightarrow f^{[n]} \in \sum_{\substack{\beta_0, \dots, \beta_n \leq 0 \\ \sum \beta_j = \alpha - n}} S^{\beta_0}(\mathbb{R}) \boxtimes_i \dots \boxtimes_i S^{\beta_n}(\mathbb{R}).$$

For each component $\phi \in S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R})$ in the (finite) decomposition, we have

$$\|\phi\|_{S^{\beta_0}(\mathbb{R})\boxtimes_i\cdots\boxtimes_i S^{\beta_n}(\mathbb{R}),k_0,\ldots,k_n} \lesssim \sum_{r=0}^{n+\sum_{j=0}^n k_j+2} \|f\|_{T^{\alpha}(\mathbb{R}),r}.$$

3. Let $\alpha \geq n$. Then

$$f \in T^{\alpha}(\mathbb{R}) \Rightarrow f^{[n]} \in \sum_{\sum \beta_j = \alpha - n} S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R}).$$

For each component $\phi \in S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R})$ in the (finite) decomposition, we have

$$\|\phi\|_{S^{\beta_0}(\mathbb{R})\boxtimes_i\cdots\boxtimes_i S^{\beta_n}(\mathbb{R}),k_0,\ldots,k_n} \lesssim \sum_{r=0}^{n+\sum_{j=0}^n k_j+2} \|f\|_{T^{\alpha}(\mathbb{R}),r}.$$

Proof. 1. For $g \in C_c^{\infty}(\mathbb{R})$ with almost analytic extension \tilde{g} , we have

$$g(x) = -\frac{1}{\pi} \int_{\mathbb{C}} \frac{\partial \tilde{g}}{\partial \overline{z}} (z-x)^{-1} dz,$$

and hence

$$g^{[n]}(\lambda_0, \dots, \lambda_n) = \frac{(-1)^n}{\pi} \int_{\mathbb{C}} \frac{\partial \tilde{g}}{\partial \overline{z}} (z - \lambda_0)^{-1} \cdots (z - \lambda_n)^{-1} dz.$$
(3.4)

Now take $f \in T^{\alpha}(\mathbb{R})$ with $\alpha \leq n$ and with almost analytic extension \tilde{f} . Directly from Definition 2.3.1, writing $\sigma(z) := \tau(\frac{\Im(z)}{\langle \Re(z) \rangle})$, it follows that (cf. [Dav95a, Section 2.2])

$$\frac{\partial \tilde{f}}{\partial \overline{z}} = \frac{1}{2} \left(\sum_{r=0}^{N} f^{(r)}(\Re(z)) \frac{(i\Im(z))^r}{r!} \right) (\sigma_x(z) + i\sigma_y(z)) + \frac{1}{2} f^{(N+1)}(\Re(z)) \frac{(i\Im(z))^N}{n!} \sigma(z).$$

We define

$$U := \{ z \in \mathbb{C} : \langle \Re(z) \rangle < |\Im(z)| < 2 \langle \Re(z) \rangle \}, \qquad V := \{ z \in \mathbb{C} : 0 \le |\Im(z)| < 2 \langle \Re(z) \rangle \},$$

and note that the support of σ is contained in V, while the support of σ_x and σ_y are contained in U. More precisely,

$$|\sigma_x(z) + i\sigma_y(z)| \lesssim \frac{1}{\langle \Re(z) \rangle} \chi_U(z).$$

Therefore we have the estimate [Dav95b, Lemma 1]

$$\begin{split} \int_{\mathbb{C}} \left| \frac{\partial \tilde{f}}{\partial \bar{z}} \right| |z - \lambda_0|^{-1} \cdots |z - \lambda_n|^{-1} dz \\ &\lesssim \sum_{r=0}^N \int_U |f^{(r)}(\Re(z))| |\Im(z)|^{r-n-1} \langle \Re(z) \rangle^{-1} dz + \int_V |f^{(N+1)}(\Re(z))| |\Im(z)|^{N-n-1} dz \\ &\lesssim \sum_{r=0}^{N+1} \int_{\mathbb{R}} |f^{(r)}(x)| \langle x \rangle^{r-n-1} dx = \sum_{r=0}^{N+1} \|f\|_{T^n(\mathbb{R}), r}, \end{split}$$

where the last estimate (integration over the imaginary direction) is justified when $N \ge n+1$.

Hence the integral

$$\frac{(-1)^n}{\pi} \int_{\mathbb{C}} \frac{\partial \tilde{f}}{\partial \overline{z}} (z - \lambda_0)^{-1} \cdots (z - \lambda_n)^{-1} dz$$

converges.

Since [Dav95b, Lemma 6] gives that $C_c^{\infty}(\mathbb{R})$ is dense in $T^{\alpha}(\mathbb{R})$, the Lebesgue dominated convergence theorem then gives that the identity (3.4) extends to all $f \in T^{\alpha}(\mathbb{R}), \alpha \leq n$, i.e.

$$f^{[n]}(\lambda_0,\ldots,\lambda_n) = \frac{(-1)^n}{\pi} \int_{\mathbb{C}} \frac{\partial \tilde{f}}{\partial \overline{z}} (z-\lambda_0)^{-1} \cdots (z-\lambda_n)^{-1} dz.$$

In order to show that this is a decomposition as described in Definition 3.2.1, with $a_j(\lambda_j, z) = (z - \lambda_j)^{-1}$, we will now estimate the expressions

$$\|(z-\cdot)^{-1}\|_{S^{\beta}(\mathbb{R}),k} = \sup_{\lambda \in \mathbb{R}} \langle \lambda \rangle^{k-\beta} \left| \frac{\partial^{k}}{\partial \lambda^{k}} (z-\lambda)^{-1} \right|$$

$$\lesssim \left(\sup_{\lambda \in \mathbb{R}} \langle \lambda \rangle^{-\beta} |z-\lambda|^{-1} \right) \cdot \left(\sup_{\lambda \in \mathbb{R}} \langle \lambda \rangle^{k} |z-\lambda|^{-k} \right).$$

$$(3.5)$$

Note that, for $\lambda \in \mathbb{R}$ and $z \in \mathbb{C} \setminus \mathbb{R}$,

$$\frac{\langle \lambda \rangle}{|z-\lambda|} = \frac{|\lambda \pm i|}{|z-\lambda|} \le 1 + \frac{|z \pm i|}{|z-\lambda|} \le 1 + \frac{\langle z \rangle}{|\Im(z)|},$$
(3.6)

as $\min(|z+i|, |z-i|) \leq \langle z \rangle$. Next, we estimate

$$\sup_{\lambda \in \mathbb{R}} \langle \lambda \rangle^{-\beta} |z - \lambda|^{-1}$$

for $-1 \leq \beta \leq 0$. We estimate the supremum over $\lambda > 1$, $|\lambda| \leq 1$ and $\lambda < -1$ separately. First, for $|\lambda| \leq 1$ we have $1 \leq \langle \lambda \rangle \leq \sqrt{2}$, and so

$$\sup_{|\lambda| \le 1} \langle \lambda \rangle^{-\beta} |z - \lambda|^{-1} \lesssim \frac{1}{|\Im(z)|}.$$

For $\lambda > 1$, we have $\langle \lambda \rangle \leq \sqrt{2}\lambda$, so that

$$\sup_{\lambda>1} \langle \lambda \rangle^{-\beta} |z-\lambda|^{-1} \lesssim \sup_{\lambda>1-\Re(z)} \frac{(\lambda+\Re(z))^{-\beta}}{\left(\lambda^2+\Im(z)^2\right)^{\frac{1}{2}}}.$$

Writing $v = (\lambda, \Im(z)) \in \mathbb{R}^2$, then by using Cauchy–Schwarz for the inner product on \mathbb{R}^2 , we have

$$\sup_{\lambda>1-\Re(z)} \frac{(\lambda+\Re(z))^{-\beta}}{\left(\lambda^2+\Im(z)^2\right)^{\frac{1}{2}}} = \sup_{\lambda>1-\Re(x)} \frac{\left(v\cdot\left(1,\frac{\Re(z)}{\Im(z)}\right)\right)^{-\beta}}{\|v\|}$$
$$\leq \sup_{\lambda>1-\Re(z)} \frac{\|(1,\frac{\Re(z)}{\Im(z)})\|^{-\beta}}{\|v\|^{1+\beta}}$$
$$\leq \frac{|z|^{-\beta}}{|\Im(z)|}.$$

For $\lambda < -1$ we have a similar estimate, and hence combined we have

$$\sup_{\lambda \in \mathbb{R}} \langle \lambda \rangle^{-\beta} |z - \lambda|^{-1} \lesssim \frac{1}{|\Im(z)|} \max(1, |z|^{-\beta}) \le \frac{\langle z \rangle^{-\beta}}{|\Im(z)|}.$$
(3.7)

Combining (3.5), (3.6) and (3.7) we get an estimate

$$\sup_{\lambda \in \mathbb{R}} \langle \lambda \rangle^{k-\beta} \left| \frac{\partial^k}{\partial \lambda^k} (z-\lambda)^{-1} \right| \lesssim \frac{\langle z \rangle^{-\beta}}{|\Im(z)|} \left(1 + \frac{\langle z \rangle}{|\Im(z)|} \right)^k.$$

Let $-1 \leq \beta_0, \ldots, \beta_n \leq 0$. Taking the inequality above and proceeding as before with $N \geq n + 1 + \sum_{j=0}^n k_j$, we have

$$\begin{split} \int_{\mathbb{C}} \left| \frac{\partial \tilde{f}}{\partial \overline{z}} \right| \left(\sup_{\lambda_0 \in \mathbb{R}} \langle \lambda_0 \rangle^{k_0 - \beta_0} \left| \frac{\partial^{k_0}}{\partial \lambda_0^{k_0}} (z - \lambda_0)^{-1} \right| \right) \cdots \left(\sup_{\lambda_n \in \mathbb{R}} \langle \lambda_n \rangle^{k_n - \beta_n} \left| \frac{\partial^{k_n}}{\partial \lambda_n^{k_n}} (z - \lambda_0)^{-1} \right| \right) dz \\ \lesssim \sum_{r=0}^{N+1} \int_{\mathbb{R}} |f^{(r)}(x)| \langle x \rangle^{r-n-1-\sum_{j=0}^n \beta_j} dx < \infty. \end{split}$$

This converges in particular for $\sum_{j=0}^{n} \beta_j = \alpha - n$. Since $-n - 1 \leq \sum_{j=0}^{n} \beta_j \leq 0$, this choice is possible if $-1 \leq \alpha \leq n$. We have therefore proved for $-1 \leq \alpha \leq n$, and $-1 \leq \beta_0, \ldots, \beta_n \leq 0$ such that $\sum_{j=0}^{n} \beta_j = \alpha - n$, that

$$\begin{split} \|f^{[n]}(\lambda_0,\ldots,\lambda_n)\|_{S^{\beta_0}(\mathbb{R})\boxtimes_i\cdots\boxtimes_i S^{\beta_n}(\mathbb{R}),k_0,\ldots,k_n} \\ &= \int_{\mathbb{C}} \|(z-\cdot)^{-1}\|_{S^{\beta_0}(\mathbb{R}),k_0}\cdots\|(z-\cdot)^{-1}\|_{S^{\beta_n}(\mathbb{R}),k_n} \left|\frac{\partial \tilde{f}}{\partial \overline{z}}\right| dz \\ &\lesssim \sum_{r=0}^{n+\sum_{j=0}^n k_j+2} \|f\|_{T^n(\mathbb{R}),r} \le \sum_{r=0}^{n+\sum_{j=0}^n k_j+2} \|f\|_{T^\alpha(\mathbb{R}),r}. \end{split}$$

2. For $f \in T^{\alpha}(\mathbb{R})$, $-1 \leq \alpha \leq n$, we have by the first part of the lemma that for each $n \in \mathbb{N}$,

$$f^{[n]} \in S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R})$$

where each β_j can be chosen to lie in the interval [-1, 0], and $\sum \beta_j = \alpha - n$. For $f \in T^{\alpha}(\mathbb{R})$ with $\alpha \leq -1$, we can write $f = g \cdot (x+i)^{-k}$ where $g \in T^{\beta}(\mathbb{R})$, $-1 \leq \beta \leq 0$ and $k \in \mathbb{N}$. The Leibniz rule for divided differences dictates

$$f^{[n]}(\lambda_0,\ldots,\lambda_n) = \sum_{l=0}^n g^{[l]}(\lambda_0,\ldots,\lambda_l) \left((x+i)^{-k} \right)^{[n-l]}(\lambda_l,\ldots,\lambda_n).$$

From part 1 and the explicit form of the divided differences of $((x+i)^{-k})^{[n]}$ we therefore conclude that each term is an element of

$$\sum_{\substack{\beta_0,\ldots,\beta_n\leq 0\\\sum\beta_j=\alpha-k-n}} S^{\beta_0}(\mathbb{R})\boxtimes_i\cdots\boxtimes_i S^{\beta_n}(\mathbb{R}),$$

with the required estimate of norms.

3. This follows analogously to assertion 2, by analysing $(g(x+i)^k)^{[n]}$ for $g \in T^{\alpha}(\mathbb{R})$ with $-1 \leq \alpha \leq 0$.

Remark 3.3.2. The proof of Lemma 3.3.1 in fact shows that if $f \in C^{n+2}(\mathbb{R})$ such that $||f||_{T^{\beta}(\mathbb{R}),k} < \infty$ for $k = 0, \ldots, n+2$, then given any spectral measures E_0, \ldots, E_n , we have

$$\|f^{[n]}\|_{L^{\beta_0}_{\infty}(E_0)\hat{\otimes}_i\cdots\hat{\otimes}_i L^{\beta_n}_{\infty}(E_n)} \le \|f^{[n]}\|_{S^{\beta_0}(\mathbb{R})\boxtimes_i\cdots\boxtimes_i S^{\beta_n}(\mathbb{R}),0,\dots,0} \le \sum_{k=0}^{n+2} \|f\|_{T^{\beta}(\mathbb{R}),k} < \infty.$$

For n = 0, the space of functions that satisfy this condition closely resembles the space $\mathfrak{F}_m(\mathbb{R})$ used in [Car+16] in the context of double operator integrals.

Theorem 3.3.3. Let H_0, \ldots, H_n be such that each $H_i \in \operatorname{op}^h(\Theta), h > 0$, is Θ -elliptic and symmetric, and let $f \in C^{n+2}(\mathbb{R})$ such that $\|f\|_{T^\beta(\mathbb{R}),k} < \infty$, $k = 0, \ldots, n+2$ for some $\beta \in \mathbb{R}$. Then for any $X_i \in \operatorname{op}^{r_i}(\Theta), i = 1, \ldots, n$,

$$T_{f^{[n]}}^{H_0,\ldots,H_n}(X_1,\ldots,X_n) \in \mathrm{op}^{r+(\beta-n)h}(\Theta),$$

$$\|T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_n)\|_{\mathcal{H}^{s+r+(\beta-n)h}\to\mathcal{H}^s} \le C_{s,H_0,\dots,H_n}\left(\sum_{k=0}^{n+2} \|f\|_{T^{\beta}(\mathbb{R}),k}\right) \prod_{i=1}^n \|X_i\|_{\mathcal{H}^{s_i+r_i}\to\mathcal{H}^{s_i}}$$

for some $s_1,\dots,s_n \in \mathbb{R}$.

Proof. Lemma 3.3.1 and Remark 3.3.2, combined with Theorem 3.2.5. \Box

Theorem 3.3.4. Let $H_i, [\Theta, H_i] \in \operatorname{op}^0(\Theta), i = 0, \ldots, n$, be such that each $\overline{H_i}^{0,0} : \mathcal{H} \to \mathcal{H}$ is self-adjoint, and let $f \in C^{\infty}(\mathbb{R})$. Then

$$T_{f^{[n]}}^{H_0,\ldots,H_n}(X_1,\ldots,X_n) \in \operatorname{op}^{\sum_j r_j}(\Theta).$$

Proof. Lemma 3.3.1 and Remark 3.3.2, combined with Theorem 3.2.5.

Due to Lemma 3.3.1, the divided difference $f^{[n]}$ for $f \in C^{\infty}(\mathbb{R})$ is in particular a permitted symbol for Theorem 3.2.8.

3.4 MOI identities

The most important identities for our applications of our multiple operator integrals are the following. These are generalisations of the identities

$$[f(H), X] = T_{f^{[1]}}^{H,H}([H, X]);$$

$$f(H + X) - f(H) = T_{f^{[1]}}^{H + X,H}(X),$$

seeing that $f(H) = T_{f^{[0]}}^H()$.

Proposition 3.4.1. Let $a, X_1, \ldots, X_n \in \operatorname{op}(\Theta)$, let $H_i \in \operatorname{op}^{h_i}(\Theta)$, $h_i > 0$ be symmetric and Θ -elliptic and let $f \in T^{\beta}(\mathbb{R})$, $\beta \in \mathbb{R}$. Then

$$T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_j,aX_{j+1},\dots,X_n) - T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_ja,X_{j+1},\dots,X_n)$$
(3.8)
= $T_{f^{[n+1]}}^{H_0,\dots,H_j,H_j,\dots,H_n}(X_1,\dots,X_j,[H_j,a],X_{j+1},\dots,X_n);$

$$T_{f^{[n]}}^{H_0,\dots,H_n}(aX_1,\dots,X_n) - aT_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_n)$$

$$= T_{f^{[n+1]}}^{H_0,H_0,H_1,\dots,H_n}([H_0,a],X_1,\dots,X_n);$$

$$T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_n)a - T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_na)$$

$$= T_{f^{[n+1]}}^{H_0,\dots,H_n,H_n}(X_1,\dots,X_n,[H_n,a]).$$
(3.9)

Moreover, for $A \in op^{a}(\Theta), a > 0, B \in op^{b}(\Theta), b > 0$ symmetric and Θ -elliptic,

$$T_{f^{[n]}}^{H_0,\dots,H_{j-1},A,H_{j+1},\dots,H_n}(X_1,\dots,X_n) - T_{f^{[n]}}^{H_0,\dots,H_{j-1},B,H_{j+1},\dots,H_n}(X_1,\dots,X_n)$$

= $T_{f^{[n+1]}}^{H_0,\dots,H_{j-1},A,B,H_{j+1},\dots,H_n}(X_1,\dots,X_j,A-B,X_{j+1},\dots,X_n).$ (3.11)

The same assertions hold for self-adjoint $H_i, A, B \in \operatorname{op}^0(\Theta)$ such that $[\Theta, H_i], [\Theta, A], [\Theta, B] \in \operatorname{op}^0(\Theta)$ and with $f \in C^{\infty}(\mathbb{R})$.

Proof. We prove equation 3.8 in the Θ -elliptic case, the other identities and the zero-order case follow analogously. For $f \in T^{\beta}(\mathbb{R})$, the multiple operator integrals appearing in the equation are then well-defined, see Theorem 3.3.3.

Write

$$F_j(\lambda_0,\ldots,\lambda_{n+1}):=f^{[n]}(\lambda_0,\ldots,\lambda_{j-1},\lambda_{j+1},\ldots,\lambda_{n+1}),$$

and observe that

$$F_{j+1}(\lambda_0,\ldots,\lambda_{n+1})-F_j(\lambda_0,\ldots,\lambda_{n+1})=(\lambda_j-\lambda_{j+1})f^{[n+1]}(\lambda_0,\ldots,\lambda_{n+1}).$$

Hence,

$$\begin{split} T_{f^{[n+1]}}^{H_0,\dots,H_j,H_j,\dots,H_n}(X_1,\dots,X_j,[H_j,a],X_{j+1},\dots,X_n) \\ &= T_{(\lambda_j-\lambda_{j+1})f^{[n+1]}}^{H_0,\dots,H_j,H_j,\dots,H_n}(X_1,\dots,X_j,a,X_{j+1},\dots,X_n) \\ &= T_{F_{j+1}}^{H_0,\dots,H_j,H_j,\dots,H_n}(X_1,\dots,X_j,a,X_{j+1},\dots,X_n) \\ &- T_{F_j}^{H_0,\dots,H_j,H_j,\dots,H_n}(X_1,\dots,X_j,a,X_{j+1},\dots,X_n) \\ &= T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_j,aX_{j+1},\dots,X_n) \\ &- T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_ja,X_{j+1},\dots,X_n). \end{split}$$

With these identities in hand, we can show that the MOI constructed in the previous section is an element of $OP(\Theta)$ if all its components are and the symbol is a divided difference.

Theorem 3.4.2. Let H_0, \ldots, H_n be such that each $H_i \in OP^h(\Theta)$, h > 0, is symmetric and Θ -elliptic, and let $f \in T^{\beta}(\mathbb{R})$ for some $\beta \in \mathbb{R}$. For operators $X_i \in OP^{r_i}(\Theta)$, $r := \sum_{i=1}^n r_i$, we have that

$$T_{f^{[n]}}^{H_0,\ldots,H_n}(X_1,\ldots,X_n) \in \mathrm{OP}^{h(\beta-n)+r}(\Theta).$$

Similarly, if each H_i instead is such that $H_i, [\Theta, H_i] \in OP^0(\Theta)$ and H_i is self-adjoint, then for any $f \in C^{\infty}(\mathbb{R})$ we have that

$$T_{f^{[n]}}^{H_0,\ldots,H_n}(X_1,\ldots,X_n) \in \mathrm{OP}^r(\Theta).$$

Proof. We focus on the Θ -elliptic case, the zero-order case follows similarly. Taking n = 1 to ease notation, using Proposition 3.4.1 gives that

$$[\Theta, T_{f^{[1]}}^{H_0, H_1}(X_1)] = T_{f^{[2]}}^{H_0, H_0, H_1}([\Theta, H_0], X_1) + T_{f^{[1]}}^{H_0, H_1}([\Theta, X_1]) + T_{f^{[2]}}^{H_0, H_1, H_1}(X_1, [\Theta, H_1]).$$

As $[\Theta, H_i] \in \operatorname{op}^h(\Theta)$ and $[\Theta, X_1] \in \operatorname{op}^{r_1}(\Theta)$, Lemma 3.3.1 combined with Theorem 3.2.5 gives that

$$T_{f^{[2]}}^{H_0,H_0,H_1}([\Theta,H_0],X_1),T_{f^{[1]}}^{H_0,H_1}([\Theta,X_1]),T_{f^{[2]}}^{H_0,H_1,H_1}(X_1,[\Theta,H_1]) \in \operatorname{op}^{h(\beta-n)+r}(\Theta).$$

Higher commutators and n > 1 follow analogously.

We have now proven Theorem 2.2.7 and Theorem 2.3.8 which were claimed in the previous chapter: they are special cases of Theorem 3.4.2.

In the setting that $\Theta^{-1} \in \mathcal{L}_s$, s > 0, it is immediate from Theorem 3.2.5 and Theorem 2.2.3 that for $H_0, \ldots, H_n \in \operatorname{op}^h(\Theta)$, h > 0 symmetric and Θ -elliptic, and $f \in T^{\beta}(\mathbb{R})$, the multiple operator integral

$$T_{f^{[n]}}^{H_0,\dots,H_n}(X_1,\dots,X_n) \in \mathrm{op}^{(\beta-n)h+r}(\Theta)$$
 (3.12)

can be considered to be a trace-class operator on \mathcal{H} if β is small enough. Namely, we have

$$\|A\|_1 \le \|\Theta^{-s}\|_1 \|A\|_{\mathcal{H}^0 \to \mathcal{H}^s}$$

3.5 Asymptotic expansions

Through the identities proved in the previous section, the theory of multiple operator integrals lends itself well for establishing asymptotic expansions of operators. As an immediate example, we prove a noncommutative Taylor expansion for pseudodifferential operators.

Theorem 3.5.1. Let $f \in T^{\beta}(\mathbb{R})$, $H \in \operatorname{op}^{h}(\Theta)$, h > 0, Θ -elliptic and symmetric, and let $V \in \operatorname{op}^{r}(\Theta)$ be symmetric. If the order of the perturbation V is strictly smaller than that of H, i.e. r < h, we have

$$f(H+V) \sim \sum_{n=0}^{\infty} T_{f^{[n]}}^{H}(V,\dots,V),$$
 (3.13)

in the sense that

$$f(H+V) - \sum_{n=0}^{N} T_{f^{[n]}}^{H}(V, \dots, V) \in \operatorname{op}^{m_{N}}(\Theta)$$

with $m_N \downarrow -\infty$.

Proof. Using the last part of Proposition 3.4.1 with A = H + V, B = H, we have

$$f(H+V) - f(H) = T_{f^{[0]}}^{H+V}() - T_{f^{[0]}}^{H}()$$
$$= T_{f^{[1]}}^{H+V,H}(V).$$

Repeating the argument, we get

$$f(H+V) - \sum_{n=0}^{N} T_{f^{[n]}}^{H}(V, \dots, V) = T_{f^{[N+1]}}^{H+V, H, \dots, H}(V, \dots, V).$$

Now, if r < h, Theorem 3.3.3 gives that

$$T_{f^{[N+1]}}^{H+V,H,\ldots,H}(V,\ldots,V) \in \mathrm{op}^{(\beta-N-1)h+Nr}(\Theta),$$

with

$$(\beta - N - 1)h + Nr = N(r - h) + (\beta - 1)h \downarrow -\infty.$$

Note that, if H and V are commuting operators, (3.13) recovers the classic Taylor expansion formula $f(H+V) \sim \sum_{n=0}^{\infty} \frac{f^{(n)}(H)}{n!} V^n$. Hence, the expansion (3.13) can be interpreted as a type of noncommutative Taylor expansion. Each term in the expansion can itself be expanded as follows. Recall that we write $\delta_H(X) := [H, X], \ \delta_H^n(X) := \delta_H(\cdots \delta_H(\delta_H(X)) \cdots).$

Proposition 3.5.2. Let $X_i \in op^{r_i}(\Theta)$, $H \in op^h(\Theta)$, h > 0 symmetric and Θ -elliptic, and $f \in T^{\beta}(\mathbb{R})$. Then

$$T_{f^{[n]}}^{H}(X_{1},...,X_{n}) = \sum_{m=0}^{N} \sum_{m_{1}+\dots+m_{n}=m} \frac{C_{m_{1},\dots,m_{n}}}{(n+m)!} \delta_{H}^{m_{1}}(X_{1}) \cdots \delta_{H}^{m_{n}}(X_{n}) f^{(n+m)}(H) + S_{N}^{n}(X_{1},\dots,X_{n}),$$

where

$$C_{m_1,\dots,m_n} := \prod_{j=1}^n \binom{j+m_1+\dots+m_j-1}{m_j}$$

and the remainder $S_N^n(X_1, \ldots, X_n)$ is a sum of terms of the form

$$\delta_{H}^{m_{1}}(X_{1})\cdots\delta_{H}^{m_{k}}(X_{k})T_{f^{[n+N+1]}}^{H}(1,\ldots,1,\delta_{H}^{N+1-m_{1}-\cdots-m_{k}}(X_{k+1}),1,\ldots,1,X_{k+2},\ldots,X_{n}).$$

If the commutators $\delta_H^k(X_j)$ have a lower order than the expected $r_j + kh$, explicitly if

$$\delta_H^k(X_j) \in \mathrm{op}^{r_j + k(h-\varepsilon)}(\Theta)$$

for some $\varepsilon > 0$, then the above gives an asymptotic expansion

$$T_{f^{[n]}}^{H}(X_{1},\ldots,X_{n}) \sim \sum_{m=0}^{\infty} \sum_{m_{1}+\cdots+m_{n}=m} \frac{C_{m_{1},\ldots,m_{n}}}{(n+m)!} \delta_{H}^{m_{1}}(X_{1}) \cdots \delta_{H}^{m_{n}}(X_{n}) f^{(n+m)}(H),$$

in the sense that the remainder term

$$S_N^n(X_1,\ldots,X_n) \in \mathrm{op}^{m_N}(\Theta)$$

with $m_N = \sum_j r_j + (\beta - n)h - \varepsilon(N+1) \downarrow -\infty.$

The proof of this proposition is a lengthy combinatorial exercise, the computations of which are standard and appear in many proofs of the local index formula [CPRS06a, Lemma 6.11][CM95, Equation (71)][Hig04, Lemma 2.12], see also [Dal98; Pay11]. The novelty is that they can be performed in the very general context of (unbounded) MOIs. In effect, Proposition 3.5.2 is a generalisation of the cited results.

Definition 3.5.3. The multiset coefficient $\binom{n}{k}$ for $n, k \in \mathbb{N}$ is defined as

$$\binom{n}{k} := \binom{n+k-1}{k}.$$

Lemma 3.5.4. For $f \in T^{\beta}(\mathbb{R})$, $H \in \operatorname{op}^{h}(\Theta)$, h > 0 symmetric and Θ -elliptic, $X_{i} \in \operatorname{op}^{r_{i}}(\Theta)$, we have

$$T_{f^{[n+j]}}^{H}(\underbrace{1,\ldots,1}_{j},X_{1},\ldots,X_{n})$$

$$=\sum_{m=0}^{N}\left(\binom{m+1}{j}\right)\delta_{H}^{m}(X_{1})T_{f^{[n+j+m]}}^{H}(\underbrace{1,\ldots,1}_{j+1+m},X_{2},\ldots,X_{n})+R_{j,N}^{n}(X_{1},\ldots,X_{n}),$$

where

$$R_{j,N}^{n}(X_{1},\ldots,X_{n}) := \sum_{l=0}^{j} \left(\binom{N+1}{l} \right) T_{f^{[n+j+N+1]}}^{H}(\underbrace{1,\ldots,1}_{j-l},\delta_{H}^{N+1}(X_{1}),\underbrace{1,\ldots,1}_{N+1+l},X_{2},\ldots,X_{n}).$$

Proof. Multiset coefficients have the property that

$$\sum_{l=0}^{j} \left(\binom{m}{l} \right) = \left(\binom{m+1}{j} \right).$$

The assertion of the lemma follows by induction on N. For N = 0,

$$T_{f^{[n+j]}}^{H}(\underbrace{1,\ldots,1}_{j},X_{1},\ldots,X_{n})$$

$$= X_{1}T_{f^{[n+j]}}^{H}(\underbrace{1,\ldots,1}_{j+1},X_{2},\ldots,X_{n}) + \sum_{l=0}^{j}T_{f^{[n+j+1]}}^{H}(\underbrace{1,\ldots,1}_{j-l},\delta_{H}(X_{1}),\underbrace{1,\ldots,1}_{1+l},X_{2},\ldots,X_{n})$$

by applying Proposition 3.4.1 j + 1 times on X_1 .

Suppose that the assertion holds for N-1. Then

$$\begin{split} T_{f^{[n+j]}}^{H}(\underbrace{1,\ldots,1}_{j},X_{1},\ldots,X_{n}) \\ &= \sum_{m=0}^{N-1} \left(\binom{m+1}{j} \right) \delta_{H}^{m}(X_{1}) T_{f^{[n+j+m]}}^{H}(\underbrace{1,\ldots,1}_{j+1+m},X_{2},\ldots,X_{n}) \\ &+ \sum_{l=0}^{j} \left(\binom{N}{l} \right) T_{f^{[n+j+N]}}^{H}(\underbrace{1,\ldots,1}_{j-l},\delta_{H}^{N}(X_{1}),\underbrace{1,\ldots,1}_{N+l},X_{2},\ldots,X_{n}) \\ &=^{*} \sum_{m=0}^{N-1} \left(\binom{m+1}{j} \right) \delta_{H}^{m}(X_{1}) T_{f^{[n+j+m]}}^{H}(\underbrace{1,\ldots,1}_{j+1+m},X_{2},\ldots,X_{n}) \\ &+ \sum_{l=0}^{j} \left(\binom{N}{l} \right) \delta_{H}^{N}(X_{1}) T_{f^{[n+j+N]}}^{H}(\underbrace{1,\ldots,1}_{N+1+j},X_{2},\ldots,X_{n}) \\ &+ \sum_{l=0}^{j} \left(\binom{N}{l} \right) \sum_{k=0}^{j-l} T_{f^{[n+j+N+1]}}^{H}(\underbrace{1,\ldots,1}_{j-l-k},\delta_{H}^{N+1}(X_{1}),\underbrace{1,\ldots,1}_{N+1+l+k},X_{2},\ldots,X_{n}), \end{split}$$

where in the step marked with * we applied Proposition 3.4.1 j - l times on $\delta_H^N(X_1)$.

Continuing on,

$$T_{f^{[n+j]}}^{H}(\underbrace{1,\ldots,1}_{j},X_{1},\ldots,X_{n})$$

$$=\sum_{m=0}^{N-1}\left(\binom{m+1}{j}\right)\delta_{H}^{m}(X_{1})T_{f^{[n+j+m]}}^{H}(\underbrace{1,\ldots,1}_{j+1+m},X_{2},\ldots,X_{n})$$

$$+\left(\binom{N+1}{j}\right)\delta_{H}^{N}(X_{1})T_{f^{[n+j+N]}}^{H}(\underbrace{1,\ldots,1}_{N+1+j},X_{2},\ldots,X_{n})$$

$$+\sum_{l=0}^{j}\sum_{k=0}^{j-l}\left(\binom{N}{l}\right)T_{f^{[n+j+N+1]}}^{H}(\underbrace{1,\ldots,1}_{j-l-k},\delta_{H}^{N+1}(X_{1}),\underbrace{1,\ldots,1}_{N+1+l+k},X_{2},\ldots,X_{n}).$$

In the last sum, relabel r := k + l, so that

$$T_{f^{[n+j]}}^{H}(\underbrace{1,\ldots,1}_{j},X_{1},\ldots,X_{n})$$

$$=\sum_{m=0}^{N}\left(\binom{m+1}{j}\right)\delta_{H}^{m}(X_{1})T_{f^{[n+j+m]}}^{H}(\underbrace{1,\ldots,1}_{j+1+m},X_{2},\ldots,X_{n})$$

$$+\sum_{r=0}^{j}\sum_{l=0}^{r}\left(\binom{N}{l}\right)T_{f^{[n+j+N+1]}}^{H}(\underbrace{1,\ldots,1}_{j-r},\delta_{H}^{N+1}(X_{1}),\underbrace{1,\ldots,1}_{N+1+r},X_{2},\ldots,X_{n})$$

$$=\sum_{m=0}^{N}\left(\binom{m+1}{j}\right)\delta_{H}^{m}(X_{1})T_{f^{[n+j+m]}}^{H}(\underbrace{1,\ldots,1}_{j+1+m},X_{2},\ldots,X_{n})$$

$$+\sum_{r=0}^{j}\left(\binom{N+1}{r}\right)T_{f^{[n+j+N+1]}}^{H}(\underbrace{1,\ldots,1}_{j-r},\delta_{H}^{N+1}(X_{1}),\underbrace{1,\ldots,1}_{N+1+r},X_{2},\ldots,X_{n}).$$

This concludes the induction step.

Proof of Proposition 3.5.2. Apply Lemma 3.5.4 first to the first entry of $T_{f^{[n]}}^H(X_1,\ldots,X_n)$,

$$T_{f^{[n]}}^{H}(X_{1},\ldots,X_{n}) = \sum_{m_{1}=0}^{N} \left(\binom{m_{1}+1}{0} \right) \delta_{H}^{m_{1}}(X_{1}) T_{f^{[n+m_{1}]}}^{H}(\underbrace{1,\ldots,1}_{m_{1}+1},X_{2},\ldots,X_{n}) + R_{0,N}^{n}(X_{1},\ldots,X_{n}).$$

Apply Lemma 3.5.4 once more, expanding up to order $N - m_1$ instead of N.

$$T_{f^{[n]}}^{H}(X_{1},...,X_{n}) = \sum_{m_{1}=0}^{N} \sum_{m_{2}=0}^{N-m_{1}} \left(\binom{m_{1}+1}{0} \right) \left(\binom{m_{2}+1}{m_{1}+1} \right) \delta_{H}^{m_{1}}(X_{1}) \delta_{H}^{m_{2}}(X_{2}) T_{f^{[n+m_{1}+m_{2}]}}^{H}(\underbrace{1,...,1}_{2+m_{1}+m_{2}},X_{3},...,X_{n}) + \sum_{m_{1}=0}^{N} \left(\binom{m_{1}+1}{0} \right) \delta_{H}^{m_{1}}(X_{1}) R_{1+m_{1},N-m_{1}}^{n-1}(X_{2},...,X_{n}) + R_{0,N}^{n}(X_{1},...,X_{n}).$$

Repeating gives the formula

$$T_{f^{[n]}}^{H}(X_{1},...,X_{n})$$

$$=\sum_{m=0}^{N}\sum_{m_{1}+\cdots+m_{n}=m}\prod_{j=1}^{n}\left(\binom{m_{j}+1}{j-1+m_{1}+\cdots+m_{j-1}}\right)\delta_{H}^{m_{1}}(X_{1})\cdots\delta_{H}^{m_{n}}(X_{n})T_{f^{[n+m]}}^{H}(\underbrace{1,...,1}_{n+m})$$

$$+S_{N}^{n}(X_{1},...,X_{n}),$$

where

$$S_N^n(X_1, \dots, X_n) := \sum_{k=0}^{n-1} \sum_{m_1 + \dots + m_k \le N} \prod_{j=1}^k \left(\binom{m_j + 1}{j - 1 + m_2 + \dots + m_{j-1}} \right) \\ \times \delta_H^{m_1}(X_1) \cdots \delta_H^{m_k}(X_k) R_{k+m_1 + \dots + m_k, N-m_1 - \dots - m_k}^{n-k}(X_{k+1}, \dots, X_n).$$

The observation that

$$\left(\binom{n}{k}\right) = \left(\binom{k+1}{n-1}\right),\,$$

and the definition

finishes the proof of the proposition.

Putting these results together, we now have a noncommutative Taylor expansion.

Theorem 3.5.5. Let $f \in T^{\beta}(\mathbb{R})$, $H \in \operatorname{op}^{h}(\Theta)$, h > 0, Θ -elliptic and symmetric, and let $V \in \operatorname{op}^{r}(\Theta)$, r < h, be symmetric. Assume that

$$\delta_H^k(V) \in \mathrm{op}^{r+k(h-\varepsilon)}(\Theta)$$

for some $\varepsilon > 0$. Then we have the noncommutative Taylor expansion

$$f(H+V) \sim \sum_{n,m=0}^{\infty} \sum_{m_1+\dots+m_n=m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \delta_H^{m_1}(V) \cdots \delta_H^{m_n}(V) f^{(n+m)}(H).$$

Proof. For N, M > 0, Theorem 3.5.1 and Proposition 3.5.2 give

$$f(H+V) = \sum_{n=0}^{N} \sum_{m=0}^{M} \sum_{m_1+\dots+m_n=m}^{M} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \delta_H^{m_1}(V) \cdots \delta_H^{m_n}(V) f^{(n+m)}(H) + \operatorname{op}^{N(r-h)+(\beta-1)h}(\Theta) + \sum_{n=0}^{N} \operatorname{op}^{nr+(\beta-n)h-\varepsilon(M+1)}(\Theta).$$

As $(N, M) \to (\infty, \infty)$, we see that the order of the remainder decreases to $-\infty$.

The noncommutative Taylor expansion (3.13) features prominently in [Pay07] for classical pseudodifferential operators on manifolds, and in [Dal98; Pay11] in a more abstract sense. For bounded operators on Banach spaces, it has been studied in [HL24]. For related expansions for bounded operators V, see also [Han06; Sui11; Skr18; NS22].

3.6 Trace expansions and NCG

The observation in the previous section that the local index formula is closely connected to the noncommutative Taylor expansion in Theorem 3.5.5 leads us naturally to the topic of asymptotic expansions of trace formulas. In various contexts of noncommutative geometry and beyond, expansions are studied of the kind

$$\operatorname{Tr}(f(tH+tV)) \underset{t\downarrow 0}{\sim} \sum_{k=0}^{\infty} c_k t^{r_k}, \qquad (3.14)$$

for an increasing sequence $r_k \uparrow \infty$ and constants $c_k \in \mathbb{C}$, which means that as $t \downarrow 0$

$$Tr(f(tH + tV)) = \sum_{k=0}^{N} c_k t^{r_k} + O(t^{r_{N+1}})$$

for every $N \in \mathbb{R}$. Or, more generally (c.f. [EI18]),

$$\operatorname{Tr}(f(tH+tV)) \underset{t\downarrow 0}{\sim} \sum_{k=0}^{\infty} \rho_k(t), \qquad (3.15)$$

where $\rho_k(t) = O(t^{r_k})$ and

$$Tr(f(tH + tV)) = \sum_{k=0}^{N} \rho_k(t) + O(t^{r_{N+1}}).$$

To study the asymptotic expansion of expressions like

$$\mathrm{Tr}(ae^{-t(D+V)^2}),$$

we will use a modified version of Theorem 3.5.1 and Proposition 3.5.2. For this purpose, we need to make a more detailed analysis of norm bounds of MOIs and of the remainder in the Taylor expansion in Theorem 3.5.1.

Proposition 3.6.1. Let $H_i \in \operatorname{op}^h(\Theta)$ for a fixed h > 0 be symmetric and Θ -elliptic operators. If $X_i \in \operatorname{op}^{r_i}(\Theta)$, $r := \sum_{i=1}^n r_i$, $f \in T^{\alpha}(\mathbb{R})$ with $\alpha \leq n, t \leq 1$, then

$$\|T_{f^{[n]}}^{tH_0,\dots,tH_n}(X_1,\dots,X_n)\|_{\mathcal{H}^{q+r+(\alpha-n)h}\to\mathcal{H}^q} \lesssim t^{\alpha-n}, \quad q \in \mathbb{R}.$$

Rephrased, if $f \in T^{\frac{u-r}{h}+n}$, $u \leq r$, then

$$\|T_{f^{[n]}}^{tH_0,\dots,tH_n}(X_1,\dots,X_n)\|_{\mathcal{H}^{q+u}\to\mathcal{H}^q} \lesssim t^{\frac{u-r}{h}}, \quad q \in \mathbb{R}$$

Proof. Lemma 3.3.1 gives that

$$f^{[n]} \in \sum_{\beta_0 + \dots + \beta_n = \alpha - n} S^{\beta_0}(\mathbb{R}) \boxtimes_i \dots \boxtimes_i S^{\beta_n}(\mathbb{R}),$$

where each $\beta_i \leq 0$ since $\alpha \leq n$ (recall Definition 3.2.1 for this notation). Consider one of the summands, $\phi \in S^{\beta_0}(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^{\beta_n}(\mathbb{R})$. Then

$$\phi(\lambda_0,\ldots,\lambda_n) = (\lambda_0 + i)^{\beta_0} \cdots (\lambda_n + i)^{\beta_n} \cdot \psi(\lambda_0,\ldots,\lambda_n),$$

for a function $\psi \in S^0(\mathbb{R}) \boxtimes_i \cdots \boxtimes_i S^0(\mathbb{R})$, and thus, by Remark 3.2.3 and Theorem 3.2.5,

$$T_{\phi}^{tH_0,\dots,tH_n}(X_1,\dots,X_n)$$

= $T_{\psi}^{tH_0,\dots,tH_n}((tH_0+i)^{\beta_0}X_1(tH_1+i)^{\beta_1},X_2(tH_2+i)^{\beta_2},\dots,X_n(tH_n+i)^{\beta_n}).$

Corollary 2.2.4 and Theorem 3.2.5 give that

$$\|T_{\psi}^{tH_{0},\dots,tH_{n}}((tH_{0}+i)^{\beta_{0}}X_{1}(tH_{1}+i)^{\beta_{1}},X_{2}(tH_{2}+i)^{\beta_{2}},\dots,X_{n}(tH_{n}+i)^{\beta_{n}})\|_{\mathcal{H}^{q+r+(\alpha-n)h}\to\mathcal{H}^{q}}$$

$$\lesssim \|(tH_{0}+i)^{\beta_{0}}\|_{\mathcal{H}^{q_{0}+\beta_{0}h}\to\mathcal{H}^{q_{0}}}\cdots\|(tH_{n}+i)^{\beta_{n}}\|_{\mathcal{H}^{q_{n}+\beta_{n}h}\to\mathcal{H}^{q_{n}}},$$

for q_i some real numbers. Theorem 2.2.3 gives that

$$\|(tH_j+i)^{\beta_i j}\|_{\mathcal{H}^{q_j+\beta_j h}\to \mathcal{H}^{q_j}} \lesssim \sup_{x\in\mathbb{R}} |(tx+i)^{\beta_j}|\langle x\rangle^{-\beta_j} \lesssim t^{\beta_j}.$$

Therefore,

$$\|T_{\phi}^{tH_0,\dots,tH_n}(X_1,X_2,\dots,X_n)\|_{\mathcal{H}^{q+r+(\alpha-n)h}\to\mathcal{H}^q} \lesssim t^{\beta_0+\dots+\beta_k} = t^{\alpha-n}.$$

Proposition 3.6.2. Let $\Theta^{-1} \in \mathcal{L}_s$, s > 0, $f \in T^{\beta}(\mathbb{R})$, $H \in \operatorname{op}^h(\Theta)$, h > 0 Θ -elliptic and symmetric and $V \in \operatorname{op}^r(\Theta)$ symmetric. Let $h > r \ge 0$ and $\beta < -\frac{s}{h}$. For every $N \in \mathbb{N}$, we have as $t \downarrow 0$,

$$\operatorname{Tr}(f(tH+tV)) = \sum_{n=0}^{N} t^n \operatorname{Tr}(T_{f^{[n]}}^{tH}(V,\ldots,V)) + O(t^{(N+1)(1-\frac{r}{h})-\frac{s}{h}}).$$

Proof. The proof of Theorem 3.5.1 gives that

$$f(tH+tV) = \sum_{n=0}^{N} T_{f^{[n]}}^{tH}(tV,\dots,tV) + T_{f^{[N+1]}}^{tH+tV,tH,\dots,tH}(tV,\dots,tV).$$

The condition $\beta < -\frac{s}{h}$ assures that all terms on the left and right-hand side are trace-class (cf. (3.12)). Furthermore, we have $f \in T^{\beta}(\mathbb{R}) \subseteq T^{(N+1)(1-\frac{r}{h})-\frac{s}{h}}(\mathbb{R})$ so that Proposition 3.6.1 provides that

$$\|T_{f^{[N+1]}}^{tH+tV,tH,...,tH}(tV,...,tV)\|_{1} \leq t^{N+1} \|T_{f^{[N+1]}}^{tH+tV,tH,...,tH}(V,...,V)\|_{\mathcal{H}^{-s}\to\mathcal{H}^{0}} \leq t^{(N+1)(1-\frac{r}{h})-\frac{s}{h}}.$$

This proposition makes it clear that to determine the coefficients of asymptotic expansions of the type (3.14) or (3.15), it suffices to study the asymptotic expansions of the multiple operator integral

$$\mathrm{Tr}(T_{f^{[n]}}^{tH}(V,\ldots,V)),$$

which we do with Proposition 3.5.2.

Theorem 3.6.3. Let $\Theta^{-1} \in \mathcal{L}_s$, s > 0, $f \in T^{\beta}(\mathbb{R})$, $H \in \operatorname{op}^h(\Theta)$, h > 0 symmetric and Θ -elliptic, $V \in \operatorname{op}^r(\Theta)$ symmetric. If $h > r \ge 0$, $\beta \le -\frac{s}{h}$, and $\delta^n_H(V) \in \operatorname{op}^{r+n(h-\varepsilon)}(\Theta)$, then $\operatorname{Tr}(f(tH+tV))$ admits an asymptotic expansion as $t \downarrow 0$ of type (3.15) given by

$$\operatorname{Tr}(f(tH+tV)) = \sum_{n=0}^{N} \sum_{m=0}^{N} \sum_{m=0}^{N} \sum_{m_1+\dots+m_n=m} t^{n+m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}\left(\delta_H^{m_1}(V) \cdots \delta_H^{m_n}(V) f^{(n+m)}(tH)\right) + O(t^{m_N}),$$

where $m_N := (N+1) \min\left(\frac{\varepsilon}{h}, (1-\frac{r}{h})\right) - \frac{s}{h}$, so that $m_N \uparrow \infty$ as $N \to \infty$.

Proof. Combining Propositions 3.6.2 and 3.5.2, we have that

$$\operatorname{Tr}(f(tH+tV)) = \sum_{n=0}^{N} \sum_{m=0}^{N} t^{n+m} \sum_{m_1+\dots+m_n=m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}\left(\delta_H^{m_1}(V) \cdots \delta_H^{m_n}(V) f^{(n+m)}(tH)\right) + \sum_{n=0}^{N} t^n \operatorname{Tr}(S_{N,t}^n(V,\dots,V)) + O(t^{(N+1)(1-\frac{r}{h})-\frac{s}{h}}),$$

where $S_{N,t}^n(V,\ldots,V)$ is a sum of terms of the form

$$t^{N+1}\delta_{H}^{m_{1}}(V)\cdots\delta_{H}^{m_{k}}(V)T_{f^{[n+N+1]}}^{tH}(1,\ldots,1,\delta_{H}^{N+1-m_{1}-\dots-m_{k}}(V),1,\ldots,1,V,\ldots,V)$$

We then estimate

$$\begin{split} \left\| t^{N+1} \delta_{H}^{m_{1}}(V) \cdots \delta_{H}^{m_{k}}(V) T_{f^{[n+N+1]}}^{tH}(1, \dots, 1, \delta_{H}^{N+1-m_{1}-\dots-m_{k}}(V), 1, \dots, 1, V, \dots, V) \right\|_{1} \\ \lesssim t^{N+1} \left\| \delta_{H}^{m_{1}}(V) \cdots \delta_{H}^{m_{k}}(V) T_{f^{[n+N+1]}}^{tH}(1, \dots, 1, \delta_{H}^{N+1-m_{1}-\dots-m_{k}}(V), 1, \dots, 1, V, \dots, V) \right\|_{\mathcal{H}^{-s} \to \mathcal{H}^{0}} \\ \leq t^{N+1} \left\| \delta_{H}^{m_{1}}(V) \cdots \delta_{H}^{m_{k}}(V) \right\|_{\mathcal{H}^{kr+(m_{1}+\dots+m_{k})(h-\varepsilon)} \to \mathcal{H}^{0}} \\ \times \left\| T_{f^{[n+N+1]}}^{tH}(1, \dots, 1, \delta_{H}^{N+1-m_{1}-\dots-m_{k}}(V), 1, \dots, 1, V, \dots, V) \right\|_{\mathcal{H}^{-s} \to \mathcal{H}^{kr+(m_{1}+\dots+m_{k})(h-\varepsilon)}}. \end{split}$$

Applying Proposition 3.6.1 then provides that

$$\|S_{N,t}^{n}(V,\dots,V)\|_{1} \lesssim t^{N+1-\frac{s+kr+(m_{1}+\dots+m_{k})(h-\varepsilon)}{h}-\frac{(n-k)r+(N+1-m_{1}-\dots-m_{k})(h-\varepsilon)}{h}}{=t^{-\frac{s+nr}{h}+(N+1)\frac{\varepsilon}{h}},$$

and hence

$$\sum_{n=0}^{N} t^n \operatorname{Tr}(S_{N,t}^n(V,\ldots,V)) \lesssim \sum_{n=0}^{N} t^{n(1-\frac{r}{h})+(N+1)\frac{\varepsilon}{h}-\frac{s}{h}}.$$

Defining

$$m_N := (N+1)\min\left(\frac{\varepsilon}{h}, \left(1-\frac{r}{h}\right)\right) - \frac{s}{h}$$

concludes the proof.

This expansion in fact partially answers a specific open problem posed by Eckstein and Iochum in [EI18]. Given a spectral triple $(\mathcal{A}, \mathcal{H}, D)$ it is a common assumption to require the existence of an asymptotic expansion as $t \downarrow 0$ of

$$\operatorname{Tr}(ae^{-tD^2})$$

where $a \in \mathcal{A}$. Their question is whether the existence of asymptotic expansions of

$$\operatorname{Tr}(ae^{-t(D+V)^2})$$

can be deduced for suitable perturbations V from this, and whether it could be enough for $\text{Tr}(e^{-tD^2})$ to admit an asymptotic expansion.

First, we address the second part of the question by giving an explicit example where the asymptotic expansion of $\text{Tr}(e^{-tD^2})$ provides no control over the expansions of $\text{Tr}(ae^{-tD^2})$.

Theorem 3.6.4. [EI18, Theorem 3.2] For a bounded operator a and invertible positive operator D such that $D^{-1} \in \mathcal{L}_s$, s > 0, the existence of an asymptotic expansion

$$\operatorname{Tr}(ae^{-tD^2}) \underset{t\downarrow 0}{\sim} \sum_{k=0}^{\infty} \rho_k(t),$$

where

$$\rho_k(t) := \sum_{z \in X_k} \left(\sum_{n=0}^d c_{n,k} \log^n t \right) t^{-z},$$

with $c_{n,k} \in \mathbb{C}$ and for suitable sets $X_k \subset \mathbb{C}$ (for details, see [EI18, Theorem 3.2]), implies the existence of a meromorphic continuation of

$$\zeta_{D^2,a}(s) := \operatorname{Tr}(a|D|^{-2s})$$

to the complex plane, with poles of order at most d+1 located at points in $\bigcup_{k=0}^{\infty} X_k \subset \mathbb{C}$.

Example 3.6.5. Let $\mathcal{A} = \ell_{\infty}(\mathbb{Z}_{\geq 1})$, $\mathcal{H} = \ell_2(\mathbb{Z}_{\geq 1})$ where \mathcal{A} is represented on \mathcal{H} by pointwise multiplication, and let D be the diagonal operator on \mathcal{H} given by

$$De_n = ne_n, \quad n \ge 1.$$

This is a spectral triple for trivial reasons: \mathcal{A} acts on \mathcal{H} by bounded operators, and [D, a] = 0 for all $a \in \mathcal{A}$. Despite being atypical, $(\mathcal{A}, \mathcal{H}, D)$ satisfies most of the assumptions commonly made in the literature in terms of smoothness or summability. The algebra \mathcal{A} is not separable, but all of the following arguments can be performed in a separable (even finite dimensional) subalgebra of \mathcal{A} .

It is a classical result that we have the asymptotic expansion

$$\operatorname{Tr}(e^{-tD^2}) = \sum_{n=1}^{\infty} e^{-tn^2} \underset{t\downarrow 0}{\sim} \frac{\sqrt{\pi}}{2} t^{-\frac{1}{2}} - \frac{1}{2},$$

see for example [Gil04, Lemma 3.1.3]. Nonetheless, the functions $Tr(a|D|^{-2s})$ for $a \in A$ are very badly behaved. For example, let

$$a := \sum_{n=2}^{\infty} \frac{1}{\log n} e_n \in \mathcal{A},$$

so that

$$\zeta_{a,D^2}(s) = \operatorname{Tr}(aD^{-2s}) = \sum_{n=2}^{\infty} \frac{1}{\log n} n^{-2s}, \quad \Re(s) > \frac{1}{2},$$

which is holomorphic on $\Re(s) > \frac{1}{2}$. Now,

$$\frac{d}{ds}\zeta_{a,D^2}(s) = -2\sum_{n=2}^{\infty} n^{-2s}$$

= 2 - 2\zeta(2s), $\Re(s) > \frac{1}{2}$,

where ζ is the Riemann zeta function which has a simple pole at 1. Therefore, $\zeta_{a,D^2}(s) + \log(2s-1)$ is the antiderivative of an entire function, which implies that $\zeta_{a,D^2}(s) = -\log(2s-1) + f(s)$ where f(s) is entire [Rud87, Theorem 10.14]. We conclude that ζ_{a,D^2} does not admit a meromorphic extension to the complex plane, and thus $\operatorname{Tr}(ae^{-tD^2})$ does not have an asymptotic expansion as $t \downarrow 0$ of the type in Theorem 3.6.4.

An even more pathological example is

$$b := \sum_{n=2}^{\infty} \frac{\Lambda(n)}{\log(n)} e_n$$

where Λ is the von Mangoldt function which satisfies

$$\Lambda(n) := \begin{cases} \log(p) & \text{if } n = p^k \text{ for } p \text{ prime}; \\ 0 & \text{otherwise.} \end{cases}$$

A classical formula asserts that [Tit86, p.4]

$$\zeta_{b,D}(s) = \operatorname{Tr}(b|D|^{-s}) = \sum_{n=2}^{\infty} \frac{\Lambda(n)}{\log(n)} n^{-s} = \log \zeta(s), \quad \Re(s) > 1$$

which is badly behaved at every zero of ζ and at s = 1.

To answer the first part of the question, we deduce the following.

Corollary 3.6.6. Let $(\mathcal{A}, \mathcal{H}, D)$ be a regular s-summable spectral triple, s > 0. Let $V, P \in \mathcal{B}, V$ self-adjoint and bounded, where \mathcal{B} is the algebra generated by \mathcal{A} and D. Then for all $f \in T^{\beta}(\mathbb{R})$ with $\beta < -s$, the expressions

$$\operatorname{Tr}(f(tD+tV)), \quad \operatorname{Tr}(Pe^{-t(D+V)^2}) \quad and \quad \operatorname{Tr}(Pe^{-t|D+V|})$$

admit an asymptotic expansion as $t \downarrow 0$ given respectively by

$$\operatorname{Tr}(f(tD+tV)) = \sum_{n=0}^{N} \sum_{m=0}^{N} \sum_{m=0}^{N} \sum_{m_1+\dots+m_n=m} t^{n+m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}\left(\delta_D^{m_1}(V) \cdots \delta_D^{m_n}(V) f^{(n+m)}(tD)\right) + O(t^{N+1-s}),$$

where C_{m_1,\ldots,m_n} is the same as in Proposition 3.5.2,

$$\operatorname{Tr}(Pe^{-t(D+V)^2}) = \sum_{n=0}^{N} \sum_{m=0}^{N} \sum_{m=0}^{N} \sum_{m_1+\dots+m_n=m}^{N} (-t)^{n+m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}(PA^{(m_1)} \cdots A^{(m_n)} \exp(-tD^2)) + O(t^{\frac{N+1-s}{2}}),$$

where $A := DV + VD + V^2$, and $A^{(m)} := \delta^m_{D^2}(A)$, and

$$\operatorname{Tr}(Pe^{-t|D+V|}) = \sum_{n=0}^{N} \sum_{m=0}^{N} \sum_{m=0}^{N} \sum_{m_1+\dots+m_n=m}^{N} (-t)^{n+m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}(P\delta_{|D|}^{m_1}(B) \cdots \delta_{|D|}^{m_n}(B) \exp(-t|D|)) + O(t^{(N+1)(1-\varepsilon)-s}),$$

where B := |D + V| - |D| and $\varepsilon > 0$ can be picked arbitrarily small.

Proof. Given a regular s-summable spectral triple $(\mathcal{A}, \mathcal{H}, D)$, write $\Theta := (1 + D^2)^{\frac{1}{2}}$. Let $V, P \in \mathcal{B}, V$ self-adjoint and bounded, where \mathcal{B} is the algebra generated by \mathcal{A} and D. If $f \in T^{\beta}(\mathbb{R})$ with $\beta < -s$, Theorem 3.6.3 immediately gives that

$$\operatorname{Tr}(f(tD+tV)) = \sum_{n=0}^{N} \sum_{m=0}^{N} \sum_{m=0}^{N} \sum_{m_1+\dots+m_n=m} t^{n+m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}\left(\delta_D^{m_1}(V) \cdots \delta_D^{m_n} + (V)f^{(n+m)}(tD)\right) + O(t^{N+1-s}).$$

Regarding the expansion of $\operatorname{Tr}(Pe^{-t(D+V)^2})$ we have that $D^2 \in \operatorname{OP}^2(\Theta)$, $A \in \operatorname{OP}^1(\Theta)$ since $(\mathcal{A}, \mathcal{H}, D)$ is regular. As $[D^2, A] = [\Theta^2, A] = \Theta[\Theta, A] + [\Theta, A]\Theta$, we have that $A^{(m)} \in \operatorname{OP}^{1+m}(\Theta)$. Furthermore, $\mathcal{B} \subseteq \operatorname{op}(\Theta)$. The proof of this corollary is then the same as the proof of Theorem 3.6.3. Filling in

$$m_N = (N+1)\min\left(\frac{\varepsilon}{h}, \left(1-\frac{r}{h}\right)\right) - \frac{s}{h} = \frac{N+1-s}{2}$$

gives the order of the error term.

For the expansion of $\operatorname{Tr}(Pe^{-t|D+V|})$, while $|D| \in \operatorname{OP}^1(\Theta)$ due to Theorem 2.2.3, to conclude something similar for |D+V| we have to do more work. Note that D has discrete spectrum since $(1+D^2)^{-\frac{1}{2}} \in \mathcal{L}_s$. If $V \in \operatorname{OP}^0(\Theta)$ is self-adjoint, then D+V has real discrete spectrum too. Hence we can modify the function $x \mapsto |x|$ slightly on a small neighbourhood around x = 0 to get a smooth function f which has the property that f(x) = |x| on $\sigma(D+V) \cup \sigma(D)$, and $f \in S^1(\mathbb{R})$ since the second and higher derivatives of f are all compactly supported. Using Theorem 3.3.3 and the observation that $S^1(\mathbb{R}) \subseteq T^{1+\varepsilon}(\mathbb{R})$ for all $\varepsilon > 0$, we have

$$|D+V| \in \mathrm{OP}^{1+\varepsilon}(\Theta), \ |D+V| - |D| = T_{f^{[1]}}^{D+V,D}(V) \in \mathrm{OP}^{\varepsilon}(\Theta).$$

Therefore, we get

$$\operatorname{Tr}(Pe^{-t|D+V|}) = \sum_{n=0}^{N} \sum_{m=0}^{N} \sum_{m=0}^{N} \sum_{m_1+\dots+m_n=m}^{N} (-t)^{n+m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}(P\delta_{|D|}^{m_1}(B) \cdots \delta_{|D|}^{m_n}(B) \exp(-t|D|)) + O(t^{(N+1)(1-\varepsilon)-s}).$$

where B := |D + V| - |D| and $\varepsilon > 0$ can be chosen arbitrarily small.

Apart from providing a perturbative expansion of the spectral action, Corollary 3.6.6 shows that if for all $P \in \mathcal{B}$ we have an expansion

$$\operatorname{Tr}(Pe^{-tD^2}) \underset{t\downarrow 0}{\sim} \sum_{k=0}^{\infty} c_k(P) t^{r_k}$$
(3.16)

for constants $c_k(P) \in \mathbb{C}$, then for all $P \in \mathcal{B}$ self-adjoint and bounded there exist constants $c_k(P, V) \in \mathbb{C}$ such that

$$\operatorname{Tr}(Pe^{-t(D+V)^2}) \underset{t\downarrow 0}{\sim} \sum_{k=0}^{\infty} c_k(P,V)t^{r_k}.$$

Similarly, if each $\text{Tr}(Pe^{-tD^2})$ admits an asymptotic expansions of the type in Theorem 3.6.4, then $\text{Tr}(Pe^{-t(D+V)^2})$ does too.

Remark 3.6.7. Corollary 3.6.6 can be modified to work for non-unital spectral triples. Given a spectral triple $(\mathcal{A}, \mathcal{H}, D)$ with non-unital algebra \mathcal{A} writing $\Theta = (1 + D^2)^{\frac{1}{2}}$, if one assumes instead of $\Theta^{-1} \in \mathcal{L}_s$ that there exists $p \ge 1$ such that $a\Theta^{-s} \in \mathcal{L}_1$ for all $a \in \mathcal{A} \cup [D, \mathcal{A}]$ and s > p as is proposed in [CGRS14], then we also have $a \cdot \operatorname{op}^{-s}(\Theta) \in \mathcal{L}_1$ for s > p. It follows that as $t \downarrow 0$

$$\operatorname{Tr}(af(tD+tV)) = \sum_{n=0}^{N} \sum_{m=0}^{N} t^{n+m} \sum_{m_1+\dots+m_n=m} \frac{C_{m_1,\dots,m_n}}{(n+m)!} \operatorname{Tr}\left(a\delta_D^{m_1}(V)\cdots\delta_D^{m_n}(V)f^{(n+m)}(tD)\right) + O(t^{N+1-p})$$

for $a \in \mathcal{A} \cup [D, \mathcal{A}]$, $V \in \mathcal{B}$ bounded and self-adjoint, and $f \in T^{\beta}(\mathbb{R})$, $\beta < -p$.

We can now answer the first part of the question by Eckstein and Iochum. More precisely, in [EI18, Chapter 5] the question is asked when, for a spectral triple $(\mathcal{A}, \mathcal{H}, D)$, the existence of an asymptotic expansion of $\text{Tr}(e^{-t|D|})$ implies the existence of an expansion of $\text{Tr}(e^{-t|D+V|})$ for a suitable perturbation V. Corollary 3.6.6 compared with Example 3.6.5 suggests that this is not generally possible. We illustrate this with the following example.

Example 3.6.8. Let us revisit Example 3.6.5 where $(\mathcal{A}, \mathcal{H}, D) = (\ell_{\infty}(\mathbb{Z}_{\geq 1}), \ell_2(\mathbb{Z}_{\geq 1}), D)$, and D is defined by

$$De_n = ne_n, \quad n \ge 1.$$
We take as before

$$a := \sum_{n=2}^{\infty} \frac{1}{\log n} e_n.$$

Noting that |D + a| - |D| = a, we can apply Corollay 3.6.6 to get (in this situation we can choose $\varepsilon = 0$)

$$\operatorname{Tr}(e^{-t|D+a|}) = \operatorname{Tr}(e^{-tD}) - t\operatorname{Tr}(ae^{-tD}) + O(t)$$

Taking the Mellin transform we get for $\Re(s) > 1$ (see [EI18, Proposition 2.10])

$$\operatorname{Tr}(|D+a|^{-s}) = \frac{1}{\Gamma(s)} \int_0^\infty t^{s-1} \operatorname{Tr}(e^{-t|D+a|}) dt$$
$$= \frac{1}{\Gamma(s)} \int_0^1 t^{s-1} \operatorname{Tr}(e^{-t|D+a|}) dt + \frac{1}{\Gamma(s)} \int_1^\infty t^{s-1} \operatorname{Tr}(e^{-t|D+a|}) dt.$$

Since

$$\operatorname{Tr}(e^{-t|D+a|}) \le \operatorname{Tr}(e^{-tD}) = (e^t - 1)^{-1} \le 2e^{-t}, \quad t \ge 1,$$

we have that

$$s \mapsto \frac{1}{\Gamma(s)} \int_1^\infty t^{s-1} \operatorname{Tr}(e^{-t|D+a|}) ds$$

is holomorphic. It follows that

$$\operatorname{Tr}(|D+a|^{-s}) = \frac{1}{\Gamma(s)} \int_0^1 t^{s-1} \operatorname{Tr}(e^{-t|D+a|}) dt + \operatorname{holo}_{\mathbb{C}}(s)$$
$$= \frac{1}{\Gamma(s)} \int_0^1 t^{s-1} \operatorname{Tr}(e^{-tD}) - t \operatorname{Tr}(ae^{-tD}) dt + \operatorname{holo}_{\Re(s)>-1}(s)$$
$$= \zeta_{0,D}(s) - s\zeta_{a,D}(s+1) + \operatorname{holo}_{\Re(s)>-1}(s).$$

Since $s \mapsto \zeta_{a,D}(s+1)$ does not extend holomorphically to any punctured neighbourhood of s = 0 as the computations in Example 3.6.5 show, we conclude that $\zeta_{a,D}(s) =$ $\operatorname{Tr}(|D+a|^{-s})$ does not admit a meromorphic extension to the entire complex plane. By Theorem 3.6.4,

$$\operatorname{Tr}(e^{-t|D+a|})$$

does not admit an asymptotic expansion of the type listed in the theorem.

We conclude by remarking that the existence of an asymptotic expansion of $\text{Tr}(Pe^{-tD^2})$ for $P \in \mathcal{B}$ of the type in Theorem 3.6.4 is guaranteed for commutative spectral triples [GS95, Theorem 2.7], and by the same theorem also for almost commutative spectral triples [Sui25, Chapter 10].

Part II

Dixmier Trace Formulas

Chapter 4

Connes' integral formula and quantum ergodicity

I am extremely impressed by your taste displayed in the opening epigraphs to each chapter.

Fedor Sukochev

This chapter is based on [HM24b], joint work with Edward McDonald. Thanks are extended towards Nigel Higson for helpful comments and suggestions, and Magnus Goffeng for pointing out the condition [D, A] being bounded is sufficient for Lemma 4.2.1. We also acknowledge Eric Leichtnam, Qiaochu Ma, and Raphaël Ponge for their assistance. The main results of this chapter are a formula for the noncommutative integral in Theorem 4.1.7, a noncommutative Szegő formula in Theorem 4.2.2, and a link between NCG and Quantum Ergodicity in Theorem 4.4.11.

For applications of NCG in physics and numerical computations in NCG, it is important to know how well spectral triples can be approximated by a finite truncation, since this is all we can measure physically or compute numerically. A physical detector in a measurement for example is typically only effective in a certain energy range. Hence for a spectral triple $(\mathcal{A}, \mathcal{H}, D)$ it makes sense to consider a spectral projection $P = \chi_I(D)$, and truncate the triple into the form

$$(P\mathcal{A}P, P\mathcal{H}, PD).$$

Now, PAP is not generally an algebra anymore. Connes and Van Suijlekom introduced the concept of operator system spectral triples for this purpose [CS21], a generalisation of usual spectral triples.

Definition 4.0.1. A unital operator system spectral triple $(\mathcal{A}, \mathcal{H}, D)$ consists of a unital space \mathcal{A} of bounded operators on a Hilbert space \mathcal{H} such that its norm closure is *-invariant, D is a self-adjoint operator on \mathcal{H} with compact resolvent, and for all $T \in \mathcal{A}$ we have that $T \operatorname{dom}(D) \subseteq \operatorname{dom}(D)$ and [D, T] extends to a bounded operator.

Developments in this direction were made in [DLM14; GS20; GS21; CS22; DLL22; Hek22; GS23; Rie23; LS24; Sui24a; Sui24b] amongst others.

We will connect this paradigm with Connes' (normalised) noncommutative integral

$$a \mapsto \frac{\operatorname{Tr}_{\omega}(a\langle D \rangle^{-d})}{\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})}, \quad a \in B(\mathcal{H}),$$

$$(4.1)$$

recall that $\langle x \rangle := (1 + |x|^2)^{\frac{1}{2}}$. Namely, we will show that given a finite-rank spectral projection $P_{\lambda} := \chi_{[-\lambda,\lambda]}(D)$ where $\chi_{[-\lambda,\lambda]}$ is the indicator function of the interval $[-\lambda,\lambda] \subseteq \mathbb{R}$, the functional

$$P_{\lambda}aP_{\lambda} \mapsto \frac{\operatorname{Tr}(P_{\lambda}aP_{\lambda})}{\operatorname{Tr}(P_{\lambda})}, \quad a \in B(\mathcal{H}),$$

$$(4.2)$$

approximates the noncommutative integral on spectrally truncated compact spectral triples (Proposition 4.1.1, Theorem 4.1.7). This is a result in the spirit of [Ste19], where finite-rank approximations of zeta residues are given. We however do not assume the existence of a full asymptotic expansion of the heat trace. Instead, we focus our efforts on the computation of the first term of this expansion, which is the noncommutative integral.

The language involved is closely tied to the field of quantum ergodicity, the inception of which can largely be credited to Shnirelman, Zelditch and Colin de Verdière [Shn74; Col85; Zel87]. For reviews of this field, see [Zel10; Zel17]. Quantum ergodicity is a property of an

operator which can mean various things. A common definition is that, given a compact Riemannian manifold M and a positive self-adjoint operator Δ on $L_2(M)$ with compact resolvent, the operator Δ is said to be quantum ergodic if for every orthonormal basis $\{e_n\}_{n=0}^{\infty}$ of $L_2(M)$ consisting of eigenfunctions of Δ there exists a density one subsequence $J \subseteq \mathbb{N}$ such that for all zero-order pseudodifferential operators $Op(\sigma)$ with principal symbol $\sigma \in C^{\infty}(S^*M)$,

$$\lim_{J\ni j\to\infty} \langle e_j, \operatorname{Op}(\sigma) e_j \rangle_{L_2(M)} = \int_{S^*M} \sigma \, d\nu,$$

and where ν is the measure on the cotangent sphere S^*M induced by the Riemannian metric. In this context, a density one subsequence means that

$$\frac{\#J \cap \{0, \dots, n\}}{n+1} \to 1, \quad n \to \infty.$$

Quantum ergodicity implies in particular that the functions $|e_j|^2$ become uniformly distributed over M as $J \ni j \to \infty$, in the sense that the measures $|e_j|^2 d\nu_g$ converge to $d\nu_g$ in the weak*-topology, see Figure 4.1.

Although quantum ergodicity shares a philosophical link with NCG — emerging from a functional-analytic approach to ergodic geodesic flow on compact Riemannian manifolds — there has yet to be made an explicit connection between the two fields, despite their contemporary development. We will show in Section 4.4 that our results on the noncommutative integral on truncated spectral triples provide the means with which the gap can be bridged.

We furthermore propose a straightforward noncommutative generalisation of the property of ergodic geodesic flow on compact Riemannian manifolds for spectral triples, and explore what some results from the field of quantum ergodicity provide in this context. Our definition of ergodicity is known in the study of C^* -dynamical systems as uniqueness of the vacuum state, and hence a result by Zelditch [Zel96] can now be recognised as an NCG version of the classical result by Colin de Verdière that ergodicity of the geodesic flow implies quantum ergodicity of the Laplace–Beltrami operator [Col85], see Theorem 4.4.11 below.



CHAPTER 4. CONNES' INTEGRAL FORMULA AND QUANTUM ERGODICITY

Figure 4.1: An illustration of 64 eigenfunctions of the Laplacian on a rose-shaped domain with Dirichlet boundary conditions, corresponding to the 64 smallest eigenvalues (counting multiplicities). Generated in Python with the finite-element method, using FEniCSx [Bar+23]. Quantum ergodicity would imply that there exists a density one subsequence of eigenfunctions $\{e_j\}_{j\in J}$ along which the measures $|e_j|^2 d\lambda$ converge to the uniform (Lebesgue) measure $d\lambda$ in the weak*-topology.

Additionally, we will draw from a result of Widom [Wid79] on the asymptotic behaviour of the functional (4.2), which directly implies a Szegő limit formula for spectral triples that satisfy the Weyl law (Theorem 4.2.2). This provides that for all self-adjoint $A \in B(\mathcal{H})$ which map dom |D| into itself and such that [D, A] is bounded,

$$\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \cdot \omega \circ M\left(\frac{\operatorname{Tr}(f(P_{\lambda_n} A P_{\lambda_n}))}{\operatorname{Tr}(P_{\lambda_n})}\right) = \operatorname{Tr}_{\omega}(f(A) \langle D \rangle^{-d}), \quad f \in C(\mathbb{R}), \ f(0) = 0.$$

Here, $M : \ell_{\infty} \to \ell_{\infty}$ is a logarithmic averaging operator, and $\omega \in \ell_{\infty}^*$ is an extended limit. Details are provided in Section 4.2. Note that we use the short-hand notation $\omega \circ M(a_n)$ for $\omega \circ M(\{a_n\}_{n=1}^{\infty})$. We remark that this result provides the insight that Szegő limit theorems can be interpreted as versions of Connes' integral formula.

An outline of this chapter is as follows. We first explore and make precise the relation between the functionals (4.1) and (4.2) in Section 4.1. Section 4.2 provides the mentioned Szegő limit theorem for NCG. Next, we discuss a way of interpreting the functional (4.2) when the noncommutative integral (4.1) is not defined, for example in θ -summable or Li₁summable spectral triples. Namely, we relate the functional (4.2) to a functional that is sometimes called the Fröhlich functional, which has been studied extensively in [GRU19] as a KMS state. Finally, in Section 4.4 we exhibit our study in quantum ergodicity and its relation to NCG through our results on the noncommutative integral.

4.1 Integration on truncated spectral triples

Let us fix a closed self-adjoint operator D on a separable Hilbert space \mathcal{H} such that $\langle D \rangle^{-d} \in \mathcal{L}_{1,\infty}$, where d > 0 and $\langle x \rangle := (1 + |x|^2)^{\frac{1}{2}}$. We fix an extended limit $\omega \in \ell_{\infty}^*$ and assume that $\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) > 0$. We write $P_{\lambda} := \chi_{[-\lambda,\lambda]}(D)$. This situation is modeled after (compact) *d*-dimensional spectral triples $(\mathcal{A}, \mathcal{H}, D)$.

We first provide the most straight-forward approach to the noncommutative integral on truncated triples, using standard techniques that are employed in quantum ergodicity [Col85]. We write

$$f(t) \sim Ct^{-c}$$

to mean

$$\lim_{t \to 0} t^{\alpha} f(t) = C.$$

Proposition 4.1.1. Let $a \in B(\mathcal{H})$. If there exist constants $C, C(a) \in \mathbb{R}$ with

$$\operatorname{Tr}(e^{-tD^2}) \sim Ct^{-\frac{d}{2}}, \quad \operatorname{Tr}(ae^{-tD^2}) \sim C(a)t^{-\frac{d}{2}},$$

then

$$\operatorname{Tr}_{\omega}(a\langle D\rangle^{-d}) = \operatorname{Tr}_{\omega}(\langle D\rangle^{-d}) \lim_{\lambda \to \infty} \frac{\operatorname{Tr}(P_{\lambda}aP_{\lambda})}{\operatorname{Tr}(P_{\lambda})}.$$

Proof. By [LSZ21, Corollary 8.1.3] we have that

$$C = \Gamma(\frac{d}{2} + 1) \operatorname{Tr}_{\omega}(\langle D \rangle^{-d}), \quad C(a) = \Gamma(\frac{d}{2} + 1) \operatorname{Tr}_{\omega}(a \langle D \rangle^{-d}).$$

Recall that we assume $\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) > 0$. An application of the Hardy–Littlewood Tauberian theorem [Fel71, Theorem XII.5.2] to the function $\operatorname{Tr}(e^{-tD^2})$ shows that

$$\operatorname{Tr}(P_{\lambda}) \sim \operatorname{Tr}_{\omega}(\langle D \rangle^{-d})\lambda^{d}, \quad \lambda \to \infty.$$

Applying the theorem again to the function $\operatorname{Tr}(ae^{-tD^2})$ then gives that $\lim_{\lambda \to \infty} \frac{\operatorname{Tr}(P_{\lambda}aP_{\lambda})}{\operatorname{Tr}(P_{\lambda})}$ exists and is equal to $\frac{\operatorname{Tr}_{\omega}(a\langle D \rangle^{-d})}{\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})}$.

Remark 4.1.2. The Hardy-Littlewood Tauberian theorem implies that the condition $\operatorname{Tr}(e^{-tD^2}) \sim Ct^{-\frac{d}{2}}$ as $t \to 0$ is equivalent to $\lambda(k, D^2) \sim \widetilde{C}k^{\frac{2}{d}}$ as $k \to \infty$ [Fel71, Theorem XII.5.2].

Definition 4.1.3. We say that D^2 (as fixed at the start of this section) satisfies a Weyl law if $\operatorname{Tr}(e^{-tD^2}) \sim Ct^{-\frac{d}{2}}$, and it satisfies a local Weyl law for an operator $a \in B(\mathcal{H})$ if $\operatorname{Tr}(ae^{-tD^2}) \sim C(a)t^{-\frac{d}{2}}$.

See [MSZ22] for an investigation of the validity of the (local) Weyl law for spectral triples, and [Pon23] for an extensive study of Weyl's law in relation to Connes' integral formula. The latter, work by Ponge, answers some questions regarding Weyl laws and the noncommutative integral related to measurability of operators. Although the local Weyl laws hold for Riemannian manifolds and a wide class of spectral triples [GS95; EI18; MSZ22], we have seen an example of a spectral triple in which such behaviour does not hold in Chapter 3, see Example 3.6.5. In the remainder of this section we show what can be deduced without this condition. We now fix an orthonormal basis $\{e_n\}_{n=0}^{\infty}$ of eigenvectors of |D|, ordered such that the corresponding eigenvalues $\{\lambda_n\}_{n=0}^{\infty}$ are non-decreasing.

Lemma 4.1.4. Let $A \in B(\mathcal{H})$. Then

$$\operatorname{Tr}_{\omega}(A\langle D\rangle^{-d}) = \omega \left(\frac{1}{\log(n+2)} \sum_{k=0}^{n} \langle \lambda_k \rangle^{-d} \langle e_k, Ae_k \rangle\right).$$

If D^2 satisfies Weyl's law, i.e. $\lambda_k \sim Ck^{\frac{1}{d}}$, this simplifies to

$$\operatorname{Tr}_{\omega}(A\langle D\rangle^{-d}) = \operatorname{Tr}_{\omega}(\langle D\rangle^{-d}) \cdot \omega \left(\frac{1}{\log(n+2)} \sum_{k=0}^{n} \frac{\langle e_k, Ae_k \rangle}{k+1}\right)$$

Proof. The first part is [LSZ21, Corollary 7.1.4(c)], i.e. a corollary of Theorem 1.6.4, the second claim is [LSZ21, Theorem 7.1.5(a)] or [LS11]. \Box

What appears in the lemma above is the logarithmic mean $M: \ell_{\infty} \to \ell_{\infty}$, defined by

$$M: x \mapsto \left\{ \frac{1}{\log(n+2)} \sum_{k=0}^{n} \frac{x_k}{k+1} \right\}_{n=0}^{\infty}.$$

This can be compared with the Cesàro mean

$$C: x \mapsto \left\{ \frac{1}{n+1} \sum_{k=0}^{n} x_k \right\}_{n=0}^{\infty}.$$

Lemma 4.1.5. For any sequence $x \in \ell_{\infty}$, we have

$$(M(x))_n = (M \circ C(x))_n + o(1), \quad n \to \infty.$$

Proof. For $x \in \ell_{\infty}$ and $k \ge 0$ we have

$$\frac{x_k}{k+1} = \left(\frac{1}{k+1}\sum_{l=0}^k x_l\right) - \frac{k}{k+1}\left(\frac{1}{k}\sum_{l=0}^{k-1} x_l\right)$$
$$= (C(x))_k - (C(x))_{k-1} + \frac{1}{k+1}(C(x))_{k-1}.$$

Hence, as $n \to \infty$

$$(M(x))_n = \frac{1}{\log(n+2)} \sum_{k=0}^n \frac{x_k}{k+1}$$

= $\frac{1}{\log(n+2)} \left((C(x))_n + \sum_{k=1}^n \frac{1}{k+1} (C(x))_{k-1} \right)$
= $(M \circ T \circ C(x))_n + o(1),$

where $T: (x_0, x_1, x_2, ...) \mapsto (0, x_0, x_1, ...)$ is the right-shift operator on ℓ_{∞} . Finally, for any bounded sequence $a \in \ell_{\infty}$, we have that

$$(M \circ T(a))_n - (M(a))_n = o(1), \quad n \to \infty,$$

which can be found in [LSZ21, Lemma 6.2.12].

Since both M and C are regular transformations in Hardy's terminology [Har49, Chapter III], meaning that $M(x)_n \to c$ whenever $x_n \to c$, it is a consequence of Lemma 4.1.5 that for $x \in \ell_{\infty}$, if $C(x)_n \to c$ then $M(x)_n \to c$ as $n \to \infty$. We introduce one more crucial lemma. Namely, writing Q_n for the projection onto $\{e_0, \ldots, e_n\}$, we want to switch freely between

$$\frac{\operatorname{Tr}(P_{\lambda}aP_{\lambda})}{\operatorname{Tr}(P_{\lambda})}, \quad \frac{\operatorname{Tr}(Q_{n}aQ_{n})}{\operatorname{Tr}(Q_{n})}$$

The first can be written as $\frac{\operatorname{Tr}(Q_{N(\lambda)}aQ_{N(\lambda)})}{\operatorname{Tr}(Q_{N(\lambda)})}$, where $N(\lambda)$ is the greatest $k \geq 0$ such that $\lambda_k \leq \lambda$, and thus can be interpreted as a subsequence of the second. The following lemma can therefore be applied, which appeared as [Aza+22, Lemma 4.8], i.e. the published version of Chapter 5, in a slightly weaker form and in a different context.

Lemma 4.1.6. Let $\phi : \mathbb{N} \to \mathbb{R}_{>0}$ be an increasing function such that $\phi(n) \to \infty$ as $n \to \infty$, let $\{a_k\}_{k \in \mathbb{N}} \subseteq \mathbb{R}$ be a sequence such that $\left\{\frac{1}{\phi(n)}\sum_{k=0}^{n}|a_k|\right\}_{n=0}^{\infty}$ is bounded, and let $\{k_0, k_1, \ldots\}$ be an infinite, increasing sequence of positive integers such that

$$\lim_{n \to \infty} \frac{\phi(k_{n+1})}{\phi(k_n)} = 1,$$

r

and

$$\frac{1}{\phi(k_n)}\sum_{k=k_{n-1}+1}^{k_n}|a_k|=o(1),\quad n\to\infty.$$

Labeling $k_{i_n} := \min\{k_i : k_i \ge n\}$, we have that

$$\frac{1}{\phi(n)}\sum_{k=0}^{n}a_{k} = \frac{1}{\phi(k_{i_{n}})}\sum_{k=0}^{k_{i_{n}}}a_{k} + o(1), \quad n \to \infty.$$

Proof. Without loss of generality, we can assume that $\{a_k\}_{k\in\mathbb{N}}$ is a positive sequence. We have

$$\frac{1}{\phi(n)} \sum_{k=1}^{n} a_k - \frac{1}{\phi(k_{i_n})} \sum_{k=1}^{k_{i_n}} a_k \le \left(\frac{\phi(k_{i_n})}{\phi(k_{i_n-1})} - 1\right) \frac{1}{\phi(k_{i_n})} \sum_{k=1}^{k_{i_n}} a_k = o(1);$$

$$\frac{1}{\phi(k_{i_n})} \sum_{k=1}^{k_{i_n}} a_k - \frac{1}{\phi(n)} \sum_{k=1}^{n} a_k \le \frac{1}{\phi(k_{i_n})} \sum_{k=k_{i_n-1}+1}^{k_{i_n}} a_k = o(1).$$

We can now prove the main result of this section.

Theorem 4.1.7. Let $A \in B(\mathcal{H})$. If D^2 satisfies Weyl's law (Definition 4.1.3), then

$$\frac{\operatorname{Tr}_{\omega}(A\langle D\rangle^{-d})}{\operatorname{Tr}_{\omega}(\langle D\rangle^{-d})} = (\omega \circ M) \Big(\langle e_n, Ae_n \rangle \Big) = (\omega \circ M) \left(\frac{\operatorname{Tr}(Q_n A Q_n)}{\operatorname{Tr}(Q_n)}\right) = (\omega \circ M) \left(\frac{\operatorname{Tr}(P_{\lambda_n} A P_{\lambda_n})}{\operatorname{Tr}(P_{\lambda_n})}\right)$$

If furthermore Q is an operator with $\bigcap_{n\geq 0} \operatorname{dom}(D^n) \subseteq \operatorname{dom} Q$ such that for some $s \geq -d$, $Q\langle D \rangle^{-s}$ extends to a bounded operator, e.g. if $Q \in \operatorname{op}^s(\langle D \rangle)$, we have

$$\operatorname{Tr}_{\omega}(Q) = \omega \left(\frac{\operatorname{Tr}(P_{\lambda_n} Q P_{\lambda_n})}{\log(\operatorname{Tr}(P_{\lambda_n}))} \right), \quad s = -d;$$
$$\frac{\operatorname{Tr}_{\omega}(Q\langle D\rangle^{-s-d})}{\left(\operatorname{Tr}_{\omega}(\langle D\rangle^{-d})\right)^{\frac{s}{d}+1}} = \left(\frac{s}{d} + 1\right) \omega \circ M \left(\frac{\operatorname{Tr}(P_{\lambda_n} Q P_{\lambda_n})}{\operatorname{Tr}(P_{\lambda_n})^{\frac{s}{d}+1}}\right), \quad s > -d$$

Proof. The first equality in the first equation appeared in Lemma 4.1.4, the second equality is a consequence of Lemma 4.1.5 and the trivial identity

$$\frac{\operatorname{Tr}(Q_n A Q_n)}{\operatorname{Tr}(Q_n)} = \frac{1}{n+1} \sum_{k=0}^n \langle e_k, A e_k \rangle.$$

The last equality follows from Lemma 4.1.6 when taking $\phi(n) = n + 1$, since the Weyl law gives that $\frac{N(\lambda_n)}{N(\lambda_{n+1})} \to 1$. The assumption

$$\frac{1}{N(\lambda_n)} \sum_{k=N(\lambda_{n-1})+1}^{N(\lambda_n)} \langle e_k, Ae_k \rangle = o(1), \quad n \to \infty$$

in Lemma 4.1.6 is satisfied, since

$$\frac{1}{N(\lambda_n)}\sum_{k=N(\lambda_{n-1})+1}^{N(\lambda_n)} |\langle e_k, Ae_k\rangle| \le ||A||_{\infty} \frac{N(\lambda_n) - N(\lambda_{n-1})}{N(\lambda_n)} = o(1), \quad n \to \infty.$$

Now take an operator Q with $\bigcap_{n\geq 0} \operatorname{dom}(D^n) \subseteq \operatorname{dom} Q$ such that $Q\langle D \rangle^{-s}$ extends to a bounded operator. For s = -d, the given formula for $\operatorname{Tr}_{\omega}(Q)$ is a combination of Lemma 4.1.4 and Lemma 4.1.6. For $s \neq -d$, first, due to Weyl's law

$$\langle \lambda_k \rangle^{-s-d} = \left(\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \right)^{\frac{s}{d}+1} (k+1)^{-\frac{s}{d}-1} + o\left((k+1)^{-\frac{s}{d}-1} \right), \quad k \to \infty$$

and hence, since $(k+1)^{-\frac{s}{d}}\langle e_k, Qe_k\rangle$ is bounded, we have that

$$\langle \lambda_k \rangle^{-s-d} \langle e_k, Qe_k \rangle = \left(\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \right)^{\frac{s}{d}+1} (k+1)^{-\frac{s}{d}-1} \langle e_k, Qe_k \rangle + o\left((k+1)^{-1} \right), \quad k \to \infty.$$

Now applying Lemma 4.1.4 and then Lemma 4.1.5,

$$\begin{aligned} \operatorname{Tr}_{\omega}(Q\langle D\rangle^{-s-d}) &= \omega \left(\frac{1}{\log(n+2)} \sum_{k \leq n} \langle \lambda_k \rangle^{-s-d} \langle e_k, Qe_k \rangle \right) \\ &= \left(\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \right)^{\frac{s}{d}+1} \omega \circ M \left((n+1)^{-\frac{s}{d}} \langle e_n, Qe_n \rangle \right) \\ &= \left(\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \right)^{\frac{s}{d}+1} \omega \circ M \left(\frac{1}{n+1} \sum_{k \leq n} (k+1)^{-\frac{s}{d}} \langle e_k, Qe_k \rangle \right). \end{aligned}$$

Using Abel's summation formula, as $n \to \infty$

$$\frac{1}{n+1} \sum_{k \le n} (k+1)^{-\frac{s}{d}} \langle e_k, Qe_k \rangle = (n+1)^{-\frac{s}{d}-1} \sum_{k \le n} \langle e_k, Qe_k \rangle -\frac{1}{n+1} \sum_{k \le n-1} \left((k+2)^{-\frac{s}{d}} - (k+1)^{-\frac{s}{d}} \right) \sum_{j \le k} \langle e_j, Qe_j \rangle.$$

By Taylor's formula, we have

$$(k+2)^{-\frac{s}{d}} - (k+1)^{-\frac{s}{d}} + \frac{s}{d}(k+1)^{-\frac{s}{d}-1} = \frac{s}{d}\left(\frac{s}{d}+1\right)\int_0^1 (1-\theta)(k+1+\theta)^{-\frac{s}{d}-2}\,d\theta.$$

Therefore

$$\frac{1}{n+1} \sum_{k \le n} (k+1)^{-\frac{s}{d}} \langle e_k, Qe_k \rangle$$
$$= (n+1)^{-\frac{s}{d}-1} \sum_{k \le n} \langle e_k, Qe_k \rangle + \frac{s}{d} C \left(\left\{ (k+1)^{-\frac{s}{d}-1} \sum_{j \le k} \langle e_j, Qe_j \rangle \right\}_{k=0}^{\infty} \right)_n$$
$$- \frac{s}{d} \left(\frac{s}{d} + 1 \right) \int_0^1 (1-\theta) C \left(\left\{ (k+1+\theta)^{-\frac{s}{d}-2} \sum_{j \le k} \langle e_j, Qe_j \rangle \right\}_{k=0}^{\infty} \right)_n d\theta,$$

where $C : \ell_{\infty} \to \ell_{\infty}$ is the Cesàro operator. Since $(j+1)^{-\frac{s}{d}} \langle e_j, Qe_j \rangle$ is bounded and s > -d we have

$$\left| (k+1+\theta)^{-\frac{s}{d}-2} \sum_{j \le k} \langle e_j, Q e_j \rangle \right| = O((k+1)^{-1}), \quad k \to \infty.$$

Thus

$$\frac{1}{n+1} \sum_{k \le n} (k+1)^{-\frac{s}{d}} \langle e_k, Qe_k \rangle = (n+1)^{-\frac{s}{d}-1} \sum_{k \le n} \langle e_k, Qe_k \rangle + \frac{s}{d} C \left(\left\{ (k+1)^{-\frac{s}{d}-1} \sum_{j \le k} \langle e_j, Qe_j \rangle \right\}_{k=0}^{\infty} \right)_n + O\left(\frac{\log(n+2)}{n+1}\right).$$

Using Lemma 4.1.5 again, we have

$$\operatorname{Tr}_{\omega}(Q\langle D\rangle^{-s-d}) = \left(\operatorname{Tr}_{\omega}(\langle D\rangle^{-d})\right)^{\frac{s}{d}+1} \left(1 + \frac{s}{d}\right) \omega \circ M\left(\frac{1}{(n+1)^{\frac{s}{d}+1}} \sum_{k \le n} \langle e_k, Qe_k \rangle\right).$$

To apply Lemma 4.1.6, taking $\phi(n) = (n+1)^{\frac{s}{d}+1}$ and $k_n = N(\lambda_n)$, we need to check that

$$\frac{1}{N(\lambda_n)^{\frac{s}{d}+1}} \sum_{k=N(\lambda_{n-1})+1}^{N(\lambda_n)} |\langle e_k, Qe_k \rangle| \lesssim \frac{1}{N(\lambda_n)^{\frac{s}{d}+1}} \sum_{k=N(\lambda_{n-1})+1}^{N(\lambda_n)} k^{\frac{s}{d}}$$
$$\lesssim \frac{N(\lambda_n)^{\frac{s}{d}+1} - N(\lambda_{n-1})^{\frac{s}{d}+1}}{N(\lambda_n)^{\frac{s}{d}+1}}$$
$$= o(1), \quad n \to \infty.$$

Hence Lemma 4.1.6 applies, and we conclude that

$$\operatorname{Tr}_{\omega}(Q\langle D\rangle^{-s-d}) = \left(\frac{s}{d} + 1\right) \left(\operatorname{Tr}_{\omega}(\langle D\rangle^{-d})\right)^{\frac{s}{d}+1} \omega \circ M\left(\frac{\operatorname{Tr}(P_{\lambda_n}QP_{\lambda_n})}{\operatorname{Tr}(P_{\lambda_n})^{\frac{s}{d}+1}}\right).$$

As an obvious consequence of Theorem 4.1.7, if for $A \in B(\mathcal{H})$

$$\frac{\operatorname{Tr}(P_{\lambda}AP_{\lambda})}{\operatorname{Tr}(P_{\lambda})}$$

converges, it follows that, provided D^2 satisfies Weyl's law, the limit must necessarily be the noncommutative integral of A. Furthermore, if the noncommutative integral is independent of ω , meaning that $A\langle D \rangle^{-d}$ is Dixmier measurable (see e.g. [LSZ21; LMSZ23; Pon23]) one can replace $\omega \circ M$ by $\lim \circ M$ on the right hand sides of Theorem 4.1.7. Finally, with a Weyl law, for self-adjoint $A \in B(\mathcal{H})$ we have

$$\liminf_{k \to \infty} \langle e_k, Ae_k \rangle \le \liminf_{\lambda \to \infty} \frac{\operatorname{Tr}(P_\lambda A P_\lambda)}{\operatorname{Tr}(P_\lambda)} \le \frac{\operatorname{Tr}_\omega(A \langle D \rangle^{-d})}{\operatorname{Tr}_\omega(\langle D \rangle^{-d})} \le \limsup_{\lambda \to \infty} \frac{\operatorname{Tr}(P_\lambda A P_\lambda)}{\operatorname{Tr}(P_\lambda)} \le \limsup_{k \to \infty} \langle e_k, Ae_k \rangle$$

All results achieved in this section are different flavours of the observation that the noncommutative integral is the limit point — in a weak, averaging notion — of the sequence $\{\langle e_k, Ae_k \rangle\}_{k=0}^{\infty}$. For the circle \mathbb{S}^1 this is not surprising; given $f = \sum_{k=-\infty}^{\infty} a_k e_k \in L_1(\mathbb{S}^1)$ in Fourier basis, we have for every $k \in \mathbb{Z}$

$$\langle e_k, M_f e_k \rangle = a_0 = \int_{\mathbb{S}^1} f(t) \, dt.$$

More generally, Proposition 4.1.1 combined with Connes' integral formula (Theorem 1.4.3) and Lemma 4.1.6 shows that for any *d*-dimensional closed Riemannian manifold M with volume form ν_g we have that the Cesàro mean of the sequence

$$\langle e_k, M_f e_k \rangle, \quad f \in C(M)$$

converges to $\int_M f \, d\nu_g$. This fact is precisely what started investigations into quantum ergodicity. Recall that this covers the study of to what extent the matrix elements $\langle e_k, M_f e_k \rangle$ themselves converge to an integral of f. More details will be provided in Section 4.4.

Previously, in [LPS10; LS11; LS221] it had already been observed that for spectral triples $(\mathcal{A}, \mathcal{H}, D)$ where D^2 satisfies Weyl's law that if the noncommutative integral

$$\frac{\operatorname{Tr}_{\omega}(a\langle D\rangle^{-d})}{\operatorname{Tr}_{\omega}(\langle D\rangle^{-d})}, \quad a \in \mathcal{A},$$

is independent of ω , then

$$\frac{1}{\log(n+2)}\sum_{k=0}^{n}\frac{\langle e_k, ae_k\rangle}{k+1}, \quad a \in \mathcal{A},$$

converges as $n \to \infty$, which was interpreted as being related to quantum ergodicity.

In quantum ergodicity and related fields, there is a vast literature on the properties and asymptotics of the operators $P_{\lambda}aP_{\lambda}$. Through the results established in this section, the link with Connes' integral formula unlocks this literature for study from the perspective of noncommutative geometry. One result from this cross-pollination is a Szegő limit theorem for truncated spectral triples.

4.2 Szegő limit theorem

Szegő proved various limit theorems concerning determinants of Toeplitz matrices, inspired by a conjecture by Pólya and after work on these determinants by Toeplitz, Caratheodory and Fejér, see [Sze15] and references therein. Much later, Widom provided a generalisation of these results with a simplified proof [Wid79], see also [LS96] for a version for elliptic selfadjoint (pseudo)differential operators on manifolds without boundary. We now provide a translation of the results of Widom into noncommutative geometry. We thank Magnus Goffeng for pointing out that instead of requiring that [|D|, A] is bounded, it suffices to assume in the following lemma that [D, A] is bounded.

Lemma 4.2.1 ([Wid79]). Let D^2 satisfy Weyl's law (Definition 4.1.3), and let $A \in B(\mathcal{H})$ and $B \in B(\mathcal{H})$ map dom |D| into itself, and be such that [D, A] and [D, B] are bounded. Then

$$\lim_{\lambda \to \infty} \frac{\operatorname{Tr}(P_{\lambda}A(1-P_{\lambda})BP_{\lambda})}{\operatorname{Tr}(P_{\lambda})} = 0.$$

Proof. First, [D, A] being bounded implies that $[\langle D \rangle^{\frac{1}{2}}, A]$ is bounded due to our results in Chapter 3, namely Theorem 3.3.3 and Proposition 3.4.1 (alternatively, see [GVF01, Lemma 10.13]). Hence, replacing D by $\langle D \rangle^{\frac{1}{2}}$, we can assume that D is positive and that [|D|, A] and [|D|, B] are bounded. Then, by the Cauchy-Schwarz inequality, an equivalent formulation of the statement is that for every B such that [|D|, B] is bounded, we have

$$\lim_{\lambda \to \infty} \frac{\|P_{\lambda}B(1-P_{\lambda})\|_{HS}^2}{\operatorname{Tr}(P_{\lambda})} = 0$$

where $\|\cdot\|_{HS}$ is the Hilbert–Schmidt norm. The following argument is essentially due to Widom [Wid79], see also [Gui79, Lemma 3.4].

Let N > 0. A quick computation [Wid79, p. 145] shows that

$$||P_{\lambda}B(1-P_{\lambda+N})||_{HS}^{2} \le N^{-2} ||P_{\lambda}[|D|, B](1-P_{\lambda+N})||_{HS}^{2}$$

By the triangle inequality, we have

$$\begin{aligned} \|P_{\lambda}B(1-P_{\lambda})\|_{HS}^{2} &\leq 2\|P_{\lambda}B(P_{\lambda+N}-P_{\lambda})\|_{HS}^{2} + 2\|P_{\lambda}B(1-P_{\lambda+N})\|_{HS}^{2} \\ &\leq 2\|B\|_{\infty}^{2}\operatorname{Tr}(P_{\lambda+N}-P_{\lambda}) + 2N^{-2}\operatorname{Tr}(P_{\lambda})\|[|D|,B]\|_{\infty}^{2}. \end{aligned}$$

Weyl's law implies that

$$\operatorname{Tr}(P_{\lambda+N} - P_{\lambda}) = o(\operatorname{Tr}(P_{\lambda})), \quad \lambda \to \infty,$$

and hence

$$\limsup_{\lambda \to \infty} \frac{\|P_{\lambda}B(1-P_{\lambda})\|_{HS}^2}{\operatorname{Tr}(P_{\lambda})} \le 2N^{-2}\|[|D|,B]\|_{\infty}^2.$$

Since N is arbitrary, this completes the proof.

Following Widom [Wid79] further, Lemma 4.2.1 can be combined with the characterisation of Connes' integral theorem in Theorem 4.1.7 into a Szegő limit theorem.

Theorem 4.2.2. Let D^2 satisfy Weyl's law (Definition 4.1.3), and let $A \in B(\mathcal{H})$ be self-adjoint and such that it maps dom |D| into itself and [D, A] is bounded. Then

$$\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \cdot (\omega \circ M) \left(\frac{\operatorname{Tr}(f(P_{\lambda_n} A P_{\lambda_n}))}{\operatorname{Tr}(P_{\lambda_n})} \right) = \operatorname{Tr}_{\omega}(f(A) \langle D \rangle^{-d}), \quad f \in C(\mathbb{R}), \ f(0) = 0.$$

If for every positive integer k there is some constant $C_k \in \mathbb{R}$ with

$$\operatorname{Tr}(A^k e^{-tD^2}) \sim C_k t^{-\frac{d}{2}},$$

then for every $f \in C(\mathbb{R})$ with f(0) = 0 we have

$$\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \lim_{\lambda \to \infty} \frac{\operatorname{Tr}(f(P_{\lambda}AP_{\lambda}))}{\operatorname{Tr}(P_{\lambda})} = \operatorname{Tr}_{\omega}(f(A)\langle D \rangle^{-d})$$

Proof. We sketch the proof of the stronger identity

$$\operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \cdot (\omega \circ M) \left(\frac{\operatorname{Tr}(P_{\lambda_n} f(P_{\lambda_n} A P_{\lambda_n}) P_{\lambda_n})}{\operatorname{Tr}(P_{\lambda_n})} \right) = \operatorname{Tr}_{\omega}(f(A) \langle D \rangle^{-d}), \quad f \in C(\mathbb{R}).$$

$$(4.3)$$

Lemma 4.2.1 gives that

$$\lim_{\lambda \to \infty} \frac{\operatorname{Tr} \left(P_{\lambda} A^k P_{\lambda} - (P_{\lambda} A P_{\lambda})^k \right)}{\operatorname{Tr} (P_{\lambda})} = 0, \quad k \ge 1,$$
(4.4)

which implies equation (4.3) for polynomial f through Theorem 4.1.7. An application of the Stone–Weierstrass theorem provides an extension to continuous functions. Details can be found in [Wid79].

The second assertion for polynomial functions f is a combination of (4.4) and Proposition 4.1.1. If f is a continuous function on \mathbb{R} with f(0) = 0, let $\varepsilon > 0$ and choose a polynomial function p with p(0) = 0 such that

$$\|f-p\|_{L_{\infty}([-\|A\|_{\infty},\|A\|_{\infty}])} < \varepsilon.$$

Then

$$\frac{\operatorname{Tr}((f-p)(P_{\lambda}AP_{\lambda}))}{\operatorname{Tr}(P_{\lambda})}\Big| < \varepsilon$$

and

$$\operatorname{Tr}_{\omega}((f-p)(A)\langle D\rangle^{-d})| \leq \varepsilon \|\langle D\rangle^{-d}\|_{1,\infty}$$

Hence

$$\limsup_{\lambda \to \infty} \left| \operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \frac{\operatorname{Tr}(f(P_{\lambda}AP_{\lambda}))}{\operatorname{Tr}(P_{\lambda})} - \operatorname{Tr}_{\omega}(f(A)\langle D \rangle^{-d}) \right| \leq 2\varepsilon \|\langle D \rangle^{-d}\|_{1,\infty}.$$

Since ε is arbitrary, this implies the result.

4.3 Fröhlich functional

So far, we have considered situations modeled after *d*-dimensional spectral triples, where $\langle D \rangle^{-d} \in \mathcal{L}_{1,\infty}$. There are many examples of spectral triples that do not satisfy this condition, however. Instead, one could consider the property of θ -summability, which says that $\operatorname{Tr}(e^{-tD^2}) < \infty$ for all t > 0, or Li₁-summability which requires $\operatorname{Tr}(e^{-t|D|}) < \infty$ for t large enough.

For this section, we therefore assume that D is a self-adjoint operator with compact resolvent, but we do not assume Weyl laws. The functional

$$a \mapsto \lim_{t \to \beta} \frac{\operatorname{Tr}(ae^{-t|D|})}{\operatorname{Tr}(e^{-t|D|})}, \quad a \in B(\mathcal{H}).$$

which is sometimes called the Fröhlich functional after [FGR98; CFF93], has been studied extensively in the literature [GM18; GRU19]. We highlight the relation between this functional and the one that has been the object of study in this chapter.

Proposition 4.3.1. Assume there exists $\beta \geq 0$ such that $\operatorname{Tr}(e^{-t|D|}) < \infty$ for $t > \beta$ and $\lim_{t \searrow \beta} \operatorname{Tr}(e^{-t|D|}) = \infty$. Then, for any extended limit $\omega \in \ell_{\infty}^*$ there exists an extended limit $\hat{\omega}_{D,\beta} \in \ell_{\infty}^*$ depending on D and β such that

$$\omega\left(\frac{\operatorname{Tr}(ae^{-(\beta+\frac{1}{n})|D|})}{\operatorname{Tr}(e^{-(\beta+\frac{1}{n})|D|})}\right) = \hat{\omega}_{D,\beta}\left(\frac{\operatorname{Tr}(P_{\lambda_n}aP_{\lambda_n})}{\operatorname{Tr}(P_{\lambda_n})}\right), \quad a \in B(\mathcal{H}).$$

Furthermore,

$$\lim_{\lambda \to \infty} \frac{\operatorname{Tr}(P_{\lambda} a P_{\lambda})}{\operatorname{Tr}(P_{\lambda})} = \lim_{t \to \beta} \frac{\operatorname{Tr}(a e^{-t|D|})}{\operatorname{Tr}(e^{-t|D|})}, \quad a \in B(\mathcal{H}),$$

in the sense that if the LHS limit exists, then the RHS limit exists and the equality holds.

Proof. Write $\{r_k\}_{k=0}^{\infty}$ for the eigenvalues of |D| counted without multiplicity so that $r_0 < r_1 < \cdots$. Observe the identity $r_k = \lambda_{N(r_k)}$ where $N(\lambda) := \#\{k : \lambda_k \leq \lambda\}$ is the spectral counting function of |D| and $\{\lambda_n\}_{n=0}^{\infty}$ are the eigenvalues of |D| counted with multiplicity. Then,

$$\begin{aligned} \frac{\mathrm{Tr}(ae^{-(\beta+\frac{1}{n})|D|})}{\mathrm{Tr}(e^{-(\beta+\frac{1}{n})|D|})} &= \frac{1}{\mathrm{Tr}(e^{-(\beta+\frac{1}{n})|D|})} \sum_{k=0}^{\infty} \left(\sum_{\lambda_j=r_k} \langle e_j, ae_j \rangle \right) e^{-(\beta+\frac{1}{n})r_k} \\ &= \frac{1}{\mathrm{Tr}(e^{-(\beta+\frac{1}{n})|D|})} \sum_{k=0}^{\infty} \left(\mathrm{Tr}(P_{r_k}) \frac{\mathrm{Tr}(P_{r_k}aP_{r_k})}{\mathrm{Tr}(P_{r_k})} - \mathrm{Tr}(P_{r_{k-1}}) \frac{\mathrm{Tr}(P_{r_{k-1}}aP_{r_{k-1}})}{\mathrm{Tr}(P_{r_{k-1}})} \right) e^{-(\beta+\frac{1}{n})r_k}. \end{aligned}$$

Hence if we define $\hat{\omega}_{D,\beta} \in (\ell_{\infty})^*$ by

$$\hat{\omega}_{D,\beta}(b_n) := \omega \left(\frac{1}{\mathrm{Tr}(e^{-(\beta + \frac{1}{n})|D|})} \sum_{k=0}^{\infty} \left(\mathrm{Tr}(P_{r_k}) b_{N(r_k)} - \mathrm{Tr}(P_{r_{k-1}}) b_{N(r_{k-1})} \right) e^{-(\beta + \frac{1}{n})r_k} \right), \quad b \in \ell_{\infty},$$

we have by construction that

$$\omega\left(\frac{\operatorname{Tr}(ae^{-(\beta+\frac{1}{n})|D|})}{\operatorname{Tr}(e^{-(\beta+\frac{1}{n})|D|})}\right) = \hat{\omega}_{D,\beta}\left(\frac{\operatorname{Tr}(P_{\lambda_n}aP_{\lambda_n})}{\operatorname{Tr}(P_{\lambda_n})}\right), \quad a \in B(\mathcal{H}).$$

Crucially, $\hat{\omega}_{D,\beta}$ is an extended limit if and only if $\lim_{t \searrow \beta} \operatorname{Tr}(e^{-t|D|}) = \infty$, see [Har49, Theorem III.2]. The second assertion of the proposition is proved through the continuous version of the cited theorem, namely [Har49, Theorem III.5].

Finally, if the limit

$$\lim_{\lambda \to \infty} \frac{\operatorname{Tr}(P_{\lambda} a P_{\lambda})}{\operatorname{Tr}(P_{\lambda})}$$

exists, then all extended limits on the sequence $\frac{\text{Tr}(P_{\lambda_n}aP_{\lambda_n})}{\text{Tr}(P_{\lambda_n})}$ coincide, and so we conclude

$$\lim_{\lambda \to \infty} \frac{\operatorname{Tr}(P_{\lambda} a P_{\lambda})}{\operatorname{Tr}(P_{\lambda})} = \lim_{t \to \beta} \frac{\operatorname{Tr}(a e^{-t|D|})}{\operatorname{Tr}(e^{-t|D|})}, \quad a \in B(\mathcal{H}).$$

Writing $P_D := \chi_{[0,\infty)}$ and applying the above results to $P_D D$ instead of D, we have that

$$\omega\left(\frac{\operatorname{Tr}(P_Dae^{-(\beta+\frac{1}{n})D})}{\operatorname{Tr}(P_De^{-(\beta+\frac{1}{n})D})}\right) = \hat{\omega}_{D,\beta}\left(\frac{\operatorname{Tr}(\chi_{[0,\lambda_n]}(D)a\chi_{[0,\lambda_n]}(D))}{\operatorname{Tr}(\chi_{[0,\lambda_n]}(D))}\right), \quad a \in B(\mathcal{H}),$$

which is a functional that is extensively studied in [GRU19]. In particular, it defines a KMS state of inverse temperature β on the Toeplitz algebra generated by a Li₁-summable spectral triple $(\mathcal{A}, \mathcal{H}, D)$ satisfying some extra conditions.

4.4 Noncommutative ergodicity

Quantum ergodicity began as a study of geodesic flow on manifolds through abstract operator theoretical language. On a closed Riemannian manifold (M, g) we can define the geodesic flow as a map $G_t : SM \to SM$, where SM is the unit sphere in the tangent bundle of the manifold M. For a point $(x, v) \in SM$, one simply takes the unique geodesic $\gamma : \mathbb{R} \to M$ with $\gamma(0) = x$ and $\gamma'(0) = v$, and defines $G_t(x, v) := (\gamma(t), \gamma'(t))$. This flow is said to be ergodic if every measurable function $f \in L_{\infty}(SM)$ which is fixed by the flow (i.e. $f \circ G_t = f$ almost everywhere) is constant almost everywhere. Equivalently, the geodesic flow can be defined on S^*M , the unit sphere in the cotangent bundle.

Let $\{e_k\}_{k=0}^{\infty}$ be any orthonormal basis of eigenvectors of the Laplace–Beltrami operator Δ_g and let $P_{\lambda} := \chi_{[0,\lambda]}(-\Delta_g)$. Related to the result derived in Section 4.1, it is known that for compact Riemannian manifolds we have that

$$\frac{\operatorname{Tr}(P_{\lambda}\operatorname{Op}(a)P_{\lambda})}{\operatorname{Tr}(P_{\lambda})} \to \int_{S^*M} a \, d\nu,$$

where $a \in C^{\infty}(S^*M)$ and Op(a) is a classical pseudodifferential operator with principal symbol a. Colin de Verdière showed [Col85] that this fact can be used, if M has ergodic geodesic flow, to show that there exists a density one subsequence $\{e_j\}_{j\in J}$ of $\{e_k\}_{k=0}^{\infty}$, meaning that $\frac{\#(J\cap\{0,\dots,n\})}{n+1} \to 1$, such that

$$\lim_{J\ni j\to\infty} \langle e_j, \operatorname{Op}(a)e_j \rangle = \int_{S^*M} a \, d\nu.$$

This and related properties are called quantum ergodicity of the operator Δ_g . See also Figure 4.1.

Before we start to put quantum ergodicity results into a noncommutative geometrical context, let us observe first that our labours in Section 4.1 provide a result in the other direction. The Weyl measure of an operator, which is the relevant measure for quantum ergodicity [CHT18, Section 4], clearly admits a Dixmier trace formula.

Definition 4.4.1. Let M be a manifold equipped with a nonvanishing density ρ , and let Δ be a self-adjoint positive operator on $L_2(M,\rho)$ with compact resolvent. Let $\{e_k\}_{k=0}^{\infty}$ be an orthonormal basis of $L_2(M,\rho)$ consisting of eigenvectors of Δ with corresponding eigenvalues $\{\lambda_k\}_{k=0}^{\infty}$. If

$$\lim_{\lambda \to \infty} \frac{1}{N(\lambda)} \sum_{\lambda_k \le \lambda} \langle e_k, M_f e_k \rangle$$

exists for all $f \in C_c(M)$, then there exists a measure μ_{Δ} such that

$$\lim_{\lambda \to \infty} \frac{1}{N(\lambda)} \sum_{\lambda_k \le \lambda} \langle e_k, M_f e_k \rangle = \int_M f \, d\mu_{\Delta}.$$

This measure is called the local Weyl measure of Δ .

Proposition 4.4.2. If Δ as in Definition 4.4.1 satisfies Weyl's law

$$\lambda(k,\Delta) \sim Ck^{\frac{2}{d}}$$

for some $0 < d \in \mathbb{R}$ and admits a local Weyl measure μ_{Δ} , then

$$\operatorname{Tr}_{\omega}(M_f(1+\Delta)^{-\frac{d}{2}}) = \operatorname{Tr}_{\omega}((1+\Delta)^{-\frac{d}{2}}) \int_M f \, d\mu_{\Delta}.$$
(4.5)

Proof. Consequence of Theorem 4.1.7.

This is relevant for sub-Riemannian manifolds, in which case one can take Δ to be the sub-Laplacian and μ_{Δ} is not necessarily the usual volume form on the manifold M. Notably, a rescaling of this measure was found very recently in [KSZ24b] to be a spectrally correct sub-Riemannian volume of M, additionally providing in that context a generalisation of the above Dixmier trace formula to any normalised continuous trace ϕ . This measure is studied extensively in this context in [CHT18] as well.

We will now shift our attention to results in quantum ergodicity which are interesting when viewed from the perspective of noncommutative geometry. To start, we provide an analogue of ergodicity of the geodesic flow – a property a compact Riemannian manifold can have, which we should therefore be able to see as a property of a spectral triple. For this purpose we recall the following construction and theorem by Connes [Con95]. For a spectral triple $(\mathcal{A}, \mathcal{H}, D)$ we write Ψ^0 for the set of operators admitting an asymptotic expansion

$$P = b_0 + b_{-1} \langle D \rangle^{-1} + b_{-2} \langle D \rangle^{-2} + \cdots, \quad b_j \in \mathcal{B},$$

with \mathcal{B} generated by \mathcal{A} and $\delta^n(\mathcal{A})$, where $\delta(a) := [|D|, a]$. The asymptotic expansion is taken in the sense of Chapters 2 and 3, which means here that the difference between Pand the *n*th partial summand is an element of $\operatorname{op}^{-n}(\langle D \rangle)$. Note that we do not include the operators $[D, \mathcal{A}]$ in \mathcal{B} . For $A \in B(\mathcal{H})$, we write $\sigma_t(A) := e^{it|D|}Ae^{-it|D|}$, $t \in \mathbb{R}$.

Theorem 4.4.3. For a unital regular spectral triple $(\mathcal{A}, \mathcal{H}, D)$, where regular means that $\delta^n(a) \in B(\mathcal{H})$ for all $a \in \mathcal{A}$, $n \in \mathbb{N}$, define

$$S^*\mathcal{A} := C^* \left(\bigcup_{t \in \mathbb{R}} \sigma_t (\Psi^0) + K(\mathcal{H}) \right) / K(\mathcal{H}).$$

This C^* -algebra comes equipped with automorphisms

$$\sigma_t(A + K(\mathcal{H})) := e^{it|D|} A e^{-it|D|} + K(\mathcal{H}).$$

For $(\mathcal{A}, \mathcal{H}, D) \simeq (C^{\infty}(M), L_2(S), D_M)$, the Dirac spectral triple associated with a compact Riemannian spin manifold, we have $S^*\mathcal{A} \simeq C(S^*M)$. Furthermore, if $A = Op^F(\sigma) \in \Psi^0$ for $\sigma \in C^{\infty}(S^*M)$ we have $\sigma_t(A + K(\mathcal{H})) = Op^F(\sigma \circ G_t) + K(\mathcal{H})$ where G_t is the geodesic flow on S^*M . The identity $\sigma_t(\operatorname{Op}^F(\sigma)) = \operatorname{Op}^F(\sigma \circ G_t) + K(\mathcal{H})$ is known as Egorov's theorem, see e.g. [Zel17, Section 9.2][Zwo12, Section 11.1]. This provides the basis for interpreting σ_t as an analogue of geodesic flow even in the noncommutative case.

In Theorem 4.4.3, it is important that the operators $[D, \mathcal{A}]$ are not included in Ψ^0 . For illustration, in the commutative case the principal symbol of |D| acts as a scalar on the vector bundle S, meaning that \mathcal{B} and hence Ψ^0 can be regarded as acting on $L_2(M)$ instead of $L_2(S)$. The isomorphism $S^*C^{\infty}(M) \cong C(S^*M)$ in Theorem 4.4.3 is then an extension of the symbol map $\sigma: \Psi^0_{cl} \to C^{\infty}(S^*M)$ on classical pseudodifferential operators on M.

Remark 4.4.4. For a (unital) spectral triple, $\langle D \rangle^{-1}$ is compact and hence

$$S^*\mathcal{A} = C^*\left(\bigcup_{t\in\mathbb{R}}\sigma_t(\mathcal{B}) + K(\mathcal{H})\right)/K(\mathcal{H}).$$

Furthermore, since for $b \in \mathcal{B}$ we have the convergence in norm

$$\lim_{t \to 0} \frac{\sigma_t(b) - b}{t} = [|D|, b],$$

we in fact have

$$S^*\mathcal{A} = C^*\left(\bigcup_{t\in\mathbb{R}}\sigma_t(\mathcal{A}) + K(\mathcal{H})\right)/K(\mathcal{H}).$$

A few examples of this construction are given in [GL98]. In the context of foliations of manifolds, it has been covered in [Kor05].

The automorphisms σ_t provide an action of \mathbb{R} on the C^* -algebra $S^*\mathcal{A}$, and this noncommutative cotangent sphere is thus an example of a C^* -dynamical system.

Definition 4.4.5. A C^{*}-dynamical system (\mathcal{A}, G, α) consists of a C^{*}-algebra \mathcal{A} , a locally compact group G, and a strongly continuous representation $\alpha : G \to \operatorname{Aut}(\mathcal{A})$.

There is a vast literature on C^* -dynamical systems, see [BR87, Section 2.7] for a start. In particular it has been a popular object of study in the field of quantum ergodicity, see e.g. [Zel96].

In Connes's point of view, the automorphisms σ_t on $S^*\mathcal{A}$ are the noncommutative analogue of geodesic flow. Recall that for compact manifolds this flow is said to be ergodic if the only measurable functions that are invariant almost everywhere under the geodesic flow are the functions that are constant almost everywhere. This definition is measure-theoretic in nature, and to translate it into a statement on spectral triples we therefore define the noncommutative L_2 -space on $S^*\mathcal{A}$, which corresponds with $L_2(S^*M)$ in the commutative case.

Proposition 4.4.6. Let $(\mathcal{A}, \mathcal{H}, D)$ be a unital regular spectral triple where D^2 satisfies Weyl's law (Definition 4.1.3). The functional

$$\tau(A + K(\mathcal{H})) := \frac{\operatorname{Tr}_{\omega}(A\langle D \rangle^{-d})}{\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})},$$

defines a finite positive trace on $S^*\mathcal{A}$.

Proof. This is well-known. We remark that the traciality of τ in fact follows from Theorem 4.1.7, Widom's Lemma 4.2.1, and the trivial identity

$$\operatorname{Tr}(P_{\lambda}AP_{\lambda}BP_{\lambda}) = \operatorname{Tr}(P_{\lambda}BP_{\lambda}AP_{\lambda}).$$

Definition 4.4.7. We define $L_2(S^*\mathcal{A})$ as the Hilbert space \mathcal{H}_{τ} in the GNS construction $(\pi_{\tau}, \mathcal{H}_{\tau})$. Explicitly, writing $I = \{A + K(\mathcal{H}) \in S^*\mathcal{A} : \tau(A^*A) = 0\}$, we define

$$L_2(S^*\mathcal{A}) := \overline{S^*\mathcal{A}/I}^{\|\cdot\|_{L_2}},$$

where the completion is taken in the semi-norm $||A + I||_{L_2} = (\tau(A^*A))^{\frac{1}{2}}$. The space $L_2(S^*A)$ is a Hilbert space with inner product defined via

$$\langle A+I, B+I \rangle_{L_2} := \tau(B^*A), \quad A, B \in S^*\mathcal{A}.$$

Notation 4.4.8. In accordance with the paradigm called the C^{*}-algebraic approach to the principal symbol [Cor79; SZ18; MSZ19; KSZ24a], we write sym for the quotient map

sym :
$$C^* \left(\bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A}) + K(\mathcal{H}) \right) \to S^* \mathcal{A}$$

 $A \mapsto A + K(\mathcal{H}),$

which is understood as a symbol map. Writing furthermore π for the other quotient mapping

$$\pi: S^* \mathcal{A} \to L_2(S^* \mathcal{A})$$
$$A \mapsto A + I,$$

where I is as in Definition 4.4.7, we will furthermore use the notation

$$\operatorname{sym}_{L_2} := \pi \circ \operatorname{sym} : C^* \left(\bigcup_{t \in \mathbb{R}} \sigma_t (\mathcal{A}) + K(\mathcal{H}) \right) \to L_2(S^*\mathcal{A}).$$

- **Example 4.4.9.** 1. For the Dirac spectral triple coming from a compact Riemannian spin manifold, $(C^{\infty}(M), L_2(S), D_M)$, we have that $S^*C^{\infty}(M) \simeq C(S^*M)$ with $\tau_{S^*\mathcal{A}} = \int_{S^*M}$. Hence $L_2(S^*C^{\infty}(M)) \simeq L_2(S^*M)$. The action σ_t agrees with the usual geodesic flow.
 - Given an even dimensional compact Riemannian manifold, and a finite dimensional spectral triple (A_F, H_F, D_F), we have for the almost commutative manifold (see Section 1.6)

$$(\mathcal{A} := C^{\infty}(M) \otimes \mathcal{A}_F, L_2(S) \otimes \mathcal{H}_F, D := D_M \otimes 1 + \gamma_M \otimes D_F),$$

that $S^*\mathcal{A} \simeq C(S^*M) \otimes \mathcal{A}_F$ with $\tau_{S^*\mathcal{A}} = \int_{S^*M} \otimes \operatorname{Tr.}$ Hence $L_2(S^*\mathcal{A}) \simeq L_2(S^*M) \otimes HS_F$, where HS_F is simply \mathcal{A}_F equipped with the Hilbert–Schmidt norm. The automorphisms σ_t act as $\sigma_t^M \otimes 1$, where σ_t^M is the usual geodesic flow on S^*M . This corrects [GL98, Lemma 2.2].

3. For the noncommutative torus $(C^{\infty}(\mathbb{T}^d_{\theta}), L_2(\mathbb{T}^d_{\theta}) \otimes \mathbb{C}^{N_d}, D)$ (see Section 1.6), we have that $S^*C(\mathbb{T}^d_{\theta}) \simeq C(\mathbb{T}^d_{\theta}) \otimes C(\mathbb{S}^{d-1})$ with $\tau_{S^*\mathbb{T}^d_{\theta}} = \tau_{\mathbb{T}^d_{\theta}} \otimes \int_{\mathbb{S}^{d-1}} and \otimes is$ the minimal C^* -tensor product. Hence $L_2(S^*\mathbb{T}^d_{\theta}) \simeq L_2(\mathbb{T}^d_{\theta}) \otimes L_2(\mathbb{S}^{d-1})$. The automorphisms σ_t act as

$$\sigma_t(u_k \otimes g) = u_k \otimes \exp(it \, k \cdot x)g, \quad t \in \mathbb{R}, k \in \mathbb{Z}^d, x \in \mathbb{S}^{d-1} \subseteq \mathbb{R}^d, g \in C(\mathbb{S}^{d-1}).$$

4. Let \mathcal{A} be the Toeplitz algebra, i.e. the C^{*}-algebra generated by the shift operator on $\ell_2(\mathbb{N})$, and let D be the operator on $\ell_2(\mathbb{N})$ defined on the standard basis $\{e_j\}_{j\in\mathbb{N}}$

$$D: e_j \mapsto j e_j, \quad j \in \mathbb{N}.$$

For the spectral triple $(\mathcal{A}, \ell_2(\mathbb{N}), D)$, we have $S^*\mathcal{A} \simeq C(\mathbb{S}^1)$ with $\tau_{S^*\mathcal{A}} = \int_{\mathbb{S}^1}$. Hence $L_2(S^*\mathcal{A}) \simeq L_2(\mathbb{S}^1)$. The automorphism σ_t is given by rotation.

Proof. (1) can be found in [Con95].

(2): Since $|D| = \sqrt{D_M^2 \otimes 1 + 1 \otimes D_F^2}$, it follows that $|D| - |D_M| \otimes 1$ is a compact operator on $L_2(S) \otimes \mathcal{H}_F$. We will show this with the double operator integrals from Chapter 3. First, one can omit the kernels of |D| and $|D_M| \otimes 1$ from the Hilbert space as the projection onto the kernel of either operator is finite-rank and thus compact. Both operators have compact resolvent. Hence, after this modification, the function $f(x) = \sqrt{x}$ is smooth on a neighbourhood of the spectra of the operators |D| and $|D_M| \otimes 1$. Define the Sobolev spaces $\mathcal{H}^s := \operatorname{dom} |D_M|^s \otimes 1$, $s \in \mathbb{R}$, and apply Proposition 3.4.1 and Theorem 3.3.3 to find that, for $0 < \varepsilon < 1$,

$$|D| - |D_M| \otimes 1 = T_{f^{[1]}}^{D^2, D_M^2 \otimes 1} (1 \otimes D_F^2) \in \mathrm{op}^{-1+\varepsilon}(|D_M| \otimes 1) \subseteq K(\mathcal{H}),$$

It now follows from Duhamel's formula (see e.g. [ACDS09, Lemma 5.2]) that

$$e^{it|D|} - e^{it(|D_M| \otimes 1)} = it \int_0^1 e^{ist|D|} (|D| - |D_M| \otimes 1) e^{i(1-s)t(|D_M| \otimes 1)} \, ds \in K(\mathcal{H}).$$

Therefore,

$$\bigcup_{t\in\mathbb{R}}\sigma_t(\mathcal{B})+K(\mathcal{H})=\bigcup_{t\in\mathbb{R}}\sigma_t^M(\mathcal{B}_M)\otimes\mathcal{A}_F+K(\mathcal{H}),$$

where σ_t^M is the geodesic flow on M and \mathcal{B}_M the algebra generated by $\delta_{|D_M|}^n(C^{\infty}(M)) \subseteq B(L_2(M))$. We conclude that $S^*\mathcal{A} \simeq C(S^*M) \otimes \mathcal{A}_F$. This proof shows that the action σ_t on $C(S^*M) \otimes \mathcal{A}_F$ is given by $\sigma_t = \sigma_t^M \otimes 1$.

(3): Although the Hilbert space of the spectral triple is $L_2(\mathbb{T}^d_{\theta}) \otimes \mathbb{C}^{N_d}$, since $|D| = \sqrt{-\Delta} \otimes 1$ (see Section 1.6) similarly to the manifold case \mathcal{B} acts trivially on the \mathbb{C}^{N_d} -component. We can therefore make the identification $\mathcal{B} \subseteq B(L_2(\mathbb{T}^d_{\theta}))$. In fact, we claim that $\overline{\mathcal{B}}^{\|\cdot\|}$ is a C^* -algebra stable under the action $\sigma_t(\cdot) = e^{it\sqrt{-\Delta}}(\cdot)e^{-it\sqrt{-\Delta}}$, and therefore

$$S^*C(\mathbb{T}^d_\theta) \simeq (\overline{\mathcal{B}}^{\|\cdot\|} + K(L_2(\mathbb{T}^d_\theta))) / K(L_2(\mathbb{T}^d_\theta)).$$

The claim holds since formally $\sigma_t(a) = \sum_{k=0}^{\infty} \frac{(it)^k}{k!} \delta^k(a)$, and this sum is actually norm convergent for $a \in \text{Poly}(\mathbb{T}^d_{\theta}) := \text{span}\{u_n\}_{n \in \mathbb{Z}^d}$. Denoting the generated *-algebra $\mathcal{B}_{poly} :=$ $\langle a, \delta^n(a) \rangle_{a \in \operatorname{Poly}(\mathbb{T}^d_a)}$ we therefore have

$$\sigma_t: \mathcal{B}_{poly} \to \overline{\mathcal{B}}^{\|\cdot\|}.$$

Since $\operatorname{Poly}(\mathbb{T}^d_{\theta})$ is dense in $C^{\infty}(\mathbb{T}^d_{\theta})$ and σ_t is an isometry on $B(L_2(\mathbb{T}^d_{\theta}))$, it is easily seen that this implies that σ_t maps $\overline{\mathcal{B}}^{\|\cdot\|}$ into itself.

By construction $C(\mathbb{T}^d_{\theta})$ is represented on $L_2(\mathbb{T}^d_{\theta})$ as bounded left-multiplication operators (denote the representation π_1), and $C(\mathbb{S}^{d-1})$ is as well via the representation

$$\pi_2(g) = g\Big(\frac{D_1}{\sqrt{-\Delta}}, \dots, \frac{D_d}{\sqrt{-\Delta}}\Big), \quad g \in C(\mathbb{S}^{d-1}),$$

where $D_i : u_k \mapsto k_i u_k$. It is shown in [MSZ19] that, writing $\Pi(C(\mathbb{T}^d_{\theta}), C(\mathbb{S}^{d-1}))$ for the C^* -algebra generated by $\pi_1(C(\mathbb{T}^d_{\theta}))$ and $\pi_2(C(\mathbb{S}^{d-1}))$ inside $B(L_2(\mathbb{T}^d_{\theta}))$, we have

$$\Pi(C(\mathbb{T}^d_\theta), C(\mathbb{S}^{d-1})) / K(L_2(\mathbb{T}^d_\theta)) \simeq C(\mathbb{T}^d_\theta) \otimes C(\mathbb{S}^{d-1}).$$

To determine that $S^*\mathbb{T}^d_\theta \simeq C(\mathbb{T}^d_\theta) \otimes C(\mathbb{S}^{d-1})$, it therefore suffices to show that

$$\overline{\mathcal{B}}^{\|\cdot\|} + K(L_2(\mathbb{T}^d_\theta)) = \Pi(C(\mathbb{T}^d_\theta), C(\mathbb{S}^{d-1})) + K(L_2(\mathbb{T}^d_\theta)) \subseteq B(L_2(\mathbb{T}^d_\theta)).$$

To start, it is immediately obvious that $\pi_1(C(\mathbb{T}^d_\theta)) \subseteq \overline{\mathcal{B}}^{\|\cdot\|}$. Next, the operators $\frac{D_i}{\sqrt{-\Delta}}$ generate $\pi_2(C(\mathbb{S}^{d-1}))$ as a C^* -algebra, and we claim that

$$u_{e_j}^*[\sqrt{-\Delta}, u_{e_j}] - \frac{D_j}{\sqrt{-\Delta}} \in K(L_2(\mathbb{T}_{\theta}^d)),$$

where $e_j \in \mathbb{Z}^d$ is the standard unit vector. This would imply

$$\Pi(C(\mathbb{T}^d_\theta), C(\mathbb{S}^{d-1})) + K(L_2(\mathbb{T}^d_\theta)) \subseteq \overline{\mathcal{B}}^{\|\cdot\|} + K(L_2(\mathbb{T}^d_\theta)).$$

The claim is proven by writing

$$\left(u_{e_j}^*\left[\sqrt{-\Delta}, u_{e_j}\right] - \frac{D_j}{\sqrt{-\Delta}}\right)u_k = \left(|k + e_j| - |k| - \frac{k_j}{|k|}\right)u_k.$$

Now define

$$f(t,k) := |k + te_j|, \quad k \in \mathbb{Z}^d, t \in \mathbb{R},$$

and note that its derivatives in the t variable are

$$f'(t,k) = \frac{k_j + t}{|k + te_j|}, \quad f''(t,k) = \frac{|k + te_j|^2 - (k_j + t)^2}{|k + te_j|^3}.$$

Hence

$$\left(u_{e_j}^* \left[\sqrt{-\Delta}, u_{e_j} \right] - \frac{D_j}{\sqrt{-\Delta}} \right) u_k = (f(1,k) - f(0,k) - f'(0,k)) u_k$$
$$= \int_0^1 (1-t) f''(t,k) \, dt \cdot u_k.$$

From the form of $f^{\prime\prime}(t,k)$ above, we therefore have

$$|f(1,k) - f(0,k) - f'(0,k)| \in c_0(\mathbb{Z}^d),$$

which indeed shows that $u_{e_j}^*[\sqrt{-\Delta}, u_{e_j}] - \frac{D_j}{\sqrt{-\Delta}}$ is a compact operator.

For the other direction, the above arguments already show that

$$[\sqrt{-\Delta}, u_{e_j}] \in \Pi(C(\mathbb{T}^d_\theta), C(\mathbb{S}^{d-1})) + K(L_2(\mathbb{T}^d_\theta)).$$

Since by explicit computation

$$u_{e_j}^* \delta_{\sqrt{-\Delta}}^n(u_{e_j}) = (u_{e_j}^* [\sqrt{-\Delta}, u_{e_j}])^n,$$

and since $u_{e_j}^m = u_{me_j}$, we have that

$$\delta_{\sqrt{-\Delta}}^n(u_k) \in \Pi(C(\mathbb{T}^d_\theta), C(\mathbb{S}^{d-1})) + K(L_2(\mathbb{T}^d_\theta)), \quad n \in \mathbb{Z}_{\geq 0}, k \in \mathbb{Z}^d,$$

and hence

$$\overline{\mathcal{B}}^{\|\cdot\|} \subseteq \Pi(C(\mathbb{T}^d_\theta), C(\mathbb{S}^{d-1})) + K(L_2(\mathbb{T}^d_\theta)).$$

In conclusion, $S^* \mathbb{T}^d_{\theta} \simeq C(\mathbb{T}^d_{\theta}) \otimes C(\mathbb{S}^{d-1})$. For the automorphism σ_t , first note that

$$\sigma_t(g(\frac{D_1}{\sqrt{-\Delta}},\ldots,\frac{D_d}{\sqrt{-\Delta}})) = g(\frac{D_1}{\sqrt{-\Delta}},\ldots,\frac{D_d}{\sqrt{-\Delta}}), \quad g \in C(\mathbb{S}^{d-1}).$$

Next,

$$e^{it\sqrt{-\Delta}}u_{e_j}e^{-it\sqrt{-\Delta}}u_k = e^{it(|k+e_j|-|k|)}u_{e_j}u_k$$

= $u_{e_j}\exp(itu_{e_j}^*[\sqrt{-\Delta}, u_{e_j}])u_k$
= $u_{e_j}\exp\left(it\frac{D_j}{\sqrt{-\Delta}}\right)u_k + u_{e_j}\left(\exp(itu_{e_j}^*[\sqrt{-\Delta}, u_{e_j}]) - \exp\left(it\frac{D_j}{\sqrt{-\Delta}}\right)\right)u_k.$

We have already seen that $u_{e_j}^*[\sqrt{-\Delta}, u_{e_j}] - \frac{D_j}{\sqrt{-\Delta}} \in K(L_2(\mathbb{T}^d_\theta))$, and hence as in the proof of (2) it follows from Duhamel's formula that

$$\exp(itu_{e_j}^*[\sqrt{-\Delta}, u_{e_j}]) - \exp\left(it\frac{D_j}{\sqrt{-\Delta}}\right) \in K(L_2(\mathbb{T}_{\theta}^d)).$$

Thus, we see that

$$\sigma_t(u_n \otimes g) = u_n \otimes \exp(it \, n \cdot x)g, \quad t \in \mathbb{R}, n \in \mathbb{Z}^d, x \in \mathbb{S}^{d-1} \subseteq \mathbb{R}^d, g \in C(\mathbb{S}^{d-1}).$$

(4): It is well-known that, after identifying $\ell_2(\mathbb{N})$ with the Hardy space H^2 , any element in the Toeplitz algebra \mathcal{A} can be written as $T_{\phi} + K$, where T_{ϕ} is the Toeplitz operator with symbol $\phi \in C(\mathbb{S}^1)$ and $K \in K(\ell_2(\mathbb{N}))$, see e.g. [Mur90, Section 3.5]. By an explicit computation, it can be seen that

$$e^{it|D|}T_{\phi}e^{-it|D|} = T_{\phi \circ R_t},$$

where R_t is rotation by the angle t. Hence $\sigma_t(\mathcal{A}) = \mathcal{A}$, and

$$S^*\mathcal{A} = \mathcal{A}/K(\ell_2(\mathbb{N})) \simeq C(\mathbb{S}^1).$$

For the noncommutative integral, we can use the diagonal formula in Theorem 4.1.7, so that for an arbitrary element $T_{\phi} + K \in \mathcal{A}$,

$$\operatorname{Tr}_{\omega}((T_{\phi}+K)\langle D\rangle^{-1}) = \omega \circ M(\langle e_k, (T_{\phi}+K)e_k\rangle) = \int_{\mathbb{S}^1} \phi(t) \, dt.$$

Observe that the automorphism σ_t on $S^*\mathcal{A}$ extends to a unitary operator $\sigma_t \in B(L_2(S^*\mathcal{A}))$.

Definition 4.4.10. We say that $(\mathcal{A}, \mathcal{H}, D)$ is classically ergodic if for $a \in L_2(S^*\mathcal{A})$, we have $\sigma_t(a) = a$ for all $t \in \mathbb{R}$ if and only if $a = \lambda \cdot 1 \in L_2(S^*\mathcal{A})$ for some $\lambda \in \mathbb{C}$.

The construction of $L_2(S^*\mathcal{A})$ has now reached its goal; for spectral triples derived from compact Riemannian manifolds, this definition is precisely the usual definition of ergodicity of the geodesic flow.

We now immediately claim the following theorem, the NCG analogue of the classic result in quantum ergodicity by Shnirelman, Zelditch, and Colin de Verdière [Shn74; Col85; Zel87]. **Theorem 4.4.11.** Let $(\mathcal{A}, \mathcal{H}, D)$ be a unital regular spectral triple with local Weyl laws for $a \in \mathcal{A}$ (Definition 4.1.3). Assume that the closure of \mathcal{A} in $B(\mathcal{H})$ is separable. If the triple is classically ergodic, then for every basis $\{e_n\}_{n=0}^{\infty}$ of eigenvectors of |D| there exists a density one subset $J \subseteq \mathbb{N}$ such that

$$\lim_{J\ni j\to\infty} \langle e_j, ae_j \rangle = \frac{\operatorname{Tr}_{\omega}(a\langle D \rangle^{-d})}{\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})}, \quad a \in \mathcal{A}.$$

Proof. Classical ergodicity of $(\mathcal{A}, \mathcal{H}, D)$ means precisely that the C*-dynamical system $(S^*\mathcal{A}, \mathbb{R}, \sigma_t)$ has a unique vacuum state in the terminology of [Zel96]. Hence, due to Proposition 4.1.1, the theorem is a consequence of [Zel96, Lemma 2.1].

This theorem, while its mathematical core is already an established result in quantum ergodicity, gives a fresh perspective on the criterion of a C^* -dynamical system having a 'unique vacuum state'. And while the vast majority of results in the paper [Zel96] are formulated for 'quantised abelian' C^* -dynamical systems, which in our case would mean $S^*\mathcal{A}$ is represented as a commutative algebra on $L_2(S^*\mathcal{A})$, the philosophy of noncommutative geometry provides solid reason to study not quantised abelian C^* -dynamical systems but ones with a unique vacuum state, as proposed by Zelditch [Zel96].

Example 4.4.12. We continue Example 4.4.9.

- 1. The canonical spectral triple corresponding to a compact Riemannian spin manifold, $(C^{\infty}(M), L_2(S), D_M)$ is classically ergodic if and only if M has ergodic geodesic flow.
- Any nontrivial almost commutative manifold (C[∞](M) ⊗ A_F, L₂(S) ⊗ H_F, D_M ⊗ 1 + γ_M ⊗ D_F) is not classically ergodic. Note that this corrects [Zel96, Corollary (3.1)], which was already known to experts to be false.
- 3. The noncommutative torus, like the commutative torus, is not classically ergodic.
- The spectral triple of the Toeplitz algebra is classically ergodic. See [Zel96, Example (D)] for a generalisation.

Proof. (1): See [Col85].

(2): Since σ_t acts on $L_2(S) \otimes HS_F$ by $\sigma_t^M \otimes 1$, any element of the form $1 \otimes a$ is a fixed point of σ_t .

(3): It follows from Example 4.4.9 that for any $f \in C(\mathbb{S}^{d-1})$, the element $\operatorname{sym}_{L_2}(1 \otimes f) \in L_2(S^*\mathbb{T}^d_{\theta})$ is a fixed point of σ_t .

(4): Since the only rotationally invariant functions in $L_2(\mathbb{S}^1)$ are the constant functions, the claim follows.

We note that the well-studied examples of spectral triples in noncommutative geometry often possess a high degree of symmetry, and in geometric examples a high degree of symmetry can obstruct ergodicity.

We now conclude the paper by giving some equivalent conditions for classical ergodicity. First, we invoke von Neumann's mean ergodic theorem.

Proposition 4.4.13. For any $a \in L_2(S^*\mathcal{A})$, there exists $a_{avg} \in L_2(S^*\mathcal{A})$ such that, putting

$$a_T := \frac{1}{T} \int_0^T \sigma_t(a) \, dt,$$

we have

$$\lim_{T \to \infty} \|a_T - a_{avg}\|_{L_2} \to 0.$$

Furthermore,

$$\langle 1, a_{avg} \rangle_{L_2} = \langle 1, a \rangle_{L_2},$$

and the map $a \mapsto a_{avg}$ is L₂-continuous.

Proof. The existence of a_{avg} and the L_2 -convergence of a_T to a_{avg} follows from von Neumann's mean ergodic theorem. Next, since σ_t is a unitary operator with $\sigma_t(1) = 1$, we have

$$\langle 1, \sigma_t(a) \rangle_{L_2} = \langle 1, a \rangle_{L_2}$$

Hence,

$$\begin{aligned} \left| \langle 1, a_{avg} \rangle_{L_2} - \langle 1, a \rangle_{L_2} \right| &\leq \left| \langle 1, a_{avg} \rangle_{L_2} - \langle 1, a_T \rangle_{L_2} \right| + \underbrace{\left| \langle 1, a_T \rangle_{L_2} - \langle 1, a \rangle_{L_2} \right|}_{=0} \\ &= \left| \langle 1, a_{avg} - a_T \rangle_{L_2} \right| \\ &\leq \|a_T - a_{avg}\|_{L_2}, \end{aligned}$$

which converges to 0 as $T \to \infty$ due to the first part.

Finally, for the continuity of $a \mapsto a_{avg}$, note that

$$||a_T||_{L_2} \le \frac{1}{T} \int_0^T ||\sigma_t(a)||_{L_2} dt = ||a||_{L_2},$$

and taking the limit $T \to \infty$,

$$\|a_{avg}\|_{L_2} \le \|a\|_{L_2}.$$

Remark 4.4.14. Since $S^*\mathcal{A}$ is represented as bounded operators on $L_2(S^*\mathcal{A})$, we can consider the von Neumann algebra $\pi_{\tau}(S^*\mathcal{A})''$ in $B(L_2(S^*\mathcal{A}))$, denoted as $L_{\infty}(S^*\mathcal{A})$, to which τ extends as a faithful normal tracial state. We can define the noncommutative L_p spaces $L_p(S^*\mathcal{A}) := L_p(\tau)$ for $1 \leq p \leq \infty$ via standard constructions (we recover $L_2(S^*\mathcal{A})$) for p = 2). This is precisely how the spaces $L_p(\mathbb{T}^d_{\theta})$ are constructed [LMSZ23, Section 3.5]. It is possible to show that $\sigma_t : L_p(S^*\mathcal{A}) \to L_p(S^*\mathcal{A})$ are isometries for all $1 \leq p \leq \infty$, and the averages in Proposition 4.4.13 exist and converge in every $L_p(S^*\mathcal{A})$.

Proposition 4.4.15. Given a unital regular spectral triple $(\mathcal{A}, \mathcal{H}, D)$ satisfying Weyl's law, the following are equivalent:

- 1. the spectral triple is classically ergodic;
- 2. for all $a \in L_2(S^*\mathcal{A})$,

$$a_{avg} = \langle 1, a \rangle_{L_2} \cdot 1;$$

3. writing

$$A_T := \frac{1}{T} \int_0^T \sigma_t(A) \, dt, \quad A \in \Big\langle \bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A}) \Big\rangle,$$

where $\left\langle \bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A}) \right\rangle$ is the *-algebra generated by $\bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A})$, for all $A \in \left\langle \bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A}) \right\rangle$ we have

$$\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})^{2} \cdot \lim_{T \to \infty} \omega \circ M\left(\langle e_{k}, |A_{T}|^{2} e_{k} \rangle\right) = \left| \operatorname{Tr}_{\omega}(A \langle D \rangle^{-d}) \right|^{2}.$$

Proof. (1) \Leftrightarrow (2) is easily seen from the fact that a_{avg} is a fixed point of σ_t . Next, if $A \in \left\langle \bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A}) \right\rangle$, then by Theorem 4.1.7 it follows that

$$\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})^{2} \cdot \omega \circ M\left(\langle e_{k}, |A_{T}|^{2} e_{k} \rangle\right) = \operatorname{Tr}_{\omega}(\langle D \rangle^{-d}) \operatorname{Tr}_{\omega}(|A_{T}|^{2} \langle D \rangle^{-d})$$
$$= \operatorname{Tr}_{\omega}(\langle D \rangle^{-d})^{2} \langle \operatorname{sym}_{L_{2}}(A_{T}), \operatorname{sym}_{L_{2}}(A_{T}) \rangle_{L_{2}}.$$

Since $\operatorname{sym}_{L_2}(A_T) = \operatorname{sym}_{L_2}(A)_T$, Proposition 4.4.13 gives that

$$\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})^{2} \cdot \lim_{T \to \infty} \omega \circ M\left(\langle e_{k}, |A_{T}|^{2} e_{k} \rangle\right) = \operatorname{Tr}_{\omega}(\langle D \rangle^{-d})^{2} \langle \operatorname{sym}_{L_{2}}(A)_{avg}, \operatorname{sym}_{L_{2}}(A)_{avg} \rangle_{L_{2}}.$$
(4.6)

(2) \Rightarrow (3): This now follows from Equation (4.6) and the identity $\langle 1, \operatorname{sym}_{L_2}(A) \rangle_{L_2} = \frac{\operatorname{Tr}_{\omega}(A\langle D \rangle^{-d})}{\operatorname{Tr}_{\omega}(\langle D \rangle^{-d})}.$

(3) \Rightarrow (2): For $a \in L_2(S^*\mathcal{A})$, Proposition 4.4.13 gives that $\langle a_{avg}, 1 \rangle_{L_2} = \langle a, 1 \rangle_{L_2}$, and hence

$$\begin{split} \|a_{avg} - \langle 1, a \rangle \cdot 1\|_{L_2}^2 &= \langle a_{avg}, a_{avg} \rangle_{L_2} - \langle a_{avg}, 1 \rangle_{L_2} \overline{\langle 1, a \rangle_{L_2}} - \langle 1, a_{avg} \rangle_{L_2} \langle 1, a \rangle_{L_2} + |\langle 1, a \rangle_{L_2}|^2 \\ &= \langle a_{avg}, a_{avg} \rangle_{L_2} - |\langle 1, a \rangle_{L_2}|^2. \end{split}$$

Therefore, assumption (3) combined with Equation (4.6) gives for all $A \in \left\langle \bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A}) \right\rangle$,

$$\operatorname{sym}_{L_2}(A)_{avg} = \langle 1, \operatorname{sym}_{L_2}(A) \rangle_{L_2} \cdot 1.$$

The image of $\left\langle \bigcup_{t \in \mathbb{R}} \sigma_t(\mathcal{A}) \right\rangle$ under the map sym_{L_2} being dense in $L_2(S^*\mathcal{A})$, and the map $a \to a_{avg}$ being L_2 -continuous, we can conclude that

$$a_{avg} = \langle 1, a \rangle_{L_2} \cdot 1$$

for all $a \in L_2(S^*\mathcal{A})$.

Chapter 5

The density of states on discrete spaces

Best wishes and remember: WORK HARDER!!!

Fedor Sukochev

This chapter is an adaptation of [Aza+22], joint work with Nurulla Azamov, Edward McDonald, Fedor Sukochev, and Dmitriy Zanin. The main result in this chapter is a Dixmier trace formula for the density of states on discrete metric spaces, Theorem 5.0.1.

Given a metric space (X, d) with a Borel measure, and a self-adjoint operator H on $L_2(X)$, a Dixmier trace formula for the density of states is an equality of two Borel measures on \mathbb{R} that can be associated with H. On the one hand, we can fix a weight $w : X \to \mathbb{C}$ such that $M_w \in \mathcal{L}_{1,\infty}$, which implies that $f(H)M_w \in \mathcal{L}_{1,\infty}$ for all $f \in C_c(\mathbb{R})$, inducing a measure

$$\operatorname{Tr}_{\omega}(f(H)M_w) = \int_{\mathbb{R}} f \, d\nu_1, \quad f \in C_c(\mathbb{R}),$$
(5.1)

where $\omega \in \ell_{\infty}^*$ is an extended limit. This measure ν_1 is guaranteed to exist due to the Riesz-Markov-Kakutani theorem, and a priori it might depend on our choice of ω . On

the other hand, we can look more geometrically towards the measure based on the limits

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(f(H) M_{\chi_{B(x_0, R)}}) = \int_{\mathbb{R}} f \, d\nu_2, \quad f \in C_c(\mathbb{R}),$$
(5.2)

where $x_0 \in X$ is some chosen basepoint and $|B(x_0, R)|$ is the measure of the closed ball $B(x_0, R)$. For the definition of this measure to make sense we must require that the volume of the balls $B(x_0, R)$ is finite for all R > 0 and that $f(H)M_{\chi_{B(x_0,R)}} \in \mathcal{L}_1$, but even then the limits in equation (5.2) are not guaranteed to exist. If the limits *do* converge, we say that H admits a density of states (DOS), and we commonly write ν_H for ν_2 .

In the paper [AMSZ20], it has been proven for the choice $X = \mathbb{R}^d$, $w(x) = (1 + |x|^2)^{-\frac{d}{2}}$, and $H = -\Delta + M_V$ with $V \in L_{\infty}(\mathbb{R}^d)$ real-valued, that the measures (5.1) and (5.2) are equal up to a constant depending on d (and independent of ω). Hence, the measure (5.1) in that setting is in fact *independent* of ω . It was explained in Section 1.5 that the case V = 0 is a Fourier transform of Connes' integral formula (Theorem 1.4.3), the general case is substantially more work.

In this chapter, we prove this Dixmier trace formula for countable discrete metric spaces. This comes with an additional condition on the metric space, a condition that ensures the volumes $|B(x_0, R)|$ grow in a regular and sub-exponential manner as $R \to \infty$. Namely, we will require that

$$\frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} \to 1, \quad k \to \infty,$$

where $\{r_k\}_{k=0}^{\infty}$ is the set $\{d(y, x_0) : y \in X\}$ (without multiplicities) ordered in increasing manner. We will call this Property (C). The main theorem of this chapter is then the following, the proof of which is specific to the discrete case and is based on recent advances in operator theory and the notion of V-modulated operators hatched in the theory of singular traces, see [KLPS13] or [LSZ21, Section 7.3] and Section 1.6.

Theorem 5.0.1. Let (X, d_X) be a countably infinite discrete metric space such that every ball contains at most finitely many points, and let $x_0 \in X$. Then the image of the map $d_X(\cdot, x_0) : X \to \mathbb{R}_{\geq 0}$ is a collection of isolated points which can be ordered in an increasing way, denote this by $\{r_k\}_{k \in \mathbb{N}} \subseteq \mathbb{R}$. Suppose that

$$\lim_{k \to \infty} \frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 1.$$
 (C)

Then for any positive, radially strictly decreasing function $w : X \to \mathbb{C}$ with $M_w \in \mathcal{L}_{1,\infty}$ we have that for every extended limit $\omega \in \ell_{\infty}^*$

$$\operatorname{Tr}_{\omega}(TM_w) = \operatorname{Tr}_{\omega}(M_w) \lim_{k \to \infty} \frac{\operatorname{Tr}(TM_{\chi_{B(x_0, r_k)}})}{|B(x_0, r_k)|},$$
(5.3)

for all bounded linear operators T on $\ell_2(X)$ for which the limit on the right-hand side exists.

In particular, if H is a self-adjoint, possibly unbounded, operator on $\ell_2(X)$ admitting a density of states measure ν_H , then for all extended limits ω ,

$$\operatorname{Tr}_{\omega}(f(H)M_w) = \int_{\mathbb{R}} f \, d\nu_H, \quad f \in C_c(\mathbb{R}).$$
(5.4)

Because $\{r_k\}_{k=0}^{\infty}$ denotes all possible distances $d_X(x_0, \cdot)$ in the discrete space X, the limit on the right-hand side of equation (5.3) exists if and only if the continuous limit

$$\lim_{R \to \infty} \frac{\operatorname{Tr}(TM_{\chi_{B(x_0,R)}})}{|B(x_0,R)|}$$

exists, and these limits are necessarily equal. This is therefore in line with the definition of the density of states (5.2).

In comparison with the Euclidean case, we do not have to make any assumptions on the operator T beyond the existence of the existence of the limit (5.3), whereas [AMSZ20] is specific to the case $T = f(-\Delta + M_V)$. Furthermore, [AMSZ20] has the requirement $d \geq 2$ on the dimension of the space \mathbb{R}^d , an analogue of which is not necessary in the discrete setting. Finally, Theorem 5.0.1 holds for any choice of radially symmetric strictly decreasing function $w : X \to \mathbb{C}$ such that $M_w \in \mathcal{L}_{1,\infty}$ (the latter can be written more briefly as $w \in \ell_{1,\infty}(X)$). In particular, we can always pick

$$w(x) := \frac{1}{1 + |B(x_0, d_X(x_0, x))|}, \quad x \in X,$$

for which $\operatorname{Tr}_{\omega}(M_w) = 1$, see Corollary 5.2.3. This provides that the constant of proportionality in (5.3), $\operatorname{Tr}_{\omega}(M_w)$, for this choice of weight w is independent of the choice of ω . Hence, for this choice of w, when T = f(H) for an operator H admitting a density of states, this ensures that the measure (5.1) is independent of the choice of ω as in the Euclidean case.

By definition, independence of the measure (5.1) with respect to ω is equivalent to the operators $f(H)M_w$ being Dixmier measurable. Let us conclude this introduction with an example illustrating that this is a strictly weaker requirement than H admitting a density of states.

Example 5.0.2. Let $X = \mathbb{N}$ with the usual metric d(x, y) = |x - y|, and take the basepoint $x_0 = 0$. On $\ell_2(\mathbb{N})$, a diagonal operator $H(e_n) = \lambda_n e_n$ admits a density of states if the limits

$$\lim_{n \to \infty} \frac{1}{|B(x_0, n)|} \operatorname{Tr}(f(H) M_{\chi_{B(x_0, n)}}) = \lim_{n \to \infty} \frac{1}{n+1} \sum_{k=0}^n f(\lambda_k), \quad f \in C_c(\mathbb{R}),$$

exist.

r

With the choice $w(n) := \frac{1}{n+1}$, we have that $M_w \in \mathcal{L}_{1,\infty}$. Then, for $f \in C(\mathbb{R})$, we have that $f(H)M_w \in \mathcal{L}_{1,\infty}$ is Dixmier measurable if and only if the limit

$$\lim_{n \to \infty} \frac{1}{\log(2+n)} \sum_{k=0}^{n} \lambda(k, f(H)M_w)$$

exists [LSZ21, Theorem 9.1.2(c)], where $\lambda(f(H)M_w)$ is an eigenvalue sequence of $f(H)M_w$ ordered in decreasing modulus. Since $f(H)M_w$ is M_w -modulated (see Section 1.6 or [KLPS13] and [LSZ21, Section 7.3]), we have through Theorem 1.6.4 that

$$\sum_{k=0}^{n} \lambda(k, f(H)M_w) = \sum_{k=0}^{n} \frac{f(\lambda_k)}{k+1} + O(1), \quad n \to \infty.$$

Hence, $f(H)M_w$ is Dixmier measurable if and only if the limit

$$\lim_{n \to \infty} \frac{1}{\log(n+2)} \sum_{k=0}^{n} \frac{f(\lambda_k)}{k+1},$$

exists.

With the choice

$$\lambda_n := \begin{cases} 0, & n \in [2^{2m} + 1, 2^{2m+1}], & m = 0, 1, 2, \dots, \\ 1, & n \in [2^{2m-1} + 1, 2^{2m}], & m = 1, 2, 3, \dots, \end{cases}$$
the diagonal operator H does not admit a density of states, but $f(H)M_w$ is Dixmier measurable for all $f \in C(\mathbb{R})$.

Proof. Observe that $\frac{1}{n+1} \sum_{k=0}^{n} \lambda_n$ does not converge as $n \to \infty$, since

$$\frac{1}{2^{2m}+1} \sum_{k=0}^{2^{2m}} \lambda_n = \frac{1}{2^{2m}+1} \sum_{k=0}^{2m} (-2)^k$$
$$= \frac{1}{2^{2m}+1} \left(\frac{1-(-2)^{2m+1}}{1+2}\right)$$
$$\to \frac{2}{3};$$
$$\frac{1}{2^{2m+1}+1} \sum_{k=0}^{2^{2m+1}} \lambda_n = \frac{1}{2^{2m+1}+1} \sum_{k=0}^{2m} (-2)^k$$
$$= \frac{1}{2^{2m+1}+1} \left(\frac{1-(-2)^{2m+1}}{1+2}\right)$$
$$\to \frac{1}{3},$$

where we used the closed-form formula for the sum of a geometric series. Hence H cannot admit a density of states.

Next, we show that $\frac{1}{\log(2+n)} \sum_{k=0}^{n} \frac{\lambda_k}{k+1}$ converges as $n \to \infty$. For convenience, we will prove the convergence of $\frac{1}{\log(n)} \sum_{k=1}^{n} \frac{\lambda_k}{k}$, taking n > 2, and we will use that the harmonic number $H(n) = \sum_{k=1}^{n} \frac{1}{k}$ has the expansion $H(n) = \log(n) + \gamma + \frac{1}{2n} + O\left(\frac{1}{n^2}\right)$, where γ is the Euler-Mascheroni constant [Knu97, Section 1.2.7]. For $m \ge 1$,

$$\begin{aligned} \frac{1}{\log(2^{2m})} \sum_{k=1}^{2^{2m}} \frac{\lambda_k}{k} &= \frac{1}{2m\log(2)} \left(1 + \sum_{k=1}^m H(2^{2k}) - H(2^{2k-1}) \right) \\ &= \frac{1}{2m\log(2)} \sum_{k=1}^m \log(2)(2k - (2k-1)) + \\ &+ \frac{1}{2m\log(2)} + \frac{1}{2m\log(2)} \sum_{k=1}^m \frac{1}{4k} - \frac{1}{4k-2} + O\left(\frac{1}{k^2}\right) \\ &\to \frac{1}{2}, \end{aligned}$$

where for the last term we used that for any sequence that converges to zero, its Cesàro

mean also converges to zero. Likewise,

$$\frac{1}{\log(2^{2m+1})} \sum_{k=1}^{2^{2m+1}} \frac{\lambda_k}{k} = \frac{1}{(2m+1)\log(2)} \left(1 + \sum_{k=1}^m H(2^{2k}) - H(2^{2k-1}) \right)$$
$$= \frac{m}{2m+1} + \frac{1}{(2m+1)\log(2)} + \frac{1}{(2m+1)\log(2)} \sum_{k=1}^m \frac{1}{4k} - \frac{1}{4k-2} + O\left(\frac{1}{k^2}\right)$$
$$\to \frac{1}{2}.$$

Note that $\frac{1}{\log(n)} \sum_{k=1}^{n} \frac{\lambda_k}{k}$ increases monotonically on $n \in [2^{2m-1}+1, 2^{2m}]$ and decreases monotonically on $n \in [2^{2m}+1, 2^{2m+1}]$. Therefore the above shows that $\frac{1}{\log(n)} \sum_{k=1}^{n} \frac{\lambda_k}{k} \rightarrow \frac{1}{2}$, and hence also $\frac{1}{\log(n)} \sum_{k=1}^{n} \frac{1-\lambda_k}{k} \rightarrow \frac{1}{2}$. Hence, for $f \in C(\mathbb{R})$,

$$\frac{1}{\log(2+n)}\sum_{k=0}^{n}\frac{f(\lambda_k)}{k}\to \frac{f(0)+f(1)}{2}, \quad n\to\infty,$$

which thus shows that $f(H)M_w$ is Dixmier measurable.

Do note that the above example is not a (discrete) Schrödinger type operator. It is unknown whether these two conditions still differ when restricting to such operators. This is also an open question for the Euclidean case, and in [AMSZ20] it was conjectured that these conditions are in fact equivalent for Schrödinger operators on Euclidean space.

The structure of the rest of the chapter is as follows. First we will discuss Property (C) in Section 5.1 and give examples of spaces where it is satisfied. In a way, we will return to the origin of the density of states, and show that crystals are included, the original source of the concept of a DOS (see [HBE87] and a very early use of the DOS in this context in 1929 by F. Bloch [Blo29]). Other concrete examples taken from physics will be considered too, demonstrating that condition (C) appearing in this theorem is very natural. Section 5.2 is the heart of the chapter where we prove the main result, Theorem 5.0.1. Finally, in Section 5.4 we apply the Dixmier trace formula for the DOS to provide a new proof of the equivariance under translations for the DOS on lattices.

5.1 Metric condition

The main theorem of this chapter is applicable to countably infinite discrete metric spaces such that every ball contains at most finitely many points and that also satisfy property (C) holds. Namely, we require that

$$\lim_{k \to \infty} \frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 1,$$
(C)

where $\{r_k\}_{k\in\mathbb{N}}$ is the increasing sequence created by ordering the set $\{d_X(x_0, y) : y \in X\}$ in increasing manner (which results in a sequence $r_k \to \infty$ since every ball in X contains at most finitely many points).

First, observe that property (C) is a condition on the so-called *crystal ball sequence* $\{|B(x_0, r_k)|\}_{k \in \mathbb{N}}$ of the metric space X [CS97], or alternatively after defining $S(x_0, r_k) := B(x_0, r_k) \setminus B(x_0, r_{k-1})$ it is a condition on the *coordination sequence* $\{|S(x_0, r_k)|\}_{k \in \mathbb{N}}$ [Bru79; CS97; OKe95].

To build some intuition, consider the following comment by J.E. Littlewood. Upon encountering the condition $\lim_{n\to\infty} \frac{\lambda_{n+1}}{\lambda_n} = 1$ he remarks [Lit11]: "[This condition is] satisfied when λ_n is any function of less order than $e^{\varepsilon n}$ for all values of ε , which increases in a regular manner. When, however, $\lambda_n > e^{\varepsilon n}$, the theorem breaks down altogether." This observation is apt, indeed our restriction on the metric space X is a strictly stronger assumption than sub-exponential growth of the sequence $\{|B(x_0, r_k)|\}_{k\in\mathbb{N}}$ (with respect to k), but exactly what kind of regular growth plus subexponential growth would imply condition (C) is difficult to pin down.

There is an equivalent description of property (C), the proof of which can be found in a recent paper by Cipriani and Sauvageot [CS23, Proposition 2.9]. The proposition they prove is slightly different, but the given proof is immediately applicable to the following.

Proposition 5.1.1. Let (X, d_X) be an infinite, discrete metric space such that each ball contains at most finitely many points, choose some point $x_0 \in X$ and order $\{d_X(x_0, y) :$

 $y \in X$ in increasing manner to define the sequence $\{r_k\}_{k \in \mathbb{N}}$. Then

$$\lim_{k \to \infty} \frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 1$$

if and only if

$$|B(x_0, r)| \sim \varphi(r)$$

for some continuous function $\varphi : \mathbb{R}_+ \to \mathbb{R}_+$.

Proof. The proof is exactly the same as in [CS23, Proposition 2.9] after replacing $N_L(x)$ by $|B(x_0, r)|$ and M_k by $|B(x_0, r_k)|$.

Remark 5.1.2. For the Cayley graph of the free group \mathbb{F}_2 we have $r_k = k$, $|B(x_0, k)| = 2^k$ and hence $|B(x_0, r)| = 2^{\lfloor r \rfloor}$, but

$$\frac{|B(x_0,r)|}{2^r} = 2^{\lfloor r \rfloor - r}$$

which does not converge as $r \to \infty$. This illustrates that $|B(x_0, r)| \sim \varphi(r)$ for some continuous function φ is a stronger assumption that one might expect.

In the same paper, another condition is given which is sufficient for property (C) to be satisfied [CS23, Proposition 2.8].

Proposition 5.1.3. Let X be a metric space as in Proposition 5.1.1. If $|B(x_0, r_k)| \sim f(k)$ (letting now $k \to \infty$ over the integers) for a function $f \in C^1(0, \infty)$ such that $\frac{f'(x)}{f(x)} \to 0$ as $x \to \infty$, then

$$\lim_{k \to \infty} \frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 1.$$

In particular, if $|B(x_0, r_k)|$ is a polynomial in k, then $\frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} \to 1$ as $k \to \infty$.

Proof. See [CS23, Proposition 2.8].

To be used later on, we also provide the following lemma.

Lemma 5.1.4. Let X be a metric space as in Proposition 5.1.1. If there exist constants C_1, C_2, d and R such that for $r_k > R$ we have

$$C_1 k^d < |S(x_0, r_k)| < C_2 k^d, (5.5)$$

then X has property (C).

Proof. If $C_1k^d < |S(x_0, r_k)| < C_2k^d$, we can deduce that also $|B(x_0, r_k)| \ge \frac{C_1}{d+1}k^{d+1} + O(k^d)$ for $r_k > R$, and therefore

$$\lim_{k \to \infty} \frac{|S(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 0,$$
(5.6)

which is equivalent with

$$\lim_{k \to \infty} \frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 1.$$

As a final general comment, when writing condition (C) in the manner of Equation (5.6), it resembles a type of Følner condition. In particular, it is reminiscent of work by Adachi and Sunada on the DOS on amenable groups where a closely related property is the subject of interest, namely Property (P) in [AS93, Proposition 1.1], also compare with [AS93, Lemma 3.2].

5.1.1 Solid matter

Consider any kind of rigid matter whose atoms or molecules are arranged in Euclidean space in such a way that it can be described by a tiling of that space. To be precise, we mean a tiling generated by only a finite selection of different tiles, with each type of tile having a fixed arrangement of atoms within (at least 1). Any crystal can be described in this a way using only one tile by considering its underlying Bravais lattice [AM76, Chapter 4], but the definition above includes quasicrystals [SBGC84; LS84]. For the approach of quasicrystals by tilings see for example [Hof95; Jar89; Nel86]. Specifically, [Hof93] establishes the existence of the integrated DOS, which is the existence of the function $\lambda \mapsto \nu_H(-\infty, \lambda)$, for every self-adjoint vertex-pattern-invariant operator on aperiodic selfsimilar tilings. If we define the set X of the metric space (X, d_X) as the atoms or molecules of the material and impose the induced Euclidean metric, then we claim that this space has property (C).

Proposition 5.1.5. Let X be a discrete subset of \mathbb{R}^d with the inherited Euclidean metric, such that X can be defined by a tiling as described above. Then (X, d_X) has property (C).

Proof. Without loss of generality, assume that the diameters of the tiles are all less than 1. Hence $r_{k+1} \in (r_k, r_k + 2]$, and therefore it suffices to show that

$$\frac{|B(x_0, r_k + 2) \setminus B(x_0, r_k)|}{|B(x_0, r_k)|} \xrightarrow{k \to \infty} 0.$$

Now, the number of vertices contained in $B(x_0, r_k + 2) \setminus B(x_0, r_k)$ is bounded from above by some constant times $(r_k)^{d-1}$: if the smallest tile has volume V, and each tile contains at most n atoms, and $\tilde{B}(x_0, r_k + 2)$ denotes the ball in \mathbb{R}^d , then there can be at most $n \frac{|\tilde{B}(x_0, r_k + 2) \setminus \tilde{B}(x_0, r_k)|}{V}$ vertices in $B(x_0, r_k + 2) \setminus B(x_0, r_k)$, which is bounded by $C_1(r_k)^{d-1}$.

If the volume of the biggest tile is W, the number of tiles that are *fully* contained in $B(x_0, r_k)$ is similarly bounded from below by $\frac{|\tilde{B}(x_0, r_k-1)|}{W}$ because we assumed that the diameter of the tiles is less than 1. Recall that we assumed that each tile contains at least one atom. Then the number of atoms in $B(x_0, r_k)$ can be bounded from below by $\frac{|\tilde{B}(x_0, r_k-1)|}{W}$, which is of the form $C_2(r_k)^d + O((r_k)^{d-1})$. Hence indeed

$$0 \le \frac{|B(x_0, r_k + 2) \setminus B(x_0, r_k)|}{|B(x_0, r_k)|} \le \frac{C_1(r_k)^{d-1}}{C_2(r_k)^d + O((r_k)^{d-1})} \xrightarrow{k \to \infty} 0,$$

and we see that this metric space satisfies condition (C).

5.1.2 Crystals

In another approach, we can take the atoms of crystals as the vertices of a graph and define our discrete metric space (X, d_X) as this graph with shortest-path metric (as a first observation, note that for any graph with the shortest-path metric, we have $r_k = k$). Common choices for such a construction are the contact graph [CS97] and the Voronoi graph [CS93, p. 33]. See for example the very recent paper [POO21] based on the model [OP05] which shows the existence of the DOS measure on \mathbb{Z} for a suitable family of Dirac operators, the very recent [PRCB21] or [Pas80] for an older result.

For crystals specifically, such a graph Γ comes with a free \mathbb{Z}^d action such that the quotient graph Γ/\mathbb{Z}^d is finite. This is a simple observation by considering the underlying Bravais lattice of any crystal [AM76, Chapter 4]. A recent advancement by Y. Nakamura, R. Sakamoto, T. Masea and J. Nakagawa [NSMN21] concerns exactly such (even possibly directed) graphs Γ with a free \mathbb{Z}^d action such that Γ/\mathbb{Z}^d is finite. Namely, these authors have proven that the coordination sequence $\{|S(x_0,k)|\}_{k\in\mathbb{N}}$ is then of quasi-polynomial type, by which they mean the following. A quasi-polynomial is defined as a function $p: \mathbb{N} \to \mathbb{Z}$ with $p(k) = c_m(k)k^m + c_{m-1}k^{m-1} + \cdots + c_0(k)$ where $c_i(k)$ are all periodic with an integral period. Equivalently, it is a function such that for some integer N

$$p(k) = \begin{cases} p_0(k) & k \equiv 0 \mod N \\ p_1(k) & k \equiv 1 \mod N \\ \vdots \\ p_{N-1}(k) & k \equiv N-1 \mod N \end{cases}$$

where p_0, \ldots, p_{N-1} are polynomials. A function of quasi-polynomial type is then defined as a function $f : \mathbb{N} \to \mathbb{Z}$ such that there exists some quasi-polynomial p and an integer M such that f(n) = p(n) for all $n \ge M$. See also [Sta86, Section 4.4].

Proposition 5.1.6. Let Γ be a graph with a free \mathbb{Z}^d action such that the quotient graph Γ/\mathbb{Z}^d is finite. Then (Γ, d_{Γ}) , where we take d_{Γ} as the shortest-path metric, has property (C).

Proof. By [NSMN21, Theorem 1.1] we have that $|S(x_0, k)|$ is a quasi-polynomial, denote this quasi-polynomial by p(k) and its constituent polynomials by p_1, \ldots, p_{N-1} . Suppose that the polynomial p_r is one with maximal degree, i.e. $\deg(p_i) \leq \deg(p_r) = t$ for all $i = 0, \ldots, N-1$. Then there exist some constants C_1, C_2 and L such that $p(k) \leq C_1 k^t$ for all $k \geq L$, and also $p_r(k) = C_2 k^t + O(k^{t-1})$. Now take $k \in \mathbb{Z}_{\geq 0}$ arbitrary, and define $a \in \mathbb{Z}_{\geq 0}$ as the smallest positive integer such that k = mN + r + a for some $m \in \mathbb{Z}$. It follows that $a \in \{0, \dots, N-1\}$, and hence note that $mN = k - r - a \ge k - 2N$, i.e. $\frac{k}{N} - 2 \le m \le \frac{k}{N}$.

$$\sum_{n=1}^{k} p(n) \ge \sum_{n=0}^{m} p_r(nN+r)$$

= $\sum_{n=0}^{m} C_2(nN+r)^t + O((m+1)(nN+r)^{t-1})$
= $C_2 \sum_{n=0}^{m} (nN)^d + O(k^t)$
= $C_3 k^{t+1} + O(k^t).$

Therefore we have that

$$0 \leq \frac{p(k+1)}{\sum_{n=1}^{k} p(n)} \leq \frac{C_1(k+1)^t}{C_3 k^{t+1} + O(k^t)} \xrightarrow{k \to \infty} 0,$$

i.e. $\frac{|S(x_0,k+1)|}{|B(x_0,k)|}$ converges to zero as $k \to \infty$.

5.1.3 The integer lattice

From the previous two subsections it follows that respectively $(\mathbb{Z}^d, \|\cdot\|_2)$ and $(\mathbb{Z}^d, \|\cdot\|_1)$ have property (C), where we define

$$\|v\|_p := \left(\sum_{i=1}^d |v_i|^p\right)^{1/p}$$

for $1 \leq p < \infty$ and for $p = \infty$

$$\|v\|_{\infty}:=\sup_{i=1,\ldots,d}|v_i|\,.$$

Even though the base space of these two metric spaces is the same, the difference between these is the domain undergoing the thermodynamic limit in the definition of the DOS, which can make a difference as demonstrated in [AMSZ22]. As mentioned, the existence of the DOS in the case \mathbb{Z} has been established for a suitable family of Dirac operators [OP05]. Another example is that, using $(\mathbb{Z}^d, \|\cdot\|_{\infty})$ as a model, the existence of a surface DOS was established for a quantum model with a surface [EKSS88].

In fact, $(\mathbb{Z}^d, \|\cdot\|_p)$ has property (C) for all $1 \leq p \leq \infty$. This is simply because in the metric space $(\mathbb{Z}^d, \|\cdot\|_p)$ we have $|B(0,r)| = V_p(d)r^d + O(r^{d-1})$ where $V_p(d)$ denotes the volume of the ℓ_p unit ball in \mathbb{R}^d , and Proposition 5.1.1 then implies condition (C).

5.1.4 Quasicrystals

Analogously to Subsection 5.1.2 one can consider graphs constructed from quasicrystals, but these will be aperiodic by definition [Jar89; Nel86]. For some investigations of the DOS on quasicrystals, see [Hof93; Hof95].

On a case-by-case basis, there are some aperiodic tilings for which condition (C) can be expected to hold on their vertex graph. Firstly, a Penrose tiling. This proposition is entirely based on recent results by A. Shutov and A. Maleev [SM15; SM18].

Proposition 5.1.7. Consider a 2D Penrose tiling in the construction of [SM15], which is a Penrose tiling with five-fold symmetry with respect to a chosen origin 0. Consider the graph induced by this tiling (i.e. with the same vertices and edges as the tiles) and take this graph with the shortest-path metric as the definition of the metric space (X, d_X) . Then this metric space has property (C).

Proof. For this particular tiling, Shutov and Maleev showed [SM18] that

$$|S(0,k)| = C(n)n + o(n)$$

where C(n), denoting $\tau = (1 + \sqrt{5})/2$, takes a value between $10\tau^{-2} \approx 3.8$ and $10\tau^{-2} + (5/2)\tau^{-1} \approx 5.4$ depending on *n*. By Lemma 5.1.4, we can then conclude that the vertex graph of this Penrose tiling satisfies condition (C).

Remark 5.1.8. Similar asymptotic behaviour can be expected when taking a different base point, or different Penrose tilings, but this is still an open problem. Numerical data supports this conjecture for Penrose tilings [BG06], as well as that this metric condition will be satisfied by aperiodic tilings like the Ammann-Beenker tiling [BG06], a certain class of quasi-periodic self-similar tilings [SMZ10] and two-dimensional quasiperiodic Ito-Ohtsuki tilings [SM08]. To this author's knowledge, no quasiperiodic tiling is known that can serve as a counter-example to property (C).

5.1.5 Percolation

A successful model for studying conduction properties of a crystal with impurities via the density of states is percolation (in this context also called quantum percolation) [AO82; Cha+86; KM06; Ves05; YN87], see also [Gri99, Section 13.2] for a general approach. In broadest generality, percolation describes the study of a statistical procedure on a graph, which means adding, removing or otherwise manipulating edges or vertices based on some probabilistic method [Gri99, Chapter 1].

Let us focus on so-called bond percolation on the graph \mathbb{Z}^d . Choosing some chance $0 \leq p \leq 1$, we declare each edge on the graph \mathbb{Z}^d to be *open* with probability p (in an independent manner), and closed with probability 1 - p. It is a well-known fact that there exists a phase transition at a critical probability $p_c(d)$ [Gri99, Chapter 1]. Namely, for $p > p_c(d)$ there exists almost surely a unique infinite cluster of vertices connected by open edges, while for $p < p_c(d)$ almost surely all clusters of vertices that are connected by open edges are finite.

In the super-critical region, meaning $p > p_c(d)$, one can wonder if our metric condition holds on the infinite cluster, meaning that we take (X, d_X) as the infinite cluster with induced shortest-path metric. This turns out to be true, which follows quite directly from a result by R. Cerf and M. Théret [CT16].

Proposition 5.1.9. Let X be the (almost surely unique) infinite cluster on \mathbb{Z}^d after supercritical bond percolation (with $p > p_c(d)$), and let d_X be the shortest path metric on X. Then (X, d_X) has property (C).

Proof. We first set up the exact situation as in [CT16]. Denote the set of edges in our graph \mathbb{Z}^d by \mathbb{E}^d , and consider a family of i.i.d. random variables $(t(e), e \in \mathbb{E}^d)$ taking values in $[0, \infty]$ (including ∞), with common distribution F. To be very precise, for each

variable t(e) we take $[0, \infty]$ as sample space, we define a σ -algebra by declaring $A \subseteq [0, \infty]$ measurable if $A \setminus \{\infty\}$ is Lebesgue measurable in \mathbb{R} , and $F : [0, \infty] \to [0, 1]$ is a measurable function that defines the distribution of the variable t(e).

These variables can be interpreted as the time it takes to travel along the corresponding edge. Then consider the random extended metric on \mathbb{Z}^d by defining for $x, y \in \mathbb{Z}^d$

$$T(x,y) = \inf\{\sum_{e \in \gamma} t(e) : \gamma \text{ is a path from } x \text{ to } y\}$$

Note that if the distribution F is such that $F(\{1\}) = p$, $F(\{\infty\}) = 1 - p$, T(x, y) is always either a positive integer or infinite and defines precisely the usual induced shortest-path metric on \mathbb{Z}^d after bond percolating on \mathbb{Z}^d with chance p. Also observe that generally, if $F([0,\infty)) > p_c(d)$, there exists almost surely a unique infinite connected cluster of vertices [CT16].

Now define $B(x_0,t) := \{y \in \mathbb{Z}^d : T(x_0,y) \leq t\}$. Then by [CT16, Theorem 5(ii)], if $F([0,\infty)) > p_c(d), F(\{0\}) < p_c(d)$ and x_0 is a vertex on the infinite cluster, as $t \to \infty$ we have almost surely

$$\frac{|B(x_0,t)|}{t^d} \to C$$

for some constant $C \in \mathbb{R}$. Returning to the distribution $F(\{1\}) = p$, $F(\{\infty\}) = 1 - p$, we can then conclude that

$$\frac{|B(x_0, k+1)|}{|B(x_0, k)|} \frac{k^d}{(k+1)^d} \to 1$$

as $k \to \infty$, and hence

$$\frac{|B(x_0, k+1)|}{|B(x_0, k)|} \to 1.$$

Remark 5.1.10. Site percolation is similar to bond percolation, but as the name suggests we would assign each vertex to be open with probability p and closed with probability 1 - p [Gri99, Section 1.6]. Arguably, this is a more physically suitable model of an alloy. The argument used above to prove the cited result for bond percolation can be used for site percolation as well, but this author is not aware of an explicit demonstration.

5.2 Dixmier trace formula for the DOS

In this section we prove Theorem 5.0.1 based on a series of lemmas.

To start off, we will demonstrate the existence of a function w such that M_w has positive Dixmier trace as required by Theorem 5.0.1. For that purpose we first need a lemma generalising the fact that $\sum_{k=1}^{n} \frac{1}{k} \sim \log(n)$.

Lemma 5.2.1. Let $\{a_k\}_{k\in\mathbb{N}}$ be a divergent, increasing sequence of strictly positive real numbers, such that $\lim_{k\to\infty} \frac{a_{k+1}}{a_k} = 1$. We use the convention that $a_{-1} = 0$. Then

$$\sum_{j=0}^{k} \frac{a_j - a_{j-1}}{a_j} \sim \log(a_k), \quad k \to \infty.$$

Proof. Note first that $\sum_{j=0}^{k} \frac{a_j - a_{j-1}}{a_j}$ equals 1 plus a lower Riemann sum of the integral $\int_{a_0}^{a_k} \frac{dx}{x}$. Hence,

$$\sum_{j=0}^{k} \frac{a_j - a_{j-1}}{a_j} \le \log(a_k) + 1 - \log(a_0).$$

Next, since $\lim_{k\to\infty} \frac{a_{k+1}}{a_k} = 1$, for $\varepsilon > 0$ we can choose K such that for k > K we have $\frac{a_k}{a_{k-1}} < 1 + \varepsilon$, which can be rearranged to $\frac{1}{a_{k-1}} - \frac{1}{a_k} < \frac{\varepsilon}{a_k}$. Note that $\log(a_k) - \log(a_0) \le \sum_{j=1}^k \frac{a_j - a_{j-1}}{a_{j-1}}$ since it is an upper Riemann sum of the integral $\int_{a_0}^{a_k} \frac{dx}{x}$. Therefore, for k > K,

$$\log(a_k) - \log(a_0) \le \sum_{j=1}^k \frac{a_j - a_{j-1}}{a_{j-1}}$$

= $\sum_{j=1}^k \frac{a_j - a_{j-1}}{a_j} + \sum_{j=1}^K (a_j - a_{j-1}) \left(\frac{1}{a_{j-1}} - \frac{1}{a_j}\right)$
+ $\sum_{j=K+1}^k (a_j - a_{j-1}) \left(\frac{1}{a_{j-1}} - \frac{1}{a_j}\right)$
< $(1 + \varepsilon) \sum_{j=1}^k \frac{a_j - a_{j-1}}{a_j} + \sum_{j=1}^K (a_j - a_{j-1}) \left(\frac{1}{a_{j-1}} - \frac{1}{a_j}\right)$

Dividing by $\log(a_k)$ (assuming without loss of generality that $a_k \neq 1$) and taking the lim inf, we get

$$1 \le (1+\varepsilon) \liminf_{k \to \infty} \frac{1}{\log(a_k)} \sum_{j=1}^k \frac{a_j - a_{j-1}}{a_j}.$$

Combined with our earlier estimate, we can conclude that

$$\sum_{j=0}^{k} \frac{a_j - a_{j-1}}{a_j} \sim \log(a_k), \quad k \to \infty.$$

We will now make use of Lemma 4.1.6, which originally appeared in a slightly weaker form in the paper this chapter is based on [Aza+22, Lemma 4.8]. For convenience, let us restate it here.

Lemma 5.2.2. Let $\phi : \mathbb{N} \to \mathbb{R}_{>0}$ be an increasing function such that $\phi(n) \to \infty$ as $n \to \infty$, let $\{a_k\}_{k \in \mathbb{N}} \subseteq \mathbb{R}$ be a sequence such that $\left\{\frac{1}{\phi(n)}\sum_{k=0}^{n}|a_k|\right\}_{n=0}^{\infty}$ is bounded, and let $\{k_0, k_1, \ldots\}$ be an infinite, increasing sequence of positive integers such that

$$\lim_{n \to \infty} \frac{\phi(k_{n+1})}{\phi(k_n)} = 1,$$

and

$$\frac{1}{\phi(k_n)} \sum_{k=k_{n-1}+1}^{k_n} |a_k| = o(1), \quad n \to \infty.$$

Labeling $k_{i_n} := \min\{k_i : k_i \ge n\}$, we have that

$$\frac{1}{\phi(n)}\sum_{k=0}^{n}a_{k} = \frac{1}{\phi(k_{i_{n}})}\sum_{k=0}^{k_{i_{n}}}a_{k} + o(1), \quad n \to \infty.$$

Corollary 5.2.3. Let (X, d_X) be a countably infinite discrete metric space such that every ball contains at most finitely many points, satisfying property (C). Let $x_0 \in X$, and define $w(x) := (1 + |B(x_0, d_X(x, x_0))|)^{-1}$. Then $w \in \ell_{1,\infty(X)}$, M_w is Dixmier measurable, and $\operatorname{Tr}_{\omega}(M_w) = 1$ for all extended limits $\omega \in \ell_{\infty}^*$.

Proof. Applying Lemma 5.2.1 to the sequence $a_k = 1 + |B(x_0, r_k)|$, we obtain

$$\lim_{k \to \infty} \frac{1}{\log(2 + |B(x_0, r_k)|)} \sum_{j=1}^k |\partial B(x_0, r_j)| (1 + |B(x_0, r_j)|)^{-1} = 1.$$

By Lemma 5.2.2, it follows that

$$\lim_{k \to \infty} \frac{1}{\log(2+k)} \sum_{j=1}^{k} \mu(k, M_w) = 1.$$

We now present a modified Toeplitz lemma, which follows from much the same proof as in Shiryaev [Shi96, Chapter IV, §3, Lemma 1].

Lemma 5.2.4 (Toeplitz lemma). Let $\{c_n\}_{n=0}^{\infty}$ be a sequence of non-negative numbers, and let $\{z_n\}_{n=0}^{\infty}$ be a sequence of complex numbers such that $z_n \to L \in \mathbb{C}$. If $d_n = \sum_{k=0}^n c_k$ diverges, then:

$$\sum_{k=0}^{n} c_k z_k = Ld_n + o(d_n)$$

as $n \to \infty$.

Proof. Let $\varepsilon > 0$ and take K > 0 sufficiently large such that if k > K then $|z_k - L| < \varepsilon$. For any n > K, rewriting the left hand side of the equality above as

$$\sum_{k=0}^{n} c_k z_k = \sum_{k=0}^{n} c_k L + \sum_{k=0}^{K} c_k (z_k - L) + \sum_{k=K+1}^{n} c_k (z_k - L),$$

we see that

$$\left|\frac{1}{d_n}\sum_{k=0}^n c_k(z_k-L)\right| \le \frac{1}{d_n}\sum_{k=0}^K c_k|z_k-L| +\varepsilon.$$

Since $d_n \to \infty$ as $n \to \infty$, it follows that:

$$\limsup_{n \to \infty} \left| \frac{1}{d_n} \sum_{k=0}^n c_k (z_k - L) \right| \le \varepsilon.$$

Since ε is arbitrary, we have:

$$\sum_{k=0}^{n} c_k(z_k - L) = o(d_n)$$

and this completes the proof.

Lemma 5.2.5. Let $\{x_k\}_{k=0}^{\infty}$ and $L \in \mathbb{C}$ be such that as $n \to \infty$,

$$\sum_{k=0}^{n} x_k = Ln + o(n).$$

Let $\{a_n\}_{n=0}^{\infty}$ be a sequence of non-negative numbers such that:

- 1. $\{a_n\}_{n=0}^{\infty}$ is non-increasing;
- 2. $\sup_{k\geq 0} ka_k < \infty$, that is, the sequence is in $\ell_{1,\infty}$;

3. $b_n := \sum_{k=0}^n a_k$ diverges.

Then

$$\sum_{k=0}^{n} a_k x_k = Lb_n + o(b_n)$$

as $n \to \infty$.

Proof. Let $y_n = \sum_{k=0}^{n-1} x_k$ with $y_0 = 0$. Abel's summation formula gives

$$\sum_{k=0}^{n} a_k x_k = \sum_{k=0}^{n} a_k (y_{k+1} - y_k)$$

= $y_{n+1} a_n - \sum_{k=1}^{n} (a_k - a_{k-1}) y_k$
= $y_{n+1} a_n + \sum_{k=1}^{n} \frac{y_k}{k} \cdot k(a_{k-1} - a_k).$

By assumption (1) we have $a_{k-1} \ge a_k$ so the sequence $c_k := k(a_{k-1} - a_k)$ is non-negative, and moreover as $n \to \infty$,

$$\sum_{k=1}^{n} c_k = b_{n-1} - na_n \to \infty,$$

since by assumption (3) $b_n \to \infty$ and by assumption (2) na_n is bounded. Therefore Lemma 5.2.4 applies to c_k and $z_k := \frac{y_k}{k}$, since by assumption $\lim_{k\to\infty} \frac{y_k}{k} = L$, and hence it follows that

$$\sum_{k=1}^{n} \frac{y_k}{k} \cdot k(a_{k-1} - a_k) = L(b_{n-1} - na_n) + o(b_{n-1} - na_n)$$

as $n \to \infty$. Thus,

$$\sum_{k=0}^{n} a_k x_k = \frac{y_{n+1}}{n} \cdot na_n + L(b_{n-1} - na_n) + o(b_{n-1} - na_n)$$
$$= Lb_{n-1} + o(b_{n-1} - na_n),$$

where in the last equality we have absorbed the vanishing term $\left(\frac{y_{n+1}}{n} - L\right) na_n$ into $o(b_{n-1} - na_n)$. Since by assumption (2) the sequence $\{na_n\}_{n=1}^{\infty}$ is bounded, it follows that:

$$\sum_{k=0}^{n} a_k x_k = Lb_n + o(b_n).$$

We could also write the result of Lemma 5.2.5 as:

$$\lim_{n \to \infty} \frac{\sum_{k=0}^{n} x_k}{\sum_{k=0}^{n} 1} = \lim_{n \to \infty} \frac{\sum_{k=0}^{n} a_k x_k}{\sum_{k=0}^{n} a_k}$$

whenever the left hand side exists and $\{a_k\}_{k=0}^{\infty}$ satisfies the stated assumptions.

Our next aim is to prove Lemma 5.2.6, which is the crux of the proof of Theorem 5.0.1. The proof is based on the notion of a V-modulated operator from [KLPS13] or [LSZ21, Section 7.3], see also Section 1.6. Recall that, since $0 < W \in \mathcal{L}_{1,\infty}$, we have that TW is W-modulated for all $T \in B(\mathcal{H})$. Applying Theorem 1.6.4, we have as $n \to \infty$,

$$\sum_{k=0}^{n} \lambda(k, TW) = \sum_{k=0}^{n} \langle e_k, Te_k \rangle \mu(k, W) + O(1).$$
 (5.7)

Lemma 5.2.6. Let $0 < W \in \mathcal{L}_{1,\infty}(\mathcal{H}) \setminus \mathcal{L}_1(\mathcal{H})$, and let $\{e_k\}_{k=0}^{\infty}$ be an orthonormal basis such that $We_k = \mu(k, W)e_k$ for all $k \ge 0$. If T is a bounded operator such that

$$\sum_{k=0}^{n} \langle e_k, Te_k \rangle = Ln + o(n), \quad n \to \infty$$

where $L \in \mathbb{C}$ then:

$$\sum_{k=0}^{n} \lambda(k, TW) = L\left(\sum_{k=0}^{n} \mu(k, W)\right) + o\left(\sum_{k=0}^{n} \mu(k, W)\right).$$
(5.8)

Moreover,

$$\operatorname{Tr}_{\omega}(TW) = \operatorname{Tr}_{\omega}(W)L$$

for all extended limits ω .

Proof. By the assumption $0 < W \in \mathcal{L}_{1,\infty} \setminus \mathcal{L}_1$, Lemma 5.2.5 applies with $a_k = \mu(k, W)$ and $x_k = \langle e_k, Te_k \rangle$, so

$$\sum_{k=0}^{n} \langle e_k, Te_k \rangle \mu(k, W) = L\left(\sum_{k=0}^{n} \mu(k, W)\right) + o\left(\sum_{k=0}^{n} \mu(k, W)\right), \ n \to \infty.$$

Thus (5.7) yields the first equality (5.8).

To obtain the result concerning Dixmier traces, we divide both sides of (5.8) by $\log(n+2)$ to get:

$$\frac{1}{\log(n+2)} \sum_{k=0}^{n} \lambda(k, TW) = L\left(\frac{1}{\log(n+2)} \sum_{k=0}^{n} \mu(k, W)\right) + o(1).$$

Thus if ω is an extended limit,

$$\omega\left(\left\{\frac{1}{\log(n+2)}\sum_{k=0}^{n}\lambda(k,TW)\right\}_{n=0}^{\infty}\right) = L\operatorname{Tr}_{\omega}(W).$$

The left hand side is exactly the Dixmier trace $Tr_{\omega}(TW)$.

Remark 5.2.7. The result of Lemma 5.2.6 can be written in a different way. We could say that:

$$\operatorname{Tr}_{\omega}(TW) = \operatorname{Tr}_{\omega}(W) \lim_{n \to \infty} \frac{1}{n+1} \sum_{k=0}^{n} \langle e_k, Te_k \rangle$$
(5.9)

whenever the right hand side exists.

Now Lemma 5.2.2 combined with (5.9) immediately implies the following proposition.

Proposition 5.2.8. If T is a bounded linear operator and $0 < W \in \mathcal{L}_{1,\infty}$ such that

$$\lim_{n \to \infty} \frac{|\{k \ge 0 : \mu(k, W) \ge \varepsilon_{n+1}\}|}{|\{k \ge 0 : \mu(k, W) \ge \varepsilon_n\}|} = 1.$$

for some decreasing, strictly positive sequence $\varepsilon_n \to 0$, then for all extended limits ω we have

$$\operatorname{Tr}_{\omega}(TW) = \operatorname{Tr}_{\omega}(W) \lim_{n \to \infty} \frac{\operatorname{Tr}(T\chi_{[\varepsilon_n, \infty)}(W))}{\operatorname{Tr}(\chi_{[\varepsilon_n, \infty)}(W))},$$
(5.10)

whenever the limit on the right hand side exists.

This proposition gives us the density of states formula for the discrete case. The idea is that in $\ell_2(X)$ we consider the basis $\{\delta_v\}_{v\in X}$ of indicator functions of points $p \in X$, and Wis an operator of pointwise multiplication by a function which is *radially decreasing* with respect to some point $x_0 \in X$, so that $\chi_{[\varepsilon_n,\infty)}(W)$ is the indicator function of a ball, and the limit $n \to \infty$ is equivalent to taking a limit over balls with radius going to infinity. This is where property (C) as discussed in Section 5.1 comes in, as this ensures that the premise of Proposition 5.2.8 is satisfied.

Proof of Theorem 5.0.1. Let $W = M_w$, where $w : X \to \mathbb{C}$ is a positive, radially strictly decreasing function with $M_w \in \mathcal{L}_{1,\infty}$. If W is trace-class, the theorem is trivial as both sides are zero. So, assume that $0 < W \in \mathcal{L}_{1,\infty}(\mathcal{H}) \setminus \mathcal{L}_1(\mathcal{H})$. Then $\{\delta_v\}_{v \in X}$ is a basis of normalised eigenvectors for W, with eigenvalue corresponding to δ_v equal to w(v). By assumption w(v) is a strictly decreasing function of $d_X(x_0, v)$, and hence the sets $\{v : w(v) \geq \delta\}$ are balls. In fact, if we define $\varepsilon_n = \tilde{\mu}(n, W)$, where $\tilde{\mu}(n, W)$ is the *n*th largest singular value of W counted *without* multiplicities, we have that

$$|\{k \ge 0 : \mu(k, W) \ge \varepsilon_n\}| = |\{v : w(v) \ge \varepsilon_n\}| = |B(x_0, r_n)|,$$

and we have assumed that

$$\lim_{n \to \infty} \frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 1.$$

Hence due to Lemma 5.2.6 and Lemma 5.2.2, we can then conclude that

$$\operatorname{Tr}_{\omega}(TW) = \operatorname{Tr}_{\omega}(W) \lim_{k \to \infty} \frac{1}{|B(x_0, r_k)|} \sum_{v \in B(x_0, r_k)} \langle \delta_v, T\delta_v \rangle$$

since we have assumed that the limit on the right exists. Observing that the sum on the right side is equal to $\text{Tr}(TM_{\chi_{B(0,r_k)}})$ gives (5.3).

Remark 5.2.9. If we again denote $\tilde{\mu}(n, M_w)$ as the nth largest singular value of M_w counted without multiplicities, define m_n as the multiplicity corresponding to this singular value and also define $M_n := \sum_{k=1}^n m_k$, then we have effectively imposed

$$\lim_{n \to \infty} \frac{M_{n+1}}{M_n} = 1.$$

Now compare this to the paper by Cipriani and Sauvageot [CS23] also referenced in Section 5.1. They study densely defined, nonnegative, unbounded, self-adjoint operators with exactly such a property for the multiplicities of its eigenvalues, and our $(M_w)^{-1}$ would fit such a description. The link between the DOS as considered in this chapter and the spectral weight those authors define remains unclear as of yet.

5.3 Problems of measurability

This section is mostly the work of Edward McDonald. Going back to the Toeplitz lemma 5.2.4, it is possible to get better behaviour of the convergence $\frac{1}{b_n} \sum_{k=0}^n a_k x_k \to L$ by assuming faster convergence of $x_k \to L$. For example, we have the following lemma with an obvious proof.

Lemma 5.3.1. Let a_n and b_n satisfy the same assumptions as Lemma 5.2.4. Let $x_n \in \mathbb{C}$, $n = 0, 1, \ldots$ If $x_k \to L$ sufficiently fast such that $\{a_k | x_k - L | \}_{k=0}^{\infty} \in \ell_1$, then:

$$\sum_{k=0}^{n} a_k x_k = Lb_n + O(1).$$

Lemma 5.3.2. Let a_n and b_n satisfy the same assumptions as Lemma 5.2.5. Let $x_n \in \mathbb{C}$, n = 0, 1, ... and $\sigma_n = \frac{1}{n+1} \sum_{k=0}^n x_k$. If $\sigma_n \to L$ sufficiently fast such that $\{a_k | \sigma_k - L|\}_{k=0}^{\infty} \in \ell_1$, then:

$$\sum_{k=0}^{n} a_k x_k = L b_n + O(1).$$

Proof. Let $y_n = \sum_{k=0}^{n-1} x_k$ with $y_0 = 0$. From the proof of Lemma 5.2.5 we have:

$$\sum_{k=0}^{n} a_k x_k = y_{n+1} a_n + \sum_{k=1}^{n} \sigma_k \cdot k(a_{k-1} - a_k),$$

the sequence $a'_k := k(a_{k-1} - a_k)$ is non-negative, and as $n \to \infty$,

$$\sum_{k=1}^{n} a'_k = b_{n-1} - na_n \to \infty,$$

Therefore, since by assumption $\lim_{k\to\infty} \sigma_k = L$ and $\{a_k | \sigma_k - L|\}_{k=0}^{\infty} \in \ell_1$, Lemma 5.3.1 applies with a' in place of a and σ_k in place of x_k , which gives

$$\sum_{k=1}^{n} \sigma_k \cdot k(a_{k-1} - a_k) = L(b_{n-1} - na_n) + O(1)$$

as $n \to \infty$. Thus,

$$\sum_{k=0}^{n} a_k x_k = \frac{y_{n+1}}{n} \cdot na_n + L(b_{n-1} - na_n) + O(1)$$
$$= Lb_{n-1} + O(1) = Lb_n + O(1).$$

For an operator $A \in \mathcal{L}_{1,\infty}$, there are various criteria relating the behaviour of the sequence $\sum_{k=0}^{n} \lambda(k, A)$ to the measurability of A. For example, [LSZ21, Theorem 5.1.5] implies that

$$\sum_{k=0}^{n} \lambda(k, A) - c \log(2+n) = O(1), \quad n \to \infty$$
 (5.11)

if and only if $\varphi(A) = c$ for all normalised traces φ on $\mathcal{L}_{1,\infty}$ (c.f. [LSZ21, Theorem 9.1.2]). For different classes of traces, different criteria are available, see [SSUZ15; Usa13]. **Theorem 5.3.3.** Let (X, d_X) , T and w satisfy the assumptions of Theorem 5.0.1. Let e_n be an orthonormal basis such that $M_w e_k = \mu(k, w) e_k$, where $\{\mu(k, w)\}_{k=0}^{\infty}$ is the decreasing rearrangement of w. If

$$\frac{1}{n+1}\sum_{k=0}^n \langle e_k, Te_k\rangle \to L \in \mathbb{C}$$

so fast that

$$\sum_{n=0}^{\infty} \mu(n,w) \left| \frac{1}{n+1} \sum_{k=0}^{n} \langle e_k, Te_k \rangle - L \right| < \infty$$

and there exists C > 0 such that:

$$\sum_{k=0}^{n} \mu(k, w) = C \log(2+n) + O(1),$$

then TM_w is measurable in the sense of Connes, specifically

$$\varphi(TM_w) = \operatorname{Tr}_{\omega}(M_w) \lim_{n \to \infty} \frac{1}{n+1} \sum_{k=0}^n \langle e_k, Te_k \rangle$$

for all traces φ on $\mathcal{L}_{1,\infty}$.

Proof. Since TM_w is M_w -modulated (see Definition 1.6.3 and Theorem 1.6.4), from (5.7) we have

$$\sum_{k=0}^{n} \lambda(k, TM_w) = \sum_{k=0}^{n} \langle e_k, Te_k \rangle \, \mu(k, w) + O(1).$$

Hence, in view of (5.11) to prove the claim it suffices to show that

$$\sum_{k=0}^{n} \langle e_k, Te_k \rangle \, \mu(k, w) = CL \log(2+n) + O(1).$$

By the second condition this is equivalent to

$$\sum_{k=0}^{n} \langle e_k, Te_k \rangle \, \mu(k, w) = L \sum_{k=0}^{n} \mu(k, w) + O(1),$$

so it suffices to prove this. This follows from Lemma 5.3.2 applied to $x_k = \langle e_k, Te_k \rangle$ and $a_k = \mu(k, w)$.

We could also replace the assumption with the slightly stronger assertion:

$$\left\{\frac{1}{n+1}\sum_{k=0}^{n}\langle e_k, Te_k\rangle - L\right\}_{k=0}^{\infty} \in \Lambda_{\log},$$

where Λ_{\log} is the space of sequences x such that

$$\sum_{k=0}^{\infty} \frac{\mu(k,x)}{k+1} < \infty.$$

5.4 Equivariance of the DOS under translations of the Hamiltonian

In this section, mostly the work of Edward McDonald, we will provide a straightforward application of the Dixmier formula for the DOS put forward in Theorem 5.0.1. Namely, we provide a new and original proof of the equivariance of the DOS on lattice graphs Xunder translations of the Hamiltionian. By this we mean that if U denotes a shift operator on $\ell_2(X)$ and the DOS exists for both a Hamiltonian H and the shifted UHU^* , then the DOS is equal for H and UHU^* . This fact is not hard to prove without Theorem 5.0.1, but it does provide a different perspective on the claim.

Afterwards, we will discuss some consequences of this translation equivariance.

5.4.1 Translation equivariance on lattice graphs

We will consider the example where $X = \mathbb{Z}^d$, embedded as a subset of \mathbb{R}^d with the Euclidean metric. Precisely the same reasoning applies to other lattices $X \subset \mathbb{R}^d$ (recall that a lattice in \mathbb{R}^d is the Z-linear span of d linearly independent vectors).

We take

$$w(x) := (1 + \|x\|_2)^{-d}$$
(5.12)

where $x \in \mathbb{Z}^d$ like before.

Lemma 5.4.1. For all $n \in \mathbb{Z}^d$, we have:

$$\{w(x) - w(x-n)\}_{x \in \mathbb{Z}^d} \in \ell_{\frac{d}{d+1},\infty}.$$

Proof. The difference w(x) - w(x - n) is provided by the formula:

$$w(x) - w(x - n) = \int_0^1 \langle \nabla w(x - (1 - \theta)n), n \rangle \, d\theta.$$

The gradient ∇w of w is easily computed as:

$$\frac{\partial}{\partial x_j} w(x) = -\frac{dx_j}{\|x\|_2 (1 + \|x\|_2)^{d+1}}.$$

Thus,

$$\|\nabla w(x)\|_2 = \frac{d}{(1+\|x\|_2)^{d+1}}$$

Therefore for x with $||x||_2 > ||n||$ we have

$$\begin{split} |w(x) - w(x - n)| &\leq d \, \|n\|_2 \max_{0 \leq \theta \leq 1} (1 + \|x - (1 - \theta)n)\|_2)^{-d - 1} \\ &\leq \frac{d \, \|n\|_2}{(1 + \|x\|_2 - \|n\|_2)^{d + 1}}, \end{split}$$

hence $|w(x) - w(x - n)|_{x \in \mathbb{Z}^d}$ is an element of $\ell_{\frac{d}{d+1},\infty}$.

Theorem 5.4.2. For $n \in \mathbb{Z}^d$, let U_n denote the operator on $\ell_2(\mathbb{Z}^d)$ of translation by n. Assume that $H = H_0 + M_V$ is a Hamiltonian operator such that the density of states exists for both H and $U_n H U_n^*$. Then both measures are equal.

Proof. For any $f \in C_c(\mathbb{R})$, we have $f(U_n H U_n^*) = U_n f(H) U_n^*$. Combining this with the tracial property of the Dixmier trace gives

$$\operatorname{Tr}_{\omega}(f(U_n H U_n^*) M_w) = \operatorname{Tr}_{\omega}(f(H) U_n^* M_w U_n).$$

By Lemma 5.4.1, we have $U_n^* M_w U_n - M_w \in \mathcal{L}_{\frac{d}{d+1},\infty} \subset \mathcal{L}_1$. Since the Dixmier trace vanishes on trace-class operators, it follows that $\operatorname{Tr}_{\omega}(f(U_n H U_n^*)M_w) = \operatorname{Tr}_{\omega}(f(H)M_w)$. Combining this with the Dixmier trace formula for the density of states, Theorem 5.0.1, completes the proof.

Via identical reasoning, we also have the following abstract assertion:

Theorem 5.4.3. Let (X, d_X) be an infinite discrete metric space and w be a function such that these satisfy the assumptions of Theorem 5.0.1, let γ be an isometry of X, and let $U_{\gamma}\delta_p = \delta_{\gamma(p)}$ be the corresponding unitary operator on $\ell_2(X)$. Assume that $w - w \circ \gamma \in$ $(\ell_{1,\infty})_0(X)$. Then

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \sum_{d_X(p, x_0) \le R} \langle \delta_p, T \delta_p \rangle = \lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \sum_{d_X(p, x_0) \le R} \langle \delta_p, U_\gamma^* T U_\gamma \delta_p \rangle,$$

provided both limits exist.

5.4.2 Ergodic operators

The following results are direct consequences of the translation equivariance of the DOS measure and therefore could be derived without help of Theorem 5.0.1. However, the Dixmier trace formula provides a different approach.

Let $(\Omega, \Sigma, \mathbb{P})$ be a probability space, and let Γ be a discrete amenable group of isometries of the metric space X from Theorem 5.0.1. We assume that there is a representation of Γ as automorphisms of Ω :

$$\gamma \in \Gamma \mapsto \alpha_{\gamma} \in \operatorname{Aut}(\Omega).$$

It is assumed that the action α is ergodic, in the sense that:

- 1. For every $\gamma \in \Gamma$, the automorphism α_{γ} is measure preserving;
- 2. If $E \subseteq \Omega$ is invariant under every α_{γ} , then $\mathbb{P}(E) = 0$ or $\mathbb{P}(\Omega \setminus E) = 0$.

We will use a generalisation of Birkhoff's ergodic theorem, obtained by Lindenstrauss [Lin01, Theorem 1.3]. This uses the concept of a Følner sequence, we give the definition as it is used for discrete groups.

Definition 5.4.4. Let Γ be a discrete group, and let $\{F_n\}_{n=0}^{\infty}$ be a sequence of subsets of Γ .

 If for every finite subset K ⊆ Γ and every δ > 0, there exists N sufficiently large such that if n > N, we have for all k ∈ K

$$|F_n \Delta k F_n| \le \delta |F_n|,$$

then $\{F_n\}_{n=0}^{\infty}$ is called a Følner sequence.

2. If $\{F_n\}_{n=0}^{\infty}$ satisfies (1) and furthermore for some $C \ge 1$ and for every $n \ge 0$, we have:

$$\left| \bigcup_{k \le n} F_k^{-1} F_{n+1} \right| \le C |F_{n+1}|,$$

then $\{F_n\}_{n=0}^{\infty}$ is called a tempered Følner sequence.

The existence of a Følner sequence in this sense is equivalent with the condition of Γ being discrete and amenable [Lub94, p. 23]. Also note that any Følner sequence has a tempered subsequence [Lin01, Proposition 1.4].

Lindenstrauss' pointwise ergodic theorem [Lin01, Theorem 1.3] implies that if $\{F_n\}_{n=0}^{\infty}$ is a tempered Følner sequence, then for all $f \in L_1(\Omega)$ we have:

$$\lim_{n \to \infty} \frac{1}{|F_n|} \sum_{\gamma \in F_n} f(\alpha_{\gamma} \omega) = \mathbb{E}(f).$$
(5.13)

for almost every $\omega \in \Omega$.

For $\gamma \in \Gamma$, let U_{γ} denote the induced unitary operator acting on $\ell_2(X)$ by:

$$U_{\gamma}\delta_p := \delta_{\gamma(p)}, \quad p \in X, \, \gamma \in \Gamma.$$

We will consider strongly measurable random operators $T \in L_1(\Omega, B(\ell_2(X)))$ which are compatible with α in the sense that:

$$U_{\gamma}T(\omega)U_{\gamma}^{*} = T(\alpha_{\gamma}\omega), \quad \gamma \in \Gamma$$
(5.14)

for almost all $\omega \in \Omega$.

Proposition 5.4.5. Let $T \in L_1(\Omega, B(\ell_2(X)))$ be a random operator satisfying (5.14) with respect to a group of isometries Γ of X, which admits a tempered Følner sequence $\{F_n\}_{n=0}^{\infty}$ of finite subsets, and with respect to an ergodic action α of Γ on Ω . If there exists a function $w: X \to \mathbb{R}_+$ satisfying the assumptions of Theorem 5.0.1 such that

$$w \circ \gamma - w \in (\ell_{1,\infty})_0(X)$$

for every $\gamma \in \Gamma$ then the density of states of $T(\xi)$ is non-random, in the sense that if the limit:

$$\lim_{R \to \infty} \frac{\operatorname{Tr}(T(\xi)M_{\chi_{B(x_0,R)}})}{|B(x_0,R)|}$$

exists for almost every ξ , then the limit is almost surely constant in ξ .

Proof. This is an application of the Lindenstrauss version of Birkhoff's ergodic theorem. The assumption on w and Theorem 5.4.3 imply that:

$$\operatorname{Tr}_{\omega}(T(\xi)M_w) = \operatorname{Tr}_{\omega}(T(\alpha_{\gamma}\xi)M_w), \quad \gamma \in \Gamma.$$
(5.15)

Therefore for every $n \ge 0$ we have:

$$\operatorname{Tr}_{\omega}(T(\xi)M_w) = \frac{1}{|F_n|} \sum_{\gamma \in F_n} \operatorname{Tr}_{\omega}(T(\alpha_{\gamma}\xi)M_w).$$

Note that:

$$|\operatorname{Tr}_{\omega}(T(\xi)M_w)| \le ||T(\xi)||_{\infty} ||w||_{1,\infty}.$$

Hence the function $\xi \mapsto \operatorname{Tr}_{\omega}(T(\xi)M_w)$ is integrable, due to our assumption that $T \in L_1(\Omega, B(\ell_2(X)))$, and the measurability of $\xi \mapsto \operatorname{Tr}_{\omega}(T(\xi)M_w)$ follows from the strong measurability of $\xi \mapsto T(\xi)$ and the norm continuity of $T \mapsto \operatorname{Tr}_{\omega}(TM_w)$. Hence, Lindenstrauss' ergodic theorem (5.13) applies to this function, and hence for almost every $\xi \in \Omega$ we have:

$$\lim_{n \to \infty} \frac{1}{|F_n|} \sum_{\gamma \in F_n} \operatorname{Tr}_{\omega}(T(\alpha_{\gamma}\xi)M_w) = \mathbb{E}(\operatorname{Tr}_{\omega}(TM_w)).$$

The right hand side has no dependence on $\xi \in \Omega$, and hence the limit is almost surely constant in ξ . Due to (5.15), this implies that $\operatorname{Tr}_{\omega}(T(\xi)M_w)$ is almost surely constant in ξ . Alluding to Theorem 5.0.1, we conclude that the density of states of $T(\xi)$ is almost surely constant in ξ .

In an alternative direction of inquiry, the condition (5.14) can be used in some circumstances to imply the existence of the density of states. For simplicity, we state the following condition for $X = \mathbb{Z}^d$.

Theorem 5.4.6. Let $T \in L_1(\Omega, B(\ell_2(\mathbb{Z}^d)))$ be a linear operator which satisfies (5.14) with respect to the action of \mathbb{Z}^d on itself by translations and an ergodic action α of \mathbb{Z}^d on Ω . Then for almost every $\xi \in \Omega$ there exists the limit:

$$\lim_{R \to \infty} \frac{1}{|B(0,R)|} \sum_{|n| \le R} \langle \delta_n, T(\xi) \delta_n \rangle = \mathbb{E}(\langle \delta_0, T \delta_0 \rangle).$$

Proof. We have that $U_n \delta_0 = \delta_n$, and therefore:

$$\langle \delta_n, T(\xi) \delta_n \rangle = \langle \delta_0, U_n^* T(\xi) U_n \delta_0 \rangle = \langle \delta_0, T(\alpha_{-n} \xi) \delta_0 \rangle.$$

It follows that:

$$\frac{1}{|B(0,R)|} \sum_{|n| \le R} \langle \delta_n, T(\xi) \delta_n \rangle = \frac{1}{|B(0,R)|} \sum_{|n| \le R} \langle \delta_0, T(\alpha_n \xi) \delta_0 \rangle.$$

By our assumption on T, the function $\xi \mapsto \langle \delta_0, T(\xi) \delta_0 \rangle$ belongs to $L_1(\Omega)$. Note that the sequence $F_N := B(0, N)$ is a tempered Følner sequence in \mathbb{Z}^d , and hence Lindenstrauss' ergodic theorem (5.13) implies that for almost every $\xi \in \Omega$ there exists the limit

$$\lim_{N \to \infty} \frac{1}{|B(0,N)|} \sum_{n \in B(0,N)} \langle \delta_0, T(\alpha_n \xi) \delta_0 \rangle = \mathbb{E}(\langle \delta_0, T(\xi) \delta_0 \rangle).$$

Note that the result also holds if the limit over balls $\{B(0,N)\}_{N\geq 0}$ is replaced with any other tempered Følner sequence, such as cubes $\{[-N,N]^d\}_{N\geq 0}$. The limit in every case is $\mathbb{E}(\langle \delta_0, T(\xi)\delta_0 \rangle)$, and hence does not depend on the choice of sequence of sets.

Theorem 5.4.7. Let $H(\xi) = H_0 + V_{\xi}(x)$ be a random operator on $\ell_2(\mathbb{Z}^d)$, where H_0 is a \mathbb{Z}^d -translation invariant difference operator and $V_{\xi}, \xi \in \Omega$, an iid random bounded function. Then there exists a set $\Omega_0 \subset \Omega$ of probability 1, such that for any $f \in C_c(\mathbb{R})$ and for any $\xi \in \Omega_0$ there exists the limit:

$$\lim_{R \to \infty} \frac{1}{|B(0,R)|} \sum_{|n| \le R} \langle \delta_n, f(H_{\xi}) \delta_n \rangle = \mathbb{E}(\langle \delta_0, f(H_{\xi}) \delta_0 \rangle).$$
(5.16)

Proof. Proof follows a standard argument, see e.g. [AW15, Chapter 3]. Let Σ be a countable dense subset of $C_c(\mathbb{R})$. The random operator $H(\xi)$ is ergodic and obeys (5.14) so Theorem 5.4.6 is applicable. By this theorem, for every $f \in \Sigma$ there exists a full set $\Omega_f \subset \Omega$ such that (5.16) holds for all $\xi \in \Omega_f$. Define a full set $\Omega_0 = \bigcap_{f \in \Sigma} \Omega_f$, so for every $f \in \Sigma$ and every $\xi \in \Omega_0$ the equality (5.16) holds. Choose any $g \in C_c(\mathbb{R})$ and let $f_1, f_2, \ldots \in \Sigma$ be such that $f_n \to g$ in uniform topology. Let $\varepsilon > 0$. Further we proceed by a standard $\varepsilon/3$ -trick. Let $N \in \mathbb{N}$ be such that for all $n \geq N ||f_n - g||_{\infty} < \varepsilon/3$. For f_N the equality (5.16) holds for any $\xi \in \Omega_0$. Let $R_0 > 0$ be such that for all $R > R_0$ and all $\xi \in \Omega_0$

$$\left|\frac{1}{|B(0,R)|}\sum_{|n|\leq R} \langle \delta_n, f_N(H_{\xi})\delta_n \rangle - \mathbb{E}(\langle \delta_0, f_N(H_{\xi})\delta_0 \rangle)\right| < \varepsilon/3.$$

Then for all $R > R_0$ and $\xi \in \Omega_0$ we have

$$\begin{aligned} \left| \frac{1}{|B(0,R)|} \sum_{|n| \le R} \langle \delta_n, g(H_{\xi}) \delta_n \rangle - \mathbb{E}(\langle \delta_0, g(H_{\xi}) \delta_0 \rangle) \right| \\ \le \left| \frac{1}{|B(0,R)|} \sum_{|n| \le R} \langle \delta_n, [g(H_{\xi}) - f_N(\xi)] \delta_n \rangle \right| \\ + \left| \frac{1}{|B(0,R)|} \sum_{|n| \le R} \langle \delta_n, f_N(H_{\xi}) \delta_n \rangle - \mathbb{E}(\langle \delta_0, f_N(H_{\xi}) \delta_0 \rangle) \right| \\ + \left| \mathbb{E}(\langle \delta_0, [f_N(H_{\xi}) - g(H_{\xi})] \delta_0 \rangle) \right| \\ < \varepsilon, \end{aligned}$$

where the last inequality follows from the triangle, Schwartz and $||f(H) - g(H)|| \le ||f - g||_{\infty} < \varepsilon/3$ inequalities.

Chapter 6

The density of states on manifolds

I wholeheartedly congratulate you and wish you to get better and work much harder.

Fedor Sukochev

This chapter is an adaptation of [HM24a], joint work with Edward McDonald. Section 6.6 appeared in the preprint version of that paper [HM23], but was omitted from the published article. For the content of this chapter, we are grateful to Teun van Nuland, Fedor Sukochev, and the anonymous referees of [HM24a] for helpful comments. The main results in this chapter are a Dixmier trace formula for the density of states on manifolds in Theorem 6.0.3 and a Dixmier trace formula for Roe's index on open manifolds in Theorem 6.4.5.

Like Chapter 5, this chapter revolves around a Dixmier trace formula for the density of states (DOS). The general form of this formula can be stated for a metric space (X, d) with a Borel measure, a weight $w : X \to \mathbb{C}$ such that $M_w \in \mathcal{L}_{1,\infty}$, and a basepoint $x_0 \in X$. As explained in the introduction of Chapter 5 — see also Section 1.5 — the Dixmier trace formula for the DOS that we study is the equality of the two Borel measures associated

with a self-adjoint operator H on $L_2(X)$ and an extended limit $\omega \in \ell_{\infty}^*$,

$$\operatorname{Tr}_{\omega}(f(H)M_w) = \int_{\mathbb{R}} f \, d\nu_1, \quad f \in C_c(\mathbb{R}),$$
(6.1)

and

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(f(H) M_{\chi_{B(x_0, R)}}) = \int_{\mathbb{R}} f \, d\nu_2, \quad f \in C_c(\mathbb{R}),$$
(6.2)

where the existence of the measure ν_2 , the density of states, has to be assumed. The equality of these measures up to a constant implies that when the DOS exists, the measure ν_1 does not depend on ω .

Whereas Chapter 5 (based on [Aza+22]) proves this result for the case where (X, d) is a discrete metric space with certain additional properties, here we will stay closer to the setting in the paper [AMSZ20] where this formula was first proven for $X = \mathbb{R}^d$, $d \ge 2$, $w(x) = (1 + |x|^2)^{-\frac{d}{2}}$, and $H = -\Delta + M_V$ with $V \in L_{\infty}(\mathbb{R}^d)$ real-valued.

Namely, we will consider (oriented, connected) non-compact Riemannian manifolds of bounded geometry (the definition follows below in Section 6.1). On these manifolds, the operators H that we will take into account are self-adjoint lower-bounded uniformly elliptic differential operators of order two. Additionally, like in the discrete case in Chapter 5, we require the volume of the balls $|B(x_0, R)|$ to grow in a sub-exponential and regular manner, specified in Definition 6.0.1.

Let us discuss these conditions. The DOS has been studied before on manifolds, usually in the setting where for the non-compact Riemannian manifold M one picks a discrete, finitely generated group Γ of isometries of (M, g) which acts freely and properly discontinuously on M such that the quotient M/Γ is compact, and the operator studied is a random Schrödinger operator [AS93; LPPV08; LPV04; PV02; Ves08]. It is not difficult to see that the existence of such a group is a stronger condition than requiring bounded geometry, this will be explicitly proven in Proposition 6.1.7.

Furthermore, in each of the cited papers a recurring assumption is that the group of isometries Γ is amenable. This is equivalent with the existence of an expanding family of

bounded domains $D_j \subset M$ such that

$$\lim_{j \to \infty} \frac{|\partial_h D_j|}{|D_j|} = 0, \quad \forall h > 0,$$

where $\partial_h D_j := \{x \in D_j : d(x, \partial D_j) \leq h\}$ [AS93]. This is a *weaker* condition than what we will require of the growth of the balls $B(x_0, R)$ specified in Definition 6.0.1 below. In fact, for these manifolds to satisfy our assumptions, it is necessary that Γ has subexponential growth which implies amenability. In Section 6.5 a more detailed analysis of this co-compact setting with a random Schrödinger operator will follow.

Let (M, g) be a *d*-dimensional Riemannian manifold, and let d_g be the distance function on M induced by the Riemannian metric. The Riemannian volume of the closed ball $B(x_0, r)$ is denoted by $|B(x_0, r)|$. Its boundary, $\partial B(x_0, r)$, is a (d-1)-dimensional Hausdorffmeasurable subset of M, and as such we can talk about its volume, calculated with respect to its inherited Riemannian metric. This d-1-dimensional volume we will also denote as $|\partial B(x_0, r)|$. In fact, it then holds that (see Section 6.1.2)

$$\frac{d}{dr}|B(x_0,r)| = |\partial B(x_0,r)|.$$

What we require of our manifolds is that both $|B(x_0, r)|$ and $|\partial B(x_0, r)|$ grow sufficiently slowly and regularly. Namely, we will ask that the ratios $\frac{|\partial B(x_0, r)|}{|B(x_0, r)|}$ and $\frac{\frac{d}{dr}|\partial B(x_0, r)|}{|\partial B(x_0, r)|}$ vanish as $R \to \infty$ in the following way.

Definition 6.0.1 (Property (D)). Let (M,g) be a non-compact Riemannian manifold of bounded geometry. It is said to have Property (D) if $r \mapsto V(r) := |B(x_0,r)|$ is in $C^2(\mathbb{R}_{\geq 0})$,

$$\frac{V'(r)}{V(r)} \in L_2(\mathbb{R}_{\ge 1}),\tag{6.3}$$

and

$$\lim_{r \to \infty} \frac{V''(r)}{V'(r)} = 0.$$
 (6.4)

Lemma 6.3.11 below gives in particular that Property (D) implies that $\frac{V'(r)}{V(r)} \to 0$ as $r \to \infty$ (which is not immediate from the requirement that it be in $L_2(\mathbb{R}_{\geq 1})$). If a function $f \in C^1(\mathbb{R})$ satisfies $\lim_{x\to\infty} \frac{f'(x)}{f(x)} = 0$, then $\log f(x) = o(x)$ and hence $f(x) = e^{o(x)}$. Therefore, if the manifold M satisfies Property (D), then necessarily both $|B(x_0, r)| = e^{o(r)}$ and $|\partial B(x_0, r)| = e^{o(r)}$. A quick calculation shows that Property (D) still admits volume growth of the order $|B(x_0, r)| = \exp(r^{\frac{1}{2}-\varepsilon})$, and it is not difficult to see that Property (D) is satisfied for Euclidean spaces. The conditions listed mostly serve to prevent erratic behaviour of the growth. Observe the similarity in this sense to Property (C) in Chapter 5 which required for discrete metric spaces X

$$\frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} \to 1, \quad k \to \infty,$$

where $\{r_k\}_{k=0}^{\infty}$ is the set $\{d(y, x_0) : y \in X\}$ ordered in increasing manner.

Finally, let us specify the class of operators for which the main theorem is formulated. The following definition is essentially the same as [Kor91] and the C^{∞} -bounded differential operators defined in [Shu92, Appendix 1].

Definition 6.0.2. A differential operator P on a d-dimensional Riemannian manifold M of bounded geometry is called a uniform differential operator (of order m) if it can be expressed in normal coordinates in a neighbourhood of each point $x \in M$ as

$$P = \sum_{|\alpha| \le m} a_{\alpha,x}(y) \partial_y^{\alpha},$$

and for all multi-indices β we have

$$\left|\partial_y^\beta a_{\alpha,x}(0)\right| \le C_{\alpha,\beta}, \quad |\alpha| \le m.$$

Following the notation of [Kor91], we denote this by $P \in BD^m(M)$.

Furthermore, let $\sigma_x(y,\xi) = \sum_{|\alpha|=m} a_{\alpha,x}(y)(i\xi)^{\alpha}$ be the principal symbol of P near x. We say that $P \in BD^m(M)$ is uniformly elliptic, denoted $P \in EBD^m(M)$, if there exists $\varepsilon > 0$ such that

$$|\sigma_x(0,\xi)| \ge \varepsilon \, |\xi|^m \,, \quad \xi \in \mathbb{R}^d, x \in M.$$

A similar definition applies to operators acting between sections of vector bundles of bounded geometry; see [Shu92].

Having discussed all necessary details, we can now formulate the main theorem of this chapter.

Theorem 6.0.3. Let (M, g) be a non-compact Riemannian manifold of bounded geometry with Property (D). Let $P \in EBD^2(M)$ be self-adjoint and lower-bounded, and let w be the function on M defined by

$$w(x) = \frac{1}{1 + |B(x_0, d_g(x, x_0))|}, \quad x \in M.$$

Then $f(P)M_w$ is an element of $\mathcal{L}_{1,\infty}$ for all compactly supported functions $f \in C_c(\mathbb{R})$. If P admits a density of states ν_P , we have for all extended limits $\omega \in \ell_{\infty}^*$

$$\operatorname{Tr}_{\omega}(f(P)M_w) = \int_{\mathbb{R}} f \, d\nu_P, \quad f \in C_c(\mathbb{R}).$$

Observe that Euclidean space is a manifold of bounded geometry satisfying Property (D), and Schrödinger operators $H = -\Delta + M_V$ with smooth bounded potential V are operators of the required class. Therefore, besides the smoothness assumption on V, the main result in this paper is a generalisation of [AMSZ20].

Whereas the proofs in [AMSZ20] are based on delicate singular value estimates particular to Euclidean space, and the ones in [Aza+22] (i.e. Chapter 5) are based on heavy real analysis, here we will prove a statement in abstract operator theory. It is a significant generalisation of [AMSZ20, Theorem 5.7]. In Section 6.6, we will make a comparison between this method and the proof in Chapter 5.

Theorem 6.0.4. Let W and P be linear operators, such that P is self-adjoint and lowerbounded and W is positive and bounded. Assume that for every t > 0 we have

- 1. $\exp(-tP)W \in \mathcal{L}_{1,\infty}$,
- 2. $\exp(-tP)[P,W] \in \mathcal{L}_1$.

Then, for every extended limit ω ,

$$\operatorname{Tr}_{\omega}(e^{-tP}W) = \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}(e^{-tP}\chi_{[\varepsilon,\infty)}(W)), \quad t > 0.$$

whenever the limit on the right hand side exists.

As an application of the Dixmier trace formula for the DOS, Theorem 6.0.3, we will look at Roe's index theorem on open manifolds [Roe88]. Roe's index theorem is one approach of many that extends Atiyah–Singer's index theorem [AS63] to non-compact manifolds, namely non-compact manifolds of bounded geometry that admit a *regular exhaustion*. For the precise definition we refer to [Roe88, Section 6], but for this introduction it suffices to know that Property (D) is a stronger assumption.

Given a *compact* manifold M with two vector bundles $E, F \to M$ and an elliptic differential operator $D: \Gamma(E) \to \Gamma(F)$, the local index formula equates the Fredholm index of D to the integral of a differential form given by topological data denoted here simply by $\mathbf{I}(D)$,

$$\operatorname{Ind}(D) = \int_{M} \mathbf{I}(D).$$
(6.5)

A special case of the index formula can be proved with the McKean–Singer formula, which in this case states that

$$\operatorname{Ind}(D) = \operatorname{Tr}(\eta e^{-t\mathcal{D}^2}), \quad t > 0,$$

where η is the grading operator on the bundle $E \oplus F \to M$, and \mathcal{D} is the self-adjoint operator acting on $\Gamma(E \oplus F)$ by the formula

$$\mathcal{D} = \begin{pmatrix} 0 & D^* \\ D & 0 \end{pmatrix}.$$

If M is not compact, neither side of (6.5) is well-defined in general. In the setting of a non-compact Riemannian manifold M of bounded geometry with a regular exhaustion, and with a graded Clifford bundle $S \to M$ also of bounded geometry (see Section 6.4), which comes with a natural first-order elliptic differential operator, the Dirac operator $D: \Gamma(S_+) \to \Gamma(S_-)$, Roe modifies both sides of equation (6.5) as follows. Defining a linear functional m on bounded d-forms via an averaging procedure, the right-hand side simply becomes $m(\mathbf{I}(D))$. For the left-hand side, Roe defines an algebra of uniformly smoothing operators $\mathcal{U}_{-\infty}$, and shows that elliptic operators are invertible modulo $\mathcal{U}_{-\infty}$. Recall that for operators that are invertible modulo compact operators (Fredholm operators), the Fredholm index is an element of the K-theory group $K_0(K((\mathcal{H})) = \mathbb{Z}$ [Weg93] (recall that $K(\mathcal{H})$ are the compact operators). In this case, we can similarly define an abstract index of an elliptic operator D as an element of $K_0(\mathcal{U}_{-\infty})$, by observing that $\mathcal{U}_{-\infty}$ forms an ideal in what Roe defines as uniform operators \mathcal{U} . Furthermore, using the functional m one can define a trace τ on $\mathcal{U}_{-\infty}$. This trace can be extended to a trace on the matrix algebras $M_n(\mathcal{U}^+_{-\infty})$ (with $\mathcal{U}^+_{-\infty}$ denoting the unitisation of $\mathcal{U}_{-\infty}$) by putting $\tau(1) = 0$ when passing to the unitisation, and then tensoring with the usual trace on $M_n(\mathbb{C})$. The tracial property gives that this map descends to a map called the dimension-homomorphism $\dim_{\tau} : K_0(\mathcal{U}_{-\infty}) \to \mathbb{R}$. These ingredients give Roe's index theorem:

$$\dim_{\tau}(\operatorname{Ind}(D)) = m(\mathbf{I}(D)). \tag{6.6}$$

The nature of the averaging procedure that Roe develops is such that if D^2 admits a density of states, Theorem 6.0.3 leads to a Dixmier trace reformulation of the analytical index dim_{τ}(Ind(D)). In Section 6.4 we prove

$$\dim_{\tau}(\operatorname{Ind}(D)) = \operatorname{Tr}_{\omega}(\eta \exp(-tD^2)M_w), \quad t > 0,$$

where η is the grading on S.

6.1 Preliminaries

6.1.1 Operator theory

We recall some facts about sums of left-disjoint families of operators, and prove an estimate for their \mathcal{L}_p -norm and $\mathcal{L}_{p,\infty}$ -norm. A family $\{T_j\}_{j=0}^{\infty}$ of bounded linear operators on a Hilbert space \mathcal{H} is *left-disjoint* if $T_j^*T_k = 0$ for all $j \neq k$.

Given a sequence $\{T_j\}_{j=0}^{\infty}$ of bounded linear operators on \mathcal{H} , let

$$\bigoplus_{j=0}^{\infty} T_j$$

denote the operator on $\mathcal{H} \otimes \ell_2(\mathbb{N})$ given by

$$\sum_{j=0}^{\infty} T_j \otimes e_j e_j^*$$

where $e_j e_j^*$ is the rank 1 projection onto the orthonormal basis element $e_j \in \ell_2$.

Note that

$$\mu\left(\bigoplus_{j=0}^{\infty}T_j\right) = \mu\left(\bigoplus_{j=0}^{\infty}\operatorname{diag}(\mu(T_j))\right).$$

To put it differently, the singular value sequence of the direct sum $\bigoplus_{j=0}^{\infty} T_j$ is the decreasing rearrangement of the sequence indexed by \mathbb{N}^2 given by

$$\{\mu(k,T_j)\}_{j,k=0}^{\infty}.$$

By definition, we have $||T||_{p_1,p_2} = ||\mu(T)||_{\ell_{p_1,p_2}}$. Therefore,

$$\left\| \bigoplus_{j=0}^{\infty} T_j \right\|_{p_1, p_2} = \| \{ \mu(k, T_j) \}_{j, k \ge 0} \|_{\ell_{p_1, p_2}(\mathbb{N}^2)}.$$

Now let q > 0. We have for each j that $\mu(k, T_j) \leq (k+1)^{-\frac{1}{q}} ||T_j||_{q,\infty}$. Therefore,

$$\left\| \bigoplus_{j=0}^{\infty} T_j \right\|_{p_1, p_2} \le \| \{ (1+k)^{-\frac{1}{q}} \| T_j \|_{q, \infty} \}_{j, k \ge 0} \|_{\ell_{p_1, p_2}(\mathbb{N}^2)}.$$

Lemma 4.3 of [LSZ20a] implies that if q is sufficiently small, then there exists a constant $C_{p_1,p_2,q}$ such that for any sequence $\{x_j\}_{j=0}^{\infty}$ we have

$$\|\{(1+k)^{-\frac{1}{q}}x_j\}_{j,k\geq 0}\|_{\ell_{p_1,p_2}(\mathbb{N}^2)} \le C_{p_1,p_2,q}\|\{x_j\}_{j=0}^{\infty}\|_{\ell_{p_1,p_2}}$$

Taking $x_j = ||T_j||_{q,\infty}$ and sufficiently small q (depending on p_1, p_2) we have

$$\left\| \bigoplus_{j=0}^{\infty} T_j \right\|_{p_1, p_2} \le C_{p_1, p_2, q} \|\{ \|T_j\|_{q, \infty}\}_{j=0}^{\infty} \|_{p_1, p_2}.$$
(6.7)

Combining [LSZ20a, Proposition 2.7, Lemma 2.9] gives the following result.

Lemma 6.1.1. Let $0 , and let <math>\{T_j\}_{j=0}^{\infty}$ be a left-disjoint family of bounded linear operators. There exist constants C_p , C'_p such that

$$\left\| \sum_{j=0}^{\infty} T_j \right\|_{p,\infty} \le C_p \left\| \bigoplus_{j=0}^{\infty} T_j \right\|_{p,\infty};$$
$$\left\| \sum_{j=0}^{\infty} T_j \right\|_p \le C_p' \left\| \bigoplus_{j=0}^{\infty} T_j \right\|_p.$$

A combination of (6.7) and Lemma 6.1.1 immediately yields the following:

Corollary 6.1.2. Let $0 , and let <math>\{T_j\}_{j=0}^{\infty}$ be a left-disjoint family of operators. There exist q, q' > 0 (depending on p) and constants $C_{p,q}, C'_{p,q'} > 0$ such that

$$\left\| \sum_{j=0}^{\infty} T_{j} \right\|_{p,\infty} \leq C_{p,q} \left\| \{ \|T_{j}\|_{q,\infty} \}_{j=0}^{\infty} \right\|_{\ell_{p,\infty}};$$
$$\left\| \sum_{j=0}^{\infty} T_{j} \right\|_{p} \leq C'_{p,q'} \left\| \{ \|T_{j}\|_{q',\infty} \}_{j=0}^{\infty} \right\|_{\ell_{p}}.$$

Remark 6.1.3. For p = 1, we can take any 0 < q, q' < 1.

6.1.2 Preliminaries on manifolds

All manifolds in this chapter are smooth, oriented, non-compact and connected unless stated otherwise.

Definition 6.1.4. Let (M, g) be a Riemannian manifold. The injectivity radius i(x) at a point $x \in M$ is defined as

 $i(x) := \sup\{R \in \mathbb{R}_{\geq 0} : \exp_x |_{B(0,R)} \text{ is a diffeomorphism}\},\$

where \exp_x is the exponential map at x and $B(0,R) \subseteq T_xM$ is the metric ball with radius R centered around the origin. The injectivity radius of the manifold M is defined as

$$i_g := \inf_{x \in M} i(x)$$

It is a theorem that the injectivity radius map

$$i: M \to (0, \infty]$$

 $x \mapsto i(x)$

is continuous [Sak96, Proposition III.4.13].

Definition 6.1.5. A Riemannian manifold (M,g) has bounded geometry if the injectivity radius i_g satisfies

$$i_g > 0,$$
and the Riemannian curvature tensor R and all its covariant derivatives are uniformly bounded.

This is the definition as in [Kor91, Definition 1.1], [Eic09, Chapter II] and [Eic07, Chapter 3]. Every open manifold admits a metric of bounded geometry [Gre78].

As is well-known (see e.g. [Kor91, Lemma 2.4]), bounded geometry implies that there exists $r_0 > 0$ and a countable set $\{x_j\}_{j=0}^{\infty}$ of points in M such that:

- 1. $M = \bigcup_{j=0}^{\infty} B(x_j, r_0);$
- 2. each ball $B(x_j, r_0)$ is a chart for the exponential normal coordinates based at x_j ;
- 3. the covering $\{B(x_j, r_0)\}_{j=0}^{\infty}$ has finite order, meaning that there exists N such that each ball intersects at most N other balls;
- 4. $\sup_{i} |B(x_{j}, r_{0})| < \infty$, recall that $|\cdot|$ indicates the volume;
- 5. there exists a partition of unity $\{\psi_j\}_{j=0}^{\infty}$ subordinate to $\{B(x_j, r_0)\}$ such that for every α we have $\sup_{j,x} |\partial^{\alpha} \psi_j(x)| < \infty$, where ∂^{α} is taken in the exponential normal coordinates of $B(x_j, r_0)$.

Without loss of generality, r_0 can be taken smaller or equal to 1 (see [Kor91, Lemma 2.3]).

We will refer to the scale of Sobolev spaces $\{\mathcal{H}^s(M)\}_{s\in\mathbb{R}}$ defined over M as

$$\mathcal{H}^s := \overline{\mathrm{dom}(1-\Delta_g)^{\frac{s}{2}}}^{\frac{s}{2}} \|\cdot\|_s,$$

with $\|\xi\|_s := \|(1 - \Delta_g)^{\frac{s}{2}} \xi\|_{L_2(M)}$, as in Section 1.2 and Chapter 2, see also [Kor91, Section 3] or [GS13]. We will make use of the fact that if $P \in BD^m(M)$, then $P \in \operatorname{op}^m(1 - \Delta_g)^{\frac{1}{2}}$, i.e. P defines a bounded linear operator from $\mathcal{H}^{s+m}(M)$ to $\mathcal{H}^s(M)$ for every $s \in \mathbb{R}$.

Remark 6.1.6. It follows from the bounded geometry assumption that there exist constants c, C > 0 such that for every $x \in M$ and R > 0 we have

$$|B(x,R)| \le C \exp(cR).$$

Note that for almost every r > 0, we have

$$\frac{d}{dr}|B(x,r)| = |\partial B(x,r)|.$$

See [Cha06, Proposition III.3.2 & Proposition III.5.1].

A standard example of a manifold of bounded geometry is a covering space of a compact manifold. The following is well-known, but lacking a reference we supply a proof for the reader's convenience.

Proposition 6.1.7. Let (M, g) be a complete d-dimensional Riemannian manifold. Let Γ be a discrete, finitely generated subgroup of the isometries of (M, g) which acts freely and properly discontinuously on M such that the quotient M/Γ is a compact (d-dimensional) Riemannian manifold. Then M has bounded geometry.

Proof. Since Γ acts cocompactly on M, there exists a compact set $L \subseteq M$ such that $\bigcup_{\gamma \in \Gamma} \gamma L = M$. Indeed, since the open balls $B(x,1), x \in M$ project onto an open cover of M/Γ , there exists a finite collection $\{B(x_i,1)\}_{i=1}^N$ that projects onto M/Γ . Defining $L := \bigcup_{i=1}^N \overline{B(x_i,1)}$ gives the claimed compact set L.

Set

$$i_L := \inf_{x \in L} i(x),$$

where i(x) is the injectivity radius at the point x. Since the injectivity radius is a continuous function on M, i(x) > 0 for all $x \in M$, and L is compact, it follows that $i_L > 0$. Since Γ acts by isometries, for any $\gamma \in \Gamma$ we have $i(\gamma x) = i(x)$, see for example the proof of [Sak96, Theorem III.5.4]. Therefore, $\inf_{x \in M} i(x) = i_L > 0$.

Next, since the curvature tensor R is smooth, clearly R and all its covariant derivatives are bounded on L. Let $\Phi : M \to M$ be the isometry by which $\gamma \in \Gamma$ acts. Then for all $x \in M \ \Phi_x^* : T_x M \to T_{\Phi(x)} M$ is an isomorphism, and by [Sak96, p. 41],

$$\Phi_x^*(\nabla_{\xi}\eta) = \nabla_{\Phi_x^*\xi}\Phi_x^*\eta,$$
$$\Phi_x^*(R(\eta,\xi)\zeta) = R(\Phi_x^*\eta,\Phi_x^*\xi)\Phi_x^*\zeta,$$

where $\xi, \eta, \zeta \in T_x M$. Combine the facts that R and its covariant derivatives are bounded on L, that $\bigcup_{\gamma \in \Gamma} \gamma L = M$, and that isometries preserve R and taking covariant derivatives in the above manner, and we can conclude that R and its covariant derivatives are uniformly bounded on M.

Remark 6.1.8. A non-compact Riemannian manifold of bounded geometry has infinite volume. This can be checked easily via the covering $M = \bigcup_{j=0}^{\infty} B(x_j, r_0)$ below Definition 6.1.5, and the observation that $\inf_j |B(x_j, r_0)| > 0$ [Kod88].

6.2 Proof of Theorem 6.0.4

Part of the proof in [AMSZ20] used the identity

$$\lim_{s \downarrow 1} (s-1) \int_{\mathbb{R}^d} F(x) (1+|x|^2)^{-\frac{s}{2}} dx = \lim_{R \to \infty} R^{-d} \int_{B(0,R)} F(x) dx$$
(6.8)

for any bounded measurable function F on \mathbb{R}^d such that the right hand side exists.

Equation (6.8) can be generalised in the following manner. The proof is essentially the same as [AMSZ20, Lemma 6.1].

Lemma 6.2.1. Let A and B be bounded linear operators, $B \ge 0$, such that $AB^s \in \mathcal{L}_1$ for every s > 1. Then

$$\lim_{s \downarrow 1} (s-1) \operatorname{Tr}(AB^s) = \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}(A\chi_{[\varepsilon,\infty)}(B))$$

whenever the limit on the right exists.

Proof. Writing $B^s = \int_0^{\|B\|_{\infty}} \lambda^s dE_B(\lambda)$, $\lambda^s = s \int_0^{\lambda} r^{s-1} dr$ and applying Fubini's theorem yields

$$\operatorname{Tr}(AB^{s}) = s \int_{0}^{\|B\|_{\infty}} r^{s-1} \operatorname{Tr}(A\chi_{[r,\infty)}(B)) dr$$

Assume without loss of generality that $||B||_{\infty} = 1$. Writing $F(r) := \text{Tr}(A\chi_{[r,\infty)}(B))$ our assumption is that

$$F(r) \sim \frac{c}{r}, \quad r \to 0,$$

and

$$\operatorname{Tr}(AB^s) = s \int_0^1 r^{s-1} F(r) \, dr.$$

Let $\varepsilon > 0$, and choose R > 0 sufficiently small such that if 0 < r < R then

$$|rF(r) - c| < \varepsilon.$$

We write $\operatorname{Tr}(AB^s) - \frac{c}{s-1}$ as

$$\operatorname{Tr}(AB^{s}) - \frac{c}{s-1} = c + s \int_{0}^{1} r^{s-1} F(r) - cr^{s-2} dr.$$

Therefore

$$|\operatorname{Tr}(AB^{s}) - \frac{c}{s-1}| \le |c| + s \int_{0}^{1} r^{s-2} |rF(r) - c| \, dr \le |c| + s \int_{R}^{1} r^{s-2} |rF(r) - c| \, dr + \frac{s\varepsilon}{s-1}.$$

It follows that

$$|(s-1)\operatorname{Tr}(AB^s) - c| = O(s-1) + s\varepsilon, \quad s \downarrow 1.$$

Since ε is arbitrary, this completes the proof.

We will make use of the following theorem, which is [LMSZ23, Theorem 1.3.20].

Theorem 6.2.2. Let A and B be positive bounded linear operators such that $[B, A^{\frac{1}{2}}] \in \mathcal{L}_1$ and $AB \in \mathcal{L}_{1,\infty}$. If

$$||A^{\frac{1}{2}}B^{s}||_{1} = o((s-1)^{-2}), \quad s \downarrow 1$$

then for every extended limit ω we have

$$\operatorname{Tr}_{\omega}(AB) = \lim_{s \downarrow 1} (s-1) \operatorname{Tr}(AB^s)$$

if the limit exists.

Hence, if $[B, A^{\frac{1}{2}}] \in \mathcal{L}_1$ and $||A^{\frac{1}{2}}B^s||_1 = o((s-1)^{-2})$, then by Lemma 6.2.1

$$\operatorname{Tr}_{\omega}(AB) = \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}(A\chi_{[\varepsilon,\infty)}(B))$$

whenever the limit on the right exists.

Recall the Araki–Lieb–Thirring inequality [Kos92]

$$\|AB\|_{r,\infty}^r \le e\|A^r B^r\|_{1,\infty}, \quad r > 1$$
(6.9)

and the numerical inequality

$$\|X\|_{\infty,1} = \sum_{k=0}^{\infty} \frac{\mu(k,X)}{k+1} \le \|X\|_{q,\infty} \zeta(1+\frac{1}{q}),$$
(6.10)

obtained by simply writing out the definitions. Here ζ is the Riemann zeta function, and it obeys $\zeta(1+\frac{1}{q}) \sim q$ as $q \to \infty$.

Corollary 6.2.3. Let A and B be positive bounded linear operators such that

1. $[A^{\frac{1}{2}}, B] \in \mathcal{L}_1,$ 2. $A^{\frac{1}{4}}B \in \mathcal{L}_{1,\infty}.$

Then for every extended limit ω we have

$$\operatorname{Tr}_{\omega}(AB) = \lim_{s \downarrow 1} (s-1) \operatorname{Tr}(AB^s)$$

if the limit exists.

Proof. Let 1 < s < 2. We have

$$\|A^{\frac{1}{2}}B^{s}\|_{1} \leq \|[A^{\frac{1}{2}}, B]B^{s-1}\|_{1} + \|BA^{\frac{1}{2}}B^{s-1}\|_{1} \leq \|[A^{\frac{1}{2}}, B]\|_{1}\|B\|_{\infty}^{s-1} + \|BA^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}B^{s-1}\|_{\infty,1} \leq \|[A^{\frac{1}{4}}, B^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}B^{s-1}\|_{\infty,1} \leq \|[A^{\frac{1}{4}}, B^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}B^{s-1}\|_{\infty,1} \leq \|[A^{\frac{1}{4}}, B^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|A^{\frac{1}{4}}\|_{1,\infty}\|$$

where we have used the inequality

$$||TS||_1 \le 2||T||_{1,\infty}||S||_{\infty,1}, \quad T \in \mathcal{L}_{1,\infty}, S \in \mathcal{L}_{\infty,1},$$

which can easily be checked via the definitions of these quasi-norms and the inequality $\mu(2k, TS) \leq \mu(k, T)\mu(k, S).$

By the numerical inequality (6.10) above

$$\|A^{\frac{1}{4}}B^{s-1}\|_{\infty,1} \lesssim (s-1)^{-1} \|A^{\frac{1}{4}}B^{s-1}\|_{\frac{1}{s-1},\infty}.$$

Since s < 2, we have $\frac{1}{s-1} > 1$, and hence the Araki–Lieb–Thirring inequality (6.9) delivers

$$\|A^{\frac{1}{4}}B^{s-1}\|_{\frac{1}{s-1},\infty}^{\frac{1}{s-1}} \le e\|A^{\frac{1}{4(s-1)}}B\|_{1,\infty} \le e\|A^{\frac{1}{4}}\|_{\infty}^{\frac{2-s}{s-1}}\|A^{\frac{1}{4}}B\|_{1,\infty}.$$

This verifies the assumptions of Theorem 6.2.2.

In our case, we have an operator P which is self-adjoint and lower-bounded, which means $\exp(-tP)$ is positive and bounded for all t > 0, and we take B = W. Then the assumptions become

$$[\exp(-tP), W] \in \mathcal{L}_1, \ \exp(-tP)W \in \mathcal{L}_{1,\infty}$$
(6.11)

for every t > 0.

The former condition can be modified to one which is easier to verify in geometric examples.

Lemma 6.2.4 (Duhamel's formula). Let P be a lower-bounded self-adjoint operator on a Hilbert space \mathcal{H} , and let W be a bounded operator. Then

$$[\exp(-tP), W] = -\int_0^t \exp(-sP)[P, W] \exp(-(t-s)P) \, ds.$$

Proof. The method of proof is standard, see for example [ACDS09, Lemma 5.2]. Define $F : [0,t] \to B(\mathcal{H})$ by $F(s) = \exp(-sP)W\exp(-(t-s)P)$. Since P is lower-bounded, $\exp(-sP)$ is bounded for all $s \in [0,t]$. Hence the derivative of F(s) in the strong operator topology is

$$F'(s) = -P\exp(-sP)W\exp(-(t-s)P) + \exp(-sP)WP\exp(-(t-s)P)$$
$$= -\exp(-sP)[P,W]\exp(-(t-s)P),$$

in the sense that

$$\lim_{h \to 0} \frac{1}{h} (F(s+h) - F(s))\xi = F'(s)\xi, \quad \xi \in \mathcal{H}.$$

Therefore, by the fundamental theorem of calculus for Banach space-valued functions, we

can conclude that for $\xi \in \mathcal{H}$,

$$[\exp(-tP), W]\xi = (F(t) - F(0))\xi$$
$$= \int_0^t F'(s)\xi \, ds$$
$$= -\int_0^t \exp(-sP)[P, W] \exp(-(t-s)P)\xi \, ds.$$

Lemma 6.2.5. Let P be a self-adjoint lower bounded linear operator and let W be a bounded self-adjoint operator such that [P, W] makes sense, and

$$\exp(-tP)[P,W] \in \mathcal{L}_1, \quad t > 0.$$

Then

$$[\exp(-tP), W] \in \mathcal{L}_1, \quad t > 0$$

Proof. By the Duhamel formula (Lemma 6.2.4),

$$[\exp(-tP), W] = -\int_0^t \exp(-sP)[P, W] \exp(-(t-s)P) \, ds$$
$$= -\int_0^{\frac{t}{2}} \exp(-sP)[P, W] \exp(-(t-s)P) \, ds$$
$$-\int_{\frac{t}{2}}^t \exp(-sP)[P, W] \exp(-(t-s)P) \, ds.$$

For $0 < s < \frac{t}{2}$, we have

$$\|\exp(-sP)[P,W]\exp(-(t-s)P)\|_{1} \le \|[P,W]\exp(-\frac{t}{2}P)\|_{1}$$

while for $\frac{t}{2} < s < t$,

$$\|\exp(-sP)[P,W]\exp(-(t-s)P)\|_{1} \le \|\exp(-\frac{t}{2})[P,W]\|_{1}.$$

Hence, the triangle inequality for weak integrals we have

$$\|[\exp(-tP), W]\|_1 \le t \|[P, W]\|_1.$$

Combining the results of this section completes the proof of Theorem 6.0.4.

6.3 Manifolds of bounded geometry

Let us now shift our attention to the case where we have a Riemannian manifold of bounded geometry M and a self-adjoint, lower-bounded operator $P \in EBD^2(M)$.

Estimates of the form

$$\|M_f g(-i\nabla)\|_{p,\infty} \le c_p \|f\|_p \|g\|_{p,\infty}$$

for p > 2 are sometimes called Cwikel-type estimates. Here, f and g are function on \mathbb{R}^d , see [Sim05, Chapter 4]. Similar estimates with p < 2 were obtained earlier by Birman and Solomyak [BS69]. In particular, it follows from [Sim05, Theorem 4.5] that if f is a measurable function on \mathbb{R}^d then for every t > 0 and 0 that we have

$$\|M_f e^{t\Delta}\|_p \lesssim_{t,p} \left(\sum_{k \in \mathbb{Z}^d} \|f\|_{L_{\infty}(k+[0,1)^d)}^p\right)^{\frac{1}{p}}$$

and similarly

$$||M_f e^{t\Delta}||_{p,\infty} \lesssim_{t,p} ||\{||f||_{L_{\infty}(k+[0,1)^d)}||_{p,\infty}$$

We seek similar estimates for functions f on manifolds of bounded geometry. In place of the decomposition of \mathbb{R}^d into cubes, $\mathbb{R}^d = \bigcup_{k \in \mathbb{Z}}^d [0, 1)^d + k$, we will use the partition of unity constructed according to Section 6.1.2.

Namely, we will show that for $0 , we have <math>\exp(-tP)M_f \in \mathcal{L}_{p,\infty}$ whenever $f \in \ell_{p,\infty}(L_{\infty})$, and $\exp(-tP)M_f \in \mathcal{L}_p$ whenever $f \in \ell_p(L_{\infty})$, where $\ell_{p,\infty}(L_{\infty})$ and $\ell_p(L_{\infty})$ are certain function spaces on M.

Definition 6.3.1. Let $\{x_j\}_{j=0}^{\infty}$ and r_0 be as in Section 6.1.2. Given a bounded measurable function f on M, define

$$\|f\|_{\ell_{p,\infty}(L_{\infty})} := \left\| \{\|f\|_{L_{\infty}(B(x_{j},r_{0}))}\}_{j=0}^{\infty} \right\|_{\ell_{p,\infty}}.$$

and

$$\|f\|_{\ell_p(L_{\infty})} := \left\| \{\|f\|_{L_{\infty}(B(x_j, r_0))} \}_{j=0}^{\infty} \right\|_{\ell_p}.$$

Let *B* be an open ball in \mathbb{R}^d such that at every point $x \in M$ we have a normal coordinate system $\exp_x \circ e : B \to B(x, r_0)$, where *e* is the identification of \mathbb{R}^d with the tangent space $T_x M$ and \exp_x is the Riemannian exponential map [Kor91, Proposition 1.2]. Via these maps we can, for each $x \in M$, pullback the metric *g* restricted to $B(x, r_0)$ to a metric g^x on *B*.

Definition 6.3.2. A Riemannian metric g on B can be represented uniquely by the d^2 smooth functions

$$g_{ij} := g(\partial_i, \partial_j) \in C^{\infty}(B).$$

Define $\operatorname{Riem}_b(B)$ as the set of Riemannian metrics for which the functions g_{ij} extend to smooth functions in $C^{\infty}(\overline{B})$. We equip $\operatorname{Riem}_b(B)$ with the topology induced by the embedding $\operatorname{Riem}_b(B) \subseteq (C^{\infty}(\overline{B}))^{d^2}$, where we we take the usual topology on $C^{\infty}(\overline{B})$ defined by the seminorms

$$p_N(f) := \max_{x \in \overline{B}} \{ |\partial^{\alpha} f(x)| : |\alpha| \le N \}.$$

The following is essentially [Eic91], see in particular the discussion below [Eic91, Theorem A]. See also the related statement [Roe88, Proposition 2.4].

Proposition 6.3.3. Let (M,g) be a manifold of bounded geometry. The functions g_{ij}^x considered as a family of smooth functions parametrized by i, j and by a point $x \in M$, can be extended to \overline{B} and then lie in a bounded subset of $C^{\infty}(\overline{B})$.

In principle the identification e of \mathbb{R}^d with the tangent space $T_x M$ can vary from point to point, and therefore the matrix elements g_{ij}^x are not uniquely defined. However this does not change the fact that for any given identification, the result of Proposition 6.3.3 holds.

Corollary 6.3.4. Let (M, g) be a manifold with bounded geometry, and let $\{B(x_j, r_0)\}_{j \in \mathbb{N}}$ be an open cover of M as in Section 6.1.2. The set

$$\{g^{x_j}: j \in \mathbb{N}\} \subset \operatorname{Riem}_b(B)$$

is a pre-compact subset of $C^{\infty}(\overline{B})$.

Proof. By Proposition 6.3.3, the functions $g_{ij}^{x_k}$ lie in a bounded set in $C^{\infty}(\overline{B})$. Since the closure of a bounded set is also bounded [Con90, Section IV.2], and since $C^{\infty}(\overline{B})$ has the Heine–Borel property [Rud91, Section 1.9], the assertion follows.

Given $g \in \operatorname{Riem}_b(B)$, we denote Δ_g^D the self-adjoint realisation of Δ_g on B with Dirichlet boundary conditions. Explicitly, Δ_g^D is defined as the operator associated with the closure of the quadratic form

$$q_g(u,v) = \int_B \sqrt{|g(x)|} \sum_{i,j} g^{ij}(x) \partial_i u(x) \overline{\partial_j v(x)} \, dx, \quad u,v \in C_c^\infty(B).$$

Lemma 6.3.5. Let x_j and $g^{x_j} \in \operatorname{Riem}_b(B)$ be as in Corollary 6.3.4. We have

$$\sup_{j\in\mathbb{N}} \|(1-\Delta_{g^{x_j}}^D)^{-1}\|_{\mathcal{L}_{\frac{d}{2},\infty}(L_2(B))} < \infty.$$

Proof. Let g be a metric on the closed unit ball \overline{B} , and let c_g and c_G be positive constants such that

$$c_g\left(\sum_{k=1}^d |\xi_k|^2\right) \le \sum_{k,l=1}^d \sqrt{|g(x)|} g^{kl}(x) \xi_k \overline{\xi_l} \le C_g\left(\sum_{k=1}^d |\xi_k|^2\right)$$

for all $\xi \in \mathbb{C}^d$. We will prove that there is a constant k_d such that

$$\|(1-\Delta_g^D)^{-1}\|_{\mathcal{L}^{\frac{d}{2},\infty}(L_2(B))} \le k_d c_g^{-\frac{1}{2}}.$$
(6.12)

Since

$$\inf_{i} c_{g^{x_j}} > 0,$$

(6.12) implies the result.

Let q_0 denote the Dirichlet quadratic form on B. That is,

$$q_0(u,v) := \sum_j \int_B \partial_j u(x) \overline{\partial_j v}(x) \, dx, \quad u,v \in C_c^\infty(B).$$

The Dirichlet Laplacian Δ_0^D on B is defined as the operator associated with the closure of the quadratic form q_0 (see e.g. [Sim15, Example 7.5.26]). By the definitions of q_g , c_g and C_g we have

$$c_g q_0(u, u) \le q_g(u, u) \le C_g q_0(u, u), \quad u \in C_c^{\infty}(B).$$

It follows that the form domains of the closures of q_0 and q_g coincide, we denote this space $H_0^1(B)$. The preceding inequality implies in particular that

$$c_g \| (1 - \Delta_0^D)^{\frac{1}{2}} u \|_{L_2(B)}^2 \le \| (1 - \Delta_g^D)^{\frac{1}{2}} u \|_{L_2(B)}^2, \quad u \in H_0^1(B).$$

By standard results in quadratic form theory (see e.g. [Sim15, Equation 7.5.29]), $1 - \Delta_g^D$ defines a topological linear isomorphism from $H_0^1(B)$ to its dual $(H_0^1(B))^*$. Therefore, replacing u with $(1 - \Delta_g^D)^{-1}v$ for $v \in (H_0^1)^*$, we arrive at

$$c_g \| (1 - \Delta_0^D)^{\frac{1}{2}} (1 - \Delta_g^D)^{-1} v \|_{L_2(B)}^2 \le \| (1 - \Delta_g^D)^{-\frac{1}{2}} v \|_{L_2(B)}^2, \quad v \in (H_0^1(B))^*.$$

Replacing v with $(1 - \Delta_0^D)^{\frac{1}{2}} w$ for $w \in L_2(B)$ gives

$$\|(1-\Delta_0^D)^{\frac{1}{2}}(1-\Delta_g^D)^{-1}(1-\Delta_0)^{\frac{1}{2}}w\|_{L_2(B)}^2 \le c_g^{-1}\|w\|_{L_2(B)}^2$$

Recall that $\|(1-\Delta_0^D)^{-\frac{1}{2}}\|_{\mathcal{L}_{d,\infty}(L_2(B))} < \infty$ by standard Weyl asymptotics. Therefore,

$$\begin{split} \|(1-\Delta_g^D)^{-1}\|_{\mathcal{L}_{\frac{d}{2},\infty}(L_2(B))} &\leq \|(1-\Delta_0^D)^{-\frac{1}{2}}\|_{\mathcal{L}_{d,\infty}(L_2(B))}^2 \|(1-\Delta_0^D)^{\frac{1}{2}}(1-\Delta_g^D)^{-1}(1-\Delta_0)^{\frac{1}{2}}\|_{B(L_2(B))} \\ &\leq c_g^{-\frac{1}{2}}\|(1-\Delta_0^D)^{-\frac{1}{2}}\|_{\mathcal{L}_{d,\infty}(L_2(B))}^2. \end{split}$$

Defining $k_d = \|(1 - \Delta_0^D)^{-\frac{1}{2}}\|_{\mathcal{L}_{d,\infty}(L_2(B))}^2$ completes the proof of (6.12), and hence of the Lemma.

Proposition 6.3.6. For all q > 0, there exists K > 0 such that

$$\sup_{j} \|M_{\psi_j}(1-\Delta)^{-K}\|_{q,\infty} < \infty,$$

where ψ_j is the partition of unity subordinate to $\{B(x_j, r_0)\}_{j \in \mathbb{N}}$ mentioned in Section 6.1.2.

Proof. The proof is inspired by the proof of [SZ23, Lemma 3.4.8]. Let $V_j : L_2(M) \to L_2(B, g^{x_j})$ be the partial isometry mapping $\xi \in L_2(M)$ to $V\xi(z) = f|_{B(x_j, r_0)} \circ \exp_x \circ e \in L_2(B, g^{x_j})$. Denote $V_j \psi_j := \phi_j$. Then, by construction,

$$M_{\psi_j} = V_j^* M_{\phi_j} V_j$$

and

$$(1 - \Delta)^{K} M_{\psi_{j}} = V_{j}^{*} (1 - \Delta_{g^{x_{j}}})^{K} M_{\phi_{j}} V_{j}.$$

It follows that

$$(1-\Delta)^{K} M_{\psi_{j}} V_{j}^{*} (1-\Delta_{g^{x_{j}}}^{D})^{-K} V_{j} = V_{j}^{*} (1-\Delta_{g^{x_{j}}}^{x_{j}})^{K} M_{\phi_{j}} (1-\Delta_{g^{x_{j}}}^{D})^{-K} V_{j}$$
$$= V_{j}^{*} [(1-\Delta_{g^{x_{j}}})^{K}, M_{\phi_{j}}] (1-\Delta_{g^{x_{j}}}^{D})^{-K} V_{j} + V_{j}^{*} M_{\phi_{j}} V_{j}$$
$$= [(1-\Delta)^{K}, M_{\psi_{j}}] V_{j}^{*} (1-\Delta_{g^{x_{j}}}^{D})^{-K} V_{j} + M_{\psi_{j}}.$$

Multiplying both sides by $(1 - \Delta)^{-K}$ and rearranging gives

$$(1-\Delta)^{-K}M_{\psi_j} = M_{\psi_j}V_j^*(1-\Delta_{g^{x_j}}^D)^{-K}V_j - (1-\Delta)^{-K}[(1-\Delta)^K, M_{\psi_j}]V_j^*(1-\Delta_{g^{x_j}}^D)^{-K}V_j.$$

We claim that

$$\sup_{j\in\mathbb{N}}\|(1-\Delta)^{-K}[(1-\Delta)^K, M_{\psi_j}]\|_{\infty}<\infty.$$

For every α we have $\sup_{j,x} |\partial^{\alpha}\psi_j(x)| < \infty$, where ∂^{α} is taken in the exponential normal coordinates of $B(x_j, r_0)$, and therefore $[(1 - \Delta)^K, M_{\psi_j}]$ is a uniform differential operator of order 2K - 1 with coefficients that are uniform in j. Using [Kor91, Theorem 3.9], it follows that $[(1 - \Delta)^K, M_{\psi_j}]$ is a bounded operator from $L_2(M)$ to the Sobolev space $\mathcal{H}^{1-2K}(M)$ with norm uniform in j. By [Kor91, Proposition 4.4], $(1 - \Delta)^{-K}$ is a bounded operator from that space into $L_2(M)$, and so the claim holds.

Since the norm of V_j is equal to one, via Lemma 6.3.5 we get for K large enough

$$\sup_{j \in \mathbb{N}} \| (1-\Delta)^{-K} M_{\psi_j} \|_{q,\infty} \leq \sup_{j \in \mathbb{N}} \left(\| M_{\psi_j} \|_{\infty} \cdot \| (1-\Delta_g^{x_j})^{-K} \|_{q,\infty} \right) \\
+ \sup_{j \in \mathbb{N}} \left(\| (1-\Delta)^{-K} [(1-\Delta)^K, M_{\psi_j}] \|_{\infty} \cdot \| (1-\Delta_g^{x_j})^{-K} \|_{q,\infty} \right) \\
< \infty. \qquad \Box$$

It follows from this proposition that for every q > 0 and every j, there exists K > 0 and a constant C_K independent of j such that

$$\|M_{f\psi_j}(1-\Delta)^{-K}\|_{q,\infty} \le C_K \|f\|_{L_{\infty}(B(x_j,r_0))}.$$
(6.13)

Let $\{\psi_j^{(0)}\}_{j=0}^{\infty}, \{\psi_j^{(2)}\}_{j=0}^{\infty}, \dots, \{\psi_j^{(N)}\}_{j=0}^{\infty}$ be a partition of $\{\psi_j\}_{j=0}^{\infty}$ into disjointly supported subfamilies. That is, for all $0 \le k \le N$, the functions $\{\psi_j^{(k)}\}_{j=0}^{\infty}$ are disjointly supported, and for every $j \ge 0$ there exists a unique $0 \le k \le N$ such that $\psi_j \in \{\psi_l^{(k)}\}_{l=0}^{\infty}$.

Theorem 6.3.7. Let 0 . For sufficiently large K, we have

$$||M_f(1-\Delta)^{-K}||_{p,\infty} \le C_{p,K,N} ||f||_{\ell_{p,\infty}(L_{\infty})};$$
$$||M_f(1-\Delta)^{-K}||_p \le C'_{p,K,N} ||f||_{\ell_p(L_{\infty})}.$$

Proof. We prove the first inequality, the second can be proved analogously. Let $f \in \ell_{p,\infty}(L_{\infty})$. Since $\{\psi_j\}_{j=1}^{\infty}$ is a partition of unity, we have

$$f = \sum_{j=0}^{\infty} \psi_j f = \sum_{k=0}^{N} \sum_{j=0}^{\infty} \psi_j^{(k)} f.$$

By the quasi-triangle inequality, there exists $C_{N,p}$ such that

$$\|M_f (1-\Delta)^{-K}\|_{p,\infty} \le C_{N,p} \sum_{k=0}^N \left\| \sum_{j=0}^\infty M_{\psi_j^{(k)} f} (1-\Delta)^{-K} \right\|_{p,\infty}$$

The operators $\{M_{\psi_j^{(k)}f}(1-\Delta)^{-K}\}_{j=0}^{\infty}$ are left-disjoint. Hence, Corollary 6.1.2 implies that there exists q > 0 such that

$$\|M_f(1-\Delta)^{-K}\|_{p,\infty} \le C_{N,p,q} \sum_{k=0}^N \left\| \{ \|M_{f\psi_j^{(k)}}(1-\Delta)^{-K}\|_{q,\infty} \}_{j=0}^\infty \right\|_{\ell_{p,\infty}}$$

From (6.13), it follows that if K is sufficiently large (depending on q), we have

$$\|M_f(1-\Delta)^{-K}\|_{p,\infty} \le C_{N,p,q} \left\| \{\|f\|_{L_{\infty}(B(x_j,r_0))}\}_{j=0}^{\infty} \right\|_{\ell_{p,\infty}}$$

The latter is the definition of the $\ell_{p,\infty}(L_{\infty})$ quasinorm.

Corollary 6.3.8. Let 0 . Let <math>P be a self-adjoint, lower-bounded $P \in EBD^m(M)$. We have $\exp(-tP)M_f \in \mathcal{L}_{p,\infty}$ whenever $f \in \ell_{p,\infty}(L_\infty)$, and $\exp(-tP)M_f \in \mathcal{L}_p$ whenever $f \in \ell_p(L_\infty)$.

Proof. By the preceding theorem, we already have that for $f \in \ell_{p,\infty}(L_{\infty})$ and sufficiently large K, $M_f(1-\Delta)^{-K} \in \mathcal{L}_{p,\infty}$. Noting that P+C > 0 and is therefore an invertible operator on $L_2(M)$ for some $C \in \mathbb{R}$, by Proposition 4.4 in [Kor91] it follows that $(P+C)^{-1}$ maps $\mathcal{H}^s(M)$ boundedly into $\mathcal{H}^{s+m}(M)$. By Theorem 3.9 in [Kor91], $(1-\Delta)^K$ maps $\mathcal{H}^s(M)$ boundedly into $\mathcal{H}^{s-2K}(M)$. Therefore, we can find $N \in \mathbb{N}$ large enough such that $(P+C)^{-N}(1-\Delta)^K$ is a bounded operator on $L_2(M)$. Noting that $\exp(-tP)(P+C)^N$ is a bounded operator for any N by the functional calculus on unbounded operators, the claim follows. The case for $f \in \ell_p(L_\infty)$ is proven analogously.

This corollary will eventually make it possible to apply Theorem 6.0.4 on P and $W = M_w$ for $w \in \ell_{1,\infty}(L_\infty)$ defined by $w(x) = (1 + |B(x_0, d_g(x, x_0))|)^{-1}$. We will also need to show that $\exp(-tP)[P, M_w] \in \mathcal{L}_1$, but this will require geometric conditions on the manifold M.

Lemma 6.3.9. Let (M,g) be a complete connected Riemannian manifold of bounded geometry. Then for any fixed $r \in \mathbb{R}$,

$$\liminf_{R \to \infty} \frac{|B(x_0, R - r)|}{|B(x_0, R + r)|} > 0.$$

Proof. The paper [GP11] proves that for manifolds of bounded geometry, we have for fixed $R \ge 1$,

$$\frac{1}{L} \le |B(x_0, R+2)| - |B(x_0, R+1)| \le L(|B(x_0, R+1)| - |B(x_0, R)|)$$

for some constant L > 0 independent of R.

Now with this inequality, one can show by induction that

$$\frac{B(x_0, R+k)|}{|B(x_0, R)|} \le (1+L)^k, \quad R \ge 1.$$

For k = 1,

$$\frac{|B(x_0, R+1)|}{|B(x_0, R)|} \le 1 + L \frac{|B(x_0, R)| - |B(x_0, R-1)|}{|B(x_0, R)|} \le 1 + L.$$

Suppose that $\frac{|B(x_0,R+k)|}{|B(x_0,R)|} \leq (1+L)^k$ for some k, then

$$\frac{|B(x_0, R+k+1)|}{|B(x_0, R)|} \le \frac{|B(x_0, R+k)|}{|B(x_0, R)|} + L \frac{|B(x_0, R+k)| - |B(x_0, R+k-1)|}{|B(x_0, R)|} \le (1+L)^k + L(1+L)^k = (1+L)^{k+1}.$$

Therefore,

$$\liminf_{R \to \infty} \frac{|B(x_0, R - r)|}{|B(x_0, R + r)|} = \liminf_{R \to \infty} \frac{|B(x_0, R)|}{|B(x_0, R + 2r)|} > \frac{1}{(1+L)^K},$$

where K is some integer greater than 2r.

Lemma 6.3.10. Let (M,g) be a complete connected Riemannian manifold of bounded geometry. Then the function

$$w(x) = (1 + |B(x_0, d_g(x, x_0))|)^{-1}, \quad x \in M,$$

is an element of $\ell_{1,\infty}(L_{\infty})(M)$.

Proof. We denote $d_g(x, x_0)$ by r(x). Note that for any $x \in B(x_k, r_0)$ we have $r(x) \ge |x_k| - r_0$ by the triangle inequality, and hence

$$||w||_{L_{\infty}(B(x_k,r_0))} \le (1+|B(x_0,r(x_k)-r_0)|)^{-1}.$$

Without loss of generality, order the x_j such that $r(x_1) \leq r(x_2) \leq \ldots$. Note that as in Remark 6.1.8, we have that $\inf_{j \in \mathbb{N}} |B(x_j, r_0)| > 0$ and hence

$$|B(x_0, |x_k| - r_0)| = \inf_{j \in \mathbb{N}} |B(x_j, r_0)| \cdot \frac{|B(x_0, |x_k| - r_0)|}{|B(x_0, |x_k| + r_0)|} \cdot \frac{|B(x_0, |x_k| + r_0)|}{\inf_{j \in \mathbb{N}} |B(x_j, r_0)|}.$$

By the ordering of the x_j , we know that all the balls $B(x_j, r_0)$ from j = 1 up to and including j = k are contained in $B(x_0, |x_k| + r_0)$. Hence

$$k \cdot \inf_{m \in \mathbb{N}} |B(x_m, r_0)| \le \sum_{j=1}^k |B(x_j, k)| \le (N+1) |B(x_0, |x_k| + r_0)|$$

since any ball can only intersect at most N other balls. We thus get

$$|B(x_0, |x_k| + r_0)| \ge \frac{k}{N+1} \inf_{j \in \mathbb{N}} |B(x_j, r_0)|$$

We will now show that for k large enough, $\frac{|B(x_0, |x_k| - r_0)|}{|B(x_0, |x_k| + r_0)|}$ is bounded below away from zero. By Lemma 6.3.9 we have

$$\liminf_{R \to \infty} \frac{|B(x_0, R - r_0)|}{|B(x_0, R + r_0)|} > 0,$$

and so there must be some R_0 such that for $R \ge R_0$ we have $\frac{|B(x_0, R-r_0)|}{|B(x_0, R+r_0)|} > \delta > 0$. We claim that we can take K large enough such that $|x_k| \ge R_0$ for $k \ge K$. Indeed, by analogous reasoning as before, at most $K := (N+1)\frac{|B(x_0, R_0+r_0)|}{\inf_j |B(x_j, r_0)|}$ points x_j can be inside the ball $|B(x_0, R_0)|$, thus $|x_k| > R_0$ for k > K.

Gathering the results above, we get the existence of some constant C such that for $k \ge K$ we have

$$|B(x_0, |x_k| - r_0)| \ge Ck.$$

This means that

$$||w||_{L_{\infty}(B(x_k, r_0))} \le (1 + Ck)^{-1},$$

proving that $w \in \ell_{1,\infty}(L_{\infty})(M)$.

We will now mold Property (D) into the form that we will apply it in.

Lemma 6.3.11. Let (M, g) be a complete connected Riemannian manifold of bounded geometry. If M has Property (D), that is if $V(r) := |B(x_0, r)| \in C^2(\mathbb{R}_{\geq 0})$ satisfies

$$\frac{V'(r)}{V(r)} \in L_2(\mathbb{R}_{\ge 1}),$$
 (6.14)

and

$$\frac{V''(r)}{V'(r)} \to 0, \quad r \to \infty, \tag{6.15}$$

then for every h > 0,

$$\left\{\frac{\sup_{s\in[0,h]}|\partial B(x_0,k+1+s)|}{|B(x_0,k+1)|}\right\}_{k\in\mathbb{N}}\in\ell_2(\mathbb{N}).$$
(6.16)

Proof. Using condition (6.15), let $\varepsilon > 0$ and choose R large enough so that $r \ge R$ implies that

$$V''(r) \le \varepsilon V'(r).$$

For $r \geq R$, $\delta > 0$, we have

$$V'(r+\delta) = V'(r) + \int_{r}^{r+\delta} V''(s) \, ds$$
$$\leq V'(r) + \varepsilon \delta \sup_{s \in [0,\delta]} V'(r+s).$$

Taking the supremum over $\delta \in [0, h]$ on both sides and rearranging gives

$$(1 - \varepsilon h) \sup_{s \in [0,h]} V'(r+s) \le V'(r).$$

In particular, this implies that

$$\inf_{s\in[0,h]} V'(k+s) \ge \inf_{s\in[0,h]} (1-\varepsilon h) \sup_{t\in[0,h]} V'(k+s+t) \ge (1-\varepsilon h) V'(k+h).$$

Now choose $\varepsilon < \min(1, \frac{1}{h})$, and we can estimate the L_2 -norm of $\frac{V'(r)}{V(r)}$ from below by a Riemann sum,

$$\begin{split} \int_{1}^{\infty} \left(\frac{V'(r)}{V(r)}\right)^{2} dr &\geq \sum_{k=1}^{\infty} \frac{\left(\inf_{s \in [0,1]} V'(k+s)\right)^{2}}{\left(\sup_{s \in [0,1]} V(k+s)\right)^{2}} \\ &\geq C + \sum_{k=1}^{\infty} (1-\varepsilon)^{2} \frac{V'(k+1)^{2}}{V(k+1)^{2}} \\ &\geq C + (1-\varepsilon)^{2} (1-\varepsilon h)^{2} \sum_{k=1}^{\infty} \frac{\sup_{s \in [0,h]} V'(k+1+h)^{2}}{V(k+1)^{2}}. \end{split}$$

This shows that (6.14) and (6.15) together imply (6.16)

Lemma 6.3.12. Let (M,g) be a non-compact Riemannian manifold with bounded geometry and Property (D). Then

$$\sum_{k=0}^{\infty} \frac{\sup_{s \in [-1,2]} \left| \partial B(x_0, (k+s)r_0) \right|^2}{(1+|B(x_0, (k-1)r_0)|)^2} < \infty.$$
(6.17)

Proof. Since M has Property (D), Lemma 6.3.11 gives that

$$\sum_{k=1}^{\infty} \frac{\sup_{s \in [0,3]} |\partial B(x_0, k+s)|^2}{|B(x_0, k)|^2} < \infty.$$

Recall that we can assume $r_0 \leq 1$. Then,

$$\sum_{k=0}^{\infty} \frac{\sup_{s \in [-1,2]} |\partial B(x_0, (k+s)r_0)|^2}{(1+|B(x_0, (k-1)r_0)|)^2}$$

$$= \sum_{N=0}^{\infty} \sum_{N \leq kr_0 \leq N+1} \frac{\sup_{s \in [-1,2]} |\partial B(x_0, (k+s)r_0)|^2}{(1+|B(x_0, (k-1)r_0)|)^2}$$

$$\leq \sum_{N=0}^{\infty} \sum_{N \leq kr_0 \leq N+1} \frac{\sup_{s \in [-1,2]} |\partial B(x_0, N+s)|^2}{(1+|B(x_0, N-1)|)^2}$$

$$\leq \left\lceil \frac{1}{r_0} \right\rceil \left(2 \cdot \sup_{s \in [0,3]} |\partial B(x_0, s)|^2 + \sum_{k=1}^{\infty} \frac{\sup_{s \in [0,3]} |\partial B(x_0, k+s)|^2}{|B(x_0, k)|^2} \right)$$

$$< \infty.$$

Note that in the next lemma, we do not need uniform ellipticity of the differential operators considered, although we will consider operators $L \in BD^2(M)$ which lack a constant term. The meaning of this condition is that in a system of normal coordinates (U, ϕ) , where $U = B(x, r_0)$ we have

$$L = \sum_{0 < |\alpha| \le 2} a_{\alpha,x}(y) D^{\alpha}.$$

Equivalently, L1 = 0. The important feature of these operators is that if $P \in BD^2(M)$ and f is a smooth function with uniformly bounded derivatives, then $[P, M_f] = [L, M_f]$ for a differential operator $L \in BD^2(M)$ with no constant term.

Lemma 6.3.13. Let (M,g) be a non-compact Riemannian manifold with bounded geometry satisfying Property (D) (Definition 6.0.1), and take $w : M \to \mathbb{R}$ as in Lemma 6.3.10. Then $Lw \in \ell_1(L_\infty)(M)$ for any $L \in BD^2(M)$ that lacks a constant term.

Proof. For any $x \in M$ we can take a neighbourhood of normal coordinates (U, ϕ) in which L takes the following form by definition:

$$\sum_{|\alpha|=1,2} a_{\alpha,x}(y) \partial_y^{\alpha},$$

where for any multi-index β

$$\left|\partial_y^\beta a_{\alpha,x}(0)\right| \le C_{\alpha,\beta}, \quad x \in M.$$

Now denote $r(x) := d_g(x_0, x)$, $\tilde{w}(r) = (1 + |B(x_0, r)|)^{-1}$ so that $w = \tilde{w} \circ r$. Note that \tilde{w} is a C^2 function on $[0, \infty)$ by assumption (Property (D)), and the combination of [Sak96, Lemma III.4.4] and [Sak96, Proposition III.4.8] gives that r is smooth almost everywhere with $\|\nabla r\| \leq 1$ almost everywhere. We therefore have almost everywhere

$$\begin{split} |Lw(x)| &\leq \bigg| \sum_{|\alpha|=1} a_{\alpha,x}(0) (\partial^{\alpha} (\tilde{w} \circ r \circ \phi^{-1}))(0) \bigg| + \bigg| \sum_{\substack{|\alpha|=2\\ \alpha=\beta+\gamma\\ |\beta|=|\gamma|=1}} a_{\alpha,x}(0) (\partial^{\alpha} (r \circ \phi^{-1}))(0) \tilde{w}'(r(x))) \bigg| \\ &= \bigg| \sum_{\substack{|\alpha|=2\\ \alpha=\beta+\gamma\\ |\beta|=|\gamma|=1}} a_{\alpha,x}(0) \Big(\tilde{w}''(r(x)) \cdot \partial^{\beta} (r \circ \phi^{-1})(0) \cdot \partial^{\gamma} (r \circ \phi^{-1})(0) \\ &+ \tilde{w}'(r(x)) \cdot \partial^{\beta} \partial^{\gamma} (r \circ \phi^{-1})(0) \Big) \bigg| \\ &\leq \bigg| \sum_{\substack{|\alpha|=2\\ \alpha=\beta+\gamma\\ |\beta|=|\gamma|=1}} a_{\alpha,x}(0) \bigg| \cdot |\tilde{w}'(r(x))| + \bigg| \sum_{\substack{|\alpha|=2\\ \alpha=\beta+\gamma\\ |\beta|=|\gamma|=1}} a_{\alpha,x}(0) \bigg| \cdot |\tilde{w}'(r(x))| \\ &+ C \bigg| \sum_{\substack{|\alpha|=2\\ \alpha=\beta+\gamma\\ |\beta|=|\gamma|=1}} a_{\alpha,x}(0) \bigg| \cdot |\tilde{w}'(r(x))| \\ &\leq (1+C) \cdot \Big(|\tilde{w}'(r(x))| + |\tilde{w}''(r(x))| \Big), \end{split}$$

where we have used the chain rule and

$$\left| \left(\partial^{\alpha} (r \circ \phi^{-1}) \right)(0) \right| = \left| \frac{\partial}{\partial x^{\alpha}} \right|_{x}(r) \right| \le \|\nabla r(x)\| \le 1$$

(because $(\nabla r)_{\alpha} = \sum_{\beta} g_{\alpha\beta} \partial^{\beta} r$ and in normal coordinates at zero, $g_{\alpha,\beta}(0) = \delta_{\alpha,\beta}$) in addition to

$$|\partial^{\beta}\partial^{\gamma}(r \circ \phi^{-1})(0)| \le \|\text{Hess } r\| \le C$$

for some constant C, since the Hessian of r is uniformly bounded [Pet06, Theorem 6.5.27].

By continuity, we therefore have everywhere

$$|Lw(x)| \le (1+C) \cdot (|\tilde{w}'(r(x))| + |\tilde{w}''(r(x))|).$$

Therefore

$$\begin{split} \|Lw\|_{L_{\infty}(B(x_{j},r_{0}))} &\leq (1+C) \cdot \Big(\sup_{s \in [-r_{0},r_{0}]} |\tilde{w}'(r(x_{j})+s)| + |\tilde{w}''(r(x_{j})+s)|\Big) \\ &\leq (1+C) \cdot \left(\sup_{s \in [-r_{0},r_{0}]} \frac{|\partial B(x_{0},r(x_{j})+s)|}{(1+|B(x_{0},r(x_{j})+s)|)^{2}} \right. \\ &+ \sup_{s \in [-r_{0},r_{0}]} \frac{\frac{d}{dR}\Big|_{R=r(x_{j})+s}}{(1+|B(x_{0},r(x_{j})+s)|)^{2}} \\ &+ 2\sup_{s \in [-r_{0},r_{0}]} \frac{|\partial B(x_{0},r(x_{j})+s)|^{2}}{(1+|B(x_{0},r(x_{j})+s)|)^{3}} \Big). \end{split}$$

Next, we calculate

$$\sum_{j \in \mathbb{N}} \sup_{s \in [-r_0, r_0]} \frac{|\partial B(x_0, r(x_j) + s)|}{(1 + |B(x_0, r(x_j) + s)|)^2}$$

=
$$\sum_{k=0}^{\infty} \sum_{\{j: kr_0 \le r(x_j) < (k+1)r_0\}} \sup_{s \in [-r_0, r_0]} \frac{|\partial B(x_0, r(x_j) + s)|}{(1 + |B(x_0, r(x_j) + s)|)^2}$$

$$\le \sum_{k=0}^{\infty} |\{j: kr_0 \le r(x_j) < (k+1)r_0\}| \cdot \sup_{l \in [-1,2]} \frac{|\partial B(x_0, (k+l)r_0|}{(1 + |B(x_0, (k+l)r_0)|)^2}$$

With a similar calculation as in the proof of Lemma 6.3.10, we have

$$\begin{split} \inf_{m \in \mathbb{N}} |B(x_m, r_0)| \cdot |\{j : kr_0 \le r(x_j) < (k+1)r_0\}| \\ &= \sum_{\{j : kr_0 \le r(x_j) < (k+1)r_0\}} \inf_{m \in \mathbb{N}} |B(x_m, r_0)| \\ &\le \sum_{\{j : kr_0 \le r(x_j) < (k+1)r_0\}} |B(x_j, r_0)| \\ &\le (N+1) \Big(|B(x_0, (k+2)r_0)| - |B(x_0, (k-1)r_0)| \Big) \\ &\le 3(N+1)r_0 \sup_{l \in [-1,2]} |\partial B(x_0, (k+l)r_0)|, \end{split}$$

since all the balls $B(x_j, r_0)$ with $kr_0 \leq r(x_j) \leq (k+1)r_0$ are contained in the annulus $B(x_0, (k+2)r_0) \setminus B(x_0, (k-1)r_0)$, and balls can intersect at most N other balls.

Using Lemma 6.3.12 we can infer that the expression (6.17) is finite, and so

$$\sum_{j \in \mathbb{N}} \sup_{s \in [-r_0, r_0]} \frac{|\partial B(x_0, r(x_j) + s)|}{(1 + |B(x_0, r(x_j) + s)|)^2} \le C' \sum_{k=0}^{\infty} \frac{\sup_{l \in [-1,2]} |\partial B(x_0, (k+l)r_0|^2}{(1 + |B(x_0, (k-1)r_0)|)^2} < \infty.$$

With analogous calculations, we also have that

$$\sum_{j \in \mathbb{N}} \sup_{s \in [-r_0, r_0]} \frac{\frac{d}{dR} \Big|_{R=r(x_j)+s} |\partial B(x_0, R)|}{(1+|B(x_0, r(x_j)+s)|)^2} < \infty$$

and

$$\sum_{j \in \mathbb{N}} \sup_{s \in [-r_0, r_0]} \frac{|\partial B(x_0, r(x_j) + s)|^2}{(1 + |B(x_0, r(x_j) + s)|)^3} < \infty.$$

We conclude that

$$||Lw||_{\ell_1(L_\infty)} = \sum_{j \in \mathbb{N}} ||Lw||_{L_\infty(B(x_j, r_0))} < \infty.$$

Corollary 6.3.14. Let $P \in EBD^2(M)$ be a self-adjoint lower-bounded operator. Let M be a non-compact Riemannian manifold with Property (D), and take $w : M \to \mathbb{R}$ as in Lemma 6.3.10. Then $\exp(-tP)[P, M_w] \in \mathcal{L}_1$.

Proof. Take $P \in EBD^2(M)$. Similarly to the proof of Corollary 6.3.8, since $\exp(-tP)(1 - \Delta)^K$ is a bounded operator for all K, it suffices to prove that $(1 - \Delta)^{-K-1}[P, M_w] \in \mathcal{L}_1$ for K large enough.

First note that

$$\|(1-\Delta)^{-K-1}[P, M_w]\|_1 \le \sum_{j \in \mathbb{N}} \|(1-\Delta)^{-K-1}[P, M_w]M_{\psi_j}\|_1$$

Using the expression $P = \sum_{|\alpha| \leq 2} a_{\alpha,x_j}(y) \partial_y^{\alpha}$ in the coordinate chart B_j , we have for $K \geq 0$

$$\begin{split} \|(1-\Delta)^{-K-1}[P, M_w] M_{\psi_j}\|_{1} &\leq \|(1-\Delta)^{-K-1} \sum_{|\alpha| \leq 2} [M_{a_{\alpha, x_j}} \partial^{\alpha}, M_w] M_{\psi_j}\|_{1} \\ &\leq \|(1-\Delta)^{-K-1} M_{\psi_j} M_{Dw}\|_{1} \\ &+ \left\| (1-\Delta)^{-K-1} \sum_{\substack{|\alpha| = 2\\ \alpha = \beta + \gamma\\ |\beta| = |\gamma| = 1}} M_{a_{\alpha, x_j}} (\partial^{\gamma} M_{\partial^{\beta} w} + \partial^{\beta} M_{\partial^{\gamma} w}) M_{\psi_j} \right\|_{1} \\ &\leq \| (1-\Delta)^{-K-1} M_{\psi_j} M_{Dw} \|_{1} \\ &+ \left\| (1-\Delta)^{-K-1} \sum_{|\beta| = 1} \sum_{|\gamma| = 1} M_{a_{\beta+\gamma, x_j}} \partial^{\gamma} M_{\partial^{\beta} w} M_{\psi_j} \right\|_{1} \\ &+ \left\| (1-\Delta)^{-K-1} \sum_{|\beta| = 1} M_{a_{2\beta, x_j}} \partial^{\beta} M_{\partial^{\beta} w} M_{\psi_j} \right\|_{1} \\ &=: I + II + III, \end{split}$$

where we have denoted D for the differential operator given near x_j by

$$D = \sum_{1 \le |\alpha| \le 2} a_{\alpha, x_j}(y) \partial_y^{\alpha}.$$

That is, D is equal to P without constant terms. Since ψ_j is by definition supported in B_j , we have

$$I = \|(1 - \Delta)^{-K - 1} M_{\psi_j} M_{Dw}\|_1$$

= $\|(1 - \Delta)^{-K - 1} M_{\psi_j} M_{\chi_{B_j}} M_{Dw}\|_1$
 $\leq \|(1 - \Delta)^{-K - 1} M_{\psi_j}\|_1 \|M_{\chi_{B_j}} M_{Dw}\|_\infty$
= $\|(1 - \Delta)^{-K - 1} M_{\psi_j}\|_1 \|Dw\|_{L_\infty(B_j)}.$

By Proposition 6.3.6, $\|(1-\Delta)^{-K-1}M_{\psi_j}\|_1$ is uniformly bounded in j for K large enough, and by Lemma 6.3.13, $\{\|Dw\|_{L_{\infty}(B_j)}\}_{j\in\mathbb{N}} \in \ell_1$.

$$\begin{split} II &= \| (1-\Delta)^{-K-1} \sum_{|\beta|=1} \sum_{|\gamma|=1} M_{a_{\beta+\gamma,x}} \partial^{\gamma} M_{\partial^{\beta}w} M_{\psi_{j}} \|_{1} \\ &\leq \| \sum_{|\beta|=1} \sum_{|\gamma|=1} (1-\Delta)^{-K-1} [M_{a_{\beta+\gamma,x}}, \partial^{\gamma}] M_{\chi_{B_{j}}} M_{\partial^{\beta}w} M_{\psi_{j}} \|_{1} \\ &+ \| \sum_{|\beta|=1} \sum_{|\gamma|=1} (1-\Delta)^{-K-1} \partial^{\gamma} M_{\chi_{B_{j}}} M_{a_{\beta+\gamma,x}} M_{\partial^{\beta}w} M_{\psi_{j}} \|_{1} \\ &= \| \sum_{|\beta|=1} \sum_{|\gamma|=1} (1-\Delta)^{-K-1} M_{\partial^{\gamma} a_{\beta+\gamma,x}} M_{\chi_{B_{j}}} M_{\partial^{\beta}w} M_{\psi_{j}} \|_{1} \\ &+ \| \sum_{|\beta|=1} \sum_{|\gamma|=1} (1-\Delta)^{-K-1} \partial^{\gamma} M_{\chi_{B_{j}}} M_{a_{\beta+\gamma,x}} M_{\partial^{\beta}w} M_{\psi_{j}} \|_{1} \\ &\leq \sum_{|\beta|=1} \sum_{|\gamma|=1} \| (1-\Delta)^{-K-1} M_{\psi_{j}} \|_{1} \| \partial^{\gamma} a_{\beta+\gamma,x} \|_{L_{\infty}(B_{j})} \| \partial^{\beta}w \|_{L_{\infty}(B_{j})} \\ &+ \sum_{|\beta|=1} \sum_{|\gamma|=1} \| (1-\Delta)^{-1} \partial^{\gamma} M_{\chi_{B_{j}}} \|_{\infty} \| (1-\Delta)^{-K} M_{\psi_{j}} \|_{1} \| a_{\beta+\gamma,x} \|_{L_{\infty}(B_{j})} \| \partial^{\beta}w \|_{L_{\infty}(B_{j})}. \end{split}$$

Suppose y is in the normal neighbourhood (U_x, ϕ_x) of x. Then, D being a local operator, at y (taking $\tilde{y} = \phi_x(y)$), there are two different expressions of D in normal coordinates:

$$D = \sum_{|\alpha| \le 2} a_{\alpha,x}(\tilde{y}) \partial_x^{\alpha}$$
$$= \sum_{|\beta| \le 2} a_{\beta,y}(0) \partial_y^{\beta}.$$

Hence we can express $a_{\alpha,x}(\tilde{y})$ in terms of $a_{\beta,y}(0)$ and transition functions $\frac{\partial^{\alpha} y}{\partial x^{\alpha}}$. These are uniformly bounded by the definition of a differential operator with bounded coefficients and [Kor91, Proposition 1.3]. Therefore $||a_{\beta+\gamma,x}||_{L_{\infty}(B_j)}$ and also $||\partial^{\gamma}a_{\beta+\gamma,x}||_{L_{\infty}(B_j)}$ are uniformly bounded in j. Likewise (taking $|\beta| = 1$),

$$\begin{aligned} |\partial_x^\beta w(y)| &= \Big| \sum_{|\alpha|=1} \frac{\partial^\alpha y}{\partial x^\alpha} \partial_y^\alpha w(y) \\ &\leq C_d \|\nabla r\| |\tilde{w}'(y)|, \end{aligned}$$

and hence $\{\|\partial^{\beta}w\|_{L_{\infty}(B_j)}\}_{j\in\mathbb{N}} \in \ell_1$ by the arguments in the proof of Lemma 6.3.13.

Finally,

$$\|(1-\Delta)^{-1}\partial^{\gamma}M_{\chi_{B_j}}\|_{\infty} \leq \sup_{|\alpha|=1} \|\frac{\partial^{\alpha}y}{\partial x^{\alpha}}\|_{L_{\infty}(B_j)}\|(1-\Delta)^{-1}|\nabla|\|_{\infty} < \infty,$$

uniformly in j. The same estimates hold for III.

Combining everything, we have

$$\|(1-\Delta)^{-K-1}[P, M_w]\|_1 \le \sum_{j \in \mathbb{N}} \|(1-\Delta)^{-K-1}[P, M_w]M_{\psi_j}\|_1 < \infty.$$

Gathering all results in this section, let M be a non-compact manifold of bounded geometry with Property (D). Let $P \in EBD^2(M)$ be self-adjoint and lower-bounded. Then $\exp(-tP)M_w \in \mathcal{L}_{1,\infty}$ by the Cwikel estimate in Corollary 6.3.8 and Lemma 6.3.10. Corollary 6.3.14 states that $\exp(-tP)[P, M_w] \in \mathcal{L}_1$. Theorem 6.0.4 then gives that

$$\operatorname{Tr}_{\omega}(e^{-tP}M_w) = \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}(e^{-tP}\chi_{[\varepsilon,\infty)}(M_w)),$$

if the limit on the right hand side exists. If we assume that P admits a density of states, we do in fact get the existence of the limit

$$\int_{\mathbb{R}} e^{-t\lambda} d\nu_P(\lambda) = \lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(e^{-tP} M_{\chi_{B(x_0, R)}})$$
$$= \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}(e^{-tP} \chi_{[\varepsilon, \infty)}(M_w)).$$

Note that the above calculation assumes that the volume of M is infinite, which is equivalent with M being non-compact (see Remark 6.1.8).

Hence,

$$\operatorname{Tr}_{\omega}(e^{-tP}M_w) = \int_{\mathbb{R}} e^{-t\lambda} d\nu_P(\lambda), \quad t > 0.$$

From this we can easily deduce by a density argument that

$$\operatorname{Tr}_{\omega}(f(P)M_w) = \int_{\mathbb{R}} f \, d\nu_P, \quad f \in C_c(\mathbb{R}).$$

For details, see [AMSZ20, Remark 6.3]. This concludes the proof of Theorem 6.0.3.

6.4 Roe's index theorem

The results of the preceding section were stated for operators P acting on scalar valued functions on M. Identical results, with the same proofs, apply to operators acting between sections of vector bundles of bounded geometry. In the terminology of Shubin [Shu92], a rank N vector bundle $\pi : S \to M$ is said to have bounded geometry if in every coordinate chart $\{B(x_j, r_0)\}_{j=0}^{\infty}$ (as defined in Section 6.1.2) E has a trivialisation

$$\pi^{-1}(B(x_j, r_0)) \approx B(x_j, r_0) \times \mathbb{R}^N$$

such that the transition functions

$$t_{j,k}: B(x_j, r_0) \times \mathbb{R}^N \cap B(x_k, r_0) \times \mathbb{R}^N \to B(x_j, r_0) \times \mathbb{R}^N \cap B(x_k, r_0) \times \mathbb{R}^N$$

have uniformly bounded derivatives in the exponential normal coordinates around x_j or x_k . See also [Eic09, Page 65].

For our purposes we will assume that $S \to M$ is equipped with a Hermitian metric h, which is assumed to be a C^{∞} -bounded section of the bundle $\overline{S} \otimes S$ in the terminology of Shubin [Shu92, Appendix 1]. We denote $L_2(S)$ for the Hilbert space of square integrable sections of S with respect to the volume form of M and the Hermitian metric h.

Shubin defines elliptic differential operators acting on sections of vector bundles of bounded geometry. Given a vector bundle S of bounded geometry, define $EBD^m(M, S)$ as the space of differential operators P such that in the exponential normal coordinates y around $x \in M$, we have

$$P = \sum_{|\alpha| \le m} a_{\alpha,x}(y) \partial_y^{\alpha}$$

where $a_{\alpha,x}(y)$ are $N \times N$ matrices, identified with sections of End(S) in the local trivialisation of S and a synchronous frame, and

$$\|\partial_y^\beta a_{\alpha,x}(0)\| \le C_{\alpha,\beta}, \quad |\alpha| \le m$$

where $\|\cdot\|$ is the norm on the fibre $\operatorname{End}(S)_x$ defined by the Hermitian metric h.

Given such an operator $P \in EBD^2(M, S)$, we say by analogy with the scalar-valued case that P has a density of states ν_P if for every t there exists the limit

$$\lim_{R \to \infty} \frac{1}{|B(0,R)|} \operatorname{Tr}(e^{-tP} M_{\chi_{B(0,R)}}) = \int_{\mathbb{R}} e^{-t\lambda} d\nu_P(\lambda).$$

Here, the trace is now with respect to the Hilbert space $L_2(S)$ rather than $L_2(M)$. A verbatim repetition of the proof of Theorem 6.0.3 shows the following.

Theorem 6.4.1. Let $S \to M$ be a vector bundle of bounded geometry over a non-compact Riemannian manifold of bounded geometry with Property (D). If $P \in EBD^2(M, S)$ is a self-adjoint lower bounded operator having a density of states ν_P , then for $f \in C_c(\mathbb{R})$ we have

$$\operatorname{Tr}_{\omega}(f(P)M_w) = \int_{\mathbb{R}} f(\lambda) \, d\nu_P(\lambda)$$

where $w(x) = (1 + |B(x_0, d(x, x_0))|)^{-1}$. Similarly, for all t > 0 we have

$$\operatorname{Tr}_{\omega}(\exp(-tP)M_w) = \int_{\mathbb{R}} \exp(-t\lambda) d\nu_P(\lambda).$$

In [Roe88], Roe considers (orientable) manifolds of bounded geometry that have a regular exhaustion. In this section, we will only consider manifolds that satisfy the assumptions of Theorem 6.0.3. In particular, the assumptions imply that $\lim_{R\to\infty} \frac{|\partial B(x_0,R)|}{|B(x_0,R)|} = 0$, i.e. any increasing sequence of metric balls $\{B(x_0,R_i)\}_{i=0}^{\infty}$ where $R \to \infty$ forms a regular exhaustion.

Denote the Banach space of C^1 uniformly bounded *n*-forms on M by $\Omega^n_\beta(M)$. An element m in the dual space of $\Omega^n_\beta(M)$ is said to be associated with the regular exhaustion $\{B(x_0, R_i)\}$ if for each bounded *n*-form α

$$\lim_{i \to \infty} \inf \left| \langle \alpha, m \rangle - \frac{1}{|B(x_0, R_i)|} \int_{B(x_0, R_i)} \alpha \right| = 0.$$

The algebra $\mathcal{U}_{-\infty}(M)$ consists of operators $A : C_c^{\infty}(M) \to C_c^{\infty}(M)$ such that for each $s, k \in \mathbb{R}$, A has a continuous extension to a quasilocal operator from $\mathcal{H}^s(M) \to \mathcal{H}^{s-k}(M)$. In Roe's terminology, an operator $A : \mathcal{H}^s(M) \to \mathcal{H}^{s-k}(M)$ is quasilocal if for each $K \subset M$ and each $u \in \mathcal{H}^k(M)$ supported within K,

$$\|Au\|_{\mathcal{H}^{s-k}(M\setminus \operatorname{Pen}^+(K,r))} \le \mu(r)\|u\|_{\mathcal{H}^s(M)},$$

where $\mu : \mathbb{R}^+ \to \mathbb{R}^+$ is a function such that $\mu(r) \to 0$ as $r \to \infty$, and $\text{Pen}^+(K, r)$ is the closure of $\bigcup \{B(x, r) : x \in K\}$.

Operators $A \in \mathcal{U}_{-\infty}(M)$ are represented by uniformly bounded smoothing kernels

$$Au(x) = \int k_A(x, y)u(y) \, d\nu_g(y),$$

where ν_g is the Riemannian volume form (recall that M is assumed to be oriented). Roe then defines traces on $\mathcal{U}_{-\infty}(M)$ coming from functionals associated with our regular exhaustion by

$$\tau(A) = \langle \alpha_A, m \rangle,$$

where α_A is the bounded *n*-form defined by $x \to k_A(x, x) d\nu_g$.

Recall from the introduction of this chapter that the trace τ extends to a trace on $M_n(\mathcal{U}^+_{-\infty})$, and hence descends to a dimension-homomorphism

$$\dim_{\tau} : K_0(\mathcal{U}_{-\infty}) \to \mathbb{R}.$$

Furthermore, since Roe showed that elliptic differential operators are invertible modulo $\mathcal{U}_{-\infty}$ [Roe88], one can define an abstract index of an elliptic differential operator acting on sections of a Clifford bundle as an element of $K_0(\mathcal{U}_{-\infty})$ via standard K-theory constructions. The Roe index theorem then states the following.

Theorem 6.4.2 (Roe index theorem). Let M be a Riemannian manifold, S a graded Clifford bundle on M, both with bounded geometry. Let D be the Dirac operator of S. Let m and τ be defined as above. Then

$$\dim_{\tau}(\operatorname{Ind}(D)) = m(\mathbf{I}(D)),$$

where $\mathbf{I}(D)$ is the integrand in the Atiyah–Singer index theorem.

The basis for the proof of the theorem is the following McKean–Singer formula

$$\dim_{\tau}(\mathrm{Ind}(D)) = \tau(\eta e^{-tD^2})$$

where η is the grading operator on S, and τ is now defined on operators on acting on sections of S.

The following lemma relates Roe's τ functional to the density of states. We will use the fact that if P is elliptic and self-adjoint, then the mapping

$$f \mapsto \tau(f(P))$$

is continuous on $f \in C_c(\mathbb{R})$. Indeed, by the Sobolev inequality the uniform norm of the integral kernel of f(P) is bounded above by the norm of f(P) as an operator from a Sobolev space of sufficiently negative smoothness to a Sobolev space of sufficiently positive smoothness. Since f is compactly supported, by functional calculus, the operator $f(P)(1+P)^N$ is bounded on $L_2(M,S)$ for every N, with norm depending on the width of the support of f and the uniform norm of f. Since P is elliptic, it follows from these arguments that if f is supported in [-K, K] then there is a constant C_K such that

$$|\tau(f(P))| \le C_K ||f||_{\infty}$$

See the related arguments in [Roe88, Proposition 2.9, Proposition 2.10].

Lemma 6.4.3. Let $S \to M$ be a vector bundle of bounded geometry, and let $P \in EBD^m(M,S)$ for some m > 0. Assume that P has a density of states ν_P with respect to the base-point $x_0 \in M$. If τ is associated with the exhaustion $\{B(x_0, R_i)\}_{i=0}^{\infty}$ for some sequence $R_i \to \infty$, then

$$\tau(f(P)) = \int_{\mathbb{R}} f \, d\nu_P, \quad f \in C_c(\mathbb{R}).$$

Similarly,

$$\tau(\exp(-tP)) = \int_{\mathbb{R}} \exp(-t\lambda) d\nu_P(\lambda), \quad t > 0.$$

Proof. By Theorem 6.4.1, we have

$$\begin{aligned} \operatorname{Tr}_{\omega}(\exp(-tP)M_w) &= \int_{\mathbb{R}} e^{-t\lambda} d\nu_P(\lambda) \\ &= \lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(\exp(-tP)M_{\chi_{B(x_0, R)}}) \\ &= \lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \int_{B(x_0, R)} \operatorname{tr}_{\operatorname{End}(S_x)}(K_{\exp(-tP)}(x, x)) d\nu_g(x) \\ &= \tau(\exp(-tP)), \end{aligned}$$

Since

$$f \mapsto \tau(f(P)), \quad f \in C_c(\mathbb{R})$$

is continuous in the sense described in the paragraph preceding the theorem, it follows from the Riesz theorem that there exists a measure $\mu_{\tau,P}$ on \mathbb{R} such that

$$\tau(f(P)) = \int_{\mathbb{R}} f \, d\mu_{\tau,P}, \quad f \in C_c(\mathbb{R})$$

Since $\mu_{\tau,P}$ and ν_P have identical Laplace transform, it follows that $\mu_{\tau,P} = \nu_P$.

A combination of Lemma 6.4.3 and Theorem 6.4.1 immediately yields the following:

Theorem 6.4.4. Let M be a manifold that satisfies the assumptions of Theorem 6.0.3, and let $S \to M$ be a vector bundle of bounded geometry. Let $P \in EBD^2(M, S)$, be selfadjoint and lower-bounded, and assume it admits a density of states ν_P at x_0 . Let w be the function on M defined by

$$w(x) = (1 + |B(x_0, d_g(x, x_0))|)^{-1}, \quad x \in M.$$

Then for any $f \in C_c(\mathbb{R})$ we have

$$\tau(f(P)) = \operatorname{Tr}_{\omega}(f(P)M_w)$$

for any τ associated with the regular exhaustion $\{B(x_0, R_i)\}_{i \in \mathbb{N}}$ where $R_i \to \infty$, and for any extended limit ω . Similarly,

$$\tau(\exp(-tP)) = \operatorname{Tr}_{\omega}(\exp(-tP)M_w)$$

The preceding theorem is proved under the strong assumption that P admits a density of states, which in particular implies that $\tau(\exp(-tP))$ is independent of the choice of functional m used to define τ . Addressing the question of determining which traces τ and which extended limits ω are related in this way in general is beyond the scope of this chapter.

Roe [Roe88] defines an algebra $\mathcal{U}_{-\infty}(E)$ of operators acting on sections of a vector bundle E, and τ is extended to $\mathcal{U}_{-\infty}(E)$ essentially by composing τ with the pointwise trace on $\operatorname{End}(E)$, see [Roe88] for details.

Theorem 6.4.5. Let M be a non-compact Riemannian manifold of bounded geometry with Property (D), with a graded Clifford bundle $S \to M$ of bounded geometry. Let D be a Dirac operator on M associated with the Dirac complex

$$C^{\infty}(S^+) \xrightarrow{D_+} C^{\infty}(S^-),$$

where D_+ is the restriction of D to the sections of S^+ and $D_- = D^*_+$ is its adjoint (cf. [Roe98, Chapter 11]). Let D^2 admit a density of states both when considered as an operator restricted to $L_2(S^+)$ and $L_2(S^-)$, in the sense that

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(\exp(-tD_-D_+)M_{\chi_{B(x_0, R)}}) = \int_{\mathbb{R}} e^{-t\lambda} d\nu_{D_-D_+}(\lambda), \quad t > 0,$$

for a Borel measure $\nu_{D_-D_+}$ and similarly for D_+D_- . Then for any $f \in C_c(\mathbb{R})$ such that f(0) = 1, we have

$$\dim_{\tau}(\mathrm{Ind}D) = \mathrm{Tr}_{\omega}(\eta f(D^2)M_w).$$

Proof. By [Roe88, Proposition 8.1], we have that

$$\dim_{\tau}(\operatorname{Ind} D) = \tau(\eta \exp(-tD^2)), \quad t > 0$$

where η is the grading operator

$$\eta = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix}$$

with respect to the orthogonal direct sum $L_2(S) = L_2(S^+) \oplus L_2(S^-)$.

The proof of the theorem amounts to showing that

$$\tau(\eta e^{-tD^2}) = \operatorname{Tr}_{\omega}(\eta e^{-tD^2}M_w).$$

The left-hand side is the same as

$$\tau(e^{-tD_+D_-}) - \tau(e^{-tD_-D_+})$$

while the right hand side is

$$\operatorname{Tr}_{\omega}(e^{-tD_{+}D_{-}}M_{w}) - \operatorname{Tr}_{\omega}(e^{-tD_{-}D_{+}}M_{w}).$$

Applying Theorem 6.4.1 to D_+D_- and D_-D_+ individually proves the result.

Remark 6.4.6. The index $\dim_{\tau}(\operatorname{Ind}(D))$ is computed by a version of the McKean–Singer formula [Roe88, Proposition 8.1]

$$\dim_{\tau}(\mathrm{Ind}(\mathrm{D})) = \tau(\eta \exp(-tD^2))$$

for arbitrary t > 0. One of the motivations in developing the present theorem was to give a new explanation of why the function

$$t \mapsto \tau(\eta \exp(-tD^2))$$

is independent of t. If the assumptions of Theorem 6.4.5 hold, then

$$\tau(\eta \exp(-tD^2)) = \operatorname{Tr}_{\omega}(\eta e^{-tD^2}M_w).$$

Formally differentiating the right hand side with respect to t and using the tracial property of Tr_{ω} yields

$$\frac{d}{dt} \operatorname{Tr}_{\omega}(\eta e^{-tD^2} M_w) = -\frac{1}{2} \operatorname{Tr}_{\omega}(\eta e^{-tD^2} D[D, M_w])$$

Our conditions ensure that $e^{-tD^2}D[D, M_w]$ is trace-class, and hence that

$$\frac{d}{dt} \operatorname{Tr}_{\omega}(\eta e^{-tD^2} M_w) = 0.$$

It is interesting that $\operatorname{Tr}_{\omega}(\eta e^{-tD^2}M_w)$ and the traditional heat supertrace $\operatorname{Tr}(\eta e^{-tD^2})$ on a compact manifold are both independent of t for apparently different reasons.

6.5 An example with a random operator

This section has mostly been the work of Edward McDonald.

The assumptions in Theorem 6.0.3 appear quite strong, especially the existence of the density of states. The following example of a random operator on a non-compact manifold where the density of states exists was given by Lenz, Peyerimhoff and Veselić [LPV07], generalising earlier examples in [LPV04; PV02].

Example 6.5.1. [LPV07, Example (RSM)] Let (M, g_0) be the connected Riemannian covering of a compact Riemannian manifold $X = M/\Gamma$, where Γ is an infinite group acting freely and properly discontinuously on M by isometries. Assume that there is an ergodic action α of Γ on a probability space $(\Xi, \Sigma, \mathbb{P})$, and let $\{g_{\xi}\}_{\xi \in \Xi}$ be a measurable family of metrics on M which are uniformly comparable with g_0 , in the sense that there exists A > 0 such that

$$\frac{1}{A}g_0(v,v) \le g_{\xi}(v,v) \le Ag_0(v,v)$$

for all tangent vectors v to M. Assume that the action α of Γ on Ξ is compatible with the action on Γ on M in the sense that $g_{\alpha_{\gamma}^{-1}\xi}$ is the pullback of g_{ξ} under the automorphism defined by γ .

Similarly, it is assumed that there is a measurable family $\{V_{\xi}\}_{\xi\in\Xi}$ of smooth functions on M such that

$$V_{\xi} \circ \gamma = V_{\alpha_{\gamma}^{-1}\xi}.$$

Let ν^{ξ} denote the Riemannian volume form on M corresponding to g_{ξ} , and let Δ_{ξ} be the Laplace-Beltrami operator on (M, g_{ξ}) . Then [LPV07] consider the operator on $L_2(M, \nu^{\xi})$ given by

$$H_{\xi} = -\Delta_{\xi} + M_{V_{\xi}}.$$

What is shown in [LPV07, Equation (27)] is that if Γ is amenable, then for every tempered Følner sequence $\{A_n\}_{n=0}^{\infty}$ in Γ , the limit

$$\lim_{n \to \infty} \frac{1}{|A_n|} \operatorname{Tr}_{L_2(M,\nu^{\xi}}(M_{\chi_{A_n F}} e^{-tH_{\xi}})$$

exists for almost every ξ , where F is a fundamental domain for Γ .

ı.

Recall that a we say that a Følner sequence $\{A_n\}_{n=0}^{\infty}$ is tempered if there is a constant C > 0 such that for every $n \ge 1$ we have

$$\left| \bigcup_{k < n} A_k^{-1} A_n \right| \le C |A_n|.$$

We will make one further assumption: that the measure ν^0 associated with g_0 has a

doubling property. That is, there exists a constant C such that

$$\nu^0(B_{q_0}(x_0, 2R)) \le C\nu^0(B_{q_0}(x_0, R)), \quad R > 0.$$

Note that since the identity function is a bi-Lipschitz continuous map from (M, g_0) to (M, g_{ξ}) , the same holds for the measures ν^{ξ} associated with the metrics g_{ξ} .

Proposition 6.5.2. Let (M, g_0) be as above, and let

$$w_{\xi}(x) = \frac{1}{1 + \nu^{\xi}(B_{g_{\xi}}(x_0, d_{g_{\xi}}(x, x_0)))}, \quad x \in M.$$

If M satisfies Property (D), the density for H_{ξ} exists almost surely, and for almost all $\xi \in \Xi$,

$$\operatorname{Tr}_{\omega}(f(H_{\xi})M_{w_{\xi}}) = \int_{\mathbb{R}} f(\lambda) \, d\nu_{H_{\xi}}(\lambda), \quad f \in C_{c}(\mathbb{R}),$$

for every extended limit $\omega \in \ell_{\infty}^*$.

Proof. For brevity, we will denote the measure ν^{ξ} by $|\cdot|$ and $B(x_0, R)$ for $B_{g_{\xi}}(x_0, R)$.

We will show that Property (D) implies that Γ admits a tempered Følner sequence $\{A_n\}_{n=1}^{\infty}$, and that for any bounded measurable function g on M we have

$$\lim_{k \to \infty} \frac{1}{|A_k F|} \int_{A_k F} g \, d\nu^{\xi} = \lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \int_{B(x_0, R)} g \, d\nu^{\xi} \tag{6.18}$$

if either limit exists. Recall that F is a fundamental domain for the action of Γ . Together with the results of [LPV07], this implies that the limit

$$\lim_{R \to \infty} \frac{1}{|B(x_0, R)|} \operatorname{Tr}(M_{\chi_{B(x_0, R)}} e^{-tH_{\xi}})$$

exists for every t > 0, and hence the assumptions of Theorem 6.0.3 are satisfied.

Let h > 2diam(F). For $k \ge h$, let

$$A_k = \{ \gamma \in \Gamma : \operatorname{dist}(\gamma x_0, x_0) < k - h \}$$

We claim that A_k is a tempered Følner sequence.

Note that dist $(\gamma x_0, x_0) = \text{dist}(\gamma^{-1}x_0, x_0)$, and so automatically $A_k = A_k^{-1}$. Define

$$B_k := A_k F = \bigcup_{\gamma \in A_k} \gamma F$$

First we show that $B(x_0, k - 2h) \subseteq B_k$. Indeed, if $p \in B(x_0, k - 2h)$ there exists some $\gamma \in \Gamma$ such that $p \in \gamma F$, so $\operatorname{dist}(p, x_0) \leq \operatorname{diam}(F) < h$, and thus $\operatorname{dist}(\gamma x_0, x_0) < k - 2k + h = k - h$. On the other hand, since F has diameter smaller than $\frac{h}{2}$, if $p \in B_k$ then $\operatorname{dist}(p, x_0) \leq \frac{h}{2} + k - h < k$. That is, for all $k \geq 2h$ we have

$$B(x_0, k-2h) \subset B_k \subset B(x_0, k).$$

Since the action of Γ is free, the union of the translates of F is disjoint, and

$$|B_k| = |A_k||F|, \quad k \ge 0$$

and hence

$$|B(x_0, k - 2h)| \le |F||A_k| \le |B(x_0, k)|.$$

As in the proof of Lemma 6.3.9, we know that for each h > 0 there is a constant C_h so that we have

$$\frac{|B(x_0,k)|}{|B(x_0,k-2h)|} \le C_h, \quad k > 1 + 2h$$

Therefore,

$$|A_k| \approx |B_k| \approx |B(x_0, k)|.$$

uniformly in k > 1 + 2h.

To see that A_k is Følner, let $\gamma \in \Gamma$. By the triangle inequality, there exists N > 0 such that

$$\gamma A_k \subset A_{k+N}$$

and, for k sufficiently large,

$$A_k \subset \gamma A_{k-N}$$

and therefore the symmetric difference of A_k and γA_k satisfies

$$(\gamma A_k \setminus A_k) \cup (A_k \setminus \gamma A_k) \subset A_{k+N} \setminus A_{k-N}.$$

It follows that, as $k \to \infty$,

$$\frac{(\gamma A_k \setminus A_k) \cup (A_k \setminus \gamma A_k)|}{|A_k|} \approx \frac{|B(x_0, k+N)| - |B(0, k-N)|}{|B(x_0, k)|}$$

which is vanishing as $k \to \infty$. Hence, $\{A_k\}_{k=1}^{\infty}$ is Følner.

To see that $\{A_k\}_{k=1}^{\infty}$ is tempered, it suffices to show that there is a constant C such that

$$|A_k \cdot A_k| \le C|A_k|.$$

By the triangle inequality and the fact that Γ acts isometrically we see that

$$A_k \cdot A_k \subseteq A_{2k}.$$

Therefore, by the doubling condition,

$$|A_k \cdot A_k| \le |A_{2k}| = \frac{1}{|F|} |B(x_0, 2k)| \lesssim |B(x_0, k)| \approx |A_k|.$$

uniformly in k for sufficiently large k. Hence, $\{A_k\}_{k=1}^{\infty}$ is tempered.

Finally we prove (6.18). Using the fact that $B_k \subseteq B(x_0, k)$, we write

$$\frac{1}{|B(x_0,k)|} \int_{B(x_0,k)} g \, d\nu^{\xi} = \frac{1}{|B_k|} \int_{B_k} g \, d\nu^{\xi} + \frac{|B_k| - |B(x_0,k)|}{|B(x_0,k)||B_k|} \int_{B_k} g \, d\nu^{\xi} + \frac{1}{|B(x_0,k)|} \int_{B(x_0,k) \setminus B_k} g \, d\nu^{\xi}$$

Therefore,

$$\begin{aligned} \left| \frac{1}{|B(x_0,k)|} \int_{B(x_0,k)} g \, d\nu^{\xi} - \frac{1}{|B_k|} \int_{B_k} g \, d\nu^{\xi} \right| &\leq 2 \|g\|_{\infty} \frac{|B(x_0,k)| - |B_k|}{|B(x_0,k)|} \\ &\leq 2 \|g\|_{\infty} \frac{|B(x_0,k)| - |B(x_0,k-2h)|}{|B(x_0,k)|} \end{aligned}$$

and this vanishes as $k \to \infty$. From here one easily deduces (6.18).

6.6 Discrete metric spaces revisited

The abstract operator theoretical result in Theorem 6.0.4 also recovers a weaker version of the main result from Chapter 5. This section is adapted from [HM23, Section 6], the first preprint version of [HM24a], which did not appear in the published version. As in Chapter 5, take a countably infinite discrete metric space (X, d) such that every ball contains at most finitely many points, and let $x_0 \in X$. Write $\{r_k\}_{k \in \mathbb{N}} \subseteq \mathbb{R}$ for the set of values $\{d(x, x_0) : x \in X\}$ ordered in increasing fashion. Take a positive radially symmetric strictly increasing $w : X \to \mathbb{C}$ such that $M_w \in \mathcal{L}_{1,\infty}$.

Now take a self-adjoint bounded operator $T \in B(\ell_2(X))$ and C > 0 such that T + C is strictly positive. Writing $P = \log(T + C)$, we have that $\exp(-tP) = (T + C)^t$. Then Theorem 6.0.4 — using the weaker conditions (6.11) — gives us that if for all t > 0

- 1. $(T+C)^t M_w \in \mathcal{L}_{1,\infty};$
- 2. $[(T+C)^t, M_w] \in \mathcal{L}_1,$

then we have for every extended limit $\omega \in \ell_{\infty}^*$,

$$\operatorname{Tr}_{\omega}((T+C)M_w) = \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}((T+C)\chi_{[\varepsilon,\infty)}(M_w)), \tag{6.19}$$

whenever the limit on the right hand exists. On the other hand, the main result in Chapter 5, Theorem 5.0.1, would let us conclude for all bounded $T \in B(\ell_2(X))$ (not just self-adjoint T)

$$\operatorname{Tr}_{\omega}(TM_w) = \operatorname{Tr}_{\omega}(M_w) \lim_{k \to \infty} \frac{\operatorname{Tr}(TM_{\chi_{B(x_0, r_k)}})}{|B(x_0, r_k)|},$$
(6.20)

whenever the limit on the right-hand side exists. These statements are not equivalent, namely, the existence of the limit (6.19) is a formally stronger requirement.

Recall that w is assumed to be radially strictly decreasing, so for $\varepsilon < 0$ we can pick $k \in \mathbb{N}$ such that

$$\tilde{w}(r_{k+1}) < \varepsilon \leq \tilde{w}(r_k),$$
where $w(\cdot) = \tilde{w} \circ d(x_0, \cdot)$. Then we have for positive $0 < \delta 1 < T \in B(\ell_2(X))$,

$$\varepsilon \operatorname{Tr}(T\chi_{[\varepsilon,\infty)}(M_w)) - \operatorname{Tr}_{\omega}(M_w) \frac{\operatorname{Tr}(TM_{\chi_{B(x_0,r_k)}})}{|B(x_0,r_k)|}$$

$$\leq \left(\tilde{w}(r_k) - \operatorname{Tr}_{\omega}(M_w) \frac{1}{|B(x_0,r_k)|}\right) \operatorname{Tr}(TM_{\chi_{B(x_0,r_k)}})$$

$$+ \tilde{w}(r_k) \operatorname{Tr}(TM_{\chi_{B(x_0,r_{k+1})\setminus B(x_0,r_k)})$$

$$\leq \left|\tilde{w}(r_k)|B(x_0,r_k)| - \operatorname{Tr}_{\omega}(M_w)\right| ||T||_{\infty}$$

$$+ \tilde{w}(r_k)|S(x_0,r_{k+1})|||T||_{\infty},$$

where $S(x_0, r_{k+1}) = B(x_0, r_{k+1}) \setminus B(x_0, r_k)$. Assuming that X satisfies Property (C), we have that

$$\tilde{w}(r_k)|S(x_0, r_{k+1})| = \tilde{w}(r_k)|B(x_0, r_k)|\frac{|S(x_0, r_{k+1})|}{|B(x_0, r_k)|} \to 0, \quad k \to \infty,$$

since $M_w \in \mathcal{L}_{1,\infty}$ implies that

$$\tilde{w}(r_k)|B(x_0,r_k)| \le \sup_{k\in\mathbb{N}} k\mu(k,w) < \infty,$$

where $\{\mu(k, w)\}_{k=0}^{\infty}$ is the decreasing rearrangement of w.

On the other hand we have

$$\varepsilon \operatorname{Tr}(T\chi_{[\varepsilon,\infty)}(M_w)) - \operatorname{Tr}_{\omega}(M_w) \frac{\operatorname{Tr}(TM_{\chi_{B(x_0,r_k)}})}{|B(x_0,r_k)|} \\ \geq \left(\tilde{w}(r_{k+1}) - \operatorname{Tr}_{\omega}(M_w) \frac{1}{|B(x_0,r_k)|} \right) \operatorname{Tr}(TM_{\chi_{B(x_0,r_k)}}) \\ \geq -\delta \left| \tilde{w}(r_{k+1}) |B(x_0,r_k)| - \operatorname{Tr}_{\omega}(M_w) \right|.$$

Hence, using Property (C), the condition

$$\lim_{k \to \infty} \tilde{w}(k) |B(x_0, r_k)| = \operatorname{Tr}_{\omega}(M_w)$$
(6.21)

allows going from equation (6.19) to (6.20). This does not hold for general $w \in \ell_{1,\infty}(X)$, however. Equation (6.21) requires M_w to be Dixmier measurable, but it is an even stronger condition than this [LSZ21, Theorem 9.1.6]. At least for the function $w(x) = (1 + |B(x_0, d(x, x_0))|)^{-1}$ condition (6.21) is true due to Corollary 5.2.3, and we can re-derive a weaker version of Theorem 5.0.1 using Theorem 6.0.4.

Proposition 6.6.1. Suppose $T \ge \varepsilon 1 > 0, V \in B(H)$ are bounded operators such that $[T, V] \in \mathcal{L}_1$. Then

$$[T^t, V] \in \mathcal{L}_1, \quad t > 0.$$

Proof. Let 0 < t < 1. We have

$$T^{t} = \frac{\sin(\pi t)}{\pi} \int_{0}^{\infty} \lambda^{t-1} T(\lambda + T)^{-1} d\lambda.$$

Then

$$[T^t, V] = \frac{\sin(\pi t)}{\pi} \int_0^\infty \lambda^{t-1} [T(\lambda + T)^{-1}, V] d\lambda$$
$$= \frac{\sin(\pi t)}{\pi} \int_0^\infty \lambda^{t-1} [1 - \lambda(\lambda + T)^{-1}, V] d\lambda$$
$$= \frac{\sin(\pi t)}{\pi} \int_0^\infty \lambda^t (\lambda + T)^{-1} [T, V] (\lambda + T)^{-1} d\lambda$$

Since $T \geq \varepsilon 1$, we have

$$\|(\lambda+T)^{-1}\| \le (\lambda+\varepsilon)^{-1},$$

and hence

$$\|[T^t, V]\|_1 \le \frac{\sin(\pi t)}{\pi} \int_0^\infty \lambda^t (\lambda + \varepsilon)^{-2} d\lambda \cdot \|[T, V]\|_1 < \infty.$$

To conclude, for t > 1 we can inductively apply the equality

$$[T^{t}, V] = T[T^{t-1}, V] + [T, V]T^{t-1}$$

to obtain the result.

Theorem 6.6.2. Let (X, d) be a countably infinite discrete metric space such that every ball contains at most finitely many points, and let $x_0 \in X$. Write by $\{r_k\}_{k \in \mathbb{N}}$ for the set $d(\cdot, x_0) : X \to \mathbb{R}_{\geq 0}$ ordered in increasing manner. Suppose that

$$\lim_{k \to \infty} \frac{|B(x_0, r_{k+1})|}{|B(x_0, r_k)|} = 1.$$
 (C)

Define the function $w(x) = (1 + |B(x_0, d(x, x_0))|)^{-1}$. If T is a bounded self-adjoint operator on $\ell_2(X)$ for which $[T, M_w] \in \mathcal{L}_1$, we have that for every extended limit ω ,

$$\operatorname{Tr}_{\omega}(TM_w) = \lim_{k \to \infty} \frac{\operatorname{Tr}(TM_{\chi_{B(x_0, r_k)}})}{|B(x_0, r_k)|},\tag{6.22}$$

if the limit on the right-hand side exists.

In particular, if H is a self-adjoint, possibly unbounded, operator on $\ell_2(X)$ with density of states measure ν_H , and $f \in C_c(\mathbb{R})$ and such that $[f(H), M_w] \in \mathcal{L}_1$, then for all extended limits ω

$$\operatorname{Tr}_{\omega}(f(H)M_w) = \int_{\mathbb{R}} f \, d\nu_H.$$
(6.23)

Proof. Let C > 0 such that $T + C > \delta 1 > 0$. The fact that $M_w \in \mathcal{L}_{1,\infty}$, Proposition 6.6.1 and the assumption that $[T, M_w] \in \mathcal{L}_1$ give that for all t > 0,

1. $(T+C)^t M_w \in \mathcal{L}_{1,\infty};$

2.
$$[(T+C)^t, M_w] \in \mathcal{L}_1.$$

Hence, Theorem 6.0.4 gives that

$$\operatorname{Tr}_{\omega}((T+C)M_w) = \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}((T+C)\chi_{[\varepsilon,\infty)}(M_w)),$$

whenever the limit on the right-hand side exists. Let $\varepsilon > 0$ small and take $k \in \mathbb{N}$ such that $(1 + |B(x_0, r_{k+1})|)^{-1} \le \varepsilon < (1 + |B(x_0, r_k)|)^{-1}$. Then,

$$\frac{|B(x_0, r_k)|}{1 + |B(x_0, r_{k+1})|} \le \varepsilon \operatorname{Tr}(\chi_{[\varepsilon, \infty)}(M_w)) \le \frac{|B(x_0, r_{k+1})|}{1 + |B(x_0, r_k)|}$$

Due to Property (C), we therefore have

$$\lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}(\chi_{[\varepsilon,\infty)}(M_w)) = 1 = \operatorname{Tr}_{\omega}(M_w),$$

using Corollary 5.2.3. Therefore,

$$\operatorname{Tr}_{\omega}(TM_w) = \lim_{\varepsilon \to 0} \varepsilon \operatorname{Tr}(T\chi_{[\varepsilon,\infty)}(M_w)),$$

whenever the limit on the right-hand side exists. With the computations given previously, we can conclude that

$$\operatorname{Tr}_{\omega}(TM_w) = \lim_{k \to \infty} \frac{\operatorname{Tr}(TM_{\chi_{B(x_0, r_k)}})}{|B(x_0, r_k)|},$$

if the limit on the right-hand side exists.

Finally, if H is a self-adjoint operator on $\ell_2(X)$ for which the density of states ν_H exists, we have the existence of all limits

$$\lim_{k \to \infty} \frac{\operatorname{Tr}(f(H)M_{\chi_{B(x_0,r_k)}})}{|B(x_0,r_k)|}, \quad f \in C_c(\mathbb{R})$$

If $f \in C_c(\mathbb{R})$ is such that $[f(H), M_w] \in \mathcal{L}_1$, then also the real part $\Re([f(H), M_w]) = [\Re(f)(H), M_w]$ is trace-class, and likewise the imaginary part. Since $\Re(f)(H)$ and $i\Im(f)(H)$ are self-adjoint, by the above we therefore have

$$\operatorname{Tr}_{\omega}(f(H)M_w) = \int_{\mathbb{R}} f \, d\nu_H.$$

We remark that, in the case that H is self-adjoint, bounded, and $[H, M_w]$ is trace-class, we get that $[f(H), M_w]$ is trace-class if f is operator-Lipschitz. In particular, this is the case when f is in the homogeneous Besov space $\dot{B}^1_{\infty,1}(\mathbb{R})$ [Pel90, Theorem 2].

The condition that $[T, M_w] \in \mathcal{L}_1$ is relatively weak. Since the operator $[T, M_w]$ already belongs to the commutator subspace

$$\operatorname{Com}(\mathcal{L}_{1,\infty}) := [B(H), \mathcal{L}_{1,\infty}],$$

it follows that [LSZ21, Theorem 5.1.4]

$$\sum_{k=0}^{n} \lambda(k, [T, M_w]) = O(1).$$

However, that is not quite enough for it to be trace-class. The following example has been provided by Teun van Nuland.

Example 6.6.3. Let $X = \mathbb{N}^+ = \{1, 2, \ldots\}$, so that $M_w : e_n \mapsto \frac{1}{n}e_n$. Define a bijection $\phi : \mathbb{N} \to \mathbb{N}$ by

$$2n+1 \mapsto 2^n$$

and recursively mapping each even integer 2n to the least integer that doesn't appear in $\{\phi(k): k < 2n\}$. Then define the operator $T: e_n \mapsto e_{\phi(n)}$. Observe that $T^*: e_n \mapsto e_{\phi^{-1}(n)}$. By a simple calculation,

$$[T, M_w][T, M_w]^* e_n = \left(\frac{1}{\phi^{-1}(n)} - \frac{1}{n}\right) [T, M_w] e_{\phi^{-1}(n)}$$
$$= \left(\frac{1}{\phi^{-1}(n)} - \frac{1}{n}\right)^2 e_n.$$

Hence, $[T, M_w]$ is trace-class if and only if

$$\sum_{n \in \mathbb{N}} \left| \frac{1}{\phi^{-1}(n)} - \frac{1}{n} \right| = \sum_{n \in \mathbb{N}} \left| \frac{1}{n} - \frac{1}{\phi(n)} \right| < \infty.$$

However,

$$\sum_{n \in \mathbb{N}} \left| \frac{1}{n} - \frac{1}{\phi(n)} \right| > \sum_{n \in \mathbb{N}} \left(\frac{1}{2n+1} - \frac{1}{2^n} \right),$$

which diverges. Hence $[T, M_w]$ is not trace-class.

In conclusion, Theorem 6.6.2 is significantly weaker than Theorem 5.0.1, justifying the method of proof in Chapter 5.

References

- [ABE12] W. Arendt, M. Biegert, and A. F. M. ter Elst. Diffusion determines the manifold. J. Reine Angew. Math. 667 (2012), pp. 1–25.
- [ACDS09] N. A. Azamov, A. L. Carey, P. G. Dodds, and F. A. Sukochev. Operator Integrals, Spectral Shift, and Spectral Flow. *Canadian Journal of Mathematics* 61.2 (2009), pp. 241–263.
- [ACS07] N. A. Azamov, A. L. Carey, and F. A. Sukochev. The spectral shift function and spectral flow. *Comm. Math. Phys.* 276.1 (2007), pp. 51–91.
- [AH61] M. F. Atiyah and F. Hirzebruch. Vector bundles and homogeneous spaces. Proc. Sympos. Pure Math., Vol. III. Amer. Math. Soc., Providence, RI, 1961, pp. 7–38.
- [AM76] N. W. Ashcroft and N. D. Mermin. Solid state physics. Vol. 120. Saunders, 1976.
- [AMSZ20] N. Azamov, E. McDonald, F. Sukochev, and D. Zanin. A Dixmier trace formula for the density of states. *Comm. Math. Phys.* 377.3 (2020), pp. 2597– 2628.
- [AMSZ22] N. Azamov, E. McDonald, F. Sukochev, and D. Zanin. The density of states depends on the domain. Oper. Matrices 16.4 (2022), pp. 1185–1189.
- [AMY22] I. Androulidakis, O. Mohsen, and R. Yuncken. A pseudodifferential calculus for maximally hypoelliptic operators and the Helffer-Nourrigat conjecture. arXiv preprint arXiv:2201.12060 (2022).

- [AO82] S. Alexander and R. Orbach. Density of states on fractals: "fractons". Journal de Physique Lettres 43.17 (1982), pp. 625–631.
- [AP10] A. B. Aleksandrov and V. V. Peller. Operator Hölder-Zygmund functions. Adv. Math. 224.3 (2010), pp. 910–966.
- [AP16] A. B. Aleksandrov and V. V. Peller. Operator Lipschitz functions. Uspekhi Mat. Nauk 71.4(430) (2016), pp. 3–106.
- [AS63] M. F. Atiyah and I. M. Singer. The index of elliptic operators on compact manifolds. Bull. Amer. Math. Soc. 69 (1963), pp. 422–433.
- [AS68] M. F. Atiyah and I. M. Singer. The index of elliptic operators, I. Ann. Math.
 (2) 87.3 (May 1968). A Russian translation was published in Uspehi Mat. Nauk 23:5(143). MR:236950. Zbl:0164.24001., pp. 484–530.
- [AS93] T. Adachi and T. Sunada. Density of states in spectral geometry. Comment. Math. Helv. 68.3 (1993), pp. 480–493.
- [AW15] M. Aizenman and S. Warzel. Random operators: Disorder effects on quantum spectra and dynamics. Vol. 168. Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2015, pp. xiv+326.
- [Aza+22] N. Azamov, E.-M. Hekkelman, E. McDonald, F. Sukochev, and D. Zanin. An application of singular traces to crystals and percolation. J. Geom. Phys. 179 (2022), Paper No. 104608.
- [Bar+23] I. A. Baratta, J. P. Dean, J. S. Dokken, M. Habera, J. S. Hale, C. N. Richardson, M. E. Rognes, M. W. Scroggs, N. Sime, and G. N. Wells. DOLFINx: the next generation FEniCS problem solving environment. Preprint. 2023.
- [BDF77] L. G. Brown, R. G. Douglas, and P. A. Fillmore. Extensions of C*-algebras and K-homology. Ann. of Math. (2) 105.2 (1977), pp. 265–324.
- [Bea77] R. Beals. Characterization of pseudodifferential operators and applications. Duke Math. J. 44.1 (1977), pp. 45–57.
- [Ber68] M. Berger. Le spectre des variétés riemanniennes. Rev. Roumaine Math. Pures Appl. 13 (1968), pp. 915–931.

- [BES94] J. Bellissard, A. van Elst, and H. Schulz-Baldes. The noncommutative geometry of the quantum Hall effect. J. Math. Phys. 35.10 (1994), pp. 5373– 5451.
- [BF73] R. Beals and C. Fefferman. Classes of spatially inhomogeneous pseudodifferential operators. Proc. Nat. Acad. Sci. U.S.A. 70 (1973), pp. 1500–1501.
- [BFG12] H. Bahouri, C. Fermanian-Kammerer, and I. Gallagher. Phase-space analysis and pseudodifferential calculus on the Heisenberg group. Astérisque 342 (2012), pp. vi+127.
- [BG06] M. Baake and U. Grimm. Averaged coordination numbers of planar aperiodic tilings. *Philosophical Magazine* 86.3-5 (2006), pp. 567–572.
- [BG88] R. Beals and P. Greiner. Calculus on Heisenberg manifolds. Vol. 119. Annals of Mathematics Studies. Princeton University Press, Princeton, NJ, 1988, pp. x+194.
- [BJ83] S. Baaj and P. Julg. Théorie bivariante de Kasparov et opérateurs non bornés dans les C*-modules hilbertiens. C. R. Acad. Sci. Paris Sér. I Math. 296.21 (1983), pp. 875–878.
- [BK13] J. Bourgain and A. Klein. Bounds on the density of states for Schrödinger operators. *Invent. Math.* 194.1 (2013), pp. 41–72.
- [BL76] J. Bergh and J. Löfström. Interpolation spaces. An introduction. Grundlehren der Mathematischen Wissenschaften, No. 223. Springer-Verlag, Berlin-New York, 1976, pp. x+207.
- [Bla06] B. Blackadar. Operator algebras. Vol. 122. Encyclopaedia of Mathematical Sciences. Theory of C*-algebras and von Neumann algebras, Operator Algebras and Non-commutative Geometry, III. Springer-Verlag, Berlin, 2006, pp. xx+517.
- [Bla98] B. Blackadar. K-theory for operator algebras. Second. Vol. 5. Mathematical Sciences Research Institute Publications. Cambridge University Press, Cambridge, 1998, pp. xx+300.

- [Blo29] F. Bloch. Zur Suszeptibilität und Widerstandsänderung der Metalle im Magnetfeld. Zeitschrift für Physik 53.3 (1929), pp. 216–227.
- [Bon13] J.-M. Bony. On the characterization of pseudodifferential operators (old and new). Studies in Phase Space Analysis with Applications to PDEs. New York, NY: Springer New York, 2013, pp. 21–34.
- [Bon67] R. Bonic. Some properties of Hilbert scales. Proc. Amer. Math. Soc. 18 (1967), pp. 1000–1003.
- [BP18] C. Bourne and E. Prodan. Non-commutative Chern numbers for generic aperiodic discrete systems. J. Phys. A 51.23 (2018), pp. 235202, 38.
- [BR87] O. Bratteli and D. W. Robinson. Operator algebras and quantum statistical mechanics 1. C^{*}- and W^{*}-algebras, symmetry groups, decomposition of states. Second edition. Texts and Monographs in Physics. Springer-Verlag, New York, 1987, pp. xiv+505.
- [Bru79] G. O. Brunner. The properties of coordination sequences and conclusions regarding the lowest possible density of zeolites. *Journal of Solid State Chemistry* 29.1 (1979), pp. 41–45.
- [BS03] M. S. Birman and M. Solomyak. Double operator integrals in a Hilbert space. Integral Equations Operator Theory 47.2 (2003), pp. 131–168.
- [BS66] M. Š. Birman and M. Z. Solomyak. Double Stieltjes operator integrals. Problems of mathematical physics, No. I: Spectral theory and wave processes (Russian). Izdat. Leningrad. Univ., Leningrad, 1966, pp. 33–67.
- [BS67] M. Š. Birman and M. Z. Solomyak. Double Stieltjes operator integrals. II. Problems of Mathematical Physics, No. 2, Spectral Theory, Diffraction Problems (Russian). Izdat. Leningrad. Univ., Leningrad, 1967, pp. 26–60.
- [BS69] M. Š. Birman and M. Z. Solomjak. Estimates of singular numbers of integral operators. III. Operators in unbounded domains. Vestnik Leningrad. Univ. 24.1 (1969), pp. 35–48.

- [BS73] M. Š. Birman and M. Z. Solomyak. Double Stieltjes operator integrals. III. Problems of mathematical physics, No. 6 (Russian). Izdat. Leningrad. Univ., Leningrad, 1973, pp. 27–53.
- [BS91] F. A. Berezin and M. A. Shubin. The Schrödinger equation. Vol. 66. Mathematics and its Applications (Soviet Series). Translated from the 1983 Russian edition by Yu. Rajabov, D. A. Leĭtes and N. A. Sakharova and revised by Shubin, With contributions by G. L. Litvinov and Leĭtes. Kluwer Academic Publishers Group, Dordrecht, 1991, pp. xviii+555.
- [BS98] P. Biane and R. Speicher. Stochastic calculus with respect to free Brownian motion and analysis on Wigner space. *Probab. Theory Related Fields* 112.3 (1998), pp. 373–409.
- [BT06] B. Bongioanni and J. L. Torrea. Sobolev spaces associated to the harmonic oscillator. Proc. Indian Acad. Sci. Math. Sci. 116.3 (2006), pp. 337–360.
- [Car+16] A. L. Carey, F. Gesztesy, G. Levitina, R. Nichols, D. Potapov, and F. A. Sukochev. Double operator integral methods applied to continuity of spectral shift functions. J. Spectr. Theory 6.4 (2016), pp. 747–779.
- [Car02] A. Carson. If not, winter: Fragments of Sappho. Poetry in English and Greek.Vintage Books, New York, 2002, pp. xvi+397.
- [CC96] A. H. Chamseddine and A. Connes. Universal formula for noncommutative geometry actions: unification of gravity and the standard model. *Phys. Rev. Lett.* 77.24 (1996), pp. 4868–4871.
- [CCM07] A. H. Chamseddine, A. Connes, and M. Marcolli. Gravity and the standard model with neutrino mixing. Adv. Theor. Math. Phys. 11.6 (2007), pp. 991– 1089.
- [CCS13] A. H. Chamseddine, A. Connes, and W. D. van Suijlekom. Beyond the spectral standard model: emergence of Pati-Salam unification. *Journal of High Energy Physics* 2013.11 (2013), pp. 1–36.

- [CCS15] A. H. Chamseddine, A. Connes, and W. D. van Suijlekom. Grand unification in the spectral Pati-Salam model. J. High Energy Phys. 11 (2015), 011, front matter+12.
- [CDR21] M. Chatzakou, J. Delgado, and M. Ruzhansky. On a class of anharmonic oscillators. J. Math. Pures Appl. (9) 153 (2021), pp. 1–29.
- [CF19] A. Connes and F. Fathizadeh. The term a₄ in the heat kernel expansion of noncommutative tori. Münster J. Math. 12.2 (2019), pp. 239–410.
- [CFF93] A. H. Chamseddine, G. Felder, and J. Fröhlich. Gravity in noncommutative geometry. Comm. Math. Phys. 155.1 (1993), pp. 205–217.
- [CGRS14] A. L. Carey, V. Gayral, A. Rennie, and F. A. Sukochev. Index theory for locally compact noncommutative geometries. *Mem. Amer. Math. Soc.* 231.1085 (2014), pp. vi+130.
- [Cha+86] J. T. Chayes, L. Chayes, J. R. Franz, J. P. Sethna, and S. A. Trugman. On the density of states for the quantum percolation problem. J. Phys. A 19.18 (1986), pp. L1173–L1177.
- [Cha06] I. Chavel. Riemannian geometry. Second edition. Vol. 98. Cambridge Studies in Advanced Mathematics. A modern introduction. Cambridge University Press, Cambridge, 2006, pp. xvi+471.
- [Cha84] I. Chavel. Eigenvalues in Riemannian geometry. Vol. 115. Pure and Applied Mathematics. Including a chapter by Burton Randol, With an appendix by Jozef Dodziuk. Academic Press, Inc., Orlando, FL, 1984, pp. xiv+362.
- [CHT18] Y. Colin de Verdière, L. Hillairet, and E. Trélat. Spectral asymptotics for sub-Riemannian Laplacians, I: Quantum ergodicity and quantum limits in the 3-dimensional contact case. *Duke Math. J.* 167.1 (2018), pp. 109–174.
- [CL90] R. Carmona and J. Lacroix. Spectral theory of random Schrödinger operators. Probability and its Applications. Birkhäuser Boston, Inc., Boston, MA, 1990, pp. xxvi+587.

- [CM08] A. Connes and M. Marcolli. Noncommutative geometry, quantum fields and motives. Vol. 55. American Mathematical Society Colloquium Publications. American Mathematical Society, Providence, RI; Hindustan Book Agency, New Delhi, 2008, pp. xxii+785.
- [CM14] A. Connes and H. Moscovici. Modular curvature for noncommutative twotori. J. Amer. Math. Soc. 27.3 (2014), pp. 639–684.
- [CM95] A. Connes and H. Moscovici. The local index formula in noncommutative geometry. Geom. Funct. Anal. 5.2 (1995), pp. 174–243.
- [CMPS14] M. Caspers, S. Montgomery-Smith, D. Potapov, and F. Sukochev. The best constants for operator Lipschitz functions on Schatten classes. J. Funct. Anal. 267.10 (2014), pp. 3557–3579.
- [Col85] Y. Colin de Verdière. Ergodicité et fonctions propres du laplacien. Comm. Math. Phys. 102.3 (1985), pp. 497–502.
- [Con13] A. Connes. On the spectral characterization of manifolds. J. Noncommut. Geom. 7.1 (2013), pp. 1–82.
- [Con21] A. Connes. On the Notion of Space. Global Noncommutative Geometry Seminar (Americas), Youtube. 2021. URL: https://www.youtube.com/watch? v=m20KxUB51MU#t=24m18s.
- [Con82] A. Connes. A survey of foliations and operator algebras. Operator algebras and applications, Part 1 (Kingston, Ont., 1980). Vol. 38. Proc. Sympos. Pure Math. Amer. Math. Soc., Providence, RI, 1982, pp. 521–628.
- [Con85] A. Connes. Noncommutative differential geometry. Inst. Hautes Études Sci. Publ. Math. 62 (1985), pp. 257–360.
- [Con88] A. Connes. The action functional in noncommutative geometry. Comm. Math. Phys. 117.4 (1988), pp. 673–683.
- [Con90] J. B. Conway. A course in functional analysis. Second edition. Vol. 96. Graduate Texts in Mathematics. Springer-Verlag, New York, 1990, pp. xvi+399.

- [Con94] A. Connes. Noncommutative geometry. Academic Press, Inc., San Diego, CA, 1994, pp. xiv+661.
- [Con95] A. Connes. Geometry from the spectral point of view. Lett. Math. Phys. 34.3 (1995), pp. 203–238.
- [Cor79] H. O. Cordes. Elliptic pseudodifferential operators—an abstract theory. Vol. 756.
 Lecture Notes in Mathematics. Springer, Berlin, 1979, pp. vii+331.
- [CPRS06a] A. L. Carey, J. Phillips, A. Rennie, and F. A. Sukochev. The local index formula in semifinite von Neumann algebras I. Spectral flow. Adv. Math. 202.2 (2006), pp. 451–516.
- [CPRS06b] A. L. Carey, J. Phillips, A. Rennie, and F. A. Sukochev. The local index formula in semifinite von Neumann algebras II. The even case. Adv. Math. 202.2 (2006), pp. 517–554.
- [CPRS08] A. L. Carey, J. Phillips, A. Rennie, and F. A. Sukochev. The Chern character of semifinite spectral triples. J. Noncommut. Geom. 2.2 (2008), pp. 141–193.
- [CPRS13] A. L. Carey, J. Phillips, A. Rennie, and F. A. Sukochev. The local index formula in noncommutative geometry revisited. *Noncommutative geometry* and physics. 3. Vol. 1. Keio COE Lect. Ser. Math. Sci. World Sci. Publ., Hackensack, NJ, 2013, pp. 3–36.
- [CPSZ19] M. Caspers, D. Potapov, F. Sukochev, and D. Zanin. Weak type commutator and Lipschitz estimates: resolution of the Nazarov-Peller conjecture. Amer. J. Math. 141.3 (2019), pp. 593–610.
- [CS12] A. Carey and F. Sukochev. Measurable operators and the asymptotics of heat kernels and zeta functions. J. Funct. Anal. 262.10 (2012), pp. 4582–4599.
- [CS21] A. Connes and W. D. van Suijlekom. Spectral truncations in noncommutative geometry and operator systems. *Comm. Math. Phys.* 383.3 (2021), pp. 2021– 2067.
- [CS22] A. Connes and W. D. van Suijlekom. Tolerance relations and operator systems. Acta Sci. Math. (Szeged) 88.1-2 (2022), pp. 101–129.

- [CS23] F. E. G. Cipriani and J.-L. Sauvageot. Measurability, spectral densities, and hypertraces in noncommutative geometry. J. Noncommut. Geom. 17.4 (2023), pp. 1437–1468.
- [CS93] J. H. Conway and N. J. A. Sloane. Sphere packings, lattices and groups. Second edition. Vol. 290. Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. With additional contributions by E. Bannai, R. E. Borcherds, J. Leech, S. P. Norton, A. M. Odlyzko, R. A. Parker, L. Queen and B. B. Venkov. Springer-Verlag, New York, 1993, pp. xliv+679.
- [CS97] J. H. Conway and N. J. A. Sloane. Low-dimensional lattices. VII. Coordination sequences. Proc. Roy. Soc. London Ser. A 453.1966 (1997), pp. 2369– 2389.
- [CT11] A. Connes and P. Tretkoff. The Gauss-Bonnet theorem for the noncommutative two torus. Noncommutative geometry, arithmetic, and related topics.
 Johns Hopkins Univ. Press, Baltimore, MD, 2011, pp. 141–158.
- [CT16] R. Cerf and M. Théret. Weak shape theorem in first passage percolation with infinite passage times. Ann. Inst. Henri Poincaré Probab. Stat. 52.3 (2016), pp. 1351–1381.
- [Dal67] J. L. Daleckiĭ. Infinite-dimensional elliptic operators and the corresponding parabolic equations. Uspehi Mat. Nauk 22.4(136) (1967), pp. 3–54.
- [Dal98] Y. L. Daletskiĭ. Formal operator power series and the noncommutative Taylor formula. Voronezh Winter Mathematical Schools. Vol. 184. Amer. Math. Soc. Transl. Ser. 2. Amer. Math. Soc., Providence, RI, 1998, pp. 41–61.
- [Dav95a] E. B. Davies. L_p Spectral Independence and L_1 Analyticity. Journal of the London Mathematical Society 52.1 (Aug. 1995), pp. 177–184.
- [Dav95b] E. B. Davies. The Functional Calculus. Journal of the London Mathematical Society 52.1 (Aug. 1995), pp. 166–176.
- [Dav95c] E. B. Davies. Spectral theory and differential operators. Vol. 42. Cambridge University Press, 1995.

- [DDSZ20] P. G. Dodds, T. K. Dodds, F. A. Sukochev, and D. Zanin. Arithmeticgeometric mean and related submajorisation and norm inequalities for τmeasurable operators: Part II. Integral Equations Operator Theory 92.4 (2020), Paper No. 32, 60.
- [DG75] J. J. Duistermaat and V. W. Guillemin. The spectrum of positive elliptic operators and periodic bicharacteristics. *Invent. Math.* 29.1 (1975), pp. 39– 79.
- [DIM01] S.-i. Doi, A. Iwatsuka, and T. Mine. The uniqueness of the integrated density of states for the Schrödinger operators with magnetic fields. *Math. Z.* 237.2 (2001), pp. 335–371.
- [Dix66] J. Dixmier. Existence de traces non normales. C. R. Acad. Sci. Paris Sér. A-B 262 (1966), A1107–A1108.
- [DK51] Y. L. Daleckiĭ and S. G. Kreĭn. Formulas of differentiation according to a parameter of functions of Hermitian operators. *Doklady Akad. Nauk SSSR* (N.S.) 76 (1951), pp. 13–16.
- [DK56] Y. L. Daletskiĭ and S. G. Kreĭn. Integration and differentiation of functions of Hermitian operators and applications to the theory of perturbations. *Voronež. Gos. Univ. Trudy Sem. Funkcional. Anal.* 1956.1 (1956), pp. 81– 105.
- [DLL22] F. D'Andrea, G. Landi, and F. Lizzi. Tolerance relations and quantization. Lett. Math. Phys. 112.4 (2022), Paper No. 65, 28.
- [DLM14] F. D'Andrea, F. Lizzi, and P. Martinetti. Spectral geometry with a cut-off: topological and metric aspects. J. Geom. Phys. 82 (2014), pp. 18–45.
- [DPS23] P. G. Dodds, B. de Pagter, and F. A. Sukochev. Noncommutative integration and operator theory. Vol. 349. Progress in Mathematics. Birkhäuser/Springer, Cham, 2023, pp. xi+577.
- [DS15] L. Dąbrowski and A. Sitarz. An asymmetric noncommutative torus. SIGMA Symmetry Integrability Geom. Methods Appl. 11 (2015), Paper 075, 11.

- [DS88] N. Dunford and J. T. Schwartz. Linear operators. Part I. Wiley Classics Library. General theory, With the assistance of William G. Bade and Robert G. Bartle, Reprint of the 1958 original, A Wiley-Interscience Publication. John Wiley & Sons, Inc., New York, 1988, pp. xiv+858.
- [DS99] M. Dimassi and J. Sjöstrand. Spectral asymptotics in the semi-classical limit.
 Vol. 268. London Mathematical Society Lecture Note Series. Cambridge University Press, Cambridge, 1999, pp. xii+227.
- [Dyn70] E. M. Dyn'kin. An operator calculus based on the Cauchy-Green formula, and the quasianalyticity of the classes D(h). Zap. Naučn. Sem. Leningrad. Otdel. Mat. Inst. Steklov. (LOMI) 19 (1970), pp. 221–226.
- [Dyn72] E. M. Dyn'kin. An operator calculus based on the Cauchy-Green formula. Zap. Naučn. Sem. Leningrad. Otdel. Mat. Inst. Steklov. (LOMI) 30 (1972), pp. 33–39.
- [EI18] M. Eckstein and B. Iochum. Spectral action in noncommutative geometry.
 Vol. 27. SpringerBriefs in Mathematical Physics. Springer, Cham, 2018, pp. xiv+155.
- [Eic07] J. Eichhorn. Global analysis on open manifolds. Nova Science Publishers, Inc., New York, 2007, pp. x+644.
- [Eic09] J. Eichhorn. Relative index theory, determinants and torsion for open manifolds. World Scientific Publishing Co. Pte. Ltd., Hackensack, NJ, 2009, pp. x+341.
- [Eic91] J. Eichhorn. The boundedness of connection coefficients and their derivatives. Math. Nachr. 152 (1991), pp. 145–158.
- [EKSS88] H. Englisch, W. Kirsch, M. Schroder, and B. Simon. Density of surface states in discrete models. *Physical Review Letters* 61.11 (1988), p. 1261.
- [Ell76] G. A. Elliott. On the classification of inductive limits of sequences of semisimple finite-dimensional algebras. J. Algebra 38.1 (1976), pp. 29–44.
- [Ell93] G. A. Elliott. On the classification of C*-algebras of real rank zero. J. Reine Angew. Math. 443 (1993), pp. 179–219.

- [EY19] E. van Erp and R. Yuncken. A groupoid approach to pseudodifferential calculi. J. Reine Angew. Math. 756 (2019), pp. 151–182.
- [Fel71] W. Feller. An introduction to probability theory and its applications. Vol. II.
 Second edition. John Wiley & Sons, Inc., New York-London-Sydney, 1971, pp. xxiv+669.
- [Fey51] R. P. Feynman. An operator calculus having applications in quantum electrodynamics. *Phys. Rev. (2)* 84 (1951), pp. 108–128.
- [FFF24] C. Fermanian-Kammerer, V. Fischer, and S. Flynn. A microlocal calculus on filtered manifolds. arXiv preprint arXiv:2412.17448 (2024).
- [FGR98] J. Fröhlich, O. Grandjean, and A. Recknagel. Supersymmetric quantum theory, non-commutative geometry, and gravitation. Symétries quantiques (Les Houches, 1995). North-Holland, Amsterdam, 1998, pp. 221–385.
- [FK12] F. Fathizadeh and M. Khalkhali. The Gauss-Bonnet theorem for noncommutative two tori with a general conformal structure. J. Noncommut. Geom. 6.3 (2012), pp. 457–480.
- [FK13] F. Fathizadeh and M. Khalkhali. Scalar curvature for the noncommutative two torus. J. Noncommut. Geom. 7.4 (2013), pp. 1145–1183.
- [FK19] F. Fathizadeh and M. Khalkhali. Curvature in noncommutative geometry. Advances in noncommutative geometry—on the occasion of Alain Connes' 70th birthday. Springer, Cham, 2019, pp. 321–420.
- [Fol08] G. B. Folland. Quantum field theory. Vol. 149. Mathematical Surveys and Monographs. A tourist guide for mathematicians. American Mathematical Society, Providence, RI, 2008, pp. xii+325.
- [FR14] V. Fischer and M. Ruzhansky. A pseudo-differential calculus on the Heisenberg group. C. R. Math. Acad. Sci. Paris 352.3 (2014), pp. 197–204.
- [FS74] G. B. Folland and E. M. Stein. Estimates for the $\bar{\partial}_b$ complex and analysis on the Heisenberg group. *Comm. Pure Appl. Math.* 27 (1974), pp. 429–522.

- [Gil04] P. B. Gilkey. Asymptotic formulae in spectral geometry. Studies in Advanced Mathematics. Chapman & Hall/CRC, Boca Raton, FL, 2004, pp. viii+304.
- [Gil75] P. B. Gilkey. The spectral geometry of a Riemannian manifold. J. Differential Geometry 10.4 (1975), pp. 601–618.
- [GJM22] L. Gao, M. Junge, and E. McDonald. Quantum Euclidean spaces with noncommutative derivatives. J. Noncommut. Geom. 16.1 (2022), pp. 153–213.
- [GL98] F. Golse and E. Leichtnam. Applications of Connes' geodesic flow to trace formulas in noncommutative geometry. J. Funct. Anal. 160.2 (1998), pp. 408– 436.
- [GM18] M. Goffeng and B. Mesland. Spectral triples on O_N . 2016 MATRIX annals. Vol. 1. MATRIX Book Ser. Springer, Cham, 2018, pp. 183–201.
- [GP11] R. Grimaldi and P. Pansu. Bounded geometry, growth and topology. J. Math. Pures Appl. (9) 95.1 (2011), pp. 85–98.
- [Gre78] R. E. Greene. Complete metrics of bounded curvature on noncompact manifolds. Arch. Math. (Basel) 31.1 (1978), pp. 89–95.
- [Gri99] G. Grimmett. Percolation. Second edition. Vol. 321. Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer-Verlag, Berlin, 1999, pp. xiv+444.
- [GRU19] M. Goffeng, A. Rennie, and A. Usachev. Constructing KMS states from infinite-dimensional spectral triples. J. Geom. Phys. 143 (2019), pp. 107– 149.
- [GS13] N. Große and C. Schneider. Sobolev spaces on Riemannian manifolds with bounded geometry: general coordinates and traces. *Math. Nachr.* 286.16 (2013), pp. 1586–1613.
- [GS20] L. Glaser and A. B. Stern. Understanding truncated non-commutative geometries through computer simulations. J. Math. Phys. 61.3 (2020), pp. 033507, 11.

- [GS21] L. Glaser and A. B. Stern. Reconstructing manifolds from truncations of spectral triples. J. Geom. Phys. 159 (2021), Paper No. 103921, 17.
- [GS23] M. Gielen and W. D. van Suijlekom. Operator systems for tolerance relations on finite sets. *Indag. Math.* (N.S.) 34.3 (2023), pp. 606–621.
- [GS95] G. Grubb and R. T. Seeley. Weakly parametric pseudodifferential operators and Atiyah-Patodi-Singer boundary problems. *Invent. Math.* 121.3 (1995), pp. 481–529.
- [Gui79] V. Guillemin. Some classical theorems in spectral theory revisited. Seminar on Singularities of Solutions of Linear Partial Differential Equations (Inst. Adv. Study, Princeton, N.J., 1977/78). Vol. No. 91. Ann. of Math. Stud. Princeton Univ. Press, Princeton, NJ, 1979, pp. 219–259.
- [Gui85] V. Guillemin. A new proof of Weyl's formula on the asymptotic distribution of eigenvalues. Adv. in Math. 55.2 (1985), pp. 131–160.
- [GV64] I. M. Gel'fand and N. Y. Vilenkin. Generalized functions. Vol. 4: Applications of harmonic analysis. Translated by Amiel Feinstein. Academic Press, New York-London, 1964, pp. xiv+384.
- [GVF01] J. Gracia-Bondía, J. Várilly, and H. Figueroa. *Elements of noncommutative geometry*. Birkhäuser Advanced Texts: Basler Lehrbücher [Basel Textbooks].
 Birkhäuser Boston Inc., Boston, MA, 2001, pp. xviii+685.
- [GWW92] C. Gordon, D. Webb, and S. Wolpert. Isospectral plane domains and surfaces via Riemannian orbifolds. *Invent. Math.* 110.1 (1992), pp. 1–22.
- [Han06] F. Hansen. Trace functions as Laplace transforms. Journal of Mathematical Physics 47.4 (Apr. 2006), p. 043504.
- [Har49] G. H. Hardy. *Divergent Series*. Oxford, at the Clarendon Press, 1949, pp. xvi+396.
- [HBE87] L. Hoddeson, G. Baym, and M. Eckert. The development of the quantummechanical electron theory of metals: 1928—1933. Rev. Mod. Phys. 59 (1 Jan. 1987), pp. 287–327.

- [Hek22] E.-M. Hekkelman. Truncated geometry on the circle. *Lett. Math. Phys.* 112.2 (2022), Paper No. 20, 19.
- [Hig04] N. Higson. The local index formula in noncommutative geometry. Contemporary developments in algebraic K-theory. Vol. XV. ICTP Lect. Notes. Abdus Salam Int. Cent. Theoret. Phys., Trieste, 2004, pp. 443–536.
- [HL24] L. Hartmann and M. Lesch. Divided Differences and Multivariate Holomorphic Calculus. *arXiv preprint arXiv:2411.05484* (2024).
- [HLP19a] H. Ha, G. Lee, and R. Ponge. Pseudodifferential calculus on noncommutative tori, I. Oscillating integrals. *Internat. J. Math.* 30.8 (2019), pp. 1950033, 74.
- [HLP19b] H. Ha, G. Lee, and R. Ponge. Pseudodifferential calculus on noncommutative tori, II. Main properties. *Internat. J. Math.* 30.8 (2019), pp. 1950034, 73.
- [HM23] E.-M. Hekkelman and E. McDonald. A general Dixmier formula for the density of states on open manifolds. *arXiv preprint arXiv:2304.13272v1* (2023).
- [HM24a] E.-M. Hekkelman and E. McDonald. A general Dixmier trace formula for the density of states on open manifolds. SIGMA Symmetry Integrability Geom. Methods Appl. 20 (2024), Paper No. 007, 34.
- [HM24b] E.-M. Hekkelman and E. McDonald. A noncommutative integral on spectrally truncated spectral triples, and a link with quantum ergodicity. *arXiv* preprint arXiv:2412.00628 (2024).
- [HMN24] E.-M. Hekkelman, E. McDonald, and T. D. H. van Nuland. Multiple operator integrals, pseudodifferential calculus, and asymptotic expansions. arXiv preprint arXiv:2404.16338 (2024).
- [Hof93] A. Hof. Some remarks on discrete aperiodic Schrödinger operators. J. Statist. Phys. 72.5-6 (1993), pp. 1353–1374.
- [Hof95] A. Hof. A remark on Schrödinger operators on aperiodic tilings. J. Statist. Phys. 81.3-4 (1995), pp. 851–855.

- [Hör07] L. Hörmander. The analysis of linear partial differential operators. III. Classics in Mathematics. Pseudo-differential operators, Reprint of the 1994 edition. Springer, Berlin, 2007, pp. viii+525.
- [Hör67] L. Hörmander. Pseudo-differential operators and hypoelliptic equations. Singular Integrals (Proc. Sympos. Pure Math., Chicago, Ill., 1966). Amer. Math. Soc., Providence, RI, 1967, pp. 138–183.
- [Hör69] L. Hörmander. Fourier Integral Operators: Lectures at the Nordic Summer School of Mathematics. English. Lecture notes. With foreword by Sofia Broström, comments by Johannes Sjöstrand and editorial assistance by Gerd Grubb. 1969.
- [Hör70] L. Hörmander. On the singularities of solutions of partial differential equations. Comm. Pure Appl. Math. 23 (1970), pp. 329–358.
- [Hör90] L. Hörmander. The analysis of linear partial differential operators. I. Second. Vol. 256. Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Distribution theory and Fourier analysis. Springer-Verlag, Berlin, 1990, pp. xii+440.
- [HS88] B. Helffer and J. Sjöstrand. Analyse semi-classique pour l'équation de Harper (avec application à l'équation de Schrödinger avec champ magnétique). Mém. Soc. Math. France (N.S.) 34 (1988), p. 113.
- [HS89] B. Helffer and J. Sjöstrand. Equation de Schrödinger avec champ magnétique et équation de Harper. Schrödinger Operators. Ed. by H. Holden and A. Jensen. Berlin, Heidelberg: Springer Berlin Heidelberg, 1989, pp. 118–197.
- [Ivr16] V. Ivrii. 100 years of Weyl's law. Bull. Math. Sci. 6.3 (2016), pp. 379–452.
- [Ivr80] V. Ivrii. The second term of the spectral asymptotics for a Laplace-Beltrami operator on manifolds with boundary. *Funktsional. Anal. i Prilozhen.* 14.2 (1980), pp. 25–34.
- [Jar89] M. V. Jarić, ed. Introduction to the mathematics of quasicrystals. Vol. 2.
 Aperiodicity and Order. Academic Press, Inc., Boston, MA, 1989, pp. x+226.

- [Jef04] B. Jefferies. Spectral properties of noncommuting operators. Vol. 1843. Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2004, pp. viii+184.
- [JLO88] A. Jaffe, A. Lesniewski, and K. Osterwalder. Quantum K-theory. I. The Chern character. Comm. Math. Phys. 118.1 (1988), pp. 1–14.
- [JN94] A. Jensen and S. Nakamura. Mapping Properties of Functions of Schrödinger Operators between L_p-Spaces and Besov Spaces. Spectral and scattering theory and applications. Vol. 23. Mathematical Society of Japan, 1994, pp. 187– 210.
- [Kac66] M. Kac. Can one hear the shape of a drum? Amer. Math. Monthly 73.4 (1966), pp. 1–23.
- [Kha13] M. Khalkhali. Basic noncommutative geometry. Second edition. EMS Series of Lectures in Mathematics. European Mathematical Society (EMS), Zürich, 2013, pp. xviii+239.
- [KLPS13] N. Kalton, S. Lord, D. Potapov, and F. Sukochev. Traces of compact operators and the noncommutative residue. *Adv. Math.* 235 (2013), pp. 1–55.
- [KM06] W. Kirsch and P. Müller. Spectral properties of the Laplacian on bondpercolation graphs. *Math. Z.* 252.4 (2006), pp. 899–916.
- [KM93] M. V. Karasëv and V. P. Maslov. Nonlinear Poisson brackets. Vol. 119. Translations of Mathematical Monographs. Geometry and quantization, Translated from the Russian by A. Sossinsky [A. B. Sosinskiĭ] and M. Shishkova. American Mathematical Society, Providence, RI, 1993, pp. xii+366.
- [KN65] J. J. Kohn and L. Nirenberg. An algebra of pseudo-differential operators. Comm. Pure Appl. Math. 18 (1965), pp. 269–305.
- [Knu97] D. E. Knuth. The art of computer programming. Vol. 1. Fundamental algorithms, Third edition. Addison-Wesley, Reading, MA, 1997, pp. xx+650.
- [Kod88] S. Kodani. An estimate on the volume of metric balls. Kodai Math. J. 11.2 (1988), pp. 300–305.

- [Kor05] Y. A. Kordyukov. Egorov's theorem for transversally elliptic operators on foliated manifolds and noncommutative geodesic flow. Math. Phys. Anal. Geom. 8.2 (2005), pp. 97–119.
- [Kor91] Y. A. Kordyukov. L^p-theory of elliptic differential operators on manifolds of bounded geometry. Acta Appl. Math. 23.3 (1991), pp. 223–260.
- [Kos92] H. Kosaki. An inequality of Araki-Lieb-Thirring (von Neumann algebra case). Proc. Amer. Math. Soc. 114.2 (1992), pp. 477–481.
- [KP66] S. G. Kreĭn and J. I. Petunin. Scales of Banach spaces. Uspehi Mat. Nauk 21.2(128) (1966), pp. 89–168.
- [KS18] M. Khalkhali and A. Sitarz. Gauss-Bonnet for matrix conformally rescaled Dirac. J. Math. Phys. 59.6 (2018), pp. 063505, 10.
- [KSZ24a] Y. Kordyukov, F. Sukochev, and D. Zanin. C*-algebraic approach to the principal symbol. III. J. Noncommut. Geom. 18.4 (2024), pp. 1265–1314.
- [KSZ24b] Y. Kordyukov, F. Sukochev, and D. Zanin. Principal Symbol Calculus on Contact Manifolds. Vol. 2359. Lecture Notes in Mathematics. Springer, Cham, 2024, pp. xiv+156.
- [Lan91] R. Lang. Spectral theory of random Schrödinger operators. Vol. 1498. Lecture Notes in Mathematics. A genetic introduction. Springer-Verlag, Berlin, 1991, pp. viii+125.
- [Les17] M. Lesch. Divided differences in noncommutative geometry: rearrangement lemma, functional calculus and expansional formula. J. Noncommut. Geom. 11.1 (2017), pp. 193–223.
- [Lin01] E. Lindenstrauss. Pointwise theorems for amenable groups. Invent. Math. 146.2 (2001), pp. 259–295.
- [Lit11] J. E. Littlewood. The Converse of Abel's Theorem on Power Series. Proc. London Math. Soc. (2) 9 (1911), pp. 434–448.
- [Liu18] Y. Liu. Modular curvature for toric noncommutative manifolds. J. Noncommut. Geom. 12.2 (2018), pp. 511–575.

- [Liu22] Y. Liu. Cyclic Structure behind Modular Gaussian Curvature. arXiv preprint arXiv:2201.08730 (2022).
- [LM16] M. Lesch and H. Moscovici. Modular curvature and Morita equivalence. Geom. Funct. Anal. 26.3 (2016), pp. 818–873.
- [LM89] H. B. Lawson Jr. and M.-L. Michelsohn. Spin geometry. Vol. 38. Princeton Mathematical Series. Princeton University Press, Princeton, NJ, 1989, pp. xii+427.
- [LMSZ23] S. Lord, E. McDonald, F. Sukochev, and D. Zanin. Singular traces. Vol.
 2. Trace Formulas. Vol. 46/2. De Gruyter Studies in Mathematics. Second edition. De Gruyter, Berlin, 2023.
- [Löw34] K. Löwner. Über monotone Matrixfunktionen. *Math. Z.* 38.1 (1934), pp. 177–216.
- [LPPV08] D. Lenz, N. Peyerimhoff, O. Post, and I. Veselić. Continuity properties of the integrated density of states on manifolds. Jpn. J. Math. 3.1 (2008), pp. 121– 161.
- [LPS10] S. Lord, D. Potapov, and F. Sukochev. Measures from Dixmier traces and zeta functions. J. Funct. Anal. 259.8 (2010), pp. 1915–1949.
- [LPV04] D. Lenz, N. Peyerimhoff, and I. Veselić. Integrated density of states for random metrics on manifolds. Proc. London Math. Soc. (3) 88.3 (2004), pp. 733– 752.
- [LPV07] D. Lenz, N. Peyerimhoff, and I. Veselić. Groupoids, von Neumann algebras and the integrated density of states. *Math. Phys. Anal. Geom.* 10.1 (2007), pp. 1–41.
- [LS10] S. Lord and F. Sukochev. Measure theory in noncommutative spaces. SIGMA Symmetry Integrability Geom. Methods Appl. 6 (2010), Paper 072, 36.
- [LS11] S. Lord and F. A. Sukochev. Noncommutative residues and a characterisation of the noncommutative integral. Proc. Amer. Math. Soc. 139.1 (2011), pp. 243–257.

- [LS24] M. Leimbach and W. D. van Suijlekom. Gromov-Hausdorff convergence of spectral truncations for tori. Adv. Math. 439 (2024), Paper No. 109496, 26.
- [LS84] D. Levine and P. J. Steinhardt. Quasicrystals: a new class of ordered structures. *Physical review letters* 53.26 (1984), p. 2477.
- [LS96] A. Laptev and Y. Safarov. Szegő type limit theorems. J. Funct. Anal. 138.2 (1996), pp. 544–559.
- [LSZ20a] G. Levitina, F. Sukochev, and D. Zanin. Cwikel estimates revisited. Proc. Lond. Math. Soc. (3) 120.2 (2020), pp. 265–304.
- [LSZ20b] S. Lord, F. Sukochev, and D. Zanin. A last theorem of Kalton and finiteness of Connes' integral. J. Funct. Anal. 279.7 (2020), pp. 108664, 54.
- [LSZ21] S. Lord, F. Sukochev, and D. Zanin. Singular traces. Vol. 1. Theory. Vol. 46/1. De Gruyter Studies in Mathematics. Second edition. De Gruyter, Berlin, 2021, pp. xxix+383.
- [Lub94] A. Lubotzky. Discrete groups, expanding graphs and invariant measures.
 Vol. 125. Progress in Mathematics. With an appendix by Jonathan D. Rogawski. Birkhäuser Verlag, Basel, 1994, pp. xii+195.
- [Mas76] V. P. Maslov. Operational methods. Translated from the Russian by V. Golo,N. Kulman and G. Voropaeva. Mir Publishers, Moscow, 1976, p. 559.
- [Mel83] A. Melin. Parametrix constructions for right invariant differential operators on nilpotent groups. Ann. Global Anal. Geom. 1.1 (1983), pp. 79–130.
- [Mel91] R. B. Melrose. Pseudodifferential operators, corners and singular limits. Proceedings of the International Congress of Mathematicians, Vol. I, II (Kyoto, 1990). Math. Soc. Japan, Tokyo, 1991, pp. 217–234.
- [Mel93] R. B. Melrose. The Atiyah-Patodi-Singer index theorem. Vol. 4. Research Notes in Mathematics. A K Peters, Ltd., Wellesley, MA, 1993, pp. xiv+377.
- [Mil64] J. Milnor. Eigenvalues of the Laplace operator on certain manifolds. Proc. Nat. Acad. Sci. U.S.A. 51 (1964), p. 542.

- [Mit61] B. S. Mitjagin. Approximate dimension and bases in nuclear spaces. Uspehi Mat. Nauk 16.4(100) (1961), pp. 63–132.
- [MM98] R. Mazzeo and R. B. Melrose. Pseudodifferential operators on manifolds with fibred boundaries. Vol. 2. 4. Mikio Sato: a great Japanese mathematician of the twentieth century. 1998, pp. 833–866.
- [MP49] S. Minakshisundaram and Å. Pleijel. Some properties of the eigenfunctions of the Laplace-operator on Riemannian manifolds. *Canad. J. Math.* 1 (1949), pp. 242–256.
- [MP97] B. Monthubert and F. Pierrot. Indice analytique et groupoïdes de Lie. C. R. Acad. Sci. Paris Sér. I Math. 325.2 (1997), pp. 193–198.
- [MS22] E. McDonald and F. Sukochev. Lipschitz estimates in quasi-Banach Schatten ideals. *Math. Ann.* 383.1-2 (2022), pp. 571–619.
- [MS86] G. A. Meladze and M. A. Shubin. Proper uniform pseudodifferential operators on unimodular Lie groups. *Trudy Sem. Petrovsk.* 11 (1986), pp. 74–97, 244, 246–247.
- [MS87] G. A. Meladze and M. A. Shubin. A functional calculus of pseudodifferential operators on unimodular Lie groups. *Trudy Sem. Petrovsk.* 12 (1987), pp. 164–200, 245.
- [MSZ19] E. McDonald, F. Sukochev, and D. Zanin. A C*-algebraic approach to the principal symbol II. *Math. Ann.* 374.1-2 (2019), pp. 273–322.
- [MSZ22] E. McDonald, F. Sukochev, and D. Zanin. A noncommutative Tauberian theorem and Weyl asymptotics in noncommutative geometry. *Lett. Math. Phys.* 112.4 (2022), Paper No. 77, 45.
- [Mur90] G. J. Murphy. C*-algebras and operator theory. Academic Press, Inc., Boston, MA, 1990, pp. x+286.
- [Nel86] D. R. Nelson. Quasicrystals. Scientific American 255.2 (1986), pp. 42–51.
- [Nik22] E. A. Nikitopoulos. Itô's formula for noncommutative C^2 functions of free Itô processes. *Doc. Math.* 27 (2022), pp. 1447–1507.

- [Nik23] E. A. Nikitopoulos. Multiple operator integrals in non-separable von Neumann algebras. *Journal of Operator Theory* 89.2 (2023), pp. 361–427.
- [NR10] F. Nicola and L. Rodino. Global pseudo-differential calculus on Euclidean spaces. Vol. 4. Pseudo-Differential Operators. Theory and Applications. Birkhäuser Verlag, Basel, 2010, pp. x+306.
- [NS22] T. D. H. van Nuland and W. D. van Suijlekom. Cyclic cocycles in the spectral action. J. Noncommut. Geom. 16.3 (2022), pp. 1103–1135.
- [NSMN21] Y. Nakamura, R. Sakamoto, T. Mase, and J. Nakagawa. Coordination sequences of crystals are of quasi-polynomial type. Acta Crystallographica Section A: Foundations and Advances 77.2 (2021), pp. 138–148.
- [NSZ25] T. van Nuland, F. Sukochev, and D. Zanin. Local invariants of conformally deformed non-commutative tori II: multiple operator integrals. J. Funct. Anal. 288.4 (2025), Paper No. 110754, 42.
- [NWX99] V. Nistor, A. Weinstein, and P. Xu. Pseudodifferential operators on differential groupoids. *Pacific J. Math.* 189.1 (1999), pp. 117–152.
- [OKe95] M. O'Keeffe. Coordination sequences for lattices. Zeitschrift f
 ür Kristallographie - Crystalline Materials 210.12 (1995), pp. 905–908.
- [OP05] C. R. de Oliveira and R. A. Prado. Dynamical delocalization for the 1D Bernoulli discrete Dirac operator. J. Phys. A 38.7 (2005), pp. L115–L119.
- [Pas80] L. A. Pastur. Spectral properties of disordered systems in the one-body approximation. Comm. Math. Phys. 75.2 (1980), pp. 179–196.
- [Pav71] B. Pavlov. On multidimensional integral operators. Linear Operators and Operator Equations (1971), pp. 81–97.
- [Pay07] S. Paycha. (Second) quantised resolvents and regularised traces. J. Geom. Phys. 57.5 (2007), pp. 1345–1369.
- [Pay11] S. Paycha. Noncommutative formal Taylor expansions and second quantised regularised traces. *Combinatorics and physics*. Vol. 539. Contemp. Math. Amer. Math. Soc., Providence, RI, 2011, pp. 349–376.

- [Pel06] V. V. Peller. Multiple operator integrals and higher operator derivatives. Journal of Functional Analysis 233.2 (2006), pp. 515–544.
- [Pel16] V. V. Peller. Multiple operator integrals in perturbation theory. Bulletin of Mathematical Sciences 6.1 (Apr. 2016), pp. 15–88.
- [Pel90] V. Peller. Hankel operators in the perturbation theory of unbounded selfadjoint operators. Analysis and partial differential equations. Vol. 122. Lecture Notes in Pure and Appl. Math. Dekker, New York, 1990, pp. 529–544.
- [Pet06] P. Petersen. Riemannian geometry. Second edition. Vol. 171. Graduate Texts in Mathematics. Springer, New York, 2006, pp. xvi+401.
- [PF92] L. Pastur and A. Figotin. Spectra of random and almost-periodic operators. Vol. 297. Grundlehren der mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences]. Springer-Verlag, Berlin, 1992, pp. viii+587.
- [Pon08] R. S. Ponge. Heisenberg calculus and spectral theory of hypoelliptic operators on Heisenberg manifolds. Mem. Amer. Math. Soc. 194.906 (2008), pp. viii+ 134.
- [Pon23] R. Ponge. Connes' integration and Weyl's laws. J. Noncommut. Geom. 17.2 (2023), pp. 719–767.
- [POO21] R. A. Prado, C. R. de Oliveira, and E. C. de Oliveira. Density of states and Lifshitz tails for discrete 1D random Dirac operators. *Math. Phys. Anal. Geom.* 24.3 (2021), Paper No. 30, 29.
- [PRCB21] I. Papaefstathiou, D. Robaina, J. I. Cirac, and M. C. Bañuls. Density of states of the lattice Schwinger model. *Phys. Rev. D* 104.1 (2021), Paper No. 014514, 10.
- [PS11] D. Potapov and F. Sukochev. Operator-Lipschitz functions in Schatten-von Neumann classes. Acta Math. 207.2 (2011), pp. 375–389.
- [PSS13] D. Potapov, A. Skripka, and F. Sukochev. Spectral shift function of higher order. *Invent. Math.* 193.3 (2013), pp. 501–538.

- [PV02] N. Peyerimhoff and I. Veselić. Integrated density of states for ergodic random Schrödinger operators on manifolds. Proceedings of the Euroconference on Partial Differential Equations and their Applications to Geometry and Physics (Castelvecchio Pascoli, 2000). Vol. 91. 2002, pp. 117–135.
- [PWS02] B. de Pagter, H. Witvliet, and F. A. Sukochev. Double operator integrals. J. Funct. Anal. 192.1 (2002), pp. 52–111.
- [Rie23] M. A. Rieffel. Convergence of Fourier truncations for compact quantum groups and finitely generated groups. J. Geom. Phys. 192 (2023), Paper No. 104921, 13.
- [RLL00] M. Rørdam, F. Larsen, and N. Laustsen. An introduction to K-theory for C*algebras. Vol. 49. London Mathematical Society Student Texts. Cambridge University Press, Cambridge, 2000, pp. xii+242.
- [Roe88] J. Roe. An index theorem on open manifolds. I. J. Differential Geom. 27.1 (1988), pp. 87–113.
- [Roe98] J. Roe. Elliptic operators, topology and asymptotic methods. Second edition.
 Vol. 395. Pitman Research Notes in Mathematics Series. Longman, Harlow, 1998, pp. ii+209.
- [RS76] L. P. Rothschild and E. M. Stein. Hypoelliptic differential operators and nilpotent groups. Acta Math. 137.3-4 (1976), pp. 247–320.
- [RT10] M. Ruzhansky and V. Turunen. Pseudo-differential operators and symmetries. Vol. 2. Pseudo-Differential Operators. Theory and Applications. Background analysis and advanced topics. Birkhäuser Verlag, Basel, 2010, pp. xiv+709.
- [Rud87] W. Rudin. Real and complex analysis. Third edition. McGraw-Hill Book Co., New York, 1987, pp. xiv+416.
- [Rud91] W. Rudin. Functional analysis. Second edition. International Series in Pure and Applied Mathematics. McGraw-Hill, Inc., New York, 1991, pp. xviii+424.
- [Saf97] Y. Safarov. Pseudodifferential operators and linear connections. Proc. London Math. Soc. (3) 74.2 (1997), pp. 379–416.

- [Sak96] T. Sakai. Riemannian geometry. Vol. 149. Translations of Mathematical Monographs. Translated from the 1992 Japanese original by the author. American Mathematical Society, Providence, RI, 1996, pp. xiv+358.
- [SBGC84] D. Shechtman, I. Blech, D. Gratias, and J. W. Cahn. Metallic phase with long-range orientational order and no translational symmetry. *Physical re*view letters 53.20 (1984), p. 1951.
- [Sch12] K. Schmüdgen. Unbounded self-adjoint operators on Hilbert space. Vol. 265.Graduate Texts in Mathematics. Springer, Dordrecht, 2012, pp. xx+432.
- [Sch16] H. Schulz-Baldes. Topological insulators from the perspective of non-commutative geometry and index theory. Jahresber. Dtsch. Math.-Ver. 118.4 (2016), pp. 247– 273.
- [Shi96] A. N. Shiryaev. Probability. Second edition. Vol. 95. Graduate Texts in Mathematics. Translated from the first (1980) Russian edition by R. P. Boas. Springer-Verlag, New York, 1996, pp. xvi+623.
- [Shn74] A. I. Shnirelman. Ergodic properties of eigenfunctions. Uspehi Mat. Nauk 29.6(180) (1974), pp. 181–182.
- [Shu01] M. A. Shubin. Pseudodifferential operators and spectral theory. Second edition. Translated from the 1978 Russian original by Stig I. Andersson. Springer-Verlag, Berlin, 2001, pp. xii+288.
- [Shu79] M. A. Shubin. Spectral theory and the index of elliptic operators with almostperiodic coefficients. Uspekhi Mat. Nauk 34.2(206) (1979), pp. 95–135.
- [Shu92] M. A. Shubin. Spectral theory of elliptic operators on noncompact manifolds.
 207. Méthodes semi-classiques, Vol. 1 (Nantes, 1991). 1992, pp. 5, 35–108.
- [Sim05] B. Simon. Trace ideals and their applications. Second edition. Vol. 120. Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2005, pp. viii+150.
- [Sim15] B. Simon. Operator theory. A Comprehensive Course in Analysis, Part 4.
 American Mathematical Society, Providence, RI, 2015, pp. xviii+749.

- [Sim82] B. Simon. Schrödinger semigroups. Bull. Amer. Math. Soc. (N.S.) 7.3 (1982), pp. 447–526.
- [Sjö91] J. Sjöstrand. Microlocal analysis for the periodic magnetic Schrödinger equation and related questions. *Microlocal analysis and applications (Montecatini Terme, 1989)*. Vol. 1495. Lecture Notes in Math. Springer, Berlin, 1991, pp. 237–332.
- [Skr14] A. Skripka. Asymptotic expansions for trace functionals. Journal of Functional Analysis 266.5 (2014), pp. 2845–2866.
- [Skr18] A. Skripka. Taylor asymptotics of spectral action functionals. J. Operator Theory 80.1 (2018), pp. 113–124.
- [SM08] A. V. Shutov and A. V. Maleev. Quasiperiodic plane tilings based on stepped surfaces. Acta Crystallogr. Sect. A 64.3 (2008), pp. 376–382.
- [SM15] A. V. Shutov and A. V. Maleev. Penrose tilings as model sets. Crystallography Reports 60.6 (2015), pp. 797–804.
- [SM18] A. Shutov and A. Maleev. Coordination numbers of the vertex graph of a Penrose tiling. Acta Crystallogr. Sect. A 74.2 (2018), pp. 112–122.
- [SMZ10] A. V. Shutov, A. V. Maleev, and V. G. Zhuravlev. Complex quasiperiodic selfsimilar tilings: their parameterization, boundaries, complexity, growth and symmetry. Acta Crystallographica Section A: Foundations of Crystallography 66.3 (2010), pp. 427–437.
- [SS71] M. Solomyak and V. Sten'kin. On one class of Stieltjes multiple-integral operators. *Linear Operators and Operator Equations* (1971), pp. 99–108.
- [SSUZ15] E. Semenov, F. Sukochev, A. Usachev, and D. Zanin. Banach limits and traces on $\mathcal{L}_{1,\infty}$. Adv. Math. 285 (2015), pp. 568–628.
- [ST19] A. Skripka and A. Tomskova. Multilinear operator integrals. Vol. 2250. Lecture Notes in Mathematics. Theory and applications. Springer, Cham, 2019, pp. xi+190.

- [Sta86] R. P. Stanley. Enumerative combinatorics. Vol. I. The Wadsworth & Brooks/Cole Mathematics Series. With a foreword by Gian-Carlo Rota. Wadsworth & Brooks/Cole Advanced Books & Software, Monterey, CA, 1986, pp. xiv+306.
- [Ste19] A. B. Stern. Finite-rank approximations of spectral zeta residues. Lett. Math. Phys. 109.3 (2019), pp. 565–577.
- [Ste77] V. Sten'kin. Multiple operator integrals. Izvestiya Vysshikh Uchebnykh Zavedenii. Matematika 4 (1977), pp. 102–115.
- [Str12] R. S. Strichartz. Spectral asymptotics revisited. J. Fourier Anal. Appl. 18.3 (2012), pp. 626–659.
- [Str72] R. S. Strichartz. A functional calculus for elliptic pseudo-differential operators. Amer. J. Math. 94 (1972), pp. 711–722.
- [SU16] F. Sukochev and A. Usachev. Dixmier traces and non-commutative analysis.J. Geom. Phys. 105 (2016), pp. 102–122.
- [Sui11] W. D. van Suijlekom. Perturbations and operator trace functions. J. Funct. Anal. 260.8 (2011), pp. 2483–2496.
- [Sui24a] W. D. van Suijlekom. A generalization of K-theory to operator systems. *arXiv* preprint arXiv:2409.02773 (2024).
- [Sui24b] W. D. van Suijlekom. Higher K-groups for operator systems. *arXiv preprint arXiv:2411.02981* (2024).
- [Sui25] W. D. van Suijlekom. Noncommutative geometry and particle physics. Second
 Edition. Mathematical Physics Studies. Springer, Dordrecht, 2025, pp. xiii+315.
- [SUZ13] F. Sukochev, A. Usachev, and D. Zanin. On the distinction between the classes of Dixmier and Connes-Dixmier traces. *Proc. Amer. Math. Soc.* 141.6 (2013), pp. 2169–2179.
- [SZ18] F. Sukochev and D. Zanin. A C*-algebraic approach to the principal symbol.
 I. J. Operator Theory 80.2 (2018), pp. 481–522.
- [SZ23] F. A. Sukochev and D. Zanin. The Connes character formula for locally compact spectral triples. Volume 445. Astérisque, 2023, p. 150.

- [Sza03] R. J. Szabo. Quantum field theory on noncommutative spaces. *Phys. Rep.* 378.4 (2003), pp. 207–299.
- [Sze15] G. Szegő. Ein Grenzwertsatz über die Toeplitzschen Determinanten einer reellen positiven Funktion. Math. Ann. 76.4 (1915), pp. 490–503.
- [Tay81] M. E. Taylor. Pseudodifferential operators. Vol. No. 34. Princeton Mathematical Series. Princeton University Press, Princeton, NJ, 1981, pp. xi+452.
- [Tay84] M. E. Taylor. Noncommutative microlocal analysis. I. Mem. Amer. Math. Soc. 52.313 (1984), pp. iv+182.
- [Tit86] E. C. Titchmarsh. The theory of the Riemann zeta-function. Second edition. Edited and with a preface by D. R. Heath-Brown. The Clarendon Press, Oxford University Press, New York, 1986, pp. x+412.
- [Trè67] F. Trèves. Topological vector spaces, distributions and kernels. Academic Press, New York-London, 1967, pp. xvi+624.
- [Tri78] H. Triebel. Interpolation theory, function spaces, differential operators. Vol. 18.
 North-Holland Mathematical Library. North-Holland Publishing Co., Amsterdam-New York, 1978, p. 528.
- [Tri92] H. Triebel. Theory of function spaces. II. Vol. 84. Monographs in Mathematics. Birkhäuser Verlag, Basel, 1992, pp. viii+370.
- [Usa13] A. Usachev. Subclasses of Dixmier (singular) traces and related classes of measurable operators. PhD thesis. UNSW Sydney, 2013.
- [Uuy11] O. Uuye. Pseudo-differential operators and regularity of spectral triples. *Perspectives on noncommutative geometry* 61 (2011), pp. 153–163.
- [Ves05] I. Veselić. Spectral analysis of percolation Hamiltonians. Math. Ann. 331.4 (2005), pp. 841–865.
- [Ves08] I. Veselić. Existence and regularity properties of the integrated density of states of random Schrödinger operators. Vol. 1917. Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2008, pp. x+142.

- [Voi71] E.-M. Voigt. Sappho et Alcaeus: fragmenta. Polak & van Gennep, Amsterdam, 1971, pp. ix + 507.
- [VTC87] N. N. Vakhania, V. I. Tarieladze, and S. A. Chobanyan. Probability distributions on Banach spaces. Vol. 14. Mathematics and its Applications (Soviet Series). Translated from Russian and with a preface by Wojbor A. Woyczynski. D. Reidel Publishing Co., Dordrecht, 1987, pp. xxvi+482.
- [Weg81] F. Wegner. Bounds on the density of states in disordered systems. Z. Phys. B 44.1-2 (1981), pp. 9–15.
- [Weg93] N. E. Wegge-Olsen. K-theory and C*-algebras. A friendly approach. Oxford Science Publications. The Clarendon Press, Oxford University Press, New York, 1993, pp. xii+370.
- [Wey12] H. Weyl. Das asymptotische Verteilungsgesetz der Eigenwerte linearer partieller Differentialgleichungen (mit einer Anwendung auf die Theorie der Hohlraumstrahlung). Math. Ann. 71.4 (1912), pp. 441–479.
- [Wey13] H. Weyl. Über die Randwertaufgabe der Strahlungstheorie und asymptotische Spektralgesetze. J. Reine Angew. Math. 143 (1913), pp. 177–202.
- [Wid78] H. Widom. Families of pseudodifferential operators. Topics in functional analysis (essays dedicated to M. G. Kreĭn on the occasion of his 70th birthday). Vol. 3. Adv. Math. Suppl. Stud. Academic Press, New York-London, 1978, pp. 345–395.
- [Wid79] H. Widom. Eigenvalue distribution theorems for certain homogeneous spaces.J. Functional Analysis 32.2 (1979), pp. 139–147.
- [Wid80] H. Widom. A complete symbolic calculus for pseudodifferential operators.
 Bull. Sci. Math. (2) 104.1 (1980), pp. 19–63.
- [YN87] K. Yakubo and T. Nakayama. Absence of the hump in the density of states of percolating clusters. *Physical Review B* 36.16 (1987), pp. 8933–8936.

- [Zel10] S. Zelditch. Recent developments in mathematical quantum chaos. Current developments in mathematics, 2009. Int. Press, Somerville, MA, 2010, pp. 115–204.
- [Zel17] S. Zelditch. Eigenfunctions of the Laplacian on a Riemannian manifold. Vol. 125. CBMS Regional Conference Series in Mathematics. Conference Board of the Mathematical Sciences, Washington, DC; by the American Mathematical Society, Providence, RI, 2017, pp. xiv+394.
- [Zel87] S. Zelditch. Uniform distribution of eigenfunctions on compact hyperbolic surfaces. Duke Math. J. 55.4 (1987), pp. 919–941.
- [Zel96] S. Zelditch. Quantum ergodicity of C^{*} dynamical systems. Comm. Math. Phys. 177.2 (1996), pp. 507–528.
- [ZS23] D. V. Zanin and F. Sukochev. Connes integration formula: a constructive approach. *Funct. Anal. Appl.* 57.1 (2023). Translation of Funktsional. Anal. i Prilozhen. 57 (2023), no. 1, 52–76., pp. 40–59.
- [Zwo12] M. Zworski. Semiclassical analysis. Vol. 138. Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2012, pp. xii+431.